Unravelling Pain Together

Unravelling Pain Together Patient centered transdisciplinary pa

3

ience

education

Amarins Wijma

Patient centered transdisciplinary pain neuroscience education

Amarins Wijma

UNRAVELLING PAIN TOGETHER

Patient centered transdisciplinary pain neuroscience education

Amarins Wijma

'It is never done, nor finished

it is never in perfection, yet perfect

just as life is'

UNRAVELLING PAIN TOGETHER - Amarins Wijma

ISBN/EAN:	978-94-6402-422-7
Cover design by:	Evelien Jagtman
Layout and design by:	Marilou Maes, persoonlijkproefschrift.nl
Printing:	Gildeprint Enschede, gildeprint.nl

Copyright © 2021 Amarins Wijma

All rights reserved. No part of this book may be reproduced or transmitted in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without the prior written permission of the author.

Department of Physiotherapy, Human Physiology and Anatomy Faculty of Physical Education and Physiotherapy



Doctoral program in Rehabilitation Sciences and Physiotherapy at the Vrije Universiteit Brussel.

UNRAVELLING PAIN TOGETHER Patient centered transdisciplinary pain neuroscience education

PhD dissertation presented by

Amarins Jelly Wijma

Supervisors: Prof. Dr. C. Paul van Wilgen Prof. Dr. Jo Nijs

> Co-supervisor: Dr. Doeke Keizer

Academic year 2020-2021

President of the jury:

Prof. Dr. Filip van Ginderdeuren, Vrije Universiteit Brussel

Members of the jury:

Prof. Dr. Cindy Veenhof

Prof. Dr. Rob Smeets

Prof. Dr. Anneleen Malfliet

Prof. Dr. Tom Dielens

Table of content

CHAPTER 1	General introduction	9
CHAPTER 2	To what extent is patient centeredness examined in physiotherapy and can a theoretical framework be constructed?	29
CHAPTER 3	The clinical biopsychosocial assessment in the physiotherapy encounter of patients with chronic pain	67
CHAPTER 4	Perspectives of patients with chronic pain who received transdisciplinary Pain Neuroscience Education	107
CHAPTER 5	Clinical observations of patient and therapist interactions in transdisciplinary Pain Neuroscience Education	137
CHAPTER 6	Moderators of change in self-reported signs of central sensitization following an individualized transdisciplinary pain management program	211
CHAPTER 7	General discussion	235
Summary		259
Samenvattin	g	267
Acknowledge	ements	275
About the au	thor	279
List of public	ations	283



Pain and chronic pain

Whether we want to or not, pain is an inevitable part of life. Whether it is a sprain, cut, headache or a heartache: without any doubt we will experience some form of pain at one point or another in our life. Chances are, we might even experience more severe and chronic pain. As on average one in every five persons worldwide experiences (severe) chronic pain in his/her life span^{1.4}. Whereby, chronic pain is defined as "pain that persists beyond normal time of healing and/or pain persisting for 3–6 months or longer"⁵. This makes the prevalence of chronic pain higher than any other chronic disease, including cancer, heart disease and diabetes⁶⁻⁹. Until this date chronic pain is a leading cause of disability and human suffering¹⁰. Persons with chronic pain experience the highest number of years lived with disability⁶ and a large negative impact of their pain on their quality of life^{1,3}. Even worse, severe chronic pain decreases life expectancy¹¹ and increases chances of cancer and cardiovascular mortality¹². Besides, patients with chronic pain often have comorbidities such as depression, fear and sleep problems¹³⁻¹⁶. On a financial level, chronic pain is associated with huge economic burdens for the patient, healthcare and the economy ^{3,17-21}, making it a major global healthcare problem. This enormous personal and societal impact of pain is the reason why it is proposed that chronic pain should be considered a disease in its own right²²⁻²⁴.

When looking at chronic pain as a disease in its own right, one should first look at the existing knowledge on the etiology of chronic pain. As unlike acute pain, often in chronic pain there is no longer bodily threat or 'damage' present. Yet, how can pain persist? Furthermore how should it be called? And what elements influence this pain and the patients' pain experience? The first part of this General Introduction covers the denomination of chronic pain and factors underlining the pain experience. As we answer these questions one can see that the last decades knowledge on pain mechanisms has changed and is still evolving. Subsequently, the treatment of patients with chronic pain has evolved. In the second part of this General Introduction preliminaries for the treatment of patients with chronic pain are discussed.

Etiology of chronic pain

In many patients with chronic pain, the etiology of their pain cannot, or not solely, be explained by the existence of somatic pathophysiological factors (nociceptive nor neuropathic, by inflammation, infections, autoimmune or metabolic)²⁵⁻³⁰ and the main pain mechanism model is not merely nociceptive³¹. Instead, the persistence of pain can be explained by changes in the central nervous system^{25,26,32-34}, also known as

'central sensitization' (CS) ^{24,35} or in clinical presentation known as 'nociplastic pain'³⁶. CS is defined as 'an increased responsiveness of nociceptive neurons in the central nervous system to their normal or subthreshold afferent input'^{24,35}. In patients with CS an amplification of neural signaling elicits generalized pain hypersensitivity^{25,37}. This hypersensitivity is caused by hyperexcitability of the nervous system, leading to increased responsiveness of a variety of non-harmful stimuli, such as: movement, touch, light, scents, foods and medication^{29,38}, eventually causing pain without (noxious) peripheral stimuli²⁵. In CS several neurophysiological, endocrine and immunological processes involved in neural plasticity are overly activated³⁹. Evidence exists that these patients experience prolonged pain after (potential noxious) stimulation, have pain in referred areas (widespread pain) and demonstrate symptoms as hyperesthesia, hyperalgesia and allodynia^{32,33,40}. In these patients sensitization at the spinal cord level, enhanced activity of the nociceptive descending faciliatory pathways^{41,42}, impaired conditioned pain modulation⁴³ and impaired endogenous pain inhibition³⁸ takes place. Furthermore, the central processing of patients with CS is altered and, in contrast to acute pain, different brain areas are associated with the processing of stimuli⁴⁴⁻⁴⁹. In relationship to the processing of stimuli these brain areas combined are currently known as the 'dynamic pain connectome'⁵⁰. Each individual has his/her personal 'connectome'45,48,51,52.

What's in a name?

Yet, currently there is a debate going within the scientific community, in social media and in scientific articles, about the description and definition of CS (pain) in relation to chronic pain^{36,53-56}. This discussion is based on several arguments: First, it is currently under debate that the use of the wording of CS should only be restricted to preclinical studies and not to clinical practice. This restriction is based on the following arguments: the neural response of CS is so far only measured in animals ^{54,56} and CS is only studied in humans in subjective pain measures^{39,54,56}. This statement is correct, however, are not most conditions, like non-specific neck- and back pain, depression or anxiety, measured subjectively in clinical practice? Furthermore, some argue there is an absence of evidence demonstrating how spatial extension is spreading in the central nervous system in CS⁵⁶. Nonetheless, on the other hand spread of pain sensitivity has been found in several human studies^{25,39,55}. Additionally, many studies have demonstrated that CS is a robust phenomenon occurring when induction of central facilitation of nociceptive central pathways takes place and contributes to pain hypersensitivity^{29,55}.

Second, measures of CS in clinical practice are all indirect⁵⁴, for instance by phenomena such as hyperalgesia and allodynia. These measures are based on the history taking, quantitative sensory testing⁵⁷ and self-report instruments, such as the Central Sensitization Inventory (CSI) questionnaire^{58,59}. It is discussed by some that, based on these indirect measures, clinically establishing a 'diagnosis' of CS is difficult⁵⁶. In opposition, evidence exists that upregulated and central-mediated responses by a hyperexitable nervous system are present in many patients with chronic pain^{39,60}. Besides, clinical signs of CS pathways can indeed be observed in patients with chronic pain^{33,61,62}. Therefore, to support clinical practice an algorithm for the classification of CS has been established⁵⁹. Furthermore, the International Association for the Study of Pain (IASP) terminology taskforce is working on clinical diagnostical criteria.

Third, it is discussed that CS is a continuum, i.e. signs of CS can exist in co-occurrence with nociceptive and neuropathic pain. However, signs of CS can also exist without input from a nociceptive or neuropathic component^{29,39,63,64}. But then, not every patient with chronic pain has CS and in some chronic pain populations a subgroup presents with predominant CS ^{27,29,63,65-74}. It is therefore proposed to screen the patient for their (predominant) 'pain type', while keeping in mind that an overlap between these types can occur⁵⁹.

Finally, there is presently a discussion going on about the name of CS. It is currently called dominant CS pain, central sensitivity syndromes⁷⁵, nociplastic pain³⁶ or chronic primary pain'⁷⁶. Nociplastic pain is the definition as used by the IASP terminology task force. Throughout this PhD dissertation the definition of CS is used, however, whenever CS is written it can also be read as nociplastic pain.

Factors underlining CS pain

We have now briefly reviewed the current evidence on the etiology of chronic pain, and how to call it (CS or nociplastic pain). Yet, what influences the continuation and amplification of the patients' pain experience? Even though it is not yet completely understood why some people develop CS and others do not, there are some indicators that premorbid and acute stage high sensory sensitivity and/or somatization predict CS⁷⁷. Additionally, it is known that biopsychosocial factors can contribute to increased activity in the 'dynamic pain connectome^{45,50,78-81} also known as 'cognitive and emotional senzitisation'⁸⁰. These biopsychosocial factors include, but are not limited to, maladaptive cognitions about pain, pain catastrophizing, pain related anxiety, hypervigilance, trait anxiety, trait neuroticism, kinesiophobia, depressive feelings, stress, traumatic stress disorders, adverse life events, diminished self-efficacy and inactivity/ deconditioning^{78,81-99}. Hence, as not all factors play a role in each patient it is important to identify the individual's biopsychosocial factors. Subsequent, further treatment should aim to comprehend and if possible change these (modifiable) factors in a patient centered manner.

Yet, especially when it comes to identifying and treating psychosocial factors within physiotherapy: there is a crux. Research has shown that physiotherapists struggle in this area ¹⁰⁰⁻¹⁰⁶, and for instance rarely ask about the patients' cognitions on their condition. Therefore, we conducted a narrative review, as described in this dissertation, to aid in the patient centered biopsychosocial assessment for physiotherapists encountering patients with chronic pain.

Education of pain in patients with chronic pain

In any illness, explaining the etiology in lay terms of this condition to the patient is important for the individual. To appropriately self-manage it is crucial for the individual to make sense of the biology and physiology of their condition. This is no different for patients with chronic pain¹⁰⁷. Therefore, education is recommended in the treatment for patients with chronic pain ^{108,109}. Educating patients with chronic pain in the neurophysiology of pain is called pain neuroscience education (PNE)^{107,110,111}. PNE was first mentioned on the IASP Conference in Vienna, by Louis Gifford in 1999¹¹² and first studied in a RCT by Moseley et al. in 2002¹¹³. Since then the research into PNE has shown positive effects of PNE in patients with chronic pain on their knowledge of pain and pain perceptions; decreased pain and increased endogenous pain inhibition; improved mental health, physical functioning, vitality, and self-rated disability; and a diminished passive coping, kinesiophobia and catastrophizing¹¹³⁻¹²⁴, however, only when PNE is provided in combination with other (physiotherapy) treatment modalities ¹¹³⁻¹²². When further exploring the reconceptualization of pain after group PNE, most patients, but not all, mentioned increased pain knowledge which they then translated to daily life¹²⁵. Essential in this reconceptualization is the perceived relevance for the individual, complaints being addressed and being taken seriously^{125,126}. Therefore, PNE should be performed in a patient centered manner.

Patient centered care in patients with chronic pain

Even though the importance of patient-centeredness in healthcare encounters was already recognized by philosophers, such as Kierkegaard in 1849: 'If one is truly to succeed in leading a person to a specific place, one must first and foremost take care to find him where he is and begin there. This is the secret in the entire art of helping'. Consultations in healthcare are often medically focused and there is a limited inclusion of the patients' perspective in the self-care of their chronic illnesses¹²⁷. Unlike what often happens in current practice, healthcare should be focused on the patient instead of the clinician. Reasons why we should adopt the patient centered vision, other than patient-centeredness being a moral philosophy¹²⁸, is because research has shown that improvements in patient centered care improves the therapeutic alliance, satisfaction, patient outcomes and decreases healthcare costs^{127,129-132}. These studies have changed healthcare policies: patient centered care has now been recognized as of high priority within healthcare, healthcare policies¹³³⁻¹³⁵ and identified by the World Health Organization (WHO) as an important goal for the future of medicine¹²⁹. Especially in the population of patients with chronic pain these are issues that resonate. As patients are often dissatisfied, the costs are rising, patient outcomes are often limited and self-care of their condition is of utmost importance. Therefore, patient-centeredness has possibilities to further enhance the treatment, among which PNE, of patients with chronic pain. Yet, what is patient-centeredness? The theoretical models, in form of reviews, proposing patient-centeredness have thus far, to our knowledge, been studied in overall care (The Picker Institute's principles¹³⁶), medicine ^{128,137} and nursing ¹³⁸. First described by Balint et al. in 1969, in an era where the doctor was often only focused on 'illness-oriented medicine', the author proposed to try to understand the whole person, as a unique human being¹³⁹. Probably the most commonly used framework in medicine is by Mead and Bower, describing the biopsychosocial perspective, the 'patient-as-person', sharing power and responsibility, the therapeutic alliance, and the 'doctor-as-person' ¹³⁷.

Even though there are definitions of patient-centeredness within care, medicine, nursing, occupational therapy and psychology, prior to the literature review described in this dissertation none was available for physiotherapy. Without a commonly accepted definition of patient-centeredness in physiotherapy the teachings of physiotherapists into the basis of a true personalized biopsychosocial treatment of patients with chronic pain, but also research into its' implementation and efficacy is stranded¹⁴⁰.

Transdisciplinary treatment of chronic pain

In the Netherlands, where the current dissertation and interventions took place, a stepped care principle of treatment modalities for patients with chronic pain is currently advised. In a patient-centered manner, based on the patients' preferences, nature and complexity of the complaints and effects of previous treatments, healthcare should be provided in a stepped manner¹⁴¹. In this stepped manner the therapeutic interventions of patients with chronic pain can be divided into different modalities: prevention, (often) monodisciplinary primary care, multi-, inter- or transdisciplinary secondary and tertiary care. As chronic pain is often a complex problem, effective team approaches can exceed the expertise of monodisciplinary care¹⁴². Within these team approaches differences can be observed between multi-, inter- and transdisciplinary teams (figure 1.).



Figure 1. Theoretic scheme of the differences between working multi-, inter and transdisciplinary

According to the terminology of the IASP multidisciplinary means: 'Multimodal treatment provided by practitioners from different disciplines, all the professions working separately with their own therapeutic aim for the patient and not necessarily communicating with each other'; while interdisciplinary is defined as: 'Multimodal treatment provided by a multidisciplinary team collaborating in assessment and treatment using a shared biopsychosocial model and goals, all working closely together with regular team meetings, agreement on diagnosis, therapeutic aims and plans for treatment and review'35. Transdisciplinary teamwork outperforms interdisciplinary in that the healthcare professionals collaborate intensively, with shared responsibilities^{143,144}. Even though every team member has his/her expertise (medicine, psychology or physiotherapy) there are flexible boundaries, roles and the team learns simultaneously in a nurturing biopsychosocial setting¹⁴³⁻¹⁴⁶. A large cross-sectional study by Quaschning et al. with 402 patients from 11 different rehabilitation clinics showed that team interaction is of key importance within the Model of Shared Decision Making (SDM) on patient satisfaction and treatment acceptance. Furthermore, team interaction serves as a predictor for treatment adherence and satisfaction with decisions within

the treatment¹⁴⁷. This highlights the importance of patient centered care, SDM and team interaction in rehabilitation.

So far, previous research has shown the importance of intensive collaborations, and the experiences of healthcare professionals with transdisciplinary care. However, the experience of patients with chronic pain with such an approach, nor the practice of transdisciplinary care itself has, to our knowledge, been studied. When we truly want a therapy to be patient-centered, we must first know what the patients' experiences are of that therapy (PNE). Furthermore, it is important to explore the social process of patient centered transdisciplinary PNE. When theorized which factors lead to a successful PNE, this could then be taught to other healthcare professionals working with transdisciplinary PNE.

Last, there has not yet been a study trying to comprehend the changes in outcomes following this transdisciplinary care, among which PNE, for patients with chronic pain. To comprehend under which conditions changes in outcomes of patients with chronic pain after a transdisciplinary cognitive behavioural therapy (CBT) treatment, among which PNE, might occur, a moderator analysis was performed.

Outline of the dissertation

Within this dissertation the patient centered transdisciplinary PNE and treatment for patients with chronic pain is examined. The first part of the dissertation focuses on physiotherapists. The first aim of this dissertation was to complement the existing literature of patient-centeredness, even though there are definitions of patientcenteredness within care, medicine, nursing, occupational therapy, and psychology, none was available for physiotherapy. The second aim was to aid physiotherapists, by describing a patient centered biopsychosocial assessment for physiotherapists encountering patients with chronic pain. The second part of the dissertation seeks to find an in depth understanding of the process of patient centered transdisciplinary PNE. This was done through qualitative research. Therefore, the third aim was to study and theorize the experiences of patients with chronic pain who recently received transdisciplinary PNE. The fourth aim was to comprehend the social phenomenon surrounding transdisciplinary PNE, and which factors lead to a successful transdisciplinary PNE. And the third and last part of this dissertation investigates the when and how of changes following a transdisciplinary CBT pain management treatment.

In this thesis we will therefore explore five components, each targeting one of these aims.

- Chapter 2 explores to what extent patient-centeredness is examined in physiotherapy in qualitative research and whether a theoretical framework can be constructed from this research for patient-centeredness in physiotherapy.
- Chapter 3 describes an outline of the clinical biopsychosocial assessment to be used as a practical tool by physiotherapists in the encounter of patients with chronic pain.
- Chapter 4 examines the perspectives of patients with chronic pain who recently received PNE in a transdisciplinary setting, in order to provide a theoretical model and framework of transdisciplinary PNE.
- Chapter 5 focuses on the clinical observations of transdisciplinary PNE. In order to explore, comprehend and theorize the social process between the patient, partner and healthcare professionals in the practice of transdisciplinary PNE. To answer which features and aspects contribute to the successful application of PNE.
- Chapter 6 examines whether changes in self-reported signs of CS after an individualized CBT transdisciplinary treatment program, with PNE as a treatment modality, can be explained by differences in illness perceptions and pain catastrophizing.

Chapter 7 provides a general discussion of this dissertation.

References

- 1. Breivik H, Collett B, Ventafridda V, Cohen R, Gallacher D. Survey of chronic pain in Europe: prevalence, impact on daily life, and treatment. *Eur J Pain*. 2006;10(4):287-333.
- Reid KJ, Harker J, Bala MM, et al. Epidemiology of chronic non-cancer pain in Europe: narrative review of prevalence, pain treatments and pain impact. *Curr Med Res Opin*. 2011;27(2):449-462.
- 3. Bekkering GE, Bala MM, Reid K, et al. Epidemiology of chronic pain and its treatment in The Netherlands. *Neth J Med.* 2011;69(3):141-153.
- 4. Leadley RM, Armstrong N, Lee YC, Allen A, Kleijnen J. Chronic diseases in the European Union: the prevalence and health cost implications of chronic pain. *J Pain Palliat Care Pharmacother*. 2012;26(4):310-325.
- 5. Merskey H, Bogduk N., International Association for the Study of Pain. Task Force on Taxonomy. *Classification of chronic pain: descriptions of chronic pain syndromes and definitions of pain terms.* 2 ed. the University of Michigan: IASP Press; 1994.
- GBD 2015 Disease and Injury Incidence and Prevalence Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990-2015: a systematic analysis for the Global Burden of Disease Study 2015. *Lancet.* 2016;388(10053):1545-1602.
- 7. American Diabetes Association. http://www.diabetes.org/diabetes-basics/diabetesstatistics/. Published 2012.
- 8. American Heart Association. Statistics Heart Disease and Stroke. Update: A Report From the American Heart Association. In: *Circulation*. Vol 123:e18-e209.2011:page 20.
- 9. American Cancer Society. Prevalence of Cancer. http://www.cancer.org/docroot/CRI/content/ CRI_2_6x_Cancer_Prevalence_How_Many_People_Have_Cancer.asp. Published 2014.
- 10. Goldberg DS, McGee SJ. Pain as a global public health priority. BMC Public Health. 2011;11:770.
- Torrance N, Elliott AM, Lee AJ, Smith BH. Severe chronic pain is associated with increased 10 year mortality. A cohort record linkage study. *Eur J Pain*. 2010;14(4):380-386.
- McBeth J, Symmons DP, Silman AJ, et al. Musculoskeletal pain is associated with a longterm increased risk of cancer and cardiovascular-related mortality. *Rheumatology (Oxford)*. 2009;48(1):74-77.
- 13. Tang NK, Wright KJ, Salkovskis PM. Prevalence and correlates of clinical insomnia cooccurring with chronic back pain. *J Sleep Res.* 2007;16(1):85-95.
- 14. Gatchel RJ, Peng YB, Peters ML, Fuchs PN, Turk DC. The biopsychosocial approach to chronic pain: scientific advances and future directions. *Psychol Bull.* 2007;133(4):581-624.

- 15. Kroenke K, Wu J, Bair MJ, Krebs EE, Damush TM, Tu W. Reciprocal relationship between pain and depression: a 12-month longitudinal analysis in primary care. *J Pain.* 2011;12(9):964-973.
- 16. Bahouq H, Allali F, Rkain H, Hmamouchi I, Hajjaj-Hassouni N. Prevalence and severity of insomnia in chronic low back pain patients. *Rheumatol Int.* 2013;33(5):1277-1281.
- 17. van Tulder MW, Koes BW, Bouter LM. A cost-of-illness study of back pain in The Netherlands. *Pain.* 1995;62(2):233-240.
- 18. Hoy D, Brooks P, Blyth F, Buchbinder R. The Epidemiology of low back pain. *Best Pract Res Clin Rheumatol.* 2010;24(6):769-781.
- 19. Gaskin DJ, Richard P. The economic costs of pain in the United States. *J Pain*. 2012;13(8):715-724.
- 20. Mayer S, Spickschen J, Stein KV, Crevenna R, Dorner TE, Simon J. The societal costs of chronic pain and its determinants: The case of Austria. *PLoS One.* 2019;14(3):e0213889.
- Sleed M, Eccleston C, Beecham J, Knapp M, Jordan A. The economic impact of chronic pain in adolescence: methodological considerations and a preliminary costs-of-illness study. *Pain.* 2005;119(1-3):183-190.
- Treede RD, Rief W, Barke A, et al. Chronic pain as a symptom or a disease: the IASP Classification of Chronic Pain for the International Classification of Diseases (ICD-11). *Pain*. 2019;160(1):19-27.
- 23. Baker M, Collett, B, Fischer, A, et al. Pain Proposal: Improving the Current and Future Management of Chronic Pain. A European Consensus Report. Brussels: Pfizer; 2010.
- 24. European Federation of IASP Chapters. *EFIC's declaration on chronic pain as a major healthcare problem, a disease in its own right.(presented at the European Parliament, Brussels, Belgium)* Brussels, Belgium; 2001, Oct 9.
- 25. Latremoliere A, Woolf CJ. Central sensitization: a generator of pain hypersensitivity by central neural plasticity. *J Pain*. 2009;10(9):895-926.
- 26. Woolf CJ, Salter MW. Neuronal plasticity: increasing the gain in pain. *Science*. 2000;288(5472):1765-1769.
- Meeus M, Vervisch S, De Clerck LS, Moorkens G, Hans G, Nijs J. Central sensitization in patients with rheumatoid arthritis: a systematic literature review. *Semin Arthritis Rheum*. 2012;41(4):556-567.
- 28. Nijs J, Clark J, Malfliet A, et al. In the spine or in the brain? Recent advances in pain neuroscience applied in the intervention for low back pain. *Clin Exp Rheumatol.* 2017;35 Suppl 107(5):108-115.
- 29. Nijs J, Van Houdenhove B, Oostendorp RA. Recognition of central sensitization in patients with musculoskeletal pain: Application of pain neurophysiology in manual therapy practice. *Man Ther.* 2010;15(2):135-141.

- 30. Roussel NA, Nijs J, Meeus M, Mylius V, Fayt C, Oostendorp R. Central sensitization and altered central pain processing in chronic low back pain: fact or myth? *Clin J Pain*. 2013;29(7):625-638.
- Gifford L. Pain, the Tissues and the Nervous System: A conceptual model. *Physiotherapy*. 1998;84(1):27-36.
- 32. LaMotte RH, Shain CN, Simone DA, Tsai EF. Neurogenic hyperalgesia: psychophysical studies of underlying mechanisms. *J Neurophysiol*. 1991;66(1):190-211.
- Koltzenburg M, Torebjork HE, Wahren LK. Nociceptor modulated central sensitization causes mechanical hyperalgesia in acute chemogenic and chronic neuropathic pain. *Brain*. 1994;117 (Pt 3):579-591.
- 34. Torebjork HE, Lundberg LE, LaMotte RH. Central changes in processing of mechanoreceptive input in capsaicin-induced secondary hyperalgesia in humans. *J Physiol.* 1992;448:765-780.
- 35. International Association for the Study of Pain taxonomy, updated from Pain Terms, A Current List with Definitions and Notes on Usage" *Classification of Chronic Pain, Second Edition, IASP Task Force on Taxonomy.* 2012; 209-214.
- 36. Kosek E, Cohen M, Baron R, et al. Do we need a third mechanistic descriptor for chronic pain states? *Pain.* 2016;157(7):1382-1386.
- 37. Woolf CJ, Salter MW. Neuronal plasticity: Increasing the gain in pain. *Science* 2000;288(5472):1765-9
- 38. Staud R. Abnormal endogenous pain modulation is a shared characteristic of many chronic pain conditions. *Expert Rev Neurother.* 2012;12(5):577-585.
- Woolf CJ. Central sensitization: implications for the diagnosis and treatment of pain. *Pain*. 2011;152(3 Suppl):S2-15.
- 40. Kilo S, Schmelz M, Koltzenburg M, Handwerker HO. Different patterns of hyperalgesia induced by experimental inflammation in human skin. *Brain*. 1994;117 (Pt 2):385-396.
- 41. Latremoliere A, Woolf CJ. Central Sensitization: A Generator of Pain Hypersensitivity by Central Neural Plasticity, *J of Pain*. 2009;10(9)895-926.
- 42. Meeus M, Nijs J. Central sensitization: a biopsychosocial explanation for chronic widespread pain in patients with fibromyalgia and chronic fatigue syndrome. *Clin Rheumatol.* 2007;26(4):465-473.
- 43. Lewis GN, Rice DA, McNair PJ. Conditioned pain modulation in populations with chronic pain: a systematic review and meta-analysis. *J Pain.* 2012;13(10):936-944.
- 44. Puentedura EJ, Louw A. A neuroscience approach to managing athletes with low back pain. *Phys Ther Sport.* 2012;13(3):123-133.
- 45. Moseley GL. A pain neuromatrix approach to patients with chronic pain. *Man Ther.* 2003;8(3):130-140.

- 46. Moseley GL. Widespread brain activity during an abdominal task markedly reduced after pain physiology education: fMRI evaluation of a single patient with chronic low back pain. *Aust J Physiother.* 2005;51(1):49-52.
- 47. Peyron R, Laurent B, Garcia-Larrea L. Functional imaging of brain responses to pain. A review and meta-analysis (2000). *Neurophysiol Clin.* 2000;30(5):263-288.
- 48. Flor H. The functional organization of the brain in chronic pain. *Prog Brain Res.* 2000;129:313-322.
- 49. Craggs JG, Staud R, Robinson ME, Perlstein WM, Price DD. Effective connectivity among brain regions associated with slow temporal summation of C-fiber-evoked pain in fibromyalgia patients and healthy controls. *J Pain*. 2012;13(4):390-400.
- 50. Kucyi A, Davis KD. The dynamic pain connectome. Trends in neurosciences. 2015;38(2):86-95.
- 51. Melzack R. From the gate to the neuromatrix. Pain. 1999;Suppl 6:S121-126.
- 52. Melzack R. Pain and the neuromatrix in the brain. J Dent Educ. 2001;65(12):1378-1382.
- 53. Granan LP. We do not need a third mechanistic descriptor for chronic pain states! Not yet. *Pain.* 2017;158(1):179.
- 54. van den Broeke EN. Central sensitization and pain hypersensitivity: Some critical considerations. *F1000Res.* 2018;7:1325.
- 55. Woolf CJ. What to call the amplification of nociceptive signals in the central nervous system that contribute to widespread pain? *Pain.* 2014;155(10):1911-1912.
- 56. Hansson P. Translational aspects of central sensitization induced by primary afferent activity: what it is and what it is not. *Pain.* 2014;155(10):1932-1934.
- 57. Uddin Z, MacDermid JC. Quantitative Sensory Testing in Chronic Musculoskeletal Pain. *Pain Med.* 2016;17(9):1694-1703.
- Neblett R, Hartzell MM, Cohen H, et al. Ability of the central sensitization inventory to identify central sensitivity syndromes in an outpatient chronic pain sample. *Clin J Pain*. 2015;31(4):323-332.
- Nijs J, Torres-Cueco R, van Wilgen CP, et al. Applying modern pain neuroscience in clinical practice: criteria for the classification of central sensitization pain. *Pain Physician*. 2014;17(5):447-457.
- 60. Arendt-Nielsen L, Morlion B, Perrot S, et al. Assessment and manifestation of central sensitisation across different chronic pain conditions. *Eur J Pain*. 2018;22(2):216-241.
- 61. den Boer C, Dries L, Terluin B, et al. Central sensitization in chronic pain and medically unexplained symptom research: A systematic review of definitions, operationalizations and measurement instruments. *J Psychosom Res.* 2019;117:32-40.
- 62. Zimmermann M, Herdegen T. Plasticity of the nervous system at the systematic, cellular and molecular levels: a mechanism of chronic pain and hyperalgesia. *Prog Brain Res.* 1996;110:233-259.

- 63. Staud R. Evidence for shared pain mechanisms in osteoarthritis, low back pain, and fibromyalgia. *Curr Rheumatol Rep.* 2011;13(6):513-520.
- 64. Nijs J, Meeus M, Van Oosterwijck J, et al. In the mind or in the brain? Scientific evidence for central sensitisation in chronic fatigue syndrome. *Eur J Clin Invest*. 2012;42(2):203-212.
- 65. Buchgreitz L, Egsgaard LL, Jensen R, Arendt-Nielsen L, Bendtsen L. Abnormal pain processing in chronic tension-type headache: a high-density EEG brain mapping study. *Brain.* 2008;131(Pt 12):3232-3238.
- 66. de Tommaso M, Federici A, Franco G, et al. Suggestion and pain in migraine: a study by laser evoked potentials. *CNS Neurol Disord Drug Targets*. 2012;11(2):110-126.
- 67. Paul TM, Soo Hoo J, Chae J, Wilson RD. Central hypersensitivity in patients with subacromial impingement syndrome. *Arch Phys Med Rehabil.* 2012;93(12):2206-2209.
- Mease PJ, Hanna S, Frakes EP, Altman RD. Pain mechanisms in osteoarthritis: understanding the role of central pain and current approaches to its treatment. *J Rheumatol.* 2011;38(8):1546-1551.
- 69. Suokas AK, Walsh DA, McWilliams DF, et al. Quantitative sensory testing in painful osteoarthritis: a systematic review and meta-analysis. *Osteoarthritis Cartilage*. 2012;20(10):1075-1085.
- Smart KM, Blake C, Staines A, Thacker M, Doody C. Mechanisms-based classifications of musculoskeletal pain: part 1 of 3: symptoms and signs of central sensitisation in patients with low back (+/- leg) pain. *Man Ther.* 2012;17(4):336-344.
- 71. Smart KM, Blake C, Staines A, Doody C. The Discriminative validity of "nociceptive," "peripheral neuropathic," and "central sensitization" as mechanisms-based classifications of musculoskeletal pain. *Clin J Pain*. 2011;27(8):655-663.
- 72. Malfliet A, Kregel J, Cagnie B, et al. Lack of evidence for central sensitization in idiopathic, non-traumatic neck pain: a systematic review. *Pain Physician*. 2015;18(3):223-236.
- 73. Walton DM, Kwok TS, Mehta S, et al. Cluster Analysis of an International Pressure Pain Threshold Database Identifies 4 Meaningful Subgroups of Adults With Mechanical Neck Pain. *Clin J Pain.* 2017;33(5):422-428.
- 74. Goubert D, Danneels L, Graven-Nielsen T, Descheemaeker F, Meeus M. Differences in Pain Processing Between Patients with Chronic Low Back Pain, Recurrent Low Back Pain, and Fibromyalgia. *Pain Physician*. 2017;20(4):307-318.
- 75. Yunus MB. Central sensitivity syndromes: a new paradigm and group nosology for fibromyalgia and overlapping conditions, and the related issue of disease versus illness. *Semin Arthritis Rheum.* 2008;37(6):339-352.
- 76. Treede RD, Rief W, Barke A, et al. A classification of chronic pain for ICD-11. *Pain.* 2015;156(6):1003-1007.

- Clark J, Nijs J, Yeowell G, Goodwin PC. What Are the Predictors of Altered Central Pain Modulation in Chronic Musculoskeletal Pain Populations? A Systematic Review. *Pain Physician*. 2017;20(6):487-500.
- 78. Zusman M. Forebrain-mediated sensitization of central pain pathways: 'non-specific' pain and a new image for MT. *Man Ther.* 2002;7(2):80-88.
- Lee MC, Zambreanu L, Menon DK, Tracey I. Identifying brain activity specifically related to the maintenance and perceptual consequence of central sensitization in humans. *J Neurosci.* 2008;28(45):11642-11649.
- Brosschot JF. Cognitive-emotional sensitization and somatic health complaints. Scand J Psychol. 2002;43(2):113-121.
- 81. Turk DC, Okifuji A. Psychological factors in chronic pain: evolution and revolution. *J Consult Clin Psychol*. 2002;70(3):678-690.
- 82. Gracely RH, Geisser ME, Giesecke T, et al. Pain catastrophizing and neural responses to pain among persons with fibromyalgia. *Brain.* 2004;127(Pt 4):835-843.
- 83. Hirsh AT, George SZ, Bialosky JE, Robinson ME. Fear of pain, pain catastrophizing, and acute pain perception: relative prediction and timing of assessment. *J Pain*. 2008;9(9):806-812.
- Leeuw M, Goossens ME, Linton SJ, Crombez G, Boersma K, Vlaeyen JW. The fearavoidance model of musculoskeletal pain: current state of scientific evidence. *J Behav Med.* 2007;30(1):77-94.
- 85. Vlaeyen JW, Linton SJ. Fear-avoidance and its consequences in chronic musculoskeletal pain: a state of the art. *Pain.* 2000;85(3):317-332.
- 86. Sullivan MJ, Thorn B, Rodgers W, Ward LC. Path model of psychological antecedents to pain experience: experimental and clinical findings. *Clin J Pain*. 2004;20(3):164-173.
- Evers AW, Kraaimaat FW, van Riel PL, Bijlsma JW. Cognitive, behavioral and physiological reactivity to pain as a predictor of long-term pain in rheumatoid arthritis patients. *Pain*. 2001;93(2):139-146.
- Kuehl LK, Michaux GP, Richter S, Schachinger H, Anton F. Increased basal mechanical pain sensitivity but decreased perceptual wind-up in a human model of relative hypocortisolism. *Pain.* 2010;149(3):539-546.
- McEwen BS, Kalia M. The role of corticosteroids and stress in chronic pain conditions. *Metabolism.* 2010;59 Suppl 1:S9-15.
- Rivat C, Becker C, Blugeot A, et al. Chronic stress induces transient spinal neuroinflammation, triggering sensory hypersensitivity and long-lasting anxiety-induced hyperalgesia. *Pain.* 2010;150(2):358-368.
- 91. Generaal E, Vogelzangs N, Macfarlane GJ, et al. Biological stress systems, adverse life events and the onset of chronic multisite musculoskeletal pain: a 6-year cohort study. *Ann Rheum Dis.* 2016;75:847-854

- 92. Cohen H, Neumann L, Haiman Y, Matar MA, Press J, Buskila D. Prevalence of post-traumatic stress disorder in fibromyalgia patients: overlapping syndromes or post-traumatic fibromyalgia syndrome? *Semin Arthritis Rheum.* 2002;32(1):38-50.
- 93. Sherman JJ, Turk DC, Okifuji A. Prevalence and impact of posttraumatic stress disorder-like symptoms on patients with fibromyalgia syndrome. *Clin J Pain.* 2000;16(2):127-134.
- 94. Sterling M. Differential development of sensory hypersensitivity and a measure of spinal cord hyperexcitability following whiplash injury. *Pain.* 2010;150(3):501-506.
- 95. Edwards RR, Bingham CO, 3rd, Bathon J, Haythornthwaite JA. Catastrophizing and pain in arthritis, fibromyalgia, and other rheumatic diseases. *Arthritis Rheum*. 2006;55(2):325-332.
- 96. Edwards RR, Dworkin RH, Sullivan MD, Turk DC, Wasan AD. The Role of Psychosocial Processes in the Development and Maintenance of Chronic Pain. *J Pain.* 2016;17(9 Suppl):T70-92.
- Sterling M, Hendrikz J, Kenardy J. Compensation claim lodgement and health outcome developmental trajectories following whiplash injury: A prospective study. *Pain*. 2010;150(1):22-28.
- 98. Sterling M, Chadwick BJ. Psychologic processes in daily life with chronic whiplash: relations of posttraumatic stress symptoms and fear-of-pain to hourly pain and uptime. *The Clinical journal of pain*. 2010;26(7):573-582.
- Daenen L, Nijs J, Cras P, Wouters K, Roussel N. Changes in Pain Modulation Occur Soon After Whiplash Trauma but are not Related to Altered Perception of Distorted Visual Feedback. *Pain practice: the official journal of World Institute of Pain.* 2014;14(7):588-598.
- 100. Daykin AR, Richardson B. Physiotherapists' pain beliefs and their influence on the management of patients with chronic low back pain. *Spine (Phila Pa 1976)*. 2004;29(7):783-795.
- Haggman S, Maher CG, Refshauge KM. Screening for symptoms of depression by physical therapists managing low back pain. *Phys Ther.* 2004;84(12):1157-1166.
- 102. Valjakka AL, Salanterä S, Laitila A, Julkunen J, Hagelberg NM. The association between physicians' attitudes to psychosocial aspects of low back pain and reported clinical behaviour: A complex issue. *Scandinavian Journal of Pain*. 2013;4(1):25-30.
- 103. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. *J Physiother*. 2015;61(2):68-76.
- 104. Singla M, Jones M, Edwards I, Kumar S. Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study. *Man Ther.* 2015;20(2):328-334.
- Overmeer T, Linton SJ, Boersma K. Do physical therapists recognise established risk factors? Swedish physical therapists' evaluation in comparison to guidelines. *Physiotherapy*. 2004;90(1):35-41.

- 106. van Wilgen CP, Beetsma A, Neels H, Roussel N, Nijs J. Physical therapists should integrate illness perceptions in their assessment in patients with chronic musculoskeletal pain; a qualitative analysis. *Man Ther.* 2014;19(3):229-234.
- 107. Moseley GL, Butler DS. Fifteen Years of Explaining Pain: The Past, Present, and Future. *J Pain.* 2015;16(9):807-813.
- 108. Buchbinder R, van Tulder M, Oberg B, et al. Low back pain: a call for action. *Lancet.* 2018;391(10137):2384-2388.
- 109. Foster NE, Anema JR, Cherkin D, et al. Prevention and treatment of low back pain: evidence, challenges, and promising directions. *Lancet.* 2018;391(10137):2368-2383.
- 110. Nijs J, Paul van Wilgen CP, Van Oosterwijck J, van Ittersum M, Meeus M. How to explain central sensitization to patients with 'unexplained' chronic musculoskeletal pain: practice guidelines. *Man Ther.* 2011;16(5):413-418.
- 111. Moseley GL, Butler D. S. Explain Pain. Noigroup, Adelaide, 2003, 129 pp. ISBN 097509100 2013.
- 112. Gifford L, Muncey H. Explaining Pain to Patients. Paper presented at the International Association for the Study of Pain. 1999; Vienna, Austria.
- 113. Moseley L. Combined physiotherapy and education is efficacious for chronic low back pain. *Aust J Physiother*. 2002;48(4):297-302.
- 114. Meeus M, Nijs J, Van Oosterwijck J, Van Alsenoy V, Truijen S. Pain physiology education improves pain beliefs in patients with chronic fatigue syndrome compared with pacing and self-management education: a double-blind randomized controlled trial. *Arch Phys Med Rehabil.* 2010;91(8):1153-1159.
- 115. Van Oosterwijck J, Meeus M, Paul L, et al. Pain Physiology Education Improves Health Status and Endogenous Pain Inhibition in Fibromyalgia: A Double-Blind Randomized Controlled Trial. *Clin J Pain.* 2013.
- 116. Van Oosterwijck J, Nijs J, Meeus M, et al. Pain neurophysiology education improves cognitions, pain thresholds, and movement performance in people with chronic whiplash: a pilot study. *J Rehabil Res Dev.* 2011;48(1):43-58.
- 117. Gallagher L, McAuley J, Moseley GL. A randomized-controlled trial of using a book of metaphors to reconceptualize pain and decrease catastrophizing in people with chronic pain. *Clin J Pain.* 2013;29(1):20-25.
- 118. Moseley GL. Evidence for a direct relationship between cognitive and physical change during an education intervention in people with chronic low back pain. *Eur J Pain.* 2004;8(1):39-45.
- 119. Moseley GL, Nicholas MK, Hodges PW. A randomized controlled trial of intensive neurophysiology education in chronic low back pain. *Clin J Pain*. 2004;20(5):324-330.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil*. 2011;92(12):2041-2056.

- Wood L, Hendrick PA. A systematic review and meta-analysis of pain neuroscience education for chronic low back pain: Short-and long-term outcomes of pain and disability. *Eur J Pain*. 2019;23(2):234-249.
- Watson JA, Ryan CG, Cooper L, et al. Pain neuroscience education for adults with chronic musculoskeletal pain: a mixed-methods systematic review and meta-analysis. *J Pain*. 2019;20(10)1140.e1-1140.
- Louw A, Zimney K, Puentedure EJ, et al. The efficacy of pain neuroscience education on musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract*. 2016;32(5)332-55.
- 124. Geneen LJ, Martin DJ, Adams N, et al. Effects of education to facilitate knowledge about chronic pain for adults: a systematic review with meta-analysis. *Syst Rev.* 2015;4:132.
- 125. King R, Robinson V, Ryan CG, Martin DJ. An exploration of the extent and nature of reconceptualisation of pain following pain neurophysiology education: A qualitative study of experiences of people with chronic musculoskeletal pain. *Patient Education and Counseling*. 2016;99:1389-1393.
- 126. Robinson V, King R, Ryan CG, Martin DJ. A qualitative exploration of people's experiences of pain neurophysiological education for chronic pain: The importance of relevance for the individual. *Manual Therapy.* 2016;22:56-61.
- 127. Rees S, Williams A. Promoting and supporting self-management for adults living in the community with physical chronic illness: A systematic review of the effectiveness and meaningfulness of the patient-practitioner encounter. *JBI Libr Syst Rev.* 2009;7(13):492-582.
- 128. Epstein RM, Franks P, Fiscella K, et al. Measuring patient-centered communication in patientphysician consultations: theoretical and practical issues. *Soc Sci Med.* 2005;61(7):1516-1528.
- 129. World Health Organization. WHO global strategy on people-centred and integrated health services: interim report. *World Health Organization.* 2015.
- 130. Stewart M, Brown JB, Donner A, et al. The impact of patient-centered care on outcomes. *J* Fam Pract. 2000;49(9):796-804.
- Dwamena F, Holmes-Rovner M, Gaulden CM, et al. Interventions for providers to promote a patient-centred approach in clinical consultations. *Cochrane Database Syst Rev.* 2012;12:CD003267.
- Plewnia A, Bengel J, Korner M. Patient-centeredness and its impact on patient satisfaction and treatment outcomes in medical rehabilitation. *Patient education and counseling*. 2016;99(12):2063-2070.
- 133. Institute of Medicine (US) Committee on Quality of Health Care in America. *Crossing the Quality Chasm: A New Health System for the 21st Century*. Washington (DC): National Academies Press (US); 2001.

- 134. Davis K, Schoenbaum SC, Audet AM. A 2020 vision of patient-centered primary care. *J Gen Intern Med.* 2005;20(10):953-957.
- 135. Ogden K, Barr J, Greenfield D. Determining requirements for patient-centred care: a participatory concept mapping study. *BMC Health Serv Res.* 2017;17(1):780.
- 136. Gerteis M, Edgman-Levitan, S., Daley, J., Delbanco, T.L. Through the Patient's Eyes: Understanding and Promoting Patient-Centered Care. 2 ed: Wiley; 2002.
- 137. Mead N, Bower P. Patient-centredness: a conceptual framework and review of the empirical literature. *Soc Sci Med.* 2000;51(7):1087-1110.
- Kitson A, Marshall A, Bassett K, Zeitz K. What are the core elements of patient-centred care? A narrative review and synthesis of the literature from health policy, medicine and nursing. *J Adv Nurs.* 2013;69(1):4-15.
- 139. Balint E. The possibilities of patient-centered medicine. J R Coll Gen Pract. 1969;17(82):269-276.
- 140. Cheng L, Leon V, Liang A, et al. Patient-centered care in physical therapy: definition, operationalization, and outcome measures. *Physical Therapy Reviews*. 2016;21(2):109-123.
- Dutch Pain Society. Zorgstandaard Chronische Pijn. http://www.dutchpainsociety.nl/files/ ZorgStandaard_Chronische_Pijn_versie_2017_03_28.pdf. Published 2017.
- 142. Gordon DB, Watt-Watson J, Hogans BB. Interprofessional pain education-with, from, and about competent, collaborative practice teams to transform pain care. *Pain Rep.* 2018;3(3):e663.
- 143. Shaw L, Walker R, Hogue A. The art and science of teamwork: enacting a transdisciplinary approach in work rehabilitation. *Work.* 2008;30(3):297-306.
- 144. Reilly C. Transdisciplinary approach: an atypical strategy for improving outcomes in rehabilitative and long-term acute care settings. *Rehabil Nurs*. 2001;26(6):216-220, 244.
- 145. Vyt A. Interprofessional and transdisciplinary teamwork in health care. *Diabetes Metab Res Rev.* 2008;24 Suppl 1:S106-109.
- 146. Cartmill C, Soklaridis S, David Cassidy J. Transdisciplinary teamwork: the experience of clinicians at a functional restoration program. *J Occup Rehabil.* 2011;21(1):1-8.
- 147. Quaschning K, Korner M, Wirtz M. Analyzing the effects of shared decision-making, empathy and team interaction on patient satisfaction and treatment acceptance in medical rehabilitation using a structural equation modeling approach. *Patient Educ Couns.* 2013;91(2):167-175.

To what extent is patient centeredness examined in physiotherapy and can a theoretical framework be constructed?

Patient-centeredness in physiotherapy: What does it entail? A systematic review of qualitative studies

Amarins J. Wijma^{1,2,3}, Anouck N. Bletterman⁴, Jacqui R. Clark^{3,5}, Sigrid C.J.M. Vervoort⁶, Anneke Beetsma⁷, Doeke Keizer², Jo Nijs^{1,2}, C. Paul van Wilgen^{1,2,3}

Affliations

¹Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium ²Transcare, Transdisciplinary Outpatient Treatment Centre, the Netherlands ³ Pain in Motion research group (www.paininmotion.be)

- ⁴ Fysio Stiens, Stiens, the Netherlands
- ⁵ Faculty of Health Psychology and Social Care, Manchester Metropolitan University, UK.
 ⁶ Department of Internal Medicine and Infectious Diseases, University Medical Centre Utrecht, the Netherlands
- ⁷ Department of Physiotherapy, Hanze University of Applied Sciences, School of Health Studies, Groningen, the Netherlands

Published in: Physiotherapy Theory and Practice, 33 (11), 825-840, Nov 2017

Abstract

Purpose

The literature review aimed at examining and summarizing themes related to patientcenteredness identified in qualitative research from the perspectives of patients and physiotherapists. Following the review, a secondary aim was to synthesize the themes to construct a proposed conceptual framework for utilization within physiotherapy.

Methods

A systematic search of qualitative studies was conducted including all articles up to September 2015. Methodological quality was examined with a checklist. The studies were examined for themes suggestive of the practice of patient-centeredness from the therapists' and/or the patients' perspective. Data were extracted using a data extraction form and analyzed following 'thematic synthesis'.

Results

14 articles were included. Methodological quality was high in five studies. Eight major descriptive themes and four subthemes (ST) were identified. The descriptive themes were: "Individuality" (ST 'Getting to know the patient' and ST 'Individualized treatment'), "Education", "Communication" (ST 'Non-verbal communication'), "Goal setting", "Support" (ST 'Empowerment'), "Social characteristics of a patient-centered physiotherapist", "A confident physiotherapist", and "Knowledge & Skills of a patient-centered physiotherapist".

Conclusions

Patient-centeredness in physiotherapy entails the characteristics of offering an individualized treatment, continuous communication (verbal and non-verbal), education during all aspects of treatment, working with patient defined goals in a treatment in which the patient is supported and empowered with a physiotherapist having social skills, being confident and showing specific knowledge.

Introduction

Healthcare is continuously evolving globally, one reason being the increase in incidence and prevalence of patients with (multiple) chronic diseases. In response to these changes, the complexity of healthcare is continuously expanding and the delivery of healthcare, even with all the advantages, may often be complicated, uncoordinated and unsafe¹. According to the US Institute of Medicine patient-centered care has a potential to address some of these deficits in the healthcare system. Therefore patient-centered care has a high priority in the restructuring of healthcare in the 21st century¹. The federal government of the US has established a Patient-Centered Outcomes Research Institute which underlines their recommendations for changes in healthcare. These recommendations have been developed, however, without patient participation. As Lorig (2012) suggests, *'if a service is to be patient-centered, then both the health care system and the patient have to be involved in determining what this means. Each has its own view of meaning, and patient-centered care will never be achieved if patients are not part of the solution' (p. 524)². This highlights the importance of patient-centeredness in healthcare policymaking today.*

There are many different definitions of patient-centeredness in healthcare. Patientcenteredness was first described in medicine by McWhinney as, 'the physician tries to enter the patients' world, to see the illness through the patients' eyes,'³. Patientcentered healthcare in hospital settings entails eight characteristics of care: respect for the patient's values, preferences, and expressed needs; coordinated and integrated care; clear, high-quality information and education for the patient and family; physical comfort, including pain management; emotional support and alleviation of fear and anxiety; involvement of family members and friends, as appropriate; continuity, including through care-site transitions; and access to care⁴. Probably the most commonly used framework of patient-centeredness in medicine is a model described by Mead and Bower with five interconnecting components: 1) biopsychosocial perspective, 2) the 'patient-as-person', 3) sharing power and responsibility, 4) the therapeutic alliance, and 5) the 'doctor-as-person' ⁵. Patient-centeredness has also been described as a moral philosophy of healthcare professionals to endorse high quality health care⁶.

In physiotherapy however, there is a lack of understanding surrounding the concept of patient-centeredness. Therefore, it is considered important to examine the existing literature on patient-centeredness to assist in developing a deeper understanding of the concepts and implications in physiotherapy. Mead and Bower's framework (2000) uses largely qualitative descriptive studies and it could be argued that qualitative research is the most effective way to provide an in-depth understanding of patient-centeredness perspectives.

As physiotherapists we are healthcare professionals that endorse patients' selfmanagement in which we incorporate the biopsychosocial perspective, by combining functional training for the body and coaching^{7,8}. In medicine it is known that patientcenteredness can strengthen the biopsychosocial perspective by enhancing the relationship (improving empathy, attentiveness and communication) between the healthcare professional and the patient. Furthermore patient-centered medicine shows positive effects on a range of qualitative measures relating to clarifying patients' concerns and beliefs⁹.

For the reasons outlined above, a systematic review of the available qualitative research literature related to patient-centeredness in physiotherapy was conducted. The literature review aimed to 1) examine and summarize themes related to patient-centeredness identified in qualitative research and 2) provide a framework from which to develop applications to physiotherapy. The particular phenomenon of interest was the understanding of patient-centeredness from the perspectives of patients and physiotherapists.

We only included qualitative articles as they allow for seeking meaning and understanding of a phenomenon, in this case patient-centeredness. Following the review, a secondary aim was to synthesize the themes to help construct a conceptual framework describing patient-centeredness for utilization within the context of physiotherapy. Therefore, the research question of this qualitative systematic review is:

To what extent is patient-centeredness examined in physiotherapy in qualitative research and can a theoretical framework be constructed from this research for patient-centeredness in physiotherapy?

Methods

A systematic search (Appendix 1) was conducted in PubMed (MEDLINE), EMBASE, Cochrane, PsychINFO, CINAHL, PEDro and Scopus including articles from 1970 until September 15th 2015. The time span was limited as patient-centeredness was first introduced in 1970¹⁰. In addition, the reference lists of all selected articles were screened for relevant papers not identified through the search. The search was carried out without additional limits. The PICo was used to identify the P-Population (adult patients who received physiotherapy and physiotherapists), the I-Interest (experiences) and Co-Context (physiotherapy in all settings). Based on the PICo the following search terms were used to search each of the trial registers and databases listed above: 'patient centeredness', 'patient centred', 'patient centered', 'patient oriented', 'patient focused', 'physiotherapy', 'physical therapy', 'factors', and 'aspects'. Medical Subject Headings (MESH) terms were used for patient-centered care and physiotherapy. Search terms were combined using AND and OR. Search strategies were peer reviewed by PvW and ANB.

All articles were examined for eligibility by checking the in- and exclusion criteria. Inclusion criteria were: 1) qualitative studies; 2) studies assessing patient-centeredness or aspects of patient-centeredness (or a synonym) in physiotherapy; 3) studies involving rehabilitation mentioning physiotherapy (in that case only the parts/quotes involving physiotherapy were used for this review); 4) articles written in English, Dutch or German.

Exclusion criteria where: 1) studies that examined patient-centeredness only in other medical professions besides physiotherapy; 2) articles that examined patient satisfaction only, as patient satisfaction was considered an outcome of patient-centeredness; 3) articles that involved pediatric physiotherapy (due to the triangle-relationship with children, parents and therapist); 4) studies that examined themes suggestive of the practice of patient-centeredness from the therapists' and/or the patients' perspective. Eligibility assessment of the articles was performed by one researcher (AJW). Duplicates were removed. Retrieved records were first screened on title and abstract.

The reporting of components dealing with methodological quality was assessed by AJW and ANB. A checklist based on three different checklists was created to obtain a complete methodological overview. This checklist was based on the COREQ statement

for qualitative research¹¹, the checklist used by Schoeb et al. ¹² and the checklist of the Britisch Medical Journal (BMJ)¹³. The COREQ contained non-informative items, was dichotomized and supplemented with relevant items of the checklist by Schoeb et al. and the BMJ. The development of the checklist was done by the first researcher (AJW) and reviewed by the second researcher (ANB). The full checklist is displayed in Appendix 2. For each selected paper, all the items included in the checklist were rated as Yes (Y), No (N) or unclear (?) by summing all items scored positive (scored with a Y). According to Veerbeek et al.(2011), a study has low risk for bias when it scores \geq 75% of the maximum score and at high risk for bias when it scores \leq 75%¹⁴.The methodological reviewing of the studies was done independently by AJW and ANB. Cohen's Kappa was used to assess inter-rater agreement between the two researchers assessing the study quality of the included studies¹⁵.

Data were extracted using a data extraction form, (available upon request with the first author) prior to data analysis by one reviewer, AJW. The data extraction form was pilot tested and refined. Information was extracted from each included article on: 1) characteristics of participants; 2) type of study design; 3) findings and 4) special features. Principle summary measures were aspects that describe patient-centeredness. Data synthesis was done following the method of thematic synthesis¹⁶, in which approaches from both meta-ethnography and grounded theory are used for analysis. Before data synthesis, articles were read several times to ensure familiarization with the study. Further to the free line-by-line coding of these studies, performed by the first author (AJW), the resulting 'free codes' were reviewed by ANB and PvW. In case of discrepancy across reviewers, consensus was derived by discussion between the reviewers.

The 'free codes' where then organized into related areas to construct 'descriptive' themes and 'analytical themes'. The development of the descriptive and analytical themes was performed by AJW and SCJMV and later reviewed by PvW, (available upon request). Lastly, a proposed conceptual framework was developed by AJW, ANB and PvW through brainstorm sessions based on the analytical themes and reviewed by all authors. The goal of the proposed conceptual framework to explain the interaction between the themes and to clearly state these connections. Empirical data saturation was reached by consensus between the reviewers.

Results

The flowchart of the study selection is displayed in figure 1. All 14 selected articles were qualitative studies and published in English.



2

Figure 1. Flow of studies through the review.

* Papers may have been excluded for failing to meet more than one inclusion- or exclusion criteria.

Although all the included studies collected qualitative data relevant to patientcenteredness, the methodology varied. The study designs included: grounded theory¹⁷⁻²⁰, nominal group technique²¹, ethnography^{22,23}, a descriptive qualitative approach²⁴, phenomenography²⁵, phenomenology^{26,27}, or no specific design²⁸⁻³⁰.

Study quality was assessed for each study and varied from 40% up to 75% (table 2). Five studies were defined as high quality. The inter-rater agreement between the two researchers assessing the study quality of the included studies was computed and resulted in a Cohen's Kappa of 0.511, p <0.005 which is a moderate agreement^{15,31}.

Although the agreement moderate, after discussion the reviewers agreed on the final study ratings presented in table 2.

The combined number of participants (n=231) across the included studies were recruited through physiotherapy practices and rehabilitation centers. Some studies (N= 7) included physiotherapists, others (N=5) included patients and two studies included both^{20,28} in the data collection. The participants' age ranged from 18 to 84, 4 studies did not report the participants' age^{20,23,28,30}. Data collection methods varied from observations, open interviews, semi-structured interviews, emails, and semistructured focus groups to highly structured focus groups. Study findings varied from a specific aspect of patient-centeredness to a description of patient-centeredness in physiotherapy. In table 1 an overview of study characteristics is provided.

First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
Wikman, 2006	Sweden	6 Patients rehabilitating at three different healthcare centers, 3 month or longer in rehabilitation (age 35-58, 2 male, 4 female) with neurological, circulatory and/ or orthopaedic diseases	In a room at the healthcare (4) center and at home (2)	Grounded Theory	To describe the patient's experiences of influence and participation in the rehabilitation process. Based on patient-centered care	In depth interviews	The parallel process of rehabilitation: the traditional medical model with compliance, sub ordinance and the invisible physiotherapist (at the hospital) and the individual model with being confirmed, sense of coherence, searching for information and daring to demand (within primary healthcare).	Patient's
M.O. Kidd, 2011	New Zealand	8 Musculoskeletal patients (age 20-68, 4 female and 4 male) receiving a maximum of 10 treatments	Workplace (2), home (1) or at the researchers' workplace (5)	Grounded theory	To determine patients' perspectives of components of patient-centred physiotherapy and its essential elements	Semi- structured interviews	Ability to communicate, confidence, knowledge and expertise, understanding people and an ability to relate, transparent focus on progress and outcome	Patient's
K. Cooper, 2007	Scotland, Grampian	25 Chronic low back pain patients (age 18-65, 5 male, 20 female) receiving physiotherapy in the last 6 months	At home or National Health Service hospital (not physiotherapy department)	Framework method of qualitative data analysis	 To define patient- centredness, in the context of physiotherapy for CLBP, from the patient's perspective 	Semi- structured interviews	Communication (most important), individual care, information sharing, the physiotherapist, decision- making, organization of care	Patient's

First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
M. Potter, 2003	Western	26 Current and former patients, no common complaint (age 20-79, mean 48.8 years, 10 male, 16 female) in private practice	Not described	Nominal group technique	To explore patients' perspectives regarding the qualities of a 'good' physiotherapist and to gain insight into the characteristics of good and bad experiences in private practice physiotherapy. Based on patient-centred private sector physiotherapy	Highly structured meeting process (focus group)	Communication ability (interpersonal skills, physiotherapist's manner, teaching/education) , other attributes of the physiotherapist (professional behaviour, organizational ability), characteristics of the service provided by the physiotherapist (diagnostic and treatment expertise, the environment, convenience and accessibility)	Patient's
D. Thomson, 2008	England	5 Physiotherapists working with chronic pain patients, 3-week intensive program (age 24-45, 4 female, 1 male, 2-20 years' experience) in a English National Health Service Hospital	Physiotherapists were shadowed and interviewed in an English National Health Service Hospital	c ritical ethnography	To describe and interpret the interaptists and their patients on a chronic pain unit in an English National Health Service (NHS) hospital from the perspectives of the therapists	observations	Therapist-patient interactions, communication, equality of power, rehabilitation as a risk-taking negotiating process	Therapist's

	(naniii)							
First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
A.B. Rindflesch, 2009	USA	9 Physiotherapists in acute care(3), inpatient (3) and outpatient rehabilitation (3) (age 28-56, 8 female, 1 male, 4-32 years' experience) in an Academic medical center	Onsite observation in an Academic medical center, where focus groups took place is not described	Grounded theory	To describe the practice of patient education in physical therapy among nine physical therapists from three practice areas	Focus groups and observations	Patient education is physical therapy, patient education is empowerment, the content of patient education is patient-centered , outcome of patient education is evaluated trough function	Therapist's
E. Pashley, 2010	Canada, Greater Toronto Area	10 Physiotherapists in outpatient orthopaedics (age 30-62, mean 44.4 years, 8 female, 2 male, 1,5-41 years' experience, mean 18,65 years)	Not described	Descriptive qualitative approach	(1) to describe the relevant factors that physiotherapists take into account in discontinuing treatment of adults in the outpatient orthopaedic setting and (2) to explore how these factors mediate the decision- making process	Key informant interviews and focus groups	Physiotherapists experience, funding source, facilitating self- management, negotiating patient goals and managing expectations, using objective findings, patient education	The rapis t's

Table 1. (Cor	itinued)							
First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
1. Larsson, 2010	Southern	11 Physiotherapists in different areas: orthopaedics, rheumatology, neurology, respiratory diseases and surgery (8 female, 3 male, 1-42 years' experience, median 15 years)	In a room at Health Sciences Centre at Lund University or, in 4 cases, at the respondent's workplace	Phenomenography	 To describe how physiotherapists experience client participation. Based on patient-centered care 	Semi- structured interviews	Collaboration as biopsychosocial client-centered client participation. Guidance as biomedical perspective of client participation, blocks client-centeredness. Expertise as well as biomedical perspective of client participation, but paternalistic and not client- centered.	Therapist's
E. Thornquist, 1991	Norway	Manual, psychomotor and home visiting physiotherapists	First encounters at normal work surroundings (physiotherapists practice) and at home	Not described	What do physiotherapists do to establish a relationship in encounters with patients? And more specifically: How do they relate to their patients through their bodies?	Observations (videos) and interviews	Greeting: note taking: gaze; bodily expression of caring and attentiveness; body position, orientation and closenes; manual therapy- practice: exchange of body messages; psychomotor practice: perception of body relationships	Therapist's

\sim
∇
ā
¥.
2
7
5
0
\cup
\sim
•
<u> </u>
d 1
<u> </u>
0

Table 1. (Con	tinued)							
First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
E. Leach, 2012	Australia, Queenslanc	8 Therapists 8 (occupational, speech and physiotherapy) and 5 stroke patients (age 49-84, 1 female, 4 male) in subacute rehabilitation	Emails	Not described	To examine current clinical approaches to goal-setting through the multiple disciplines of occupational therapy, speech pathology and physiotherapy within one rehabilitation facility. Specifically, it aimed to identify the degree and quality of patient input into the goalsetting process from the perspective of the therapist and compare the therapist' goals using the ICF framework	Semi- structured emails	Goal-setting approaches: therapist controlled, therapist led, patient focused. Goals identified by therapists versus perceived patient goals. Facilitators and barriers	Patient's and therapist's

(Continued)
Table 1.

42

First author, vear	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
F. Trede, 2000	Australia, Sydney	8 Physiotherapists and 7 patients with low back pain	Not described	Grounded Theory	What educational practices are currently applied and what educational theories could inform effective educational practice?	Semi- structured interviews	Professional power and compliance, hands-off attitude versus hands- on technique, the role of pain in education, and transformation from physiotherapist-centred to patient centred approaches	Patient's and therapist's
K. Harman, 2011	Canada, Nova Scotia and British Columbia	44 Physiotherapists from private practice (36 male, 8 female, mean 17.5 years' experience (range: 0.5-38 years)	Not described	Not described	To explore client education provided by physiotherapists in private practice who treat injured workers with subacute low back pain (SA-LBP)	Semi- structured focus groups	The critical importance of education, education: a multidimensional concept, understanding the physiotherapist-client relationship	Therapist's
S. Rutberg, 2013	Sweden	11 Patients with migraine (age 20-69 years, 9 female, 2 men, migraine diagnosis <1-59 years)	At the home or workplace of the participant (6), or were conducted at Luleå University of Technology (5)	Phenomenology	Exploring the lived experience of physical therapy of persons with migraine	Semi- structured interviews	Meeting a physical therapist with professional tools and a personal touch. Investing time and energy to feel better, relying on the competence of the physical therapist, wanting to be treated and to become involved as an individual, being respected in a trustful relationship	Patient's

4 ~

Table 1. (Con	tinued)							
First author, year	Country	Population	Setting	Methodology	Primary aim(s)	Methods	Major themes	Perspective
A. Hiller, 2015	Australia	9 physiotherapists (4 male, 5 female, 1,5-21 years' experience, musculoskeletal, sports, neurological, continence and pelvic floor) 52 patients (15 male, 37 female, age 20-70 years, with spinal pain, work related, knee, sports, peripheral, balance and neurological, women's and chest problems)	In private physiotherapy practices in Melbourne	Ethnographic	To first examine whether and how established models of healthcare communication (practioner-centered and patient -centered are incorporated into one-on-one consultations and second to examine physiotherapists' interpretation and understanding of their clinical communication	Observations, field notes and semi- structured interviews	Observational themes: Focus on physical aspects and pain, a consistent structure, physiotherapists lead the communication, use of casual conversation, touch as communication Physiotherapist interview theme: a sense of purpose	Therapist's

ies.
stud
nded
inclu
f the
ires o
y sco
qualit
ogical
thodol
2. Me
Table

44

No.	. Checklist item	Melander Wikman 2006	Kidd 2011	Cooper 2007	Potter 2003	Thomson 2008	Rindflesch 2009	Pashley 2010	Larsson 2010	Thornquist 1991	Leach 2012	Trede 2000	Harman 2011	Rutberg 2013	Hiller 2015
	Was the researcher	∼.	~.	~:	~:	~	c.	~	~:	~:	<u>с.</u>	~.	~:	~:	~.
	experienced or trained? *														
7	Was the	~	~	~	~	~	~	~	~	~	~	~	z	~	~
	question clearly														
	defined? \$														
\sim	Was the	~	~	~	z	~	~	Z	~	ć	z	~	<.	~	~
	methodological														
	orientation														
	suitable for														
	this research														
	question? *														
4	Was theoretical	≻	~	~	≻	~	~	~	~	ć	~	z	~	~	~
	or purposeful														
	sampling used? *														
ŝ	Was there	Z	z	~	z	z	~	z	~	Z	z	~	~	~	z
	stated														
	how many														
	participants														
	where														
	approached? *														

Table 2. (Continued)

Tal	ble 2. (Continued	()													
2°	. Checklist item	Melander Wikman 2006	Kidd 2011	Cooper 2007	Potter 2003	Thomson 2008	Rindflesch 2009	Pashley 2010	Larsson 2010	Thornquist 1991	Leach 2012	Trede 2000	Harman 2011	Rutberg 1 2013	Hiller 2015
9	Were the	~	>	~	~	~	~	~	~	z	>	z	>	~	
	important														
	characteristics														
	of the sample														
	described? *														
	Does the	~	~	~	∼.	×	z	~	z	~	~	~	~	~	
	sample produce														
	the type of														
	knowledge														
	necessary to														
	understand														
	the structures														
	and processes														
	within which the														
	individuals or														
	situations are														
	located? #														
00	Was there	~	~	~	~	×	×	~	~	×	~	~	~	´ ≻	
	stated that the														
	interview was														
	open, semi														
	structured or														
	if there were														
	focus groups? *							ĥ							

able 2. (Continued)	
•••	able 2. (

46

Š	Checklist item	Melander Wikman	Kidd 2011	Cooper 2007	Potter 2003	Thomson 2008	Rindflesch 2009	Pashley 2010	Larsson 2010	Thornquist 1991	Leach 2012	Trede 2000	Harman 2011	Rutberg 2013	Hiller 2015
		2006													
6	Were repeated	z	z	z	~	~	~	~	z	~:	>	z	z	z	>
	interviews carried out? *														
10	Were field notes	¢.	~	z	z	~	,	~	z	~	z	<i>c</i> .	2	¢.	~
	made? *														
1	Was data	~	~	z	~	Z	z	7	z	z	z	z	z	z	~
	saturation														
	discussed/														
	reached? *														
12	Were there	~	~:	~	~	Z	~	~	z	Z	~	ć	z	~	z
	two or more														
	researchers														
	that coded the														
	data?*														
10	Was software	ć:	~	~	z	~	z	~	z	Z	~	<.	~	~	~:
	used to manage														
	the data?*														
4	Did themes	~	~	~	~	~	~	~	~	~	~	~	~	~	~
	derive from the														
	data?*														
15	Were	~	~	~	~	~	~	~	~	~	≻	~	×	~	~
	participant														
	quotations														
	presented														
	to illustrate														
	themes/items?*														
				*			*								

Table 2. (Continued)

								:							
No.	. Checklist item	Melander Wikman 2006	- Kidd 2011	Cooper 2007	Potter 2003	Thomson 2008	Rindflesch 2009	Pashley 2010	Larsson 2010	Thornquist 1991	Leach 2012	Trede 2000	Harman 2011	Rutberg 2013	Hil 201
16	Were major	~	~	~	z	~	~	~	~	~	>	>	~	>	>
	themes clearly	ſ													
	findings?*	D													
17	Is the	~	~	~	z	~	z	~	Z	~	z	~	~	~	~
	description														
	described in														
	sufficient detail														
	to allow the														
	researcher or														
	the reader to														
	interpret the														
	meaning and														
	context of														
	what is being														
	researched? #														
00	Does the	~	≻	~	z	~	~	~	~	~	~	~	~	~	\succ
	researcher														
	move from														
	description														
	of the data,														
	through														
	quotations or														
	examples, to														
	an analysis and														
	interpretation														
	of their														
	meaning and														
	significance? #														

47

 Are claims beir made for the generalizability of the findings to other bodie of knowledge? (within scientif research) # 	Wikman 2006	2011	2007	2003	Thomson 2008	Rindflesch 2009	Pashley 2010	Larsson 2010	Thornquist 1991	Leach 2012	Trede 2000	Harman 2011	Rutberg 2013	Hiller 2015
in Aro chaime hoir	Z S C	>	z	z	z	z	z	Z	z	z	z	~	~	z
zo Are claims wern made for the generalizability of the findings to other populations? #	N	Z	Z	Z	Z	z	>	>	z	Z	Z	>	>	>
Overall quality in % *= original from the Y= described in the Overall quality in %.	65 COREQ statei article/good q	75 ment ¹¹ , # µuality, N⊧	70 =original fr =definitely	45 om V. Sch not descr	75 loeb et al. ¹² , § ibed or poor	55 s=original frc quality, ? = r	75 m the Britis iot clearly d	55 .h Medical J escribed in	40 ournal quality the article if it	60 checklist : is done (50 : ¹³ or not	65	75	75

In the descriptive analysis 13 descriptive themes were found. During the analytical analysis phase these were gathered into 8 major descriptive themes and 4 subthemes (ST) (two subthemes were conjoined) described below and in the proposed conceptual framework (figure 2). The descriptive themes were:

- The concept of individuality in patient-centeredness ST (1) 'Getting to know the patient'; ST (2) 'Individualized treatment';
- 2. Continuous tailored communication in lay speech
- 3. ST(3) 'Non-verbal communication'
- 4. Education during and about all aspects of the treatment
- 5. Working with patient-defined goals
- 6. A patient-centered treatment in which the patient is supported
- 7. ST(4) 'Empowerment'
- 8. Social characteristics of a patient-centered physiotherapist
- 9. A confident physiotherapist
- 10. Knowledge and skills of a physiotherapist in patient-centeredness

Individuality:

Individuality was found in all of the articles and was both from the patient's and the therapist's perspective referred to as important. This concerned specific patient-tailored education, communication and treatment. Subthemes were 'getting to know the patient' and 'individualized treatment'.

Subtheme: Getting to know the patient

It was found that both patients and physiotherapists believed that getting to know the patient as a person was important for individualization in physiotherapy. This involved getting to know patients' history, needs, preferences, personality, beliefs, values, expectations, motivation, and circumstances^{17,18,21,24-28,30} and remembering them. Patients appreciated being seen as an integration of body and soul¹⁸ and knowing the patient as a person was an essential part of this integration.

Subtheme: Individualized treatment

Patients wanted themselves, rather than the techniques, to be in the center of concern²⁰. An individualized treatment involved an individualized treatment plan so patients can learn independently²⁰, including exercises, advice and education that was

Table 2. (Continued)

composed in dialogue and collaboration with the patient^{19,24,26,27,30}. During the treatment the therapist had to be aware of the changing needs of the patient²⁶. The exercises and given advice affected patient adherence²⁷, suggesting that patient-centeredness required the physiotherapist to ensure that the patient experienced the exercises as important and individualized²⁰. Adjustments made by the physiotherapist in response to patients' feedback was experienced as important^{20,27}. Not only the content of the treatment should be individualized, but the delivery of treatment as well²⁷.

Communication

Both therapists' and patients mentioned communication as a part of patientcenteredness in all the articles. The most important aspect of communication was the need of an ongoing dialogue with patients. Moreover, the communication style should be tailored to the individual patient in clear and lay speech^{17,20,23,24,26,27}. This required openness of the therapist about themselves and the therapy, and ultimately created safety for the patient to open up^{20,26}. Personal communication and communication skills were far more important than the provision of scientific facts²⁰. By personal communication a bond was established and the therapy shifted from therapist to patient-centered²³.

Communicative abilities of a patient-centered physiotherapist meant being receptive to what the patient has to say, correctly interpreted and giving explanations in a way patients understand^{15,20}. Purposefully changing communication styles depending on the patient²³. Having the ability to explain in lay terms, directly speaking to the patient, listening, and asking appropriate questions was of importance^{17,21,24,27}.

Subtheme: Non-verbal communication

Non-verbal communication incorporated eye contact, nodding and facial expressions^{23,29,30}. This indicated interest into the patient, availability for contact and made sure the patient perceives the contact as 'being seen'²⁹. Furthermore, therapists' used their own body language and facial expression, as well as that of the patient, to establish a bond and reflect if it was indeed established^{23,30}.

Both the patients' and the therapists' experienced non-verbal communication as consisting of physiotherapists body movements. It comprised of using their hands, touch, cushions for comforting, and creating a trustful body language^{23,26,29}. Furthermore

non-verbal communication comprised of active listening to the patient and making sure that the patient was aware of this active listening^{20,21,27,29,30}.

Non-verbal communication created a sense of being respected^{23,26}, caring for the patient^{23,29}, demonstrating empathy, respect, consideration, made the patient feel at ease^{17,23}, and created room for emotions.

Education

All studies mentioned education as related to patient-centeredness. Education was mentioned as explanation about physical symptoms, the problem, intake, diagnosis, treatment, and treatment course. The content taught during education should be useful and focused on the patient's problems¹⁷. Visualizing, using metaphors and demonstrating towards the patient was found to be constructive in patient-centered education^{17,21}. Education was more than simplifying in plain language: the information had to be compatible with patients' reality, perceptions and be meaningful²⁰.

There was an interaction of this theme with Social characteristics, Communication, Individuality and Goal setting, as the content in the education should be interactively communicated in a manner that patients understand and tailored on the patients' needs and goals^{17,19,21,24,27}. Written education was not perceived as individualized and patient-centered by patients²⁷.

Goal setting

Goal setting was used by physiotherapists to activate and motivate patients, to determine what meaningful therapy would be for the patient and to set discharge limits^{19,22,24,28}. Goal setting seemed particular of physiotherapists' interest, as patients did not spontaneously mention goal setting as important for patient-centered physiotherapy. Patient-centered physiotherapists, however, tried to allow the patients to define their own goals in collaboration^{20,22,24,25}. This was done by facilitating them and guiding them, using education and dialogue to determine the patients' goals^{19,20,22,25,28}. Goals were mostly created in collaboration between the physiotherapist and the patient^{20,28}. However, some physiotherapists made no or little mention of patient-centered goals²⁴.

Support

Support from the physiotherapist consisted of a mixture of individuality, equality of responsibility, understanding, feeling important, reassuring, and empowerment^{17-22,24-27,30}. Patients valued the feeling of a physiotherapist having their back, by supporting them, relating to them and seeing them as a complete person¹⁸.

Being supportive in patient-centered physiotherapy demanded an interaction with the descriptive themes social characteristics, individualization, communication and education. As a physiotherapist could not be supportive until he knew and understood the patient (individualization). This support was established by verbal and nonverbal communication, such as touch and educating the patient. This supported and empowered the patient. This empowerment, however, could not be accomplished without the social characteristics of a patient-centered physiotherapist.

Subtheme: Empowerment

Patient-centered empowerment was mentioned as a personal feeling by the patient, where the physiotherapist tries to give responsibility and power to the patient^{18,22,30}. Strengthening of the empowerment was mostly done by touch²³, education or showing improvements in symptoms and functions^{17,19,20}. Furthermore, counseling (exploration of choices, support, encouragement and back-up) was an applied strategy^{18,20}. Being able to make an appointment quickly made patients feel empowered and helped them with coping²⁶. Physiotherapists strived for optimal patient empowerment²².

Social characteristics of a patient-centered physiotherapist

Patients described the social characteristics of a patient-centered physiotherapist as respectful, non-judgmental, non-egotistical with an open interested attitude and mind^{17,21,22,24-26}. Physiotherapists should be honest about his/her limitations and reflective of his/her own behaviour and emotions^{21,30}, put the patient's needs first and build a trusting relationship and rapport with the patient^{17,20,22,26}. This involved being friendly, supportive, considerate, patient, genuine, polite, positive, caring for the patient, having the ability to care for the patient, taking the patient seriously, believing in the patient, recognition of the patients' emotions, making a commitment to the patient and making the best effort^{17,20,22,23,26,27,30}. In essence, the therapist should understand the patient and relate to them³⁰.

However, perceiving the therapist as 'being nice' was not the only aspect of a patientcentered approach²⁷. In addition, communicative abilities of the physiotherapist were judged as important²⁶. These abilities are mentioned in the theme *"Communication"*.

Although patients appreciated getting to know the person behind the physiotherapist²⁶, a professional distance and professionalism should be maintained, as well as dedication to the profession^{17,21,26,27}. From these examples it may be suggested that patient-centeredness is all about the role the physiotherapist adopts to place the patient at the center of the treatment.

A confident physiotherapist

Both patients and physiotherapists underlined the importance of a confident physiotherapist. Besides a confident physiotherapist it was also acknowledged that the physiotherapist should inspire confidence in the patient¹⁷. Confident body language and verbal communication, and confidence in explaining to the patient were described as key ingredients^{17,26,27}. Feeling the confidence of the therapist in his/her treatment inspired confidence in the patient¹⁷ and decreased worries and fears²⁶. Furthermore, patients felt that the physiotherapist should feel confident enough to discuss any issues with their patients^{21,30}. The underlying concepts and behaviours of a confident physiotherapist were not explained in any of the studies.

Knowledge and skills of a physiotherapist in patient-centeredness

The physiotherapist should be competent enough to deal with the patient's specific disorder²⁷ and this is not only achieved by keeping skills and knowledge up to date, but also by using this knowledge and expertise with good teaching skills^{17,21,22,27}. Knowledge should be disease specific, contain-familiarity with body dysfunction and include the understanding of the patient's perspective. Besides, the therapist should have a very good understanding of the patient in order to tailor treatment^{22,25,28}.

Interestingly, physiotherapists found the greater their experience and maturity within the profession, the more they felt able to apply patient-centeredness^{19,21,24}. This may be associated with increased confidence but how experience, maturity and patient-centeredness were related was not described in detail.

Patients valued the input of physiotherapist's knowledge by means of the physiotherapist being the expert¹⁷, however did not specify this knowledge. Patients

wanted to have clear explanations, but also desired the ability to make their own or shared decisions²⁷.



Figure 2: Proposed framework for patient-centeredness in physiotherapy

The proposed conceptual framework, figure 2, was based on brainstorm sessions and consensus with multiple authors (AJW, ANB and PvW) and reviewed by all authors. During the analysis the authors discovered that patient-centeredness in physiotherapy is a dynamic concept with closely related themes and subthemes.

The analysis and brainstorm sessions uncovered that there is a difference in the themes we found: there are themes related to the physiotherapist characteristics and there are themes related to the patient-physiotherapist interaction. Figure 2 is designed according to these two differences.

The themes related to the patient-physiotherapist interaction (Individuality, Communication, Education, Goal setting, Support) are located on the left side of the proposed conceptual framework because our writing directions is from left to right, thereby suggesting that these themes are preceding the themes related to the physiotherapist characteristics. The themes (Individuality, Communication, Education, Goal setting, Support) are all of equal importance, connected, intertwined and all have an influence on each other.

Even though the patient is the most important in patient-centeredness , the physiotherapists in itself plays an important role: he/she is the one who places the patient in the center. Furthermore, the physiotherapist and his/her behaviour (Social

characteristics, Knowledge and Skills, Confidence) influences all other themes: the individuality of the therapy, communication, education, goal setting and support.

Discussion

This review identified 14 articles from qualitative studies investigating patientcenteredness in physiotherapy. From these studies a synthesis of interrelated themes ("individuality", "communication", "education", "goal setting", "support", "social characteristics of a patient-centered physiotherapist", "a confident physiotherapist", "knowledge and skills of a physiotherapist in patient-centeredness") with subthemes and a proposed conceptual framework (figure 2) of patient-centeredness in physiotherapy was made. All relevant articles related to patient-centeredness in physiotherapy were included. The findings from this review may be used to educate students and during the continuous education of clinicians. Whereby the proposed conceptual framework may be an indication and example of how the different themes interact and relate to each other: patientcenteredness in physiotherapy entails the characteristics of offering an individualized treatment, continuous communication (verbal and non-verbal), education during all aspects of treatment, working with patient-defined goals, a treatment in which the patient is supported and empowered and a physiotherapist with patient-centered social skills, confidence and knowledge.

"Individuality" concerns specific patient-tailored education, communication and treatment. "Communication" is the need for a continuous individualized dialogue with patients in clear and lay speech. When doing so patient satisfaction and therapeutic alliance improves³². Physiotherapist should be aware of these communication needs and require training during and after physiotherapy education^{33,34}. "Education" primarily involves advice about the problem, diagnosis, treatment and treatment course. "Goal setting" is used by physiotherapists to activate and motivate patients, however, was not spontaneously mentioned by patients. "Support" from the physiotherapist is seen as a mixture of individuality, equality of responsibility, understanding, reassuring and empowerment. "The social characteristics", "confidence" and "skills and knowledge" of a patient-centered physiotherapist are personal skills and encompass for instance: being able to relate to the patient, confident body language, up to date knowledge and teaching skills. This theme can be used to create awareness among physiotherapist and offers the opportunity to physiotherapists to reflect upon their attitude and behaviour.

The concepts of this review are to some extent similar to previous frameworks constructed for patient-centeredness in overall care (The Picker Institute's principles⁴), medicine^{5,6} and nursing³⁵. For instance, in all reviews individuality of the patient (*i.e. the patient as a person⁵ and respect for patients' values, preferences and expressed needs*⁴) was identified as important, which in our review was the largest theme. Furthermore, both Mead and Bower, as well as Epstein et al. included 'sharing power and responsibility' in their framework^{5,6}. This is to some extent similar to "Support" in our review. 'Patient participation and involvement' and 'the relationship between the patient and the healthcare professional' from the review of patient-centeredness in nursing³⁵ are also well represented in the themes identified in the present review, highlighting the importance of these two topics in both professions.

Unlike the frameworks in overall care, medicine and nursing, the setting/organization was not an important part of patient-centeredness in physiotherapy. The Picker Institute's principles mention the '*Involvement of family and friends'*, '*transition and continuity' and 'coordination and integration of care*⁴. In nursing '*the context where care is delivered*' implied the environment, such as policy, equipment, lack of time and deeper philosophical issues within the nurse and team³⁵. This discrepancy might be due to the different settings and hence may reflect true differences. Both, the Picker Institute and nursing frameworks are based on thorough investigations of patient-centeredness in hospitals^{4,35}, whereas most of the patients and physiotherapists from the original studies in this review work in a private practice and (sub-acute) rehabilitation settings.

From the above reflection of this review and prior research on patient-centeredness in overall care, nursing and medicine it can be argued that there are similarities as well as differences between the models. The variance between these models might reflect on true dissimilarities between the professions and settings, hence represent various forms of patient-centeredness. Therefore, it is proposed that there are distinct needs of patient-centeredness in physiotherapy compared to overall care, nursing and medicine due to professional differences. As a result, this review and proposed conceptual framework is an enhancement on prior research in overall care, nursing and medicine as it is specific for physiotherapy.

The findings of this review are also comparable to the findings of Edwards et al. about clinical reasoning strategies in physiotherapy³⁶. Their extensive grounded theory study reveals several conceptual frameworks (*'clinical reasoning strategies, cue-based*)

combining of reasoning strategies and interplay of reasoning strategies in different paradigms of knowledge generation') with subcategories³⁶. Even though their study was based on defining clinical reasoning strategies, their constructs overlap with ours. This indicates that patient-centeredness and clinical reasoning are closely connected.

There are also comparisons between the review and a recent qualitative review of O'Keefe et al. on patient-therapist interactions in musculoskeletal therapy³⁷. Whereby they found the following themes: 'physiotherapists interpersonal and communication skills' (listening, encouragement, confidence, being empathetic and friendly, and nonverbal communication), 'practical skills' (expertise and level of training, although the ability to provide good education was considered as important only by patients), 'individualized patient-centered care' (individualizing the treatment to the patient and taking patient's opinions into account), 'and organizational and environmental factors' (i.e., time and flexibility with care and appointments). Even though their aim was based on outcomes: to investigate the factors that influence the patient-physiotherapist interactions, and the aim of this review on determinants: creating a synthesis of patient-centeredness, the themes of both reviews are comparable. Suggesting that maybe in which manner a physiotherapist works patient-centered affects the outcomes of the interactions between the patient and physiotherapist. While their search terms were different both reviews included 4 articles that are the same (17,21,27,30) suggesting a great deal of overlap between the different constructs. The difference to their review and the current review, besides the focus, is that they included studies focusing on satisfaction and excluded studies that focused on physiotherapy in a rehabilitation setting. Whilst this review excluded studies focusing on satisfaction because it was suggested that satisfaction is an outcome of patient-centeredness rather than a base/determinant. Furthermore, this review included all settings and by that created an overall synthesis of patientcenteredness based on all settings in physiotherapy.

Study limitations

Due to the limited number of available studies, we included several different qualitative study designs in this review. There is a debate ongoing about combining study designs in qualitative reviews. However, the use of multiple methodologies can increase the understanding of the phenomenon/process, can compensate the limitations of individual methods³⁸, and exclusion based on qualitative methodology diminishes insight in the research topic³⁹.

We included articles that either assessed patient-centeredness or aspects of patientcenteredness (or a synonym) in physiotherapy. As a result, the primary aim of the studies included were not all based on assessing patient-centeredness. However, all studies mentioned patient-centeredness in their full text. They either had aims based on patient-centered care, used patient-centeredness as an outcome of their results, or reflected on their findings in the light of previous definitions of patient-centered care.

Within qualitative research there is debate about the preferred techniques one can use to assess the methodological quality of individual studies. For example, about saturation (included as number 11 in the methodological checklist). Saturation is a technique whereby researchers stop data collecting when no new information emerges from the data that will add to the understanding of the phenomenon under study⁴⁰). Within GT it is mentioned that data saturation is usually reached between 20-30 interviews⁴⁰. However, other researchers suggest saturation as a method to obtain methodological quality may be inapplicable⁴⁰.

The inter-rater agreement between the two researchers assessing the study quality of the included studies was 'moderate'^{14,30}. During the analysis we decided to not bring a third reviewer forward due to practical implications, however we did reach consensus on the final scoring.

One could argue that within the profession of physiotherapy many differences exist between monodisciplinary and multidisciplinary physiotherapy. Six of the 13 included articles in this review conducted research in acute or sub-acute rehabilitation, with the focus on physiotherapy. Therefore, it can be assumed that this review gives a complete overview of patient-centeredness in the different areas of physiotherapy (except for pediatric physiotherapy).

More sound qualitative research on this topic should be performed to further investigate in which manner and to what extent patient-centeredness is implemented in clinical practice. Hiller et al. were, to our knowledge, the first to investigate this with observations and interviews and found that physiotherapists' approach are more likely to be therapist-centered than patient-centered²². Qualitative research should further enhance our understanding about the perceptions of physiotherapists of patient-centeredness, see if there are differences between conditions (for instance between

non-life-threatening conditions, chronic conditions and conditions in which the patient cannot clearly communicate) and how to implement patient-centered strategies in clinical practice. These qualitative studies should contain patient-centeredness or a well-defined synonym in the title or key words to ease the search of qualitative articles⁴¹.

Additionally, the present overview calls for quantitative research to study the implementation and implications of working patient-centered in physiotherapy practice according to the provided description and framework. Not only does research show that patient-centered medicine has positive effects on clarifying patients' concerns and beliefs⁸, patient-centered medical care also reduces costs by lowering unnecessary diagnostic tests and referrals⁴². This increased effectiveness might also occur in physiotherapy and is worth studying further.

Our findings show a better understanding of the concept patient-centeredness in adult patients. This model however cannot be generalized to all health in physiotherapy conditions, for instance in patients with acute stroke or in patients with dementia of young children. Further research may focus on potential models of patient-centered strategies in these patient groups.

Conclusion

Patient-centeredness in physiotherapy is a framework containing multiple closely related themes: "individuality"; "communication"; "education"; "goal setting" and "support"; "the social characteristics", "confidence" and "skills and knowledge of a patient-centered physiotherapist". The results presented in this review provide insights into patient-centeredness in physiotherapy. A proposed conceptual framework is constructed to help physiotherapists improve their understanding of patient-centeredness. It is hoped the proposed conceptual framework developed from these study findings will assist physiotherapists in their understanding of patient-centeredness and the implications of patient-centeredness in clinical practice. Further research is needed in order to further enhance our understanding about the clinical applicability of the proposed conceptual framework and to assess the implementation and implications.

References

- Institute of Medicine (US) Committee on Quality of Health Care in America. Crossing the quality chasm: a new health system for the twenty first century. In: Washington: National Academies Press; 2001.
- 2. McWhinney IR. 'An acquaintance with particulars...'. Fam Med. 1989;21(4):296-298.
- 3. Gerteis M, Edgman-Levitan, S., Daley, J., Delbanco, T.L. *Through the Patient's Eyes: Understanding and Promoting Patient-Centered Care*. 2 ed: Wiley; 2002.
- 4. Mead N, Bower P. Patient-centredness: a conceptual framework and review of the empirical literature. *Soc Sci Med.* 2000;51(7):1087-1110.
- 5. Epstein RM, Franks P, Fiscella K, et al. Measuring patient-centered communication in patient-physician consultations: theoretical and practical issues. *Soc Sci Med.* 2005;61(7):1516-1528.
- Bandura A. Self-efficacy: toward a unifying theory of behavioral change. *Psychol Rev.* 1977;84(2):191-215.
- Bandura A, Adams NE, Beyer J. Cognitive processes mediating behavioral change. J Pers Soc Psychol. 1977;35(3):125-139.
- Dwamena F, Holmes-Rovner M, Gaulden CM, et al. Interventions for providers to promote a patient-centred approach in clinical consultations. *Cochrane Database Syst Rev.* 2012;12:CD003267.
- Balint M. Repeat prescription patients: are they an identifiable group? *Psychiatry in medicine*. 1970;1(1):3-14.
- 10. Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *International journal for quality in health care : journal of the International Society for Quality in Health Care / ISQua*. 2007;19(6):349-357.
- 11. Schoeb V, Burge E. Perceptions of patients and physiotherapists on patient participation: a narrative synthesis of qualitative studies. *Physiotherapy research international : the journal for researchers and clinicians in physical therapy.* 2012;17(2):80-91.
- 12. British Medical Journal Checklist. Available at: http://www.bmj.com/about-bmj/resourcesauthors/article-types/research/editors-checklists.
- 13. Veerbeek JM, Van Wegen EE, Harmeling-Van der Wel BC, Kwakkel G. Is accurate prediction of gait in nonambulatory stroke patients possible within 72 hours poststroke? The EPOS study. *Neurorehabilitation and neural repair*. 2011;25(3):268-274.
- 14. Fleiss JL, Cohen J. The Equivalence of Weighted Kappa and the Intraclass Correlation Coefficient as Measures of Reliability. *Educational and Psychological Measurement*. 1973;33(3):613-619.

- 15. Thomas J, Harden A. Methods for the thematic synthesis of qualitative research in systematic reviews. *BMC medical research methodology*. 2008;8:45.
- Kidd MO, Bond CH, Bell ML. Patients' perspectives of patient-centredness as important in musculoskeletal physiotherapy interactions: a qualitative study. *Physiotherapy*. 2011;97(2):154-162.
- 17. Melander Wikman A, Fältholm Y. Patient empowerment in rehabilitation: "Somebody told me to get rehabilitated". *Advances in Physiotherapy*. 2006;8(1):23-32.
- 18. Rindflesch AB. A grounded-theory investigation of patient education in physical therapy practice. *Physiotherapy theory and practice*. 2009;25(3):193-202.
- Trede FV. Physiotherapists' Approaches to Low Back Pain Education. *Physiotherapy*. 2000;86(8):427-433.
- 20. Potter M, Gordon S, Hamer P. The physiotherapy experience in private practice: the patients' perspective. *The Australian journal of physiotherapy*. 2003;49(3):195-202.
- 21. Thomson D. An ethnographic study of physiotherapists' perceptions of their interactions with patients on a chronic pain unit. *Physiotherapy theory and practice*. 2008;24(6):408-422.
- 22. Hiller A, Guillemin M, Delany C. Exploring healthcare communication models in private physiotherapy practice. *Patient Educ Couns.* 2015;98(10):1222-1228.
- Pashley E, Powers A, McNamee N, Buivids R, Piccinin J, Gibson BE. Discharge from outpatient orthopaedic physiotherapy: a qualitative descriptive study of physiotherapists' practices. *Physiotherapy Canada Physiotherapie Canada*. 2010;62(3):224-234.
- Larsson I, Liljedahl K, Gard G. Physiotherapists' experience of client participation in physiotherapy interventions: A phenomenographic study. *Advances in Physiotherapy*. 2010;12(4):217-223.
- 25. Rutberg S, Kostenius C, Ohrling K. Professional tools and a personal touch experiences of physical therapy of persons with migraine. *Disabil Rehabil.* 2013;35(19):1614-1621.
- 26. Cooper K, Smith BH, Hancock E. Patient-centredness in physiotherapy from the perspective of the chronic low back pain patient. *Physiotherapy*. 2008;94(3):244-252.
- 27. Leach E, Cornwell P, Fleming J, Haines T. Patient centered goal-setting in a subacute rehabilitation setting. *Disability and rehabilitation*. 2010;32(2):159-172.
- 28. Thornquist E. Body Communication is a Continuous Process: The First Encounter between Patient and Physiotherapist. *Scandinavian Journal of Primary Health Care*. 1991;9(3):191-196.
- 29. Harman K, Bassett R, Fenety A, Hoens AM. Client Education: Communicative Interaction between Physiotherapists and Clients with Subacute Low Back Pain in Private Practice. *Physiother Can.* 2011;63(2):212-223.
- 30. Landis JR, Koch GG. The measurement of observer agreement for categorical data. *Biometrics*. 1977;33(1):159-174.

- 31. Oliveira VC, Refshauge KM, Ferreira ML, et al. Communication that values patient autonomy is associated with satisfaction with care: a systematic review. J Physiother. 2012;58(4):215-229.
- 32. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. / Physiother. 2015;61(2):68-76.
- 33. Murray A, Hall AM, Williams GC, et al. Effect of a self-determination theory-based communication skills training program on physiotherapists' psychological support for their patients with chronic low back pain: a randomized controlled trial. Arch Phys Med Rehabil. 2015;96(5):809-816.
- 34. Kitson A, Marshall A, Bassett K, Zeitz K. What are the core elements of patient-centred care? A narrative review and synthesis of the literature from health policy, medicine and nursing. 1 Adv Nurs. 2013;69(1):4-15.
- 35. Edwards I, Jones M, Carr J, Braunack-Mayer A, Jensen GM. Clinical reasoning strategies in physical therapy. Phys Ther. 2004;84(4):312-330; discussion 331-315.
- 36. O'Keeffe M, Cullinane P, Hurley J, et al. What Influences Patient-Therapist Interactions in Musculoskeletal Physical Therapy? Qualitative Systematic Review and Meta-Synthesis. Phys Ther. 2016;96(5):609-622.
- 37. Paterson B.L. TSE, Canam C., Jillings C. . Meta-Study of Qualitative Health Research. A Practical Guide to Meta-Analysis and Meta-Synthesis. SAGE Publications Ltd; 2001.
- Booth A. Cochrane or cock-eved? how should we conduct systematic reviews of qualitative 38. research? Paper presented at the Qualitative Evidence-based Practice Conference, Taking a Critical Stance. Coventry University, May 14-16 2001
- Creswell JW. Qualitative Inquiry and Research Design, choosing among five approaches. Second 39. ed. London: Sage Publications, Inc. 2007.
- 40. O'Reilly M, Parker N. 'Unsatisfactory Saturation': a critical exploration of the notion of saturated sample sizes in qualitative research. Qualitative Research. 2013;13(2):190-197.
- 41. Jones ML. Application of systematic review methods to qualitative research: practical issues. Journal of advanced nursing. 2004;48(3):271-278.
- 42. Stewart M, Brown JB, Donner A, et al. The impact of patient-centered care on outcomes. J Fam Pract. 2000;49(9):796-804.

Appendix 1. Search strategy

Key search terms	Physiotherapy	Factors
Patient/person/client centeredness	Physical Therapy	Aspects
Patient/person/client centered care	Rehabilitation	Components
Patient/person/client centred	Remedial exercise	Features
Patient/person/client centered		Elements
Patient/person/client orientated		Parts
Patient/person/client tailored		
Patient/person/client focused		
Databases		
PubMed (MEDLINE)		

EMBASE Cochrane PsychINFO CINAHL PFDro Scopus

Search Strategy PubMed (MEDLINE):

Topic: Patient-centeredness

01. (((((((((((()())))))))) OR "patient centered care[MeSH Terms]) OR "patient centered") OR "patient centred") OR "Patient orientated") OR "patient oriented") OR "Patient tailored") OR "Patient Focused") OR "Person centeredness") OR "Person centered") OR "Person centred") OR "person orientated") OR "person oriented") OR "person tailored") OR "person focused") OR "Client centeredness") OR "Client centered") OR "Client centred") OR "client orientated") OR "client oriented") OR "client focused") OR "client tailored"

Topic: Physiotherapy

02. ((((((("physical therapy") OR physical therapist[MeSH Terms]) OR modalities, physical therapy[MeSH Terms]) OR physical therapy specialty[MeSH Terms]) OR physiotherapy) OR rehabilitation) OR rehabilitation[MeSH Terms]) OR "remedial exercise") OR remedial AND exercise) OR physical AND therapy

Topic: Factors:

03. (((((Factors) OR aspects) OR components) OR features) OR elements) OR parts 04. ((#01) AND #02) AND #03

Search Strategy EMBASE:

01. 'physiotherapy'/exp OR (physical AND 'therapy'/exp) OR 'physical therapy'/exp 02. 'patient centred' OR 'patient orientated' OR 'patient oriented' OR 'patient tailored' OR 'patient focused' OR 'person centredness' OR 'person centred' OR 'person orientated' OR 'person oriented' OR 'person tailored' OR 'person focused' OR 'client centredness' OR 'client centred' OR 'client orientated' OR 'client oriented' OR 'client focused' OR 'client tailored' OR 'patient centredness' AND [embase]/lim

03. factors OR aspects OR components OR features OR 'elements'/exp OR parts AND [embase]/ lim

04. #01 AND #02 AND #03

05. #04 AND 'qualitative research'/de(limit)

Appendix 2. Checklist methodological quality assessment

No.	Checklist item	Definition
1	Was the researcher experienced or trained?	
2	Was the research question clearly defined?	
3	Was the methodological orientation suitable for this	Grounded theory, discourse analysis,
	research question?	ethnography, phenomenology, case study
4	Was theoretical or purposeful sampling used?	
5	Was there stated how many participants where	
	approached?	
6	Were the important characteristics of the sample	Demographic data, date, where data was
	described?	collected
7	Does the sample produce the type of knowledge	Choice of informants whose knowledge or
	necessary to understand the structures and processes	experience is relevant to the substantive
	within which the individuals or situations are located?	focus and theoretical framework of the study
8	Was there stated that the interview was open, semi	
	structured or if there were focus groups?	
9	Were repeated interviews carried out?	Repeated interviews derive more information
10	Were field notes made?	
1	Was data saturation discussed/reached?	
12	Were there two or more researchers that coded the	Triangulation of coders
	data?	
13	Was software used to manage the data?	
14	Did themes derive from the data?	Themes in advance or derived from the data,
		if themes where identified in advance the
		quality of data analysis is less
15	Were participant quotations presented to illustrate	
	themes/items?	
16	Were major themes clearly presented in the findings?	
17	Is the description described in sufficient detail to allow	Appropriate presentation of primary data
	the researcher or the reader to interpret the meaning	and description of context
	and context of what is being researched?	
18	Does the researcher move from description of the	Evidence of analysis and interpretation of
	data, through quotations or examples, to an analysis	data at conceptual and theoretical level
	and interpretation of their meaning and significance?	
19	Are claims being made for the generalizability of the	Findings are related to broader theoretical
	findings to other bodies of knowledge? (within scientific	concerns and/or other empirical context
	research)	
20	Are claims being made for the generalizability of the	Findings are related to broader theoretical
	findings to other populations?	concerns and/or other empirical context

The clinical biopsychosocial assessment in the physiotherapy encounter of patients with chronic pain

Clinical biopsychosocial physiotherapy assessment of the patient with chronic pain: The first step in pain neuroscience education

Amarins J. Wijma^{1,2,3}, C. Paul van Wilgen^{1,2,3}, Mira Meeus^{3,4,5}, Jo Nijs^{1,3},

Affiliations:

¹Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium ²Transcare, Transdisciplinary Outpatient Treatment Centre, the Netherlands ³Pain in Motion International Research Group, (www.paininmotion.be), Belgium ⁴Department of Rehabilitation Sciences and Physiotherapy, Faculty of Medicine and Health Sciences, Antwerp University, Antwerp, Belgium ⁵Department of Rehabilitation Sciences and Physiotherapy, Faculty of Medicine and Health

⁵Department of Rehabilitation Sciences and Physiotherapy, Faculty of Medicine and Health Sciences, Ghent University, Ghent, Belgium

Published in: Physiotherapy Theory and Practice, 33 (5), 368-84, Jul 2016

Abstract

Purpose

Pain Neuroscience Education (PNE) is increasingly used as part of a physiotherapy treatment in patients with chronic pain. A thorough clinical biopsychosocial assessment is recommended prior to PNE to allow proper explanation of the neurophysiology of pain and the biopsychosocial interactions in an interactive and patient-centered manner. However, without clear guidelines, clinicians are left wondering how a biopsychosocial assessment should be administered.

Methods

Therefore, we provided a practical guide based on scientific research and clinical experience for the biopsychosocial assessment of patients with chronic pain in physiotherapy practice. The purpose of this paper is to describe the use of the Pain – Somatic factors – Cognitive factors – Emotional factors – Behavioural factors – Social factors – Motivation – model (PSCEBSM-model) during the intake, as well as a pain analysis sheet. This model attempts to clearly establish what the dominant pain mechanism is (predominant nociceptive, neuropathic or non-neuropathic central sensitization pain), as well as to assess the provoking and perpetuating biopsychosocial factors in the patient with chronic pain.

Conclusion

Using this approach allows the clinician to specifically classify the patient and tailor the plan of care, including PNE, to the individual patient.

Introduction

Chronic pain, also described as 'pain that persists beyond normal time of healing and/ or pain persisting for 3 to 6 months or longer'¹, is a huge global issue and major health care problem ², with a prevalence of 17 to 27% in populations all over the world ³⁻⁶. In the US chronic pain is more prevalent than diabetes, heart disease and cancer combined ⁷⁻⁹. Chronic pain is associated with increased medical costs, decreased income and huge economic burdens ^{10,11}, and has a large negative impact on the patients' quality of life ^{3,10}.

In the last few decades, evidence has shown that a more or less irreversible state of hyperexcitability within the central nervous system known as non-neuropathic central sensitization pain (CS), is present in patients with chronic pain ¹²⁻¹⁴. According to Woolf and Salter (2000) CS is operationally defined as an amplification of neural signaling within the central nervous system that elicits pain hypersensitivity ¹⁵. CS is characterized by generalized hypersensitivity of the somatosensory system ¹⁶⁻¹⁹, resulting in amplification of signaling and eventually even pain without nociceptive input.

It is known that in patients with pain syndromes such as: fibromyalgia ²⁰⁻²²; persisting traumatic neck pain ²³⁻²⁷; tension-type headache ²⁸; migraine ²⁹; subacromial impingement syndrome ³⁰; tennis elbow ^{16,17}; nonspecific arm pain ¹⁸; low back pain ^{20,31,32}; pelvic pain ^{33,34}; chronic fatigue syndrome ³⁵; osteoarthritis ^{20,36,37}; rheumatoid arthritis ³⁸; and tendinopathy ¹⁹, that pain often cannot be explained (solely) by an obvious anatomic defect or tissue damage. In fibromyalgia, chronic whiplash associated disorders, chronic fatigue syndrome and irritable bowel syndrome CS is merely the predominant underlying pain mechanism ^{20,39,40}. In other chronic pain populations, such as low back pain and osteoarthritis, a subgroup may be present with predominant CS pain ^{20,28-30,36-39,41,42}.

The neurophysiological changes in CS are related to changes in the pain neuromatrix, modulating pain processes by behavioural, emotional, social and cognitive factors. ⁴³. It is known that pain catastrophizing ⁴⁴; pain-related anxiety ⁴⁴⁻⁴⁷; trait anxiety ^{47,48} (trait anxiety is the personal level of anxiety); trait neuroticism (personal level of negative affectivity) ⁴⁹; depressive feelings and stress ⁵⁰⁻⁵²; diminished self-efficacy ⁴³; adverse life events ⁵³; and post-traumatic stress disorders ⁵⁴⁻⁵⁸ are present to varying degrees in patients with chronic pain. These can be a consequence of pain and/or can contribute
to the transition and persistence of chronic pain. Emotions, thoughts, attention and stress can influence the pain facilitating pathways ⁵⁹, thereby leading to cognitive emotional sensitization ⁶⁰. Catastrophizing, for instance, is related to activation of the pain neuromatrix, increased pain, affective distress, pain-related disability and poorer treatment outcomes ^{44,61}. Therefore, the initial examination should take into account both somatic (bottom-up, pathoanatomical, peripheral signals) and psycho-social factors (top down, dis-inhibition or pain facilitation).

Consequently, a thorough clinical biopsychosocial assessment is required to understand the process of CS and allows an individualized, patient-centered explanation including biopsychosocial interactions, also known as pain neuroscience education (PNE) ⁶²⁻⁷⁰. However, without clear guidelines clinicians are left wondering how such biopsychosocial assessment should be done and how it allows for an interactive and patient centered PNE. Therefore, the purpose of this paper is to provide a practical guide, based on scientific research and clinical experience, for the biopsychosocial assessment of patients with chronic pain in physiotherapy practice.

Intake

To facilitate the biopsychosocial intake of patients with chronic pain we suggest the use of the PSCEBSM- model (based on the SCEBS model ⁷¹ plus pain and motivation): <u>Pain – Somatic and medical factors – Cognitive factors – E</u>motional factors – <u>B</u>ehavioural factors – <u>Social factors – Motivation</u>. This model starts with examining and determining the type of pain, continues with identifying the different factors associated with chronic pain and ends with determination of the stage of motivation of the patient. A flowchart of the model for use in clinical practice is offered in Figure 1. The pain analysis sheet (Figure 2) can be used to provide a clear overview of the PSCEBSM-model, guide the content of PNE and the treatment. The use of this model takes time, modifications in clinical care, and needs adequate biopsychosocial communication skills.



Figure 1. Flowchart of the biopsychosocial assessment of patients with chronic pain



Figure 2. Pain analysis sheet

P - Type of Pain

In order to allow tailoring PNE to the underlying pain mechanisms it is important to differentiate between the 3 major pain types (nociceptive, neuropathic and CS pain) (Figure 2). An algorithm with a set of classification criteria for differentiating predominant neuropathic, nociceptive and CS pain in patients with musculoskeletal pain has been proposed by 18 pain experts from seven countries ⁷². To identify the predominant pain type two steps need to be taken. The first step entails recognizing neuropathic pain as the predominant pain type. Neuropathic pain is defined as "pain arising as a direct consequence of a lesion or disease affecting the somatosensory system" ⁷³. Table 1, adapted from ⁷² shows the clinical differentiation between predominant nociceptive, non-neuropathic CS pain and neuropathic pain. In line with the diagnostic criteria for neuropathic pain (Treede et al, 2008), central neuropathic pain can be distinguished from CS pain by the lack of damage to the nervous system in the latter group.

Table 1. Criteria for the differential classification between predominant neuropathic ^{73.75} and central sensitization pain. Adapted from Nijs et al.⁷²

Nociceptive Pain	Neuropathic Pain	Non-neuropathic CS pain
History of damage to body	History of a lesion or disease of the	No history of a lesion, damage or
tissue in the previous 6-8	nervous system, or posttraumatic/	disease of the nervous system
weeks.	postsurgical damage to the nervous	No indications from diagnostic
Pain diminishes according to the	system	examinations
natural healing phases.	Indications from diagnostic	
	examinations to reveal an anomaly	
	of the nervous system	
Related to tissue damage or	Related to a medical or systemic	No medical cause for the pain
potential damage. An ankle	cause such as, stroke, herpes,	established
sprain or almost burning a hand	diabetes, or some form of	
	neurodegenerative disease	
Local pain, most often with	Pain and sensory dysfunction is	Pain is neuroanatomical illogical
diagnostic signs such as:	neuroanatomical logical	and segmentally unrelated to the
oedema, hematomas, skin		primary source of nociception
colorations etc.		Several regions of hyperalgesia
		at sites outside and remote to
		the symptomatic area (still at
		segmentally unrelated sites)
Pain is described as sharp,	Pain is frequently described as	Pain is most frequently described
aching, or throbbing.	burning, shooting, or pricking	as vague and dull

The second step is to differentiate between predominant nociceptive and CS pain. The pain is more likely to be originated from CS if the perceived pain and disability are disproportionate to the nature of the injury or pathology ⁷² AND one of the following 2

criteria: 1) The presence of a diffuse or neuro-anatomically illogical pain distribution ⁷² that is not in accordance with dermatomes and myotomes. A wide spread pain index (also known as body diagram) can be used to assess the pain distribution by mapping the pain locations ^{76,77}. The widespread pain index, which includes 19 body regions (each region that has pain is given a point, for a range of scores from 0-19 points), can be used to aid in this process ⁷⁸. A score of 7 or greater suggests widespread pain; and 2) Hypersensitivity of senses unrelated to the musculoskeletal system ⁷², which can be assessed using the Central Sensitization Inventory (CSI) (Table 2). This includes hypersensitivity to light, sounds, smell, taste and a hypersensitive skin. The CSI appears to be a valid, reliable, usable and diagnostically relevant questionnaire assessing common symptoms and facilitating factors to CS in 25 items ^{79,80}. Based on a validation study a cutoff score of 40 points indicates the possibility that the symptoms are due to predominant CS pain ^{81,82}. However, the score of the CSI should be interpreted with caution and in accordance with the clinical symptoms of the patient. More detailed information regarding differentiating between predominant nociceptive and CS pain and how to apply this information in clinical practice can be found in the original paper⁷². Or, adopted for low back pain patients in a more recent paper ⁸³.

The outcome of the mechanism-based classification of pain types can be either predominant nociceptive, neuropathic, CS, or a mixed type of pain. The next step is to identify which factors play a role in the continuation of the patient's pain. These factors can be divided according to the other domains of the PSCEBSM-model.

	kange/ cut off score		Layenometrica	רו מנוונמו ושאתפא	authors in
					clinical practice
Central Sensitization	0-100/ >40 noints	Symptom presentation may indicate the presence	Test-retest reliability 0.82 Cronhach's alnha 0.88	Number of items: 35 Time to administer: 10 minutes	Yes
(CSI) 79,81,82		of CS pain	Sensitivity 81-82,8%, Specificity 54.8-75%		
Leeds Assessment of	0-24/≥12	Neuropathic mechanisms	Cronbach's alpha of 0.74	Number of items: 7	lf necessary
Neuropathic Symptoms and Signs	points	are likely to contribute to the patient's pain	Sensitivity 83%, Specificity 87%	Time to administer: 2-3 minutes	
(LANSS) 84		-	Good validity and reliability		
Body diagram	,	No cut of score exists	ı	Time to administer: 5 minutes	Yes
Brief Illness Perception	ı	Each item has to be	Moderate overall test-retest	Number of items: 13	Yes
Questionnaire		viewed separately	reliability	Time to administer: 5-10 minutes	
(Brief IPQ) ^{85,86}			Good concurrent, predictive	Available on: http://www.uib.no/ipq/index.html	
				-	;
Pain Catastrophizing	≥30 points	The patient is likely to	Cronbach's alpha 0.88-0.95	Number of items: 13	Yes
Scale		catastrophize if the score	Good construct, criterion,	Time to administer: 5-10 minutes	
(PCS)		is above 30	concurrent and discriminant	Subscales:	
87,88		Higher scores indicate	validity ⁸⁹	Rumination: items 8, 9, 10, 11	
		higher catastrophizing		Magnification: items 6, 7, 13 Helplessness: items 1, 2, 3, 4, 5, 12	
State-Trait Anxiety	≥39-4091	2 subscales: State and	Test-retest reliability 0.31-	Number of items: 40	lf necessary
Inventory	≥54-55 for	Trait, range of scores per	0.86	Time to administer: 10 minutes	
(STAI) (older adults ⁹²	subtest 20–80	Cronbach's alpha 0.86-0.9593	State Anxiety: current state of anxiety	
90		Higher scores indicate	Validity S-scale limited	Trait Anxiety: relatively stable aspects of "anxiety	
		greater anxiety		proneness"	

Table 2. (Continued)

ionnaire	Range/ Cut off score	Implications	Psychometrics	Practical issues	Used by the authors in clinical practice
a	≥37	The patient most likely has fear of movement Higher scores indicate greater fear of movement	Moderate construct, concurrent and predictive validity, good internal consistency and a moderate to good retest reliability ⁹⁵⁹⁶	Number of items: 17 Time to administer: 5-10 minutes The scoring on items 4, 8, 12, and 16 should be reversed	If necessary
e ce	≥19 in WAD* ⁹⁸	Above this score perceived injustice is associated with high pain severity, not returning to work, and narcotic use Higher scores indicate more perceived injustice	Cronbach's alpha 0.92 Test retest reliability 0.90- 0.98 9799 Good construct validity	Number of items: 12 Time to administer: 5-10 minutes Subscales: Blame/Unfairness: items 3, 7, 9, 10, 11, 12 Severity/Irreparability: items 1, 2, 4, 5, 6, 8	Yes
PS)		No cut off score exists Higher scores indicate less psychological flexibility	Cronbach's alpha 0.90 (avoidance), 0.75 (fusion) and 0.89 (total scale) Intercorrelation between subscales 0.46 Acceptable model fit Good construct and concurrent validity	Number of items: 16 Time to administer: 5-10 minutes Subscales: Avoidance of pain: items 2, 3, 7, 8, 9, 11, 13, 14, 15, 16 Fusion with pain thoughts: items 1, 4, 5, 6, 10, 12	If necessary
ologic scale	N 0	Indicative of "significant" or "mild" depressive symptomatology Higher scores indicate more depressive feelings	Cronbach's alpha 0.88-0.91 Test-retest reliability ICC=0.87, individual items ICC=0.11-0.73 Poor to excellent validity Sensitivity 80.0%, Specificity 69.8%	Number of items: 20 Time to administer: 5-10 minutes Subscales: Somatic-retarded activity: items 1, 2, 3, 5, 7, 11, 20 Depressed affect: items 6, 10, 14, 17, 18 Positive affect: items 4, 8, 12, 16 Interpersonal affect: items 15, 19	If necessary

Table 2. Cutoff scores and implications for the questionnaires used during the biopsychosocial intake

74

75

Questionnaire	Range/ Cut off score	: Implications	Psychometrics	Practical issues	Used by the authors in clinical practice
Patient Health Questionnaire (PHQ-2 and PHQ-9) ¹⁰⁴	PHQ-9:>10	Indicative for a depressive disorder Higher scores indicate more depressive feelings	Cronbach's alpha 0.86-0.89 PHQ-9 cut off score of > 10: Sensitivity 88%, specificity 88% Good criteria validity ^{164,105}	PHQ-2: Number of items: 2 Time to administer: 1 minute PHQ-9: Number of items: 9 Time to administer: 5-10 minutes Patients who screen positive on the PHQ-2 should be further evaluated with the PHQ-9 to determine if they meet the criteria for a depressive disorder.	If necessary
				Time to administrat. 40.45 million too	

*WAD: Patients with Whiplash Associated Disorders

S - Somatic and Medical Factors

In patients with CS, somatic and medical factors that may be present include: other (past and present) illnesses that might influence CS; nonuse or disuse of body parts; changed movement patterns; exercise capacity; and strength and muscle tension/ tonus during movements. Medication can have (positive/negative) side effects. Therefore, other medical issues and drug use should ideally be examined by a medical physician, however a physiotherapist's basic understanding of pathophysiology and medications interacting with the central nervous system is important when providing PNE. Physiotherapists are indeed capable of gathering this type of information. However, the prescription, administration and modification of medications should be done by a physician. Following the intake, a thorough physical examination should take place. It is important to recognize that in the presence of CS alters findings on clinical tests, such as the Straight Leg Raise, Upper Limb Neurodynamic Tests (ULNT) and assessments of movement or muscle strength. Due to the increased sensitivity to mechanical stimulations and changed patterns in the central nervous system in patients with CS, all physical examination tests (e.g. range of motion, strength, muscle tone, neurodynamic tests, and movement coordination) can evoke pain. Therefore, the aim of the physical examination is to support or refute the clinical picture of CS, assess movement quality, determine body movement if the manner in which the patient moves provokes symptoms consistent with CS (e.g. very guarded or with a lot of tone), and determine if there is fear of movement. In the case of positive findings, clinical reasoning skills are required to decide whether or not such physical factors are of clinical importance for the individual patient and whether or not it contributes in the persistence of CS pain. Positive findings could be: bracing when bending; holding his/ her breath while moving; increased tonus prior to movement; verbal or non-verbal signs of fear; and inconsistent movement patterns.

The physical examination is important for both the physiotherapist and patient. By assessing complaints thoroughly, both parties can be reassured that anything dangerous/serious can be ruled out and confidence is restored that the patient's pain is taken seriously.

C - Cognition/Perceptions

As discussed previously, cognitions and perceptions are important factors that might contribute to (the maintenance of) CS pain. Besides influencing the hypersensitivity in the brain by activating the pain neuromatrix ¹⁰⁶, they also influence the behavioural

and emotional factors of patients ¹⁰⁷. During history taking, the patient's perceptions and cognitions should be assessed thoroughly. Most important are his/her perceptions about the physical and mental aspects of pain as well as the consequences. Furthermore the following factors should be assessed: the expectations for care (anticipated outcome, as well as the content of the treatment); expectations regarding the prognosis of their pain; the coherence (the patients' ability to comprehend their whole situation and their capacity to use available resources to deal with their pain); and emotional representation of the pain. Cognitive patterns, such as catastrophizing, perceived injustice or perceived harm are important to recognize.

In the following section several diagnostic questionnaires are suggested to support the clinician. Not all questionnaires have to be used, rather clinicians can decide which ones to use based on their perceptions and the patient's characteristics. Table 2 provides the cutoff scores, clinical implications and psychometric information for all questionnaires.

Pain Perceptions

The Brief Illness Perception Questionnaire (Brief IPQ) can be used to assess pain perceptions of the patient. The Brief IPQ consists of 13 items and is based on the Common Sense Model of Self-regulation ¹⁰⁷ (described in *Behavioural factors*) and has a moderate overall test-retest reliability and good concurrent validity ^{85,86}. The questionnaire ends with a 3-item rank to list the personal causes of the illness. In addition, the Brief IPQ assesses the expectations for care (items 2 and 4) as well as self-efficacy (item 3). Items 6 and 7 refer to worrying about and understanding pain, respectively. With our clinical expertise patients scoring high (\geq 6) on 'worrying about their pain' and low (\leq 4) on 'understanding their pain' could potentially benefit from PNE in helps to decrease their worrying and improve the understanding of their condition.

Pain Catastrophizing

When pain catastrophizing is suspected, the Pain Catastrophizing Scale (PCS) can be used to assess the degree of pain catastrophizing. The PCS is a valid and reliable 13-item questionnaire (table 2) that examines the rumination, magnification, and helplessness patients have about their perceived ability to manage their pain ^{87,88}. In order to avoid prejudices, we propose omitting the words 'pain catastrophizing scale' from the questionnaire when handing it over to the patient. If the patient scores high (\geq 30) on

the PCS, their feelings and cognitions on catastrophizing should be acknowledged and explored in the PNE session. The patient should also be told that catastrophizing increases the activity in the pain signature in the brain and therefore increases their pain.

E - Emotional Factors

Emotional factors are related to cognitions and perceptions and include anxiety, anger, fear, depressive feelings and post-traumatic stress. Physiotherapists can specifically ask about emotional factors related to the onset of pain, such as fear of specific movements, avoidance behaviours, a psychological traumatic onset of the pain, or psychological issues including work, family, financial, or social.

Anxiety

State anxiety (related to an event) and trait anxiety (personal level of anxiety) are important factors in chronic pain. In addition to questioning the patient about anxiety, we recommend using the State-Trait Anxiety Inventory (STAI). This questionnaire has 20 items for assessing trait anxiety and 20 for state anxiety. The STAI has a good internal consistency, is reliable and has considerable construct and concurrent validity ⁹⁰. A cutoff score of 39-40 has been suggested to detect clinically significant symptoms and a higher cut score of 54–55 has been suggested for older adults ^{91,92}. If the outcome of the STAI indicates that the patient has anxiety, either state or trait, the effects of this anxiety should be explored and discussed in the PNE session.

Fear of Movement

Based on previous experiences, patients can become fearful and begin to avoid potentially painful movements. The Tampa-Scale of Kinesiophobia (TSK) is a 17-item scale that measures the *somatic focus of patients (beliefs about* underlying and serious medical problems), and *activity avoidance* (beliefs about (re)injury or increased pain). The TSK has moderate construct, concurrent and predictive validity, good internal consistency and a moderate to good retest reliability ^{95,96}. Patients scoring high on the TSK, above 37 points, are likely to have fear of movement ⁹⁴ and during the PNE session the effects of fear of movement on the pain neuromatrix in the brain (by increased activity in the hypothalamic-pituitary-adrenal axis and increased attention) should be explained.

Anger

Perceived injustice as a form of anger can be measured using the Injustice Experience Questionnaire (IEQ). Perceived injustice can have negative effects on pain, disability and treatment. For example, patients developing chronic pain following a car accident do not present with perceived injustice or anger in the acute stage, but develop it throughout the transition phase towards chronicity, with marked increased levels in the chronic stage (and not in those recovering) ¹⁰⁸. Therefore, we recommend the use of the IEQ on patients who are suspected of having anger/perceived injustice such as a patient who develops chronic pain following a car accident. The IEQ has a high internal consistency, a good construct validity and reliability ⁹⁷. If high scores (> 19) on this questionnaire are present this can be used to focus part of the PNE. By first acknowledging their feelings of anger and injustice and then explaining that such emotions sustain the pain signature in the brain and may present barriers to improvement.

Depressive Feelings

Physiotherapists are not trained to diagnose depression or other psychological states, but should be aware of their existence and role in patients with chronic pain. Depressive feelings can be assessed through self-report questionnaires. The two-item Patient Health Questionnaire-2 (PHQ-2) and nine-item PHQ-9 are commonly recommended for depression screening in clinical and research settings. The PHQ-2 is a quick and helpful screening tool for depression, with a sensitivity of 86% and specificity of 78¹⁰⁴. Patients who report more than 2 points or higher on the PHQ-2 should be further evaluated with the PHQ-9. A score of 10 or higher on the PHQ-9 detects depression. The PHQ-2 has a high sensitivity (86%), the PHQ-9 has higher specificity (91%) ¹⁰⁴. Additionally, the Center for Epidemiologic Studies Depression scale (CES-D)¹⁰¹ can be used. The CES-D is a 20 item self-report measure of depressive symptomatology during the past week. Scores of 16 or greater indicate more severe symptoms. The CES-D has a high internal consistency, good reliability, convergent and divergent validity ¹⁰⁹. Patients scoring high on this or any other scale for measuring depressive feelings need support, acknowledgement, comfort and help, each of which can be provided in part by PNE.

There is a bidirectional relationship between depression and pain ¹¹⁰, however, because patients are often fearful of being labelled ('it's in your head') we suggest explaining to patients with chronic pain that depression may be a consequence rather than a cause of chronic pain. Furthermore, we suggest explaining the interplay between pain and depression in the pain neuromatrix during PNE.

Stress

Physiotherapists are suggested to screen their patients for post-traumatic stress disorder by asking the patient about prior traumatic events and whether they frequently relive the event, avoid situations that remind them of the event or have negative changes in beliefs and feelings since the event. In addition, physiotherapists should also evaluate general levels of stress and/or stress intolerance. Stress can be related to work factors, relationships, financial stress, health related stress, etc. and should be investigated during the assessment. If relevant to the individual patient, the influence of stress on the pain neuromatrix and top down inhibitory pathways should be explained during PNE.

B - Behavioural Factors

For physiotherapists it is important to assess current behaviour and adaptations made as a consequence of pain. Both conscious and non-conscious behaviour can be the product of cognitive and emotional information when perceiving and interpreting inputs or perceived threats to health and wellbeing ^{45,107,111,112}.

Patients can be roughly divided into three subgroups: patients who demonstrate healthy behaviour (pain experience results in no/low fear, confrontation and recovery ¹¹³), avoidance (described previously) and persistence behaviour. Persisters are patients who continue to perform painful activities until completion even though the activity is perceived as too hard ¹¹⁴. In the long run, persistence behaviour can also be unhelpful and result in an extreme active-non-active pattern (also called: 'yo-yo' or 'over activityunderactivity cycling') of daily activity levels ^{115,116}. When assessing patients' behaviour in clinical practice, the majority of patients present with a mixed pattern: they avoid certain activities or movements, and simultaneously persist in others. This observation underscores the need for a thorough individual assessment and questioning of each patient individually. Patients should be questioned about their work, home and recreational activities to determine which are avoided or persisted. In addition, patients need to be asked when and why they chose to either persist or avoid the activities. An activity diary may aid in this process. There are different models explaining the above mentioned behaviour, such as the Common Sense Model of Self-regulation (CSMS), classical conditioning and operant conditioning ^{107,111,112}.

The CSMS is a model that helps understand how the perceptions, experience and impact of having a disorder might influence a patient's interpretation and response ¹⁰⁷.

Based on the perceptions a person has, he/she will present with certain behaviours in an attempt to influence the threat of a potentially painful event. After any event a person assesses whether or not the threat is diminished. If, for instance, the patient experiences lower back pain during forward bending (threat), the perceptions and emotions can change the behaviour and pattern of forward bending. The latest fear-avoidance model of Vlaeyen et al. ^{45,113} supports the CSMS and the role of pain catastrophizing in pain chronification. According to the CSMS the fear-avoidance behaviour of the patient, physical inactivity, disuse and consequent disability result from current or previous pain perceptions. Therefore, the physiotherapist should assess the impact of pain perceptions and behaviours on levels of function (work, recreation, daily activities). For example, when a patient expresses the avoidance of playing tennis due to potential back pain, the physiotherapist should ask about the patient's beliefs and emotions about what happens during this activity.

Unconscious behaviour and classical conditioning ¹¹¹ are also important. For instance, working in a stressful situation at a desk for long periods during which the patient perceives pain, the desk may become associated with the pain. The desk is a neutral stimulus, but can become associated with the pain and, in the end, can evoke pain. In a model recently proposed by Moseley and Vlaeyen ¹¹⁷, they postulate that classical conditioning can eventually result in pain from non-nociceptive impulses by stimulus generalization, called the Imprecision Hypothesis ¹¹⁷.

Behaviour and social factors may also become related through operant conditioning (changing of behaviour by the use of reinforcement, after the desired response). Operant conditioning, as described by Skinner et al.¹¹², is directly applicable to pain behaviour¹¹⁸. Operant conditioning works with positive and negative reinforcers. For example, an unconsciously positive reinforcement of the pain behaviour may occur when sympathetic attention is given to the patient, which is likely to strengthen the behaviour and increase its likelihood in the future. When behaviour is followed by negative reinforcement such as criticism, that behaviour is less likely to occur in the future and behaviour is likely to go extinct^{112,118}. In the assessment, physiotherapists should ask about avoided behaviours and how the social surroundings impact this response to identify potential positive and negative reinforcers.

S - Social Factors

Social and environmental factors that cause stress or a disbalance in the identified self of the patient can have a negative effect on pain. Social factors can be divided into: housing or living situation, social environment, work, relationship with the partner, and prior/other treatments. It is important to find out if there are components of the social factors that are helpful and supportive or stressful and unconsciously unhelpful. Other important social factors include prior/concurrent treatments and the attitudes and beliefs of these healthcare professionals (for instance: a former physiotherapist who has told the patient that his/her 'disk was out of line'). These prior/concurrent treatments, as well as advice and explanations about the patient's condition will influence their perceptions and current coping strategies. Therefore prior/concurrent treatments should be explored and communication between healthcare professionals is suggested.

Low levels of social support may present barriers to improvement in patients with chronic pain, can be a sustaining factor in CS and worsen the prognosis ^{119,120}. Unpublished results and clinical experience suggest that PNE can improve social support, especially when the therapist facilitates social support by asking the patient to bring their spouse, child or a close friend to one of the sessions. If this is not feasible, significant others can be motivated to read information about CS, such as the book 'Explain Pain'¹²¹.

M - Motivation

Determining motivation and readiness to change is vital for further treatment. The perceptions about the cause of pain and the treatment expectations are crucial to understand in order to target and modify them during the treatment ⁴³. This is especially true if the proposed treatment (including PNE) might be different from what they have heard before, and more biopsychosocial focused.

The 16-item Psychology Inflexibility in Pain Scale (PIPS) can be used to assess avoidance of pain and cognitive fusion with pain where patients get intertwined with their thoughts, and thoughts are seen as a fact, for example: 'I am my pain'. The scale has good internal consistency as well as criterion and construct validity. Furthermore, it has been reported that psychological flexibility has a mediator function in the relationship between pain and kinesiophobia, pain and disability, and acceptance and catastrophizing, meaning that these relationships are largely influenced by psychological flexibility ¹⁰⁰. The PIPS is

used to examine the patient's psychological flexibility to change. Previous research has shown that patients with chronic pain with a high degree of psychological inflexibility are likely to be non-responders in an Acceptance and Commitment Therapy (ACT) based rehabilitation ¹²². Based on the assessment and a high score on one or both scales of the PIPS, the physiotherapist might decide not to initiate treatment. Alternatively, if the PIPS score is high, one could focus intensively on PNE to change cognitions and perceptions prior to initiating the remaining parts of the rehabilitation program. Once the focused PNE has been delivered, the PIPS is scored again to see if there is a difference in score that might indicate that the patient is now ready for rehabilitation. We realize the latter is a pragmatic approach and not (yet) supported by research findings.

The stage of change model is another manner to assess the motivation for treatment and education of the patient. The start of the PNE should be tailored to the stage of change the patient is in¹²³. One of the goals of PNE is to transition patients in their stage of change when necessary, however, the starting point should be adjusted to the stage of change a patient is in. The physiotherapist has to determine which phase the patient is in, considering both the perception and emotional state of the patient. In the precontemplation phase the patient has no intention to change, he/she is not willing to adapt another explanation or another treatment or coping strategy. In the contemplation phase the patient is aware of the problem and starts thinking about changing, however, still has doubts, but is open to listen. The preparation phase is one step further: the patient is intending to take action in the next month and is more willing to listen to PNE and other new explanations. In the action phase, the patient modifies his/her behaviour, experiences, and environment in order to overcome the problems. In this very important phase the physiotherapist plays an important role in the inventory of existing or potential barriers for maintaining this new behaviour and changed perceptions. In the maintenance phase the action has been successful and the patient works to prevent relapse and to consolidate the goals for more than six months. The last phase, termination, is the phase in which people have changed and no longer need to work to prevent relapse ¹²³.

Discussion

This article describes the biopsychosocial assessment of patients with non-neuropathic CS pain in physiotherapy practice and includes a combination of clinical experience and scientific evidence. Certain aspects of this approach are scientifically validated, but certain components and combinations of components have not been studied (Type of Pain + SCEBS model + Motivation). We attempted to clearly delineate what is supported by research and what is based on expert opinion.

Similar to the recommended approach outlined in this article, Dansie and Turk have previously presented a physician guide for the assessment of patients with chronic pain ¹²⁴. Their assessment is based on three main questions: 1) What is the extent of the patient's disease or injury (physical impairment)? 2) What is the magnitude of the illness? That is, to what extent is the patient suffering, disabled, and unable to enjoy usual activities? 3) Does the individual's behaviour seem appropriate to the disease or injury, or is there any evidence of symptom amplification for a variety of psychological or social reasons?¹²⁴. Furthermore, they advise a standardized pain assessment and a brief screening interview in which the physician can screen for psychosocial problems. However, unlike the extensive description of the biopsychosocial assessment in our article they focused primarily on the assessment of pain and disability.

Diagnosis/Clinical Reasoning

For some patients with chronic pain getting a diagnosis that makes sense to them is the first step to self-management of their pain. By getting a diagnosis the pain is no longer 'in your mind', 'imagination' or 'hysteria' ¹²⁵ and has become 'legal' and acknowledged by healthcare professionals. Results by Thompson et al. show that for patients with chronic pain who 'thrive' (who live well with their pain), receiving the 'chronic pain' diagnosis by a healthcare professional, even though shocking, was the first step in their self-management ¹²⁶.

Identifying the primary mechanism contributing to that pain experience (nociception, neuropathic, CS, combination) is more important than classifying pain according to its duration. Identifying whether or not the patient has predominantly nociceptive, neuropathic or CS pain is a diagnosis in itself that offers potential treatment pathways (see Nijs et al. 2014)⁷². Based on this mechanism the further PNE can be provided (figure 3.)



Figure 3. Pain Neuroscience Education tailored to the primary pain mechanisms of the patient

Biopsychosocial Assessment: Recommendations for Further Treatment

To assess pain as a biopsychosocial phenomenon and really comprehend the essence of a patient's pain problem takes time. Obviously, it is important to focus on the changeable biopsychosocial factors while also being aware of non-changeable aspects such as personality, neuroticism and the degree of trait anxiety, which are known to be stable to some degree over time ¹²⁷⁻¹³⁰.

Knowledge of these biopsychosocial factors is essential for steering the plan of care and identifying the potential components of PNE to be used. The (psychosocial) education of the physiotherapist, including the competence, knowledge, biopsychosocial vision, interpersonal factors and 'fingerspitzengefühl' (instinct, intuitive flair, high situational awareness, ability to respond most appropriately and tactfully, etc.), combined with twoway communication and a patient-centered approach are important. Physiotherapists specialize in the assessment of function, physical activity, movements, muscle tension, etc. combined with strategies to treat these impairments. Even though questionnaires can help identify behavioural, cognitive and emotional factors (table 2), we must be reflective of our biopsychosocial view and knowledge of illness perceptions. Research has shown that physiotherapists struggle in this area¹³¹⁻¹³⁷. It is important, as health care providers, to know and respect our limits, especially when working with patients with chronic pain. Throughout the assessment physiotherapists should be aware of their limitations and ask themselves: 'is this patient (with chronic pain) in the right place here with me or should he/she be treated in a multidisciplinary setting or referred to another provider?'.

Once indications for PNE are established, individualized therapy can be initiated by explaining the biopsychosocial diagnosis to the patient; reassuring them that their pain is real and explaining why they are in pain (CS pain, neuropathic pain, nociception). Changeable factors and the receptiveness of the patient to change further guide the content and the attitude of the physiotherapist during PNE. Based on the stages of change model ¹²³ patients in the precontemplation phase need a more 'nurturing parent' role and can be more resistant and defensive. Patients who are in the contemplation phase may benefit from a 'Socratic teacher' who encourages patients to achieve insights into their own condition. If the patient is in the preparation stage, we recommend that the physiotherapist adopt the role of an 'experienced coach' who can provide a new game plan or can review and modify the patient's own plan. Patients in the action and maintenance phases benefit from a physiotherapist who becomes more of a 'consultant' who is available to provide expert advice and support ¹²³. Physiotherapists keen to learn more about this topic are referred to the cited references.

We have outlined how physiotherapists may take the first step in the successful treatment of patients with chronic pain, by motivating the patient to achieve goals and restore values and his/her identified self ^{126,138,139}. Chronic pain is complicated, and a thorough biopsychosocial intake, examination and interdisciplinary treatment plan are required for a successful treatment.

Conclusion

Prior to providing PNE and further treatment, an extensive biopsychosocial intake should be conducted. To our knowledge this is the first article describing the comprehensive biopsychosocial intake of patients with chronic pain in physiotherapy practice and is derived on scientific evidence as well as expert opinion. This approach needs to be investigated further in clinical trials with chronic pain patients.

We believe the biopsychosocial intake described here is necessary to clarify the primary type of chronic pain: predominant neuropathic, nociceptive or CS pain. This allows the physiotherapist to assess the biopsychosocial factors that may be contributing to the continuation of pain. 'Diagnosing' the patient as having CS pain, nociceptive pain, neuropathic pain or a combination, is the first step in tailoring a patient centered PNE that can aid the patient in his/her self-management process.

References

- 1. Merskey H, Bogduk N., International Association for the Study of Pain. Task Force on Taxonomy. *Classification of chronic pain: descriptions of chronic pain syndromes and definitions of pain terms.* 2 ed. the University of Michigan: IASP Press; 1994.
- 2. European Pain Federation EFIC. EFIC's declaration on pain as a major health problem, a disease in its own right. 2010. http://www.efic.org/index.asp?sub=724B97A2EjBu1C.
- 3. Breivik H, Collett B, Ventafridda V, Cohen R, Gallacher D. Survey of chronic pain in Europe: prevalence, impact on daily life, and treatment. *Eur J Pain*. 2006;10(4):287-333.
- 4. Reid KJ, Harker J, Bala MM, et al. Epidemiology of chronic non-cancer pain in Europe: narrative review of prevalence, pain treatments and pain impact. *Curr Med Res Opin.* 2011;27(2):449-462.
- 5. Leadley RM, Armstrong N, Lee YC, Allen A, Kleijnen J. Chronic diseases in the European Union: the prevalence and health cost implications of chronic pain. *J Pain Palliat Care Pharmacother*. 2012;26(4):310-325.
- Blyth FM, March LM, Brnabic AJ, Jorm LR, Williamson M, Cousins MJ. Chronic pain in Australia: a prevalence study. *Pain*. 2001;89(2-3):127-134.
- American Diabetes Association. American Diabetes Association. http://www.diabetes.org/ diabetes-basics/diabetes-statistics/. Published 2012.
- 8. American Heart Association. Statistics Heart Disease and Stroke. Update: A Report From the American Heart Association. In: *Circulation*. Vol 123:e18-e209. 2011:page 20.
- American Cancer Society. Prevalence of Cancer. http://www.cancer.org/docroot/CRI/ content/CRI_2_6x_Cancer_Prevalence_How_Many_People_Have_Cancer.asp. Published 2014.
- 10. Bekkering GE, Bala MM, Reid K, et al. Epidemiology of chronic pain and its treatment in The Netherlands. *Neth J Med.* 2011;69(3):141-153.
- 11. van Tulder MW, Koes BW, Bouter LM. A cost-of-illness study of back pain in The Netherlands. *Pain.* 1995;62(2):233-240.
- 12. Latremoliere A, Woolf CJ. Central sensitization: a generator of pain hypersensitivity by central neural plasticity. *J Pain*. 2009;10(9):895-926.
- 13. Torebjork HE, Lundberg LE, LaMotte RH. Central changes in processing of mechanoreceptive input in capsaicin-induced secondary hyperalgesia in humans. *J Physiol.* 1992;448:765-780.
- Koltzenburg M, Torebjork HE, Wahren LK. Nociceptor modulated central sensitization causes mechanical hyperalgesia in acute chemogenic and chronic neuropathic pain. *Brain.* 1994;117 (Pt 3):579-591.

- 15. Woolf CJ, Salter MW. Neuronal plasticity: increasing the gain in pain. *Science*. 2000;288(5472):1765-1769.
- 16. Fernandez-Carnero J, Fernandez-de-Las-Penas C, de la Llave-Rincon AI, Ge HY, Arendt-Nielsen L. Widespread mechanical pain hypersensitivity as sign of central sensitization in unilateral epicondylalgia: a blinded, controlled study. *Clin J Pain*. 2009;25(7):555-561.
- 17. Coombes BK, Bisset L, Vicenzino B. Thermal hyperalgesia distinguishes those with severe pain and disability in unilateral lateral epicondylalgia. *Clin J Pain*. 2012;28(7):595-601.
- Moloney N, Hall T, Doody C. Sensory hyperalgesia is characteristic of nonspecific arm pain: a comparison with cervical radiculopathy and pain-free controls. *Clin J Pain*. 2013;29(11):948-956.
- van Wilgen CP, Konopka KH, Keizer D, Zwerver J, Dekker R. Do patients with chronic patellar tendinopathy have an altered somatosensory profile? A Quantitative Sensory Testing (QST) study. Scand J Med Sci Sports. 2013;23(2):149-155.
- 20. Staud R. Evidence for shared pain mechanisms in osteoarthritis, low back pain, and fibromyalgia. *Curr Rheumatol Rep.* 2011;13(6):513-520.
- 21. Vierck CJ, Jr. Mechanisms underlying development of spatially distributed chronic pain (fibromyalgia). *Pain.* 2006;124(3):242-263.
- 22. Meeus M, Nijs J. Central sensitization: a biopsychosocial explanation for chronic widespread pain in patients with fibromyalgia and chronic fatigue syndrome. *Clin Rheumatol.* 2007;26(4):465-473.
- 23. Herren-Gerber R, Weiss S, Arendt-Nielsen L, et al. Modulation of central hypersensitivity by nociceptive input in chronic pain after whiplash injury. *Pain Med.* 2004;5(4):366-376.
- 24. Jull G, Sterling M, Kenardy J, Beller E. Does the presence of sensory hypersensitivity influence outcomes of physical rehabilitation for chronic whiplash?--A preliminary RCT. *Pain.* 2007;129(1-2):28-34.
- 25. Sterling M. Testing for sensory hypersensitivity or central hyperexcitability associated with cervical spine pain. *J Manipulative Physiol Ther.* 2008;31(7):534-539.
- 26. Sterling M, Jull G, Vicenzino B, Kenardy J. Sensory hypersensitivity occurs soon after whiplash injury and is associated with poor recovery. *Pain.* 2003;104(3):509-517.
- 27. Sterling M, Treleaven J, Edwards S, Jull G. Pressure Pain Thresholds in Chronic Whiplash Associated Disorder: Further Evidence of Altered Central Pain Processing. *Journal of Musculoskeletal Pain.* 2002;10(3):69-81.
- 28. Buchgreitz L, Egsgaard LL, Jensen R, Arendt-Nielsen L, Bendtsen L. Abnormal pain processing in chronic tension-type headache: a high-density EEG brain mapping study. *Brain.* 2008;131(Pt 12):3232-3238.
- 29. de Tommaso M, Federici A, Franco G, et al. Suggestion and pain in migraine: a study by laser evoked potentials. *CNS Neurol Disord Drug Targets*. 2012;11(2):110-126.

- 30. Paul TM, Soo Hoo J, Chae J, Wilson RD. Central hypersensitivity in patients with subacromial impingement syndrome. Arch Phys Med Rehabil. 2012;93(12):2206-2209.
- 31. Giesecke T, Gracely RH, Grant MA, et al. Evidence of augmented central pain processing in idiopathic chronic low back pain. Arthritis Rheum. 2004;50(2):613-623.
- 32. Roussel NA, Nijs J, Meeus M, Mylius V, Fayt C, Oostendorp R. Central sensitization and altered central pain processing in chronic low back pain: fact or myth? Clin J Pain. 2013;29(7):625-638.
- 33. Yang CC, Lee JC, Kromm BG, Ciol MA, Berger R. Pain sensitization in male chronic pelvic pain syndrome: why are symptoms so difficult to treat? / Urol. 2003;170(3):823-826
- 34. Farmer MA, Chanda ML, Parks EL, Baliki MN, Apkarian AV, Schaeffer Al. Brain functional and anatomical changes in chronic prostatitis/chronic pelvic pain syndrome. / Urol. 2011;186(1):117-124.
- 35. Meeus M, Nijs J, Van de Wauwer N, Toeback L, Truijen S. Diffuse noxious inhibitory control is delayed in chronic fatigue syndrome: an experimental study. Pain. 2008;139(2):439-448.
- 36. Mease Pl, Hanna S, Frakes EP, Altman RD. Pain mechanisms in osteoarthritis: understanding the role of central pain and current approaches to its treatment. I Rheumatol. 2011;38(8):1546-1551.
- 37. Suokas AK, Walsh DA, McWilliams DF, et al. Quantitative sensory testing in painful osteoarthritis: a systematic review and meta-analysis. Osteoarthritis Cartilage. 2012;20(10):1075-1085.
- Meeus M, Vervisch S, De Clerck LS, Moorkens G, Hans G, Nijs J. Central sensitization in 38. patients with rheumatoid arthritis: a systematic literature review. Semin Arthritis Rheum. 2012;41(4):556-567.
- Nijs J, Van Houdenhove B, Oostendorp RA. Recognition of central sensitization in patients 39. with musculoskeletal pain: Application of pain neurophysiology in manual therapy practice. Man Ther. 2010;15(2):135-141.
- Nijs J, Meeus M, Van Oosterwijck J, et al. In the mind or in the brain? Scientific evidence for 40. central sensitisation in chronic fatigue syndrome. Eur J Clin Invest. 2012;42(2):203-212.
- 41. Smart KM, Blake C, Staines A, Thacker M, Doody C. Mechanisms-based classifications of musculoskeletal pain: part 1 of 3: symptoms and signs of central sensitisation in patients with low back (+/- leg) pain. Man Ther. 2012;17(4):336-344.
- Smart KM, Blake C, Staines A, Doody C. The Discriminative validity of "nociceptive," 42. "peripheral neuropathic," and "central sensitization" as mechanisms-based classifications of musculoskeletal pain. Clin J Pain. 2011;27(8):655-663.
- 43. Turk DC, Okifuji A. Psychological factors in chronic pain: evolution and revolution. J Consult Clin Psychol. 2002;70(3):678-690.

- 44. Gracely RH, Geisser ME, Giesecke T, et al. Pain catastrophizing and neural responses to pain among persons with fibromyalgia. Brain. 2004;127(Pt 4):835-843.
- 45. Leeuw M. Goossens ME, Linton SI, Crombez G, Boersma K, Vlaeven IW. The fearavoidance model of musculoskeletal pain: current state of scientific evidence. / Behav Med. 2007;30(1):77-94.
- 46. Vlaeyen JW, Linton SJ. Fear-avoidance and its consequences in chronic musculoskeletal pain: a state of the art. Pain. 2000;85(3):317-332.
- 47. Hirsh AT, George SZ, Bialosky JE, Robinson ME. Fear of pain, pain catastrophizing, and acute pain perception: relative prediction and timing of assessment. / Pain. 2008;9(9):806-812.
- 48. Sullivan MJ, Thorn B, Rodgers W, Ward LC. Path model of psychological antecedents to pain experience: experimental and clinical findings. Clin J Pain. 2004;20(3):164-173.
- 49. Evers AW, Kraaimaat FW, van Riel PL, Bijlsma JW. Cognitive, behavioural and physiological reactivity to pain as a predictor of long-term pain in rheumatoid arthritis patients. Pain. 2001;93(2):139-146.
- 50. Rivat C, Becker C, Blugeot A, et al. Chronic stress induces transient spinal neuroinflammation, triggering sensory hypersensitivity and long-lasting anxiety-induced hyperalgesia. Pain. 2010;150(2):358-368.
- 51. Kuehl LK, Michaux GP, Richter S, Schachinger H, Anton F. Increased basal mechanical pain sensitivity but decreased perceptual wind-up in a human model of relative hypocortisolism. Pain. 2010;149(3):539-546.
- 52. McEwen BS, Kalia M. The role of corticosteroids and stress in chronic pain conditions. Metabolism. 2010;59 Suppl 1:S9-15.
- 53. Generaal E, Vogelzangs N, Macfarlane GJ, et al. Biological stress systems, adverse life events and the onset of chronic multisite musculoskeletal pain: a 6-year cohort study. Ann Rheum Dis. 2015.
- 54. Sterling M, Hendrikz J, Kenardy J. Compensation claim lodgement and health outcome developmental trajectories following whiplash injury: A prospective study. Pain. 2010;150(1):22-28.
- 55. Sterling M, Chadwick BJ. Psychologic processes in daily life with chronic whiplash: relations of posttraumatic stress symptoms and fear-of-pain to hourly pain and uptime. The Clinical journal of pain. 2010;26(7):573-582.
- 56. Daenen L, Nijs J, Cras P, Wouters K, Roussel N. Changes in Pain Modulation Occur Soon After Whiplash Trauma but are not Related to Altered Perception of Distorted Visual Feedback. Pain practice : the official journal of World Institute of Pain. 2014;14(7):588-598.
- 57. Cohen H, Neumann L, Haiman Y, Matar MA, Press J, Buskila D. Prevalence of post-traumatic stress disorder in fibromyalgia patients: overlapping syndromes or post-traumatic fibromyalgia syndrome? Semin Arthritis Rheum. 2002;32(1):38-50.

- 58. Sherman JJ, Turk DC, Okifuji A. Prevalence and impact of posttraumatic stress disorder-like symptoms on patients with fibromyalgia syndrome. *Clin J Pain*. 2000;16(2):127-134.
- 59. Zusman M. Forebrain-mediated sensitization of central pain pathways: 'non-specific' pain and a new image for MT. *Man Ther.* 2002;7(2):80-88.
- 60. Brosschot JF. Cognitive-emotional sensitization and somatic health complaints. *Scand J Psychol.* 2002;43(2):113-121.
- 61. Edwards RR, Bingham CO, 3rd, Bathon J, Haythornthwaite JA. Catastrophizing and pain in arthritis, fibromyalgia, and other rheumatic diseases. *Arthritis Rheum*. 2006;55(2):325-332.
- 62. Gallagher L, McAuley J, Moseley GL. A randomized-controlled trial of using a book of metaphors to reconceptualize pain and decrease catastrophizing in people with chronic pain. *Clin J Pain.* 2013;29(1):20-25.
- 63. Moseley GL. Evidence for a direct relationship between cognitive and physical change during an education intervention in people with chronic low back pain. *Eur J Pain.* 2004;8(1):39-45.
- 64. Moseley GL, Nicholas MK, Hodges PW. A randomized controlled trial of intensive neurophysiology education in chronic low back pain. *Clin J Pain*. 2004;20(5):324-330.
- 65. Moseley GL. Combined physiotherapy and education is efficacious for chronic low back pain. *Aust J Physiother*. 2002;48(4):297-302.
- 66. Meeus M, Nijs J, Van Oosterwijck J, Van Alsenoy V, Truijen S. Pain physiology education improves pain beliefs in patients with chronic fatigue syndrome compared with pacing and self-management education: a double-blind randomized controlled trial. *Arch Phys Med Rehabil.* 2010;91(8):1153-1159.
- 67. Nijs J, Meeus M, Van Oosterwijck J, et al. Treatment of central sensitization in patients with 'unexplained' chronic pain: what options do we have? *Expert Opin Pharmacother*. 2011;12(7):1087-1098.
- 68. Van Oosterwijck J, Meeus M, Paul L, et al. Pain Physiology Education Improves Health Status and Endogenous Pain Inhibition in Fibromyalgia: A Double-Blind Randomized Controlled Trial. *Clin J Pain.* 2013.
- 69. Van Oosterwijck J, Nijs J, Meeus M, et al. Pain neurophysiology education improves cognitions, pain thresholds, and movement performance in people with chronic whiplash: a pilot study. *J Rehabil Res Dev.* 2011;48(1):43-58.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil*. 2011;92(12):2041-2056.
- 71. Speckens AEMS, Ph. Van Rood, Y. Protocollaire behandeling van patiënten met onverklaarde lichamelijke klachten: Cognitieve gedragstherapie. In: Houten/Diegem, ed. *Protocollaire behandeling in de ambulante geestelijke gezondheidszorg 2.*1999:199-227.

- Nijs J, Torres-Cueco R, van Wilgen CP, et al. Applying modern pain neuroscience in clinical practice: criteria for the classification of central sensitization pain. *Pain Physician*. 2014;17(5):447-457.
- 73. Treede RD, Jensen TS, Campbell JN, et al. Neuropathic pain: redefinition and a grading system for clinical and research purposes. *Neurology*. 2008;70(18):1630-1635.
- 74. Haanpää M TR. Diagnosis and classification of neuropathic pain. AISP In: 2010;XVII(7). PCU, ed.
- 75. Haanpaa M, Attal N, Backonja M, et al. NeuPSIG guidelines on neuropathic pain assessment. *Pain.* 2011;152(1):14-27.
- Margolis RB, Tait RC, Krause SJ. A rating system for use with patient pain drawings. *Pain*. 1986;24(1):57-65.
- Margolis RB, Chibnall JT, Tait RC. Test-retest reliability of the pain drawing instrument. *Pain*. 1988;33(1):49-51.
- Wolfe F, Clauw DJ, Fitzcharles MA, et al. The American College of Rheumatology Preliminary Diagnostic Criteria for Fibromyalgia and Measurement of Symptom Severity. *Arthritis Care* & *Research.* 2010;62(5):600-610.
- 79. Mayer TG, Neblett R, Cohen H, et al. The development and psychometric validation of the central sensitization inventory. *Pain Pract.* 2012;12(4):276-285.
- Kregel J, Vuijk PJ, Descheemaeker F, et al. The Dutch Central Sensitization Inventory (CSI): Factor Analysis, Discriminative Power and Test-Retest Reliability. *Clin J Pain*. 2015.
- Neblett R, Hartzell MM, Cohen H, et al. Ability of the Central Sensitization Inventory to Identify Central Sensitivity Syndromes in an Outpatient Chronic Pain Sample. *Clin J Pain.* 2014.
- Neblett R, Hartzell MM, Cohen H, et al. Ability of the central sensitization inventory to identify central sensitivity syndromes in an outpatient chronic pain sample. *Clin J Pain*. 2015;31(4):323-332.
- Nijs J, Apeldoorn A, Hallegraeff H, et al. Low back pain: guidelines for the clinical classification of predominant neuropathic, nociceptive, or central sensitization pain. *Pain physician*. 2015;18(3):E333-346.
- 84. Bennett M. The LANSS Pain Scale: the Leeds assessment of neuropathic symptoms and signs. *Pain.* 2001;92(1-2):147-157.
- 85. Leysen M, Nijs J, Meeus M, et al. Clinimetric properties of illness perception questionnaire revised (IPQ-R) and brief illness perception questionnaire (Brief IPQ) in patients with musculoskeletal disorders: A systematic review. *Man Ther.* 2015;20(1):10-17.
- Broadbent E, Petrie KJ, Main J, Weinman J. The brief illness perception questionnaire. J Psychosom Res. 2006;60(6):631-637.

- 87. Osman A, Barrios FX, Kopper BA, Hauptmann W, Jones J, O'Neill E. Factor structure, reliability, and validity of the Pain Catastrophizing Scale. *J Behav Med.* 1997;20(6):589-605.
- Sullivan MJL, Bishop SR, Pivik J. The Pain Catastrophizing Scale: Development and validation. *Psychological Assessment*. 1995;7(4):524-532.
- Osman A, Barrios FX, Gutierrez PM, Kopper BA, Merrifield T, Grittmann L. The Pain Catastrophizing Scale: further psychometric evaluation with adult samples. *J Behav Med*. 2000;23(4):351-365.
- 90. Spielberger CD. *State-Trait Anxiety Inventory: Bibliography*. 2nd ed. Palo Alto, CA: Consulting Psychologists Press; 1989.
- 91. Knight RG, Waal-Manning HJ, Spears GF. Some norms and reliability data for the State--Trait Anxiety Inventory and the Zung Self-Rating Depression scale. *Br J Clin Psychol.* 1983;22 (Pt 4):245-249.
- Kvaal K, Ulstein I, Nordhus IH, Engedal K. The Spielberger State-Trait Anxiety Inventory (STAI): the state scale in detecting mental disorders in geriatric patients. *Int J Geriatr Psychiatry*. 2005;20(7):629-634.
- 93. Julian LJ. Measures of anxiety: State-Trait Anxiety Inventory (STAI), Beck Anxiety Inventory (BAI), and Hospital Anxiety and Depression Scale-Anxiety (HADS-A). *Arthritis Care Res* (*Hoboken*). 2011;63 Suppl 11:S467-472.
- 94. Vlaeyen JW, Kole-Snijders AM, Boeren RG, van Eek H. Fear of movement/(re)injury in chronic low back pain and its relation to behavioral performance. *Pain.* 1995;62(3):363-372.
- 95. Roelofs J, Goubert L, Peters ML, Vlaeyen JWS, Crombez G. The Tampa Scale for Kinesiophobia: further examination of psychometric properties in patients with chronic low back pain and fibromyalgia. *European Journal of Pain*. 2004;8(5):495-502.
- Swinkels-Meewisse EJCM, Swinkels RAHM, Verbeek ALM, Vlaeyen JWS, Oostendorp RAB. Psychometric properties of the Tampa Scale for kinesiophobia and the fear-avoidance beliefs questionnaire in acute low back pain. *Man Ther.* 2003;8(1):29-36.
- 97. Sullivan MJ, Adams H, Horan S, Maher D, Boland D, Gross R. The role of perceived injustice in the experience of chronic pain and disability: scale development and validation. *J Occup Rehabil.* 2008;18(3):249-261.
- Scott W, Trost Z, Milioto M, Sullivan ML. Further Validation of a Measure of Injury-Related Injustice Perceptions to Identify Risk for Occupational Disability: A Prospective Study of Individuals with Whiplash Injury. *J Occup Rehabil.* 2013;23(4):557-565.
- 99. Rodero B, Luciano JV, Montero-Marin J, et al. Perceived injustice in fibromyalgia: psychometric characteristics of the Injustice Experience Questionnaire and relationship with pain catastrophising and pain acceptance. *J Psychosom Res.* 2012;73(2):86-91.

- Wicksell RK, Lekander M, Sorjonen K, Olsson GL. The Psychological Inflexibility in Pain Scale (PIPS)--statistical properties and model fit of an instrument to assess change processes in pain related disability. *Eur J Pain*. 2010;14(7):771 e771-714.
- 101. Eaton WW, Smith, C., Ybarra, M., Muntaner, C., Tien, A. Center for Epidemiologic Studies Depression Scale: review and revision (CESD and CESD-R). In: Lawrence Erlbaum M, NJ, ed. *M.E. Maruish (Ed.) The Use of Psychological Testing for Treatment Planning and Outcomes Assessment. Instruments for Adults.* 3rd ed 2004.
- 102. Kuptniratsaikul V, Chulakadabba S, Ratanavijitrasil S. An instrument for assessment of depression among spinal cord injury patients: comparison between the CES-D and TDI. *J Med Assoc Thai.* 2002;85(9):978-983.
- 103. LaChapelle DL, Alfano DP. Revised neurobehavioral scales of the MMPI: sensitivity and specificity in traumatic brain injury. *Appl Neuropsychol.* 2005;12(3):143-150.
- Arroll B, Goodyear-Smith F, Crengle S, et al. Validation of PHQ-2 and PHQ-9 to screen for major depression in the primary care population. *Ann Fam Med.* 2010;8(4):348-353.
- 105. Kroenke K, Spitzer RL, Williams JB. The Patient Health Questionnaire-2: validity of a two-item depression screener. *Med Care*. 2003;41(11):1284-1292.
- Lee MC, Zambreanu L, Menon DK, Tracey I. Identifying brain activity specifically related to the maintenance and perceptual consequence of central sensitization in humans. *J Neurosci.* 2008;28(45):11642-11649.
- 107. Leventhal H, Brissette I, Leventhal EA. The common-sense model of self-regulation of health and illness. In: Cameron LD L, ed. *The self-regulation of health and illness behaviour*. New York: Routledge; 2003:42–65.
- 108. Ferrari R. A prospective study of perceived injustice in whiplash victims and its relationship to recovery. *Clinical rheumatology*. 2015;34(5):975-979.
- 109. Van Dam NT, Earleywine M. Validation of the Center for Epidemiologic Studies Depression Scale—Revised (CESD-R): Pragmatic depression assessment in the general population. *Psychiatry Research*. 2011;186(1):128-132.
- 110. Kroenke K, Wu J, Bair MJ, Krebs EE, Damush TM, Tu W. Reciprocal relationship between pain and depression: a 12-month longitudinal analysis in primary care. *J Pain*. 2011;12(9):964-973.
- 111. Pavlov I. Conditioned reflexes. London: Oxford University Press. 1927.
- 112. Skinner BF. *The behavior of organisms: an experimental analysis.* Oxford, England: Appleton-Century; 1938.
- 113. Crombez G, Eccleston C, Van Damme S, Vlaeyen JW, Karoly P. Fear-avoidance model of chronic pain: the next generation. *Clin J Pain*. 2012;28(6):475-483.
- 114. Huijnen IP, Verbunt JA, Peters ML, et al. Differences in activity-related behaviour among patients with chronic low back pain. *Eur J Pain*. 2011;15(7):748-755.

- 115. Harding VR, C.de C. Williams A. Activities Training: Integrating Behavioral and Cognitive Methods with Physiotherapy in Pain Management. *J Occup Rehabil*. 1998;8(1):47-60.
- Andrews NE, Strong J, Meredith PJ. Overactivity in chronic pain: is it a valid construct? *Pain*. 2015;156(10):1991-2000.
- Moseley GL, Vlaeyen JWS. Beyond nociception: the imprecision hypothesis of chronic pain. Pain. 2015;156(1):35-38.
- 118. Fordyce WE, Fowler RS, Jr., Lehmann JF, Delateur BJ, Sand PL, Trieschmann RB. Operant conditioning in the treatment of chronic pain. *Arch Phys Med Rehabil.* 1973;54(9):399-408.
- 119. DeLongis A, Holtzman S. Coping in context: the role of stress, social support, and personality in coping. *J Pers*. 2005;73(6):1633-1656.
- 120. Nijs J, Inghelbrecht E, Daenen L, et al. Long-term functioning following whiplash injury: the role of social support and personality traits. *Clinical rheumatology*. 2011;30(7):927-935.
- 121. Moseley GL, Butler D. S. Explain Pain. 2013.
- 122. Wicksell RK, Olsson GL, Hayes SC. Psychological flexibility as a mediator of improvement in Acceptance and Commitment Therapy for patients with chronic pain following whiplash. *Eur J Pain.* 2010;14(10):1059 e1051-1059 e1011.
- 123. Prochaska JO, Norcross JC. Stages of change. *Psychotherapy: Theory, Research, Practice, Training.* 2001;38(4):443-448.
- 124. Dansie EJ, Turk DC. Assessment of patients with chronic pain. Br J Anaesth. 2013;111(1):19-25.
- 125. Skuladottir H, Halldorsdottir S. The quest for well-being: self-identified needs of women in chronic pain. *Scand J Caring Sci.* 2011;25(1):81-91.
- 126. Thompson B. Cage J, Kirk R. Living well with chronic pain: A classical grounded theory. *Disabl Rehabil.* 2019; 11:1-12
- 127. Pettersson K, Brandstrom S, Toolanen G, Hildingsson C, Nylander PO. Temperament and character: prognostic factors in whiplash patients? *Eur Spine J*. 2004;13(5):408-414.
- 128. Spinhoven P, Penelo E, de Rooij M, Penninx BW, Ormel J. Reciprocal effects of stable and temporary components of neuroticism and affective disorders: results of a longitudinal cohort study. *Psychol Med.* 2014;44(2):337-348.
- 129. Prenoveau JM, Craske MG, Zinbarg RE, Mineka S, Rose RD, Griffith JW. Are anxiety and depression just as stable as personality during late adolescence? Results from a three-year longitudinal latent variable study. *J Abnorm Psychol.* 2011;120(4):832-843.
- Anusic I, Schimmack U. Stability and Change of Personality Traits, Self-Esteem, and Well-Being: Introducing the Meta-Analytic Stability and Change Model of Retest Correlations. J Pers Soc Psychol. 2016;110(5)766-781.
- 131. Daykin AR, Richardson B. Physiotherapists' pain beliefs and their influence on the management of patients with chronic low back pain. *Spine (Phila Pa 1976).* 2004;29(7):783-795.

- 132. Haggman S, Maher CG, Refshauge KM. Screening for symptoms of depression by physical therapists managing low back pain. *Phys Ther.* 2004;84(12):1157-1166.
- 133. Valjakka AL, Salanterä S, Laitila A, Julkunen J, Hagelberg NM. The association between physicians' attitudes to psychosocial aspects of low back pain and reported clinical behaviour: A complex issue. *Scandinavian Journal of Pain*.4(1):25-30.
- 134. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. *J Physiother.* 2015;61(2):68-76.
- 135. Singla M, Jones M, Edwards I, Kumar S. Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study. *Man Ther.* 2015;20(2):328-334.
- Overmeer T, Linton SJ, Boersma K. Do physical therapists recognise established risk factors? Swedish physical therapists' evaluation in comparison to guidelines. *Physiotherapy*. 2004;90(1):35-41.
- 137. van Wilgen P, Beetsma A, Neels H, Roussel N, Nijs J. Physical therapists should integrate illness perceptions in their assessment in patients with chronic musculoskeletal pain; a qualitative analysis. *Man Ther.* 2014;19(3):229-234.
- Higgins ET. Self-discrepancy: a theory relating self and affect. *Psychol Rev.* 1987;94(3):319-340.
- 139. Sutherland R, Morley S. Self-pain enmeshment: future possible selves, sociotropy, autonomy and adjustment to chronic pain. *Pain*. 2008;137(2):366-377.

Correspondence to the Perspective article "Clinical biopsychosocial physiotherapy assessment of patients with chronic pain: The first step in pain neuroscience education"

Amarins J. Wijma^{1,2,3}, C.Paul van Wilgen^{1,2,3}, Mira Meeus^{3,4,5}, Jo Nijs^{1,3}

Affiliations:

 ¹ Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium
 ² Transcare, Transdisciplinary Outpatient Treatment Centre, the Netherlands
 ³ Pain in Motion International Research Group, (www.paininmotion.be), Belgium
 ⁴ Department of Rehabilitation Sciences and Physiotherapy, Faculty of Medicine and Health Sciences, Antwerp University, Antwerp, Belgium

⁵ Department of Rehabilitation Sciences and Physiotherapy, Faculty of Medicine and Health Sciences, Ghent University, Ghent, Belgium

Published in: Physiotherapy Theory and Practice, 33 (3), 263-265, Mar 2017

We would like to thank Rob A.B. Oostendorp and his co-authors for their positive feedback on our recent article on the biopsychosocial physiotherapy assessment of patients with chronic pain ¹. In their letter to the editor Oostendorp et al. addressed important issues regarding to the familiarity of physiotherapist with the psychosocial-aspects of the biopsychosocial assessment and the usability of the Pain-Somatic-Cognitive-Emotional-Behavioural-Social-Motivation (PSCEBSM)-model. Here we take the opportunity proposed by the Editors to respond to commentary of Oostendorp et al.

First, we would like to take the opportunity to acknowledge the important contributions of Oostendorp et al. to the field of the physiotherapy assessment by their article concerning the biopsychosocial intake by manual therapists in patients with backor neck-pain². Although the biopsychosocial perspective was introduced in medicine by Engel in 1977³, the use in clinical physiotherapy practice seems to be impaired. As well-studied by Oostendorp et al. and reviewed in our article, the abilities to perform the psychosocial aspect of the biopsychosocial intake is indeed sometimes lacking and physiotherapist mainly tend to focus on the 'Bio' aspects⁴⁻¹¹. We could not agree more with Oostendorp et al. that physiotherapists primarily asses the somatic dimension. Thereby the physiotherapy practice somehow seems to cling to the Biomedical perspective, even in the treatment of chronic pain in which guidelines prescribe a biopsychosocial approach. In intend to shift this paradigm, the clinical biopsychosocial physiotherapy assessment as described in our recent article arose. This clinical biopsychosocial assessment is designed as a guideline for physiotherapists to include all aspects of the biopsychosocial perspective, and especially to not forget the psychosocial part. However, we do recognize that only reading is insufficient to allow (deep) learning and that there is a difference between learning and changing behaviour in practice. The biopsychosocial perspective is long time incorporated in multiple guidelines regarding the treatment of patients with chronic pain and in pre- and postgraduate physiotherapy courses. Yet the profession seems to struggle to merge it into their behavioural-skills in daily practice. This might be because the majority of physiotherapists have received a biomedical-focused (pre/post) graduate education.

Specific training regarding the biopsychosocial approach can facilitate physiotherapists in biopsychosocial clinical reasoning^{12,13}. Furthermore, we suggest that novel ways are investigated, in cooperation with physiotherapists, to see how we can further improve

the implementation of the biopsychosocial perspective. As we need to change this paradigm of biomedical-behaviour in physiotherapy practice.

We agree with Oostendorp et al. that the original Somatic-Cognitive-Behavioural-Social (SCEBS) method as described by Spaendonck and Bleijenberg^{14,15} differs in content from the PSCEBSM-model described in our article. In contrast to the SCEBS-method, the PSCEBSM-model contains two additional domains useful in the biopsychosocial assessment of patient with chronic pain, namely the P (Pain) factor and M (Motivation) factor. The chapter regarding the Pain-factor, contains the differentiation between three major pain types¹⁶. Identifying the primary pain mechanism allows the physiotherapist to tailor the pain neuroscience education to the patients' complaints.

The Motivation-factor was added, because motivation for a specific treatment, such as PNE, is essential to recognize and is often related to the expectations and perceptions of patients. It is suggested that a higher motivation results in better outcomes in treatments^{17,18}. However, 'motivation' in itself is an ambiguous concept containing several elements of the SCEGS-model and personal factors, social factors, factors related to the physiotherapist and factors related to the relationship between the patient and physiotherapist¹⁹. To aid physiotherapists with this ambiguous concept, we suggested adding motivation in the PSCEBSM-model. The aforementioned can be used to help investigate the treatment expectations, psychological flexibility to change and the stage of change the patient is currently in. Furthermore, physiotherapists should be aware of the influence of their own attitude towards the patient and the influence of interactive factors on the motivation of the patient.

Where the original SCEBS method focused on questions in regard to the different biopsychosocial dimensions (for instance: "Are you anxious about particular activities?" or "Are you depressed?"), the PSCEBS-model discussed in our article also includes questionnaires. The questionnaires were added for two reasons. First, to aid physiotherapists in recognizing psychosocial factors. As mentioned above physiotherapists may feel unprepared to deal with those factors and the questionnaires can be used to screen for psychosocial factors and to aid in the conversation. Second, to evaluate the treatment in an objective manner where possible.

The last focus of the commentary by Oostendorp et al. were the recommendations for further research. Our recommendation was that the (value of the) clinical

biopsychosocial assessment as described in the article requires further investigation in clinical trials. Such trials can serve to explore the added value of the biopsychosocial assessment for treatment outcome, but depending on the study aims (e.g., examining the clinimetric properties of parts of the biopsychosocial assessment), a different design (e.g., test-retest observational studies) may be needed. We acknowledge that the form and content of such research was not described in our article, as it goes beyond the scope of our paper. Furthermore, we agree with Oostendorp et al. that investigating the physiotherapy assessment of patients with chronic pain in a total scope by Quality Indicators (QI's) certainly deserves attention. As described previously a behavioural change in physiotherapists is necessary. QI's, elements of practice that can be used to assess the (change in) quality of provided care ²⁰, are appropriate evaluation measures to assess such behaviour change. Oostendorp et al. already developed a high quality and appropriate QI set in their research into the biopsychosocial history taking of manual therapists² that might be suitable to study change in biopsychosocial behaviour of physiotherapists when implementing the PSCEBSM-model. Improving patient care should be the main focus of innovations such as the PSCEBSM-model and it is a necessity that those innovations are thoroughly evaluated.

References

- Wijma AJ, van Wilgen CP, Meeus M, Nijs J. Clinical biopsychosocial physiotherapy assessment of patients with chronic pain: The first step in pain neuroscience education. *Physiother Theory Pract.* 2016;32(5):368-384.
- 2. Oostendorp RA, Elvers H, Mikolajewska E, et al. Manual physical therapists' use of biopsychosocial history taking in the management of patients with back or neck pain in clinical practice. *ScientificWorldJournal*. 2015;2015:170463.
- 3. Engel GL. The need for a new medical model: a challenge for biomedicine. *Science*. 1977;196(4286):129-136.
- 4. Daykin AR, Richardson B. Physiotherapists' pain beliefs and their influence on the management of patients with chronic low back pain. *Spine (Phila Pa 1976).* 2004;29(7):783-795.
- 5. Haggman S, Maher CG, Refshauge KM. Screening for symptoms of depression by physical therapists managing low back pain. *Phys Ther.* 2004;84(12):1157-1166.
- 6. Valjakka AL, Salanterä S, Laitila A, Julkunen J, Hagelberg NM. The association between physicians' attitudes to psychosocial aspects of low back pain and reported clinical behaviour: A complex issue. *Scandinavian Journal of Pain*.4(1):25-30.
- 7. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. *J Physiother*. 2015;61(2):68-76.
- Singla M, Jones M, Edwards I, Kumar S. Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study. *Man Ther.* 2015;20(2):328-334.
- Overmeer T, Linton SJ, Boersma K. Do physical therapists recognise established risk factors? Swedish physical therapists' evaluation in comparison to guidelines. *Physiotherapy*. 2004;90(1):35-41.
- 10. van Wilgen CP, Beetsma A, Neels H, Roussel N, Nijs J. Physical therapists should integrate illness perceptions in their assessment in patients with chronic musculoskeletal pain; a qualitative analysis. *Man Ther.* 2014;19(3):229-234.
- 11. Roussel NA, Neels H, Kuppens K, et al. History taking by physiotherapists with low back pain patients: are illness perceptions addressed properly? *Disabil Rehabil*. 2016;38(13):1268-1279.
- Synnott A, O'Keeffe M, Bunzli S, et al. Physiotherapists report improved understanding of and attitude toward the cognitive, psychological and social dimensions of chronic low back pain after Cognitive Functional Therapy training: a qualitative study. *J Physiother.* 2016;62(4):215-221.

- 13. Jacobs CM, Guildford BJ, Travers W, Davies M, McCracken LM. Brief psychologically informed physiotherapy training is associated with changes in physiotherapists' attitudes and beliefs towards working with people with chronic pain. *Br J Pain.* 2016;10(1):38-45.
- van Spaendonck K, Bleijenberg, G. Biopsychosociale klachten. Het biopsychosociale model (SCEGS) http://www.wikifysio.nl/index.php/Biopsychosociale_klachten,_SCEGS. Published 2010.
- 15. van Spaendonck K, van de Lisdonk, EH. Biopsychosociale klachten, SCEGS. In: *Het biopsychosociale ziektemodel: een proeve van onderwijskundige operationalisatie. Gezond Onderwijs-4*. Houten/Zaventem: Bohn Stafleu Van Loghum; 1995:272-276.
- Nijs J, Torres-Cueco R, van Wilgen CP, et al. Applying modern pain neuroscience in clinical practice: criteria for the classification of central sensitization pain. *Pain Physician*. 2014;17(5):447-457.
- Nosyk B, Geller J, Guh DP, et al. The effect of motivational status on treatment outcome in the North American Opiate Medication Initiative (NAOMI) study. *Drug Alcohol Depend*. 2010;111(1-2):161-165.
- Stewart JA, Egloff N, von Kanel R, Grolimund J, Studer M, Grosse Holtforth M. Motivation for Psychological Treatment Predicts Favorable Outcomes in Multimodal Interdisciplinary Treatment for Chronic Somatoform Pain. *Psychother Psychosom.* 2016;86(1):60-61.
- 19. Maclean N, Pound P. A critical review of the concept of patient motivation in the literature on physical rehabilitation. *Soc Sci Med.* 2000;50(4):495-506.
- 20. Lawrence M, Olesen F. Indicators of Quality in Health Care. *European Journal of General Practice*. 1997;3(3):103-108.

Perspectives of patients with chronic pain who received transdisciplinary Pain Neuroscience Education

What is important in transdisciplinary pain neuroscience education? The experience of patients with chronic pain: A qualitative study

Amarins J. Wijma^{1,2,3}; Caroline M. Speksnijder^{4,5}; Astrid F. Crom-Ottens ⁶; J. M. (Corine) Knulst-Verlaan⁷; Doeke Keizer²; Jo Nijs^{1,2,3}; C. Paul van Wilgen^{1,2,3}

Affiliations

¹Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium ²Transcare, Transdisciplinary Outpatient Treatment Centre, the Netherlands ³Pain in Motion research group (www.paininmotion.be)

⁴*Physiotherapy Science, Clinical Health Sciences, Utrecht, the Netherlands, University Medical Center Utrecht*

⁵Department of Oral and Maxillofacial Surgery and Special Dental Care, Utrecht, the Netherlands

⁶Winnock Groningen, the Netherlands, https://www.winnock.nl/. ⁷Slingedael Kosakoff Center, Department of physiotherapy

Published in: Disability and Rehabilitation, 40 (18) 2181-2191, Sep 2018

Abstract

Purpose

The main focus of pain neuroscience education (PNE) is to change patients' pain perceptions. Even though PNE is extensively studied, the experiences of patients regarding the PNE process are unknown. Therefore, the aim of this study was to explore these experiences.

Methods

Fifteen patients with non-specific chronic pain from a transdisciplinary treatment centre were in-depth interviewed. Data collection and analysis were performed according to Grounded Theory.

Results

Five interacting topics emerged: 1) *"the pre-PNE phase"*, involving the primary needs to provide PNE, subthemes *"a broad intake" and "the healthcare professionals"*; 2) *"a comprehensible PNE"* containing *"understandable explanation"* and *"interaction between the physiotherapist and psychologist"*; 3) *"outcomes of PNE"* including *"awareness"*, *"finding peace of mind"* and *"fewer complaints"*; 4) *"scepticism"* containing *"doubt towards the diagnosis and PNE"*, *"disagreement with the diagnosis and PNE"* and *"PNE can be confronting"*.

Conclusion

This is the first study providing insight into the constructs contributing to the PNE experience of patients with chronic pain. The results reveal the importance of the therapeutic alliance between the patient and healthcare professional, taking time, listening, providing a clear explanation, and the possible outcomes when doing so. The findings from this study can be used to facilitate healthcare professionals in providing PNE.

Introduction

Chronic pain is a huge global issue and major healthcare problem¹. With a prevalence of 17 to 27% worldwide,²⁻⁴ chronic pain is associated with increased medical costs, decreased incomes, economic burdens⁵ and a decreased quality of life in patients^{2,5}. Chronic pain is defined as pain that persists beyond normal time of healing or pain that persists for 3 to 6 months or longer⁶.

Several chronic pain syndromes cannot be explained solely by tissue damage or inflammation (nociceptive pain) nor by a lesion in the nervous system (neuropathic pain). Often patients with chronic pain present a hyper excitability within the central nervous system, known as central sensitization⁷⁻⁹. Central sensitization enlarges the neural signalling which elicits generalized pain hypersensitivity of the somatosensory system⁸. This pain hypersensitivity results in intensification of the signalling of nociceptive or neuropathic pain and even in pain without any peripheral nociceptive signals. These neurophysiologic changes are sustained by behavioural, psychological, and environmental aspects, contributing to the continuation of pain^{10,11}. In fibromyalgia, chronic whiplash, chronic fatigue syndrome, and irritable bowel syndrome, central sensitization is often the predominant mechanism¹²⁻¹⁴. In other chronic pain populations, such as low back pain and osteoarthritis, a subgroup presents a predominant central sensitization^{12,15-17}.

Pain neuroscience education (PNE) provides healthcare professionals an opportunity to explain the pain mechanism of central sensitization, its neurophysiological changes and integrates at the same time behavioural, psychological and environmental aspects which contribute to the continuation of the pain^{18,19}. It has been shown that PNE decreases pain and improves endogenous pain inhibition, mental health, physical functioning, vitality and self-rated disability, and diminishes passive coping, kinesiophobia and catastrophizing²⁰⁻²⁷. PNE addresses patients' perceptions, such as perceptions about the cause of pain, the onset of pain, or how to cope with pain¹⁹. PNE tries to increase the knowledge of pain and change inadequate pain perceptions, such as 'my back is weak', 'my vertebra is out of line', and 'there must be something physical wrong with me'. Therefore, it is important to recognize patients' experiences of the PNE process to provide a theoretical model and framework about PNE. Unfortunately, studies exploring the experience of patients with chronic pain related to PNE are essentially lacking. Identification of these experiences may further improve the knowledge of healthcare

professionals working with PNE in daily practice. Therefore, the aim of this study was to explore the experiences of patients who recently received PNE in a transdisciplinary setting in the Netherlands in order to provide a theoretical model and framework about PNE. We undertook a qualitative research to allow the theories to emerge from the perspective of patients.

Materials and Methods

Design

Based on the methods of Grounded Theory ²⁸, we conducted a qualitative study to comprehend and theorize how patients experienced PNE. Grounded Theory is a qualitative research methodology used to deeply analyse and develop theoretical explanations. In this study, these theoretical explanations emerged from the participants' unique perspectives related to the PNE phenomenon²⁹. This PNE phenomenon was grounded in data obtained from face-to-face in-depth semi-structured interviews with patients who had partaken in PNE to expose the patients' experienced concepts of PNE. Thereafter, a focus group of PNE-healthcare professionals was performed to verify the data and optimise the input. A theoretical framework was constructed based on topics obtained from these semi-structured interviews and focus group²⁹. Our study conforms to the consolidated criteria for reporting qualitative research (COREQ)³⁰.

Participants

Between January 2013 and June 2013, 15 participants were recruited by theoretical purposeful sampling ²⁹ from Transcare (Harkema, The Netherlands) a transdisciplinary outpatient treatment centre of general practitioners, psychologists, and physiotherapists treating patients with non-specific chronic pain (Tables 1 and 2). Patients were eligible to participate if they: 1) were between 18 and 85 years of age, 2) were suffering from nonspecific chronic pain as defined by the International Association for the Study of Pain (IASP)⁶, 3) were treated with PNE at Transcare, and 4) were sufficiently able to read, speak and understand Dutch. Patients were excluded from this study when they: 1) were diagnosed with specific medical condition (*e.g.*, M. Parkinson, Rheumatoid Arthritis, stroke), 2) had cognitive impairments, 3) had dementia, 4) or had a serious psychiatric condition according to the Symptom Checklist 90 (SCL-90)³¹.

Table 1. Demographics and background characteristics

Characteristics	Individual interviews	Focus group
Men	7	3
Women	8	3
Age (years; mean, range)	47 (18–62)	47 (37–57)
Experience (years; mean, range)	-	22 (16–34)
Current relational status		
Single	4	-
Living together	3	1
Married	6	5
Divorced	5	-
Education level		
Elementary school	1	-
Junior general secondary education	1	-
Senior general secondary education	2	-
Junior secondary technical education	2	-
Intermediate vocational education	6	-
Higher professional education	2	-
Higher professional education with postgraduate qualification	-	2
University without postgraduate qualification	1	-
University with postgraduate qualification	1	2
University with postgraduate qualification and PhD	-	2
Primary income source		
Student	1	-
Job	8	6
Unemployed	2	-
Disenabled to work	4	-
Location of pain		
Neck, shoulders, arm	4	-
Back and gluteal	3	-
Abdomen	2	-
Leg, hip and knee	4	-
Widespread	2	-
Treatment		
PNE	5	-
PNE + physical therapy	3	-
PNE + psychotherapy	1	-
PNE + medication	1	-
PNE + physical therapy + psychotherapy	3	-
PNE + physical therapy + medication	1	-
PNE + physical therapy + psychotherapy + medication	1	-

Table 2. Interview respondents (n=15) and key characteristics

	Respondent ^a	Age	Job	Location of pain	Duration of pain in months	Content of treatment at Transcare
1	Scott	58	Accountant	Backpain	120	PNE and Physiotherapy
2	Alice	41	Hairdresser	Neck- and shoulderpain	18	PNE and Physiotherapy
3	Lucy	62	Disenabled to work	Backpain	264	Only PNE
4	Helen	58	Housewife	Widespread pain	60	Only PNE
5	Anna	47	Management assistant	Widespread pain	16	PNE, Physiotherapy and Psychology
6	Helga	52	Geriatric nutrition assistant	Abdominal pain	276	Only PNE
,	Finch	54	Disenabled to work	Hip and legpain	48	PNE, Physiotherapy, Psychology and medication
3	Dani	18	Student	Neck- and shoulderpain	60	PNE and Physiotherapy
)	Sam	41	Deliveryman	Buttock- and legpain	24	PNE and medication
0	Freddy	35	Carpenter	Kneepain	20	Only PNE
1	Walt	33	Trader and investment advisor	Groin- and upperlegpain	180	Only PNE
2	John	62	Metalworker	Abdominal- and ribpain	120	PNE, Physiotherapy and medication
3	Amy	49	Housekeeper	Neck-, shoulder- and armpain	6	PNE and Psychology
4	Rene	56	Unemployed	Neck-shoulder- and armpain	8	PNE, Physiotherapy and Psychology
15	Wendi	40	Tax lawyer	Neck-, back and hippain	36	PNE, Physiotherapy and Psychology

^aPseudonyms used throughout the text

Patients were approached after their PNE session by the treating healthcare professionals and asked if they were willing to participate in the study. When they agreed, the first author contacted the patients and appointments were made. Furthermore, written information and informed consent were sent by email or postal mail. In the written information is stated that study participation would not influence their treatment. Confidentiality was guaranteed and all materials were handled anonymously. This study was conducted in compliance with Dutch law and the principles of the Declaration of Helsinki. All participants signed informed consent.

According to Grounded Theory, a purposeful homogenous sample was used in the first seven interviews. This homogenous sample consisted of respondents who were appointed by the healthcare professionals as having a positive attitude towards PNE. After these seven interviews, eight respondents were heterogeneously selected irrespective of their attitude towards PNE in order to obtain a realistic sample²⁹.

Pain Neuroscience Education

Transcare is a transdisciplinary outpatient treatment centre of general practitioners, psychologists, and physiotherapists treating patients with non-specific chronic pain. Transdisciplinary means that these healthcare professionals collaborate intensively, with flexible boundaries and roles, learn simultaneously, and have a shared biopsychosocial view on chronic pain^{32,33}. Transcares' aim is to provide transdisciplinary, low cost, evidence-based, patient centered PNE, and treatment for patients with chronic pain, where the initial contact with the patient is key. The treatment is focused on changing perceptions about pain and providing self-management tools to drastically reduce and minimize further medical care for chronic pain. Patients who receive treatment at Transcare first have a three-hour assessment - one hour for every discipline (general practioner, psychologist, physiotherapist). After these extensive intakes, the healthcare professionals make a biopsychosocial pain analysis with predominant pain mechanisms and contributing factors. The diagnosis of the predominant pain mechanism (i.e., nociceptive, neuropathic, central sensitization pain, or a combination of these) is based on the stepwise approach and algorithm for the classification of central sensitization pain³⁴. This algorithm can guide the therapist to determine if the pain is predominantly neuropathic or not and if not by differentiating between predominantly nociceptive pain and central sensitization. After the intake the patient receives PNE comparable to Explain Pain³⁵, adapted to the predominant pain mechanism(s) and contributing factors, using the following mode of administration: one week after the intake, the general practitioner provides the patient information about the team's biopsychosocial diagnosis in a 10-min conversation in a one-on-one setting with the patient and partner (spouse, family member, or friend). The general practitioner gives verbal information about the predominant pain mechanism(s), supported by a booklet based on the Dutch book Pain Education: A Practical Guide for Healthcare Professionals (Pijneducatie, een praktische handleiding voor [para]medici)³⁶. Two weeks after the intake, the patient and partner participate in a one-hour PNE session provided by the psychologist and physiotherapist. Once again, their pain is explained in a patientspecific manner. At first, the neuroscience explaining their pain is highlighted; secondly,

the biopsychosocial factors contributing to the persistence of pain are discussed; and finally, patient centered treatment options are discussed. Treatment options after PNE can consist of the following: none (reasons: no need for further treatment, no common ground with the healthcare professionals, patient finds it too burdensome, etc.), medication, physiotherapy, psychotherapy, or a combination of the latter two. All treatments are evidence based and patient centered. Medication therapy consists of decreasing use of ineffective medication and (if necessary) increasing centrally acting medication. Physiotherapy and psychotherapy are based on the principles of Cognitive Behavioural Therapy (CBT), including graded activity, graded exposure, relaxation techniques, mindfulness, body-oriented work, interventions for depression and anxiety, Eye Movement Desensitization and Reprocessing (EMDR), etc.

Procedure

The individual face-to-face, in-depth interviews were conducted with patients who received PNE approximately a month prior. This time interval was based on the theoretic construct which provides respondents with some time to process the bulk of information they received during the PNE sessions. Before the interviews, a semi-structured interview guide was constructed and piloted tested during two test interviews to ensure the quality of the interview guide. Table 3 contains a summary of the interview guide topics.

The interviews were contained open questions according to the 'river model', in which the interview 'flows like a river' and if a topic 'runs dry' the interviewer returns to the 'mainstream of the river'³⁷. To create an open character during the interviews prior to questioning rapport was established between the researcher and the respondents. Respondents were invited to tell the researcher to stop questioning certain topics if they felt uncomfortable or did not want to explain them any further.

Table 3. A summary of the themes mentioned in the interview guides

Initial interview guide	Interview guide after initial axial coding
Demographics	Demographics
Experiences related to the process (waiting time for intake, appointments, environment, travel distance)	-
Overall experiences of Transcare (overall experience, experience with intake, experience first PNE with the general practitioner, multidisciplinary cooperation, medication therapy, physiotherapy, psychotherapy, change in pain intensity (if applicable), change in pain perceptions)	Overall experiences of Transcare (overall experience, experience first PNE with the general practitioner, multidisciplinary cooperation, medication therapy, physiotherapy, psychotherapy, change in pain intensity (if applicable), change in pain perceptions)
-	Experiences in regard to the intake day (How did you experience the intake)
Specific experiences of PNE session with Physiotherapist and Psychologist (What was your experience, what was important in that session, what could improve, communication, interaction, attitude of healthcare professionals towards the patient, language used in the PNE, comprehensibility of the PNE, metaphors/ examples used during PNE, support/ empowerment/to have the feeling of being understood by the healthcare professionals, reassurance by the message of PNE?, repetition of PNE)	Specific experiences of PNE session with Physiotherapist and Psychologist (What was your experience, what was important in that session, what could improve, begin taken seriously, feeling understood by the healthcare professionals, being yourself, feeling comfortable, personal involvement of the healthcare professionals, having a connection with the healthcare professionals, healthcare professionals were open, personal, involved, interaction between both during the PNE, Physiotherapist and Psychologist complemented each other, observations by Psychologist, drawings made, clear explanation, why was the explanation clear, to find peace and clarity, bringing those close to you to the PNE, increased awareness, understanding your symptoms, consciousness of your body, improved self-control on the symptoms, does he/she experience fewer symptoms)
Influence of PNE on patient and daily life	Influence of PNE on patient and daily life
Anything you would like to bring in yourself?	Anything you would like to bring in yourself?

Analysis

The transcriptions of the interviews and the focus group were analysed using Grounded Theory²⁹. The interviews first were transcribed verbatim with transcription software F4 (Dr Dresing & Pehl GmbH, Marburg, Germany)³⁸. After this transcription, the data were analysed with open, axial, and selective coding according to Grounded Theory²⁹. Coding was the interpretative process in which conceptual labels were given to the data²⁹.

During the analysis, the Qualitative Analysis Guide of Leuven (QUAGOL)³⁹ was used to guide the Grounded Theory analysis process. The QUAGOL is a practice-based guide that has two parts with five stages in which the researchers are guided in a flexible and iterative constant comparing process through the analysis of data, and it respects the different phases of coding used in GT³⁹. Table 4 shows the stages we pursued during the analysis process. QSR International's NVivo 10 software (QRS International Pty Ltd., Doncaster, Australia) was used during the second stage of the QUAGOL to help shape the coding phases.

Table 4. Stages of analysis based on the QUAGOL and GT ^{28,39}

Stage Action

- 1 The transcription was first read, whereby key phrases were underlined, and the meaning of the text was interpreted tentatively.
- 2 The researcher reread the transcript in order to phrase its understanding, then set aside the transcript and wrote a narrative report. This was guided by the following question: "What are the essential characteristics of the interviewee's story that may contribute to a better insight into the research topic?"
- 3 The researcher constructed a conceptual interview scheme to provide concepts that appeared relevant to the research topic.
- 4 The appropriateness was verified by rereading the interview. The third and fourth stages were iterative.
- 5 The interview schemes were compared within and with other schemes and data of other interviews.
- 6 In the sixth stage, QSR International's NVivo 10 software was used. This was the start of the actual coding process, where open codes based on the conceptual interview schemes were listed.
- 7 The interviews were read again, and the codes were filled with significant passes of the interviews (axial coding).
- 8 The codes and topics were integrated into a meaningful conceptual framework and story-line, also known as the selective coding phase in GT.
- 9 A conceptual framework in response to the research question was created. In the last stage, the results were described on a conceptual and theoretical level, grounded in the interview data.

The first author performed the main analysis, whereby the evolving codes were repeatedly discussed with the second, third, and last author. Interviews and analysis of the interviews were performed in Dutch until data saturation was achieved, which means that no new codes or categories emerged from the data²⁹. The theory was translated into English, and an effort was made to convey the sense of the Dutch language in order to keep some perspective of context.

After the initial axial coding phase of the first seven interviews, the interview guide was adjusted based on the emerged codes (Table 3), and more detailed questions were asked to shape the next analysis phase²⁹. During and after the interviews and throughout the analysis, AJW created memos related to the evolving theory and the process, which were later used in the analysis²⁹.

After axial coding of the first seven interviews, member checks were performed. A member check is a technique whereby the data are taken back to the interviewees and discussed and tested with a few of the interviewees from whom the data were originally obtained⁴⁰. During these member checks, the initial theory, which evolved from the axial coding of the first seven interviews was discussed with two respondents of the seven respondents. The topics of the axial coding were read, and an in-depth discussion was held with both respondents. After these member checks, the initial codes of the first axial coding were adapted based on the feedback of the respondents. After development of the complete theory, a focus group was organized to discuss, optimize and verify the evolved theory of the axial coding. This focus group was conducted because the healthcare professionals have extensive knowledge of, and experience with, chronic pain and PNE. Their internal theoretical model of PNE is influenced by the expressed experiences of PNE from their prior patients. By presenting the results from the axial coding to the focus group a discussion emerged based on the axial coding themes found in the expressed experiences of the respondents and the internal frameworks of the healthcare professionals. This discussion enriched the analysis as some themes of the axial coding were confirmed, while others were adjusted by fusion or rearrangement of the themes. Nothing from the axial coding was removed as a result of the focus group. This focus group was led by the first author and consisted of healthcare professionals of Transcare (Table 1). All interviews and member checks were done by the first author. She received extensive interview training at Evers Research & Training, Rotterdam, the Netherlands.

Trustworthiness

The criteria of Lincoln and Guba⁴⁰ were used to clarify our efforts to ensure the quality of this research throughout the research process and output. Credibility (confidence in the "truth" of the findings) was ensured by: a pilot-tested interview guide, audio recordings and verbatim transcriptions of the interviews, an open non-judgmental atmosphere during the interviews, bracketing out during the data collection and analysis, member checks of the first analysis, and negative case analysis (the inclusion of respondents

with outlying experiences), peer debriefing of axial coding and theory development by the first and last author, thickly described data, and limitations about the transferability (the findings have applicability in other contexts) are made. Dependability (the findings are consistent and could be repeated) was ensured by a description and continued monitoring of the process and product of the research by the first, third, and last author and close monitoring by the other authors. Confirmability (the extent to which the findings of a study are shaped by the respondents) was ensured by the open nonjudgmental atmosphere of the interviews, bracketing out, data triangulation by semistructured interviews and a focus group, an iterative analysis, a researcher triangulation by the first and third author, and all of the coding phases were continuously discussed between the first, third, and fourth author.

Results

Respondents

Nineteen patients were asked to participate. Four patients were not willing to participate for unknown reasons. In total, 15 patients were interviewed (seven men, eight women) until saturation was reached. The average age was 47 (18 to 62) years (Tables 1 and 2). Informed consent was obtained from all respondents. The interviews lasted 35 to 86 minutes; 14 were held at the respondents' homes and one at a physiotherapy practice. Three member checks were planned, one respondent cancelled. The two member checks lasted 26 and 23 minutes. The focus group lasted 135 minutes, with six members of Transcare: one general practitioner, two psychologists, two physiotherapists, and one researcher (three men and three women, with an average age of 46 [37 to 57] years; Table 1).

Findings

Four interacting topics emerged and are depicted in Figure 1. The first topic was "the pre-PNE phase" and involved the primary needs in order to provide PNE with the respondents. Subthemes were "a broad intake" and "the healthcare professionals". The second topic, "a comprehensible PNE", contained an "understandable explanation" and the "interaction between the physiotherapist and psychologist". The third topic involved the "outcomes of PNE", including the subthemes "awareness", "finding peace of mind", and "fewer complaints". The fourth topic, "scepticism", contained "doubt towards the diagnosis and PNE", "disagreement with diagnosis and PNE", and "PNE can be confronting". Scepticism can negatively impact the third topic.



Figure 1. Conceptual model of constructs influencing the experience of PNE

The Pre-PNE Phase

A broad intake

Respondents mentioned that the intake creates conditions for PNE; it was their first encounter with and impression of the healthcare professionals. Because the intake was elaborate it made them feel that they were able to tell their complete story. Respondents also frequently stated that the intake was intense and emotionally exhausting. They were confronted with their problems, symptoms, and functional limitations for three consecutive hours. During the intake, the respondents felt the need to clarify their symptoms to both the healthcare professionals and themselves. By doing this, some already came to an increased awareness and better understanding of their complaints, symptoms, and contributing factors. Respondents sometimes took a sceptical stand towards elaborating the psychological aspects of their pain during the intake. Yet, having the intake include time with the psychologist made them more willing to accept the biopsychosocial view.

The healthcare professionals

The respondents' first impression of the healthcare professionals during the intake was crucial and mentioned as a primary condition for PNE. The healthcare professionals were perceived as empathetic, friendly, open, nice, relaxed, spontaneous, good listeners, and quickly got familiar with the respondents. The respondents experienced the healthcare professionals as interested and involved and felt that they cared for them as a person. This became apparent due to the nonverbal communication and voice tone when addressing important issues. These interpersonal aspects of the healthcare professionals made the respondents feel comfortable, at ease, connected, and understood. The interaction was perceived as pleasant, open, and honest. The interest, involvement, and concerns made the respondents feel that they were taken seriously and recognized. Therefore, they felt they could tell their story from their own perspective. They felt they were being seen rather than a bodily piece. Consequently, this increased the willingness to open up about themselves, their problems, and psychosocial factors.

Helen: "I was able to tell from my own perspective how something feels, because I felt heard. I felt that I was taken seriously. And when I get that feeling, the other one (healthcare professional) can get a clearer image of me".

Respondents often felt nervous for the intake's psychological assessment and had some reservations. However, the psychologist generally made them feel at ease, and their suspicion disappeared quickly.

Scott: "Well, I had never been to a psychologist before, so for me it was the first time, and I must admit it was 200% positive (laughs) ... I had the idea that you have to be a bit, well, crazy or very depressed to go to a psychologist. You know. And that's not true, of course. She clarified this right away and put me at ease".

The respondents felt that the healthcare professionals were competent, experts in the field of pain and knew more about chronic pain than just the knowledge from their own

profession. For instance, the physiotherapist had knowledge about the psychological aspects of pain, and the psychologist had awareness of body functions. The respondents described the healthcare professionals as a good team that collaborated intensively, whereby the healthcare professionals complemented each other.

A comprehensible PNE

Although the neurophysiology of PNE is theoretical and for some difficult to understand, respondents found the explanation good and clear, and they would not change a thing. PNE was perceived as a quest through their pain problem together with the healthcare professionals. The respondents often saw themselves and their symptoms reflected during the PNE.

Understandable explanation

The respondents mentioned that the pain mechanisms (nociceptive, neuropathic, central sensitization pain, or a combination of these) were explained step by step in a clear manner and at an easy, steady pace. The repetition of PNE (first by the general practitioner and booklet, and then by a combined session with a physiotherapist and psychologist) was perceived as important for the respondents, as it made the PNE understandable. Respondents said the plain language, the explanation on paper and the personal explanation clarified the PNE. The drawings in the booklet, which were also used during the PNE, were illustrative. The examples used during the PNE were clear and comprehensible, taken from real life and focused on the individual. Use of the burglar/fire alarm metaphor was found to be very illustrative^{19,36}. During nociceptive pain the alarm goes off because there is a burglar intruder, in central sensitization pain the alarm is so sensitive it goes off even without a burglar present. Some respondents indicated that they would not have understood the PNE without these metaphors.

Wendy: "They explained it very well, because at the general practitioner I got a blue booklet about chronic pain. About nerves and how it all works. That your body is actually a burglar alarm set incorrectly. That one I remember, when people ask me how I am doing and what was discovered, I tell them that. It [the metaphor] appeals to the imagination".

Respondents felt that they were now able to explain their pain to others, thereby receiving more recognition for their problems.

All respondents were asked to bring their partner or a close friend to the combined PNE with the physiotherapist and psychologist, which was generally valued positively. It made the explanation more understandable: *"two hear more than one, two remember more"*. Furthermore, the partner or friend could share their ideas about the contributing factors, enhancing the translation of PNE into the respondents' efforts to self-management the pain during daily life. For some, however, bringing their partner was unnecessary or even regarded as if the healthcare professionals thought that the respondents would not be honest about the treatment at home.

Interaction between the physiotherapist and psychologist

It was noticed by the respondents that there was interplay between the physiotherapist and psychologist during the session. They complemented each other. Often, one provided the explanation while the other was drawing, or one was talking and the other observed the respondent. It made the respondents feel like they kept an eye on them, checked if the provided information was understood and reflected on the respondents' thoughts and emotions. Respondents mentioned that this facilitated the understanding of the PNE and enhanced the translation to the respondents' daily self-management.

Furthermore, respondents mentioned that the presence of two healthcare professionals speaking the same language confirmed that the pain mechanism (nociceptive, neuropathic, central sensitization pain, or a combination) as explained by the healthcare professionals was actually true, making it easier to accept.

Wendy: "They are very clear and... How do I explain that? They stand behind their opinion".

Outcomes of PNE

Awareness

The PNE initiated a process of awareness in which the respondents' gained more insight in their symptoms and how to cope with their condition. Furthermore, respondents mentioned that they became more aware that they needed to handle their body more respectful. In addition, respondents mentioned that due to the PNE they gained more self-control.

Insight into symptoms

Respondents often came to find a cause and solution for their chronic pain. Clarifying their pain was important and often caused a change in pain perception and acknowledged the contributing factors. When this perception change occurred, respondents learned to acknowledge the balance and interaction between body and mind. Furthermore, they gained insight into the way their behaviour, emotions, and perceptions influenced their pain. Respondents mentioned that they became aware of the influence of previous events on pain, took life a bit easier, and learned to take their symptoms seriously.

For some, PNE was a confirmation of what they already knew, sometimes subconsciously without accepting it, and for others a means to freshen up on what they heard previously.

Consciousness of their body

As they gained more insight into their pain, the respondents learned to be more conscious of their body in a behavioural manner. They were less preoccupied and dealt with their pain more adequately. The respondents learned the influence of behaviour on their physical and psychological symptoms. In addition, they learned how to use their body more appropriately, being more conscious of tense postures and the positive influence of relaxation. Respondents mentioned that they learned to express their limits, even though it is difficult not to cross their boundaries. They experienced the positive influence of listening to their body and the advice they had received.

Gaining self-control

The insight into symptoms and improved consciousness of their body caused improved self-control and self-management. Most respondents gained more insight into their symptoms and subsequently perceived more control over their symptoms. They were able to handle their problems more adequately by accepting and learning to deal with their pain, being less occupied with pain and learning to put it at ease. The respondents mentioned that they dealt better with their pain, and that this was due to the practical translation of the PNE to the respondents' daily life.

Walt: "I think that (the education of) Transcare is good for awareness in that you don't always have to think: 'Oh, I'm in pain and I can't do anything'. It's about taking more responsibility yourself. (...) Whether physically or mentally".

Still, not all respondents experienced more self-control. Some mentioned that it was too soon and they were in too much of a psychological conflict to experience more self-control. These respondents were searching for strategies to deal with their psychological problems and pain, trying to regain balance.

Finding peace of mind

Some respondents experienced peace of mind after PNE. The acknowledgement of their pain – "*it's not in my head*", "*the pain is real*" – was important. Likewise, the explanation that there was nothing damaged was valued. In addition, for some, PNE was the last piece of the puzzle in their search for the cause and treatment of their symptoms. They were able to stop searching and found peace of mind.

Helga: "And now I found some peace of mind. Like, well, stop searching. There is, so far, nothing more to do. (...) So, well, a bit of peace of mind. Some clarity".

Some were not reassured by the explanation that there is no tissue damage accountable for their pain. These respondents found support in their perception that there must be something physically wrong.

Rene: "The reassurance is, at least that's how I interpreted it, that there is pain but no damage. And that I don't know, I don't know if there is no damage. I'm still in doubt".

Fewer symptoms

Some respondents had fewer symptoms. They attributed this to the received tips, advice, and exercises. They felt they had made progress, were able to do more, had less pain, felt less down, and slept better. A few had changing symptoms; they experienced symptom reduction, but while stressed or when performing physical effort, they temporarily experienced more pain.

Other respondents did not experience a reduction in symptoms. Some said it was too early in the process for symptom reduction. Others did not have fewer symptoms but were better able to handle them. Two respondents had more symptoms; they were hiding their pain before and were now confronted with their pain and pain-related problems. All respondents, except for two, hoped for recovery and less pain. These two had chronic pain for a substantial time period and were previously told by doctors that their pain was chronic. Even though they hoped it would diminish, they knew their pain was not going to change. They wanted an explanation for their pain and help in dealing with their pain. The other respondents found it important to know whether they had the ability to experience less pain.

Scepticism

Doubt towards the diagnosis and PNE

Most respondents were at first a little sceptical about their "diagnosis" of the predominant pain mechanism. According to the response of the focus group, this was normal. They are likely to have heard numerous explanations for their pain throughout their search for relief. Respondents often did not completely agree with the diagnosis. They knew they had chronic pain, but had doubts or were emotionally not ready to accept it. Some could not accept it as the predominant cause for their symptoms. Respondents described an internal struggle between accepting the diagnosis on the one hand and still feeling the urge to search for another (mostly physical) cause on the other hand. Respondents mentioned the conflict between the biopsychosocial perspectives of the PNE and previous physically focused treatments.

Some respondents who did not have any additional medical examinations found it hard to accept that there was nothing physically wrong without such examinations being performed.

Wendy: "Okay, so it's chronic pain. That is also reassuring in a sense that there is nothing going on. It's really double! Because neurologically there is nothing wrong, somehow you don't trust that you've not had scans made and no real medical examinations were done. On the other hand, it's comforting to hear that there is nothing serious going on".

Others were still distrustful and afraid to be disappointed due to previous negative experiences in healthcare; i.e. they were sent to various doctors, who all provided different explanations for their pain, which made them feel that they were not taken seriously, going from "pillar to post". These respondents said that *"seeing (improvement) is believing"*.

John: "Look, we still have some distrust, but that's because we were sent away by several doctors in the past with the message 'Learn to live with it'. You know, Transcare says, we know what is going on, that's it. We just hope that they are right".

Disagreement with the diagnosis and PNE

Two respondents rejected the PNE completely and did not agree with the diagnosis at all. They found PNE comprehensible but did not recognize it as applicable to themselves. They believed in a physical cause rather than a co-psychological cause for their symptoms. Their a priori expectations of Transcare were low. Nevertheless, they were satisfied with the intake, the interpersonal aspects of the healthcare professionals, the friendly environment and contact between them and the healthcare professionals, the comprehensibility of PNE, and the advice they received. They indicated that they would recommend the treatment centre to other patients with chronic pain.

One respondent missed the link between the PNE and daily activities. The respondent agreed with having central sensitization as the main pain mechanism and understood it, but for him he felt it was not helpful because he was in too much pain.

PNE can be confronting

Some respondents felt that they were rapidly marked with central sensitization as the main pain mechanism and were therefore stigmatized. They felt that the way central sensitization was explained to them was too confronting and PNE should be given more carefully. The healthcare professionals recognized that PNE can be confronting due to the nature of talking about contributing factors, which are often cognitive, behavioural, and emotional.

Discussion

In this study, we aimed to investigate the experiences of PNE in patients with chronic pain in a transdisciplinary setting. This study showed that in the process of PNE several topics and subthemes are important to patients with non-specific chronic pain.

The first identified topic was "the pre-PNE phase", the phase prior to the PNE in which the respondents met the healthcare professionals during a broad intake. The intake is described by the respondents as essential; the broad transdisciplinary assessment

already makes them more aware of the biopsychosocial view of pain. The questions during the intake of the respondents are based on the Pain, Somatic, Cognitive, Emotional, Behavioural, Social, and Motivational aspects - model^{41,42}. This model, as well as the involvement of a psychologist in the team and the pain-analysis after the intake allow the healthcare professionals to fully review all aspects of the respondent's pain problem. The questions related to this model and the involvement of the psychologist made the respondents more aware of the biopsychosocial factors related to pain. Through this assessment, the patient's perceptions and a priori expectations were already altered. Furthermore, the results showed that the patient centered, serious, interested, and friendly interpersonal approach of the healthcare professionals during the intake enhances the further treatment and the acceptance and understanding of PNE. Earlier research also showed the influence of contextual effects on health outcomes⁴³. This systematic review showed that physicians who adopt a warm, friendly, and reassuring role are more effective and showed also that the working-ground of the treatment (i.e. providing patient centered care, taking time, building rapport, and building a therapeutic alliance) enhances health outcomes^{44,45}. Another review⁴⁶ , reporting on the interaction between patients with chronic pain and healthcare professionals and its influence on the patient's self-management, describes that patients are more open when healthcare professionals are empathic, and that patients benefit from a biopsychosocial and patient centered approach. This also corresponds with our results within the topic "the pre-PNE phase".

The second identified topic, "a comprehensible PNE", comprised of "understandable explanation" and the "interaction between the physiotherapist and psychologist". Research has shown that receiving comprehensible information is necessary, as it improves the satisfaction with care, diminishes symptoms, and improves adherence⁴⁶. Until recently, PNE was only studied in a monodisciplinary (physiotherapy) setting, but not in a transdisciplinary setting. In the transdisciplinary setting of Transcare the second PNE session is given by a physiotherapist together with a psychologist. During the interviews per formed in this study, the interaction between the physiotherapist and psychologist during the PNE was mentioned by the respondents as a facilitator in the understanding of PNE, and the translation to their daily self-management. Future work should explore whether this transdisciplinary PNE is superior over monodisciplinary PNE.

The third topic involved the "outcomes of PNE", including the subthemes "awareness", "finding peace of mind", and "fewer symptoms". Respondents reported to experience

more awareness related to their insight into symptoms, better coping skills, and gained more self-control. The latter is similar to findings from a qualitative study of PNE provided in group sessions, which found that PNE in group sessions had the same reconceptualization of pain, thereby improving the pain management in eight out of 10 patients⁴⁷. Furthermore, respondents in our study often found peace of mind and some experienced fewer symptoms after PNE. The reconceptualization and clarification of the pain are known to enhance endogenous pain inhibition and neurophysiological changes in patients with chronic pain by informing the patients that their pain is real, but not dangerous, thereby reducing symptoms^{21,22,48}. However, there was a subgroup of respondents in the current study that did not experience fewer symptoms. This subgroup was not homogenous, for instance some respondents mentioned a better self-control of their pain, but did not experience fewer symptoms, while others were still trying to cope with the psychosocial factors contributing to their pain experience.

The fourth topic, "scepticism", contained "doubt towards the diagnosis and PNE", "disagreement with the diagnosis and PNE", and "PNE can be confronting". Some patients had doubt and "scepticism" towards the diagnosis, whereby a few respondents rejected their diagnoses completely. This is in accordance with a qualitative study on group PNE⁴⁷ which found that a minority of the respondents thought that PNE was irrelevant to them. These respondents reported no benefits of the group PNE. This qualitative study suggested that these respondents probably did not perceive enough relevance and benefits out of the PNE for themselves. The respondents in our study who disagreed with the PNE believed in a physical cause rather than a co-psychological cause, and their a priori expectations were low. This is in accordance with a study that found that patients' readiness to change, an increased commitment to self-management, and initial hesitation about the treatment predicts the satisfaction and likelihood of completion of the program⁴⁹⁻⁵¹.

The framework presented in this article is, to our knowledge, the first theoretical framework of the PNE experiences of respondents with chronic pain. It is constructed out of the topics found during the selective coding phase, according to Grounded Theory and the Qualitative Analysis Guide of Leuven^{28,39}. "*The pre-PNE phase*" is the first topic one encounters in the theoretical framework, as was identified by the respondents with the healthcare professionals and their biopsychosocial views. Then, as the respondents proceed from the intake to the PNE, the theoretical framework shows the topics associated with what the respondents viewed as "*a comprehensible*"

PNE". "*The pre-PNE phase*" and "*a comprehensible PNE*" both had an influence on the respondents. This influence is reflected in the connections between these topics and the topic "*outcomes of PNE*". On the lower end of the framework the last topic "*scepticism*" is displayed. The connections between this topic and "*outcomes of PNE*" display the influence of the topic "*scepticism*" on the outcomes of PNE. Furthermore, "*scepticism*" has a two-way connection with "*the pre-PNE phase*": by the intake and the friendly healthcare professionals some scepticism might reduce, and the content of the topic "*scepticism*" can also have his influence on the intake and healthcare professionals. As is the case with qualitative research, the theoretical framework was shaped through the researchers' interpretation of the results and therefore subjected to further discussion and intensification.

Reflexivity

Even though it is a strength that the researcher performing the interviews received extensive interview training at Evers Research & Training (Rotterdam, The Netherlands), and completed a course on qualitative research and analysis during her education in physiotherapy science (Clinical Health Sciences, University Medical Center Utrecht, Utrecht, The Netherlands), she had no prior experience in interviewing and undertaking gualitative research. As the researcher plays a substantial role in the research process, his/her experience has influenced the results, even though considerable effort was made by the actions mentioned in "Trustworthiness" to diminish this influence. There are some strengths and limitations to mention. Firstly, the key strength of this research is that the research question and research design to answer the research question are both highly relevant to healthcare professions working with patients with chronic pain. Secondly, there are many different visions on validity and reliability within gualitative research; however, there are efforts that can be made by researchers to ensure the trustworthiness of their research^{40.} We believe that the actions mentioned in "Trustworthiness" are strengths, as they improved the quality of our research. Nevertheless, transferability (generalizability) in qualitative research is always a delicate issue⁵². In this study, there are multiple areas that compromise the generalizability and external validity. Primarily, because the results and framework of this study are derived from one setting, instead of multiple, the generalizability and external validity is moderated. Next, the setting in itself is different from other PNE settings: there is a repetition of PNE by the general practitioner, a booklet, and combined session which is new. Furthermore, the transdisciplinary nature of the PNE, provided by a physiotherapist together with and psychologist, has not been studied before. Hence,

the effectiveness of transdisciplinary treatments requires further study and does not correspond with existing data about (monodisciplinary) PNE²⁷. Therefore, one should be careful in generalizing the study findings to monodisciplinary treatment settings that provide PNE. Thirdly, it is a strength and limitation that the data collection was performed in a clinical setting. This is a strength as it provides relevant and bottom-up research on how PNE is provided in clinical settings. However, there is a possibility that the further treatment some respondents have received at Transcare, influenced the experiences they reported on their PNE. To segregate these effects the interviewer specifically first asked the respondents about their overall experience, followed by questioning then about their experiences with the PNE sessions. Furthermore, there were also respondents included who only received PNE. We have indicated whether the respondents received further treatment and what kind of treatment in Table 2. Fourthly, the Qualitative Analysis Guide of Leuven³⁹ was used for data analysis. This relatively new tool guides the researcher through a structured data analysis³⁹. The limitation is that the Qualitative Analysis Guide of Leuven strongly recommends teamwork throughout the entire analysis, applying all steps of the Qualitative Analysis Guide of Leuven together. Due to practical issues, among which were time constraints, this was not feasible.

Future research should focus on investigating the experiences of patients with chronic pain with PNE in different settings and cultures. For instance, there is no interaction between the physiotherapist and psychologist in monodisciplinary PNE, and this may have a different effect on the experiences of patients with PNE. As for cultures, it is known that each culture has different beliefs and explanations of pain and methods to cope with pain⁵³. PNE in the Netherlands can have a different outcome than PNE in other countries. To develop a holistic understanding of PNE, it is also important that future research observes PNE in practice. Observation not only checks what respondents say in the interviews, it records events in clinical practice that respondents may not notice, may not discuss or may misinterpret. Furthermore, participant observational qualitative research also incorporates the ability to check for nonverbal expressions, interactions, and ways of communication⁵⁴.

Conclusion

This is the first study providing insight into the constructs contributing to the experience of transdisciplinary PNE in patients with chronic pain. The results reveal the importance of the therapeutic alliance between the patient and caregiver, taking time, listening, providing a clear explanation, and the possible outcomes when doing so. This takes time, involvement, good interpersonal factors, and a biopsychosocial view on pain, whereby a team interaction may enhance the outcomes following PNE. When doing so, the perceptual changes about pain will improve in a cognitive, behavioural manner.

References

- EFIC. EFIC's declaration on pain as a major health problem, a disease in its own right. 2010. http://www.efic.org/index.asp?sub=724B97A2EjBu1C.
- 2. Breivik H, Collett B, Ventafridda V, Cohen R, Gallacher D. Survey of chronic pain in Europe: prevalence, impact on daily life, and treatment. *Eur J Pain*. 2006;10(4):287-333.
- Reid KJ, Harker J, Bala MM, et al. Epidemiology of chronic non-cancer pain in Europe: narrative review of prevalence, pain treatments and pain impact. *Curr Med Res Opin*. 2011;27(2):449-462.
- 4. Leadley RM, Armstrong N, Lee YC, Allen A, Kleijnen J. Chronic diseases in the European Union: the prevalence and health cost implications of chronic pain. *J Pain Palliat Care Pharmacother*. 2012;26(4):310-325.
- Bekkering GE, Bala MM, Reid K, et al. Epidemiology of chronic pain and its treatment in The Netherlands. *Neth J Med.* 2011;69(3):141-153.
- 6. Merskey H, Bogduk N., International Association for the Study of Pain. Task Force on Taxonomy. *Classification of chronic pain: descriptions of chronic pain syndromes and definitions of pain terms.* 2 ed. the University of Michigan: IASP Press; 1994.
- 7. Latremoliere A, Woolf CJ. Central sensitization: a generator of pain hypersensitivity by central neural plasticity. *J Pain*. 2009;10(9):895-926.
- 8. Woolf CJ, Salter MW. Neuronal plasticity: increasing the gain in pain. *Science*. 2000;288(5472):1765-1769.
- 9. LaMotte RH, Shain CN, Simone DA, Tsai EF. Neurogenic hyperalgesia: psychophysical studies of underlying mechanisms. *J Neurophysiol*. 1991;66(1):190-211.
- 10. Gracely RH, Geisser ME, Giesecke T, et al. Pain catastrophizing and neural responses to pain among persons with fibromyalgia. *Brain.* 2004;127(Pt 4):835-843.
- 11. Turk DC, Okifuji A. Psychological factors in chronic pain: evolution and revolution. *J Consult Clin Psychol.* 2002;70(3):678-690.
- 12. Staud R. Evidence for shared pain mechanisms in osteoarthritis, low back pain, and fibromyalgia. *Curr Rheumatol Rep.* 2011;13(6):513-520.
- 13. Nijs J, Meeus M, Van Oosterwijck J, et al. In the mind or in the brain? Scientific evidence for central sensitisation in chronic fatigue syndrome. *Eur J Clin Invest*. 2012;42(2):203-212.
- 14. Yunus MB. Fibromyalgia and overlapping disorders: the unifying concept of central sensitivity syndromes. *Semin Arthritis Rheum*. 2007;36(6):339-356.
- 15. Paul TM, Soo Hoo J, Chae J, Wilson RD. Central hypersensitivity in patients with subacromial impingement syndrome. *Arch Phys Med Rehabil.* 2012;93(12):2206-2209.

- Meeus M, Vervisch S, De Clerck LS, Moorkens G, Hans G, Nijs J. Central sensitization in patients with rheumatoid arthritis: a systematic literature review. *Semin Arthritis Rheum*. 2012;41(4):556-567.
- 17. Nijs J, Van Houdenhove B, Oostendorp RA. Recognition of central sensitization in patients with musculoskeletal pain: Application of pain neurophysiology in manual therapy practice. *Man Ther.* 2010;15(2):135-141.
- Nijs J, Paul van Wilgen CP, Van Oosterwijck J, van Ittersum M, Meeus M. How to explain central sensitization to patients with 'unexplained' chronic musculoskeletal pain: practice guidelines. *Man Ther.* 2011;16(5):413-418.
- 19. van Wilgen CP, Keizer D. The sensitization model to explain how chronic pain exists without tissue damage. *Pain Manag Nurs.* 2012;13(1):60-65.
- 20. Meeus M, Nijs J, Van Oosterwijck J, Van Alsenoy V, Truijen S. Pain physiology education improves pain beliefs in patients with chronic fatigue syndrome compared with pacing and self-management education: a double-blind randomized controlled trial. *Arch Phys Med Rehabil.* 2010;91(8):1153-1159.
- 21. Van Oosterwijck J, Meeus M, Paul L, et al. Pain Physiology Education Improves Health Status and Endogenous Pain Inhibition in Fibromyalgia: A Double-Blind Randomized Controlled Trial. *Clin J Pain.* 2013.
- 22. Van Oosterwijck J, Nijs J, Meeus M, et al. Pain neurophysiology education improves cognitions, pain thresholds, and movement performance in people with chronic whiplash: a pilot study. *J Rehabil Res Dev.* 2011;48(1):43-58.
- 23. Gallagher L, McAuley J, Moseley GL. A randomized-controlled trial of using a book of metaphors to reconceptualize pain and decrease catastrophizing in people with chronic pain. *Clin J Pain.* 2013;29(1):20-25.
- 24. Moseley GL. Evidence for a direct relationship between cognitive and physical change during an education intervention in people with chronic low back pain. *Eur J Pain.* 2004;8(1):39-45.
- 25. Moseley GL, Nicholas MK, Hodges PW. A randomized controlled trial of intensive neurophysiology education in chronic low back pain. *Clin J Pain*. 2004;20(5):324-330.
- 26. Moseley L. Combined physiotherapy and education is efficacious for chronic low back pain. *Aust J Physiother*. 2002;48(4):297-302.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil*. 2011;92(12):2041-2056.
- 28. Bryant A CK, eds. . The sage handbook of grounded theory. London: Sage; 2007.
- 29. Corbin J SA, eds. *Basics of qualitative research: techniques and procedures for developing grounded theory.* London: Sage; 2008.

- 30. Tong A, Sainsbury P, Craig J. *Consolidated criteria for reporting qualitative research (COREQ): a* 32-item checklist for interviews and focus groups. Vol 192007.
- 31. Ettema JHM, Arrindell, W.A. *SCL-90. Symptom Checklist. Handleiding bij een multidimensionele psychopathologie-indicator.* Lisse: Swets Test Publishers; 2003.
- 32. Shaw L, Walker R, Hogue A. The art and science of teamwork: enacting a transdisciplinary approach in work rehabilitation. *Work.* 2008;30(3):297-306.
- 33. Reilly C. Transdisciplinary approach: an atypical strategy for improving outcomes in rehabilitative and long-term acute care settings. *Rehabil Nurs.* 2001;26(6):216-220, 244.
- Nijs J, Torres-Cueco R, van Wilgen CP, et al. Applying modern pain neuroscience in clinical practice: criteria for the classification of central sensitization pain. *Pain Physician*. 2014;17(5):447-457.
- 35. Moseley GL, Butler D. S. *Explain Pain.* 2013.
- Nijs J, van Wilgen CP. *Pijneducatie een praktische handleiding voor (para) medici.* 1 ed. Houten: Bohn Stafleu van Loghum; 2010.
- 37. Evers J, Boer de, F. *The Qualitative Interview: Art and Skill*. First ed. Den Haag: Eleven International Publishing; 2012.
- 38. Transcriptionsoftware f4. Deutschhausstrasse 22a, 35037 Marburg, Germany.
- 39. Dierckx de Casterle B, Gastmans C, Bryon E, Denier Y. QUAGOL: a guide for qualitative data analysis. *Int J Nurs Stud.* 2012;49(3):360-371.
- 40. Lincoln Y, Guba, EG. Naturalistic Inquiry. Newbury Park, CA1985.
- 41. Spaendonck K, van de Lisdonk, EH. Biopsychosociale klachten, SCEGS. In: *Het biopsychosociale ziektemodel: een proeve van onderwijskundige operationalisatie. Gezond Onderwijs-4.* Houten/ Zaventem: Bohn Stafleu Van Loghum; 1995:272-276.
- 42. Wijma AJ, van Wilgen CP, Meeus M, Nijs J. Clinical biopsychosocial physiotherapy assessment of patients with chronic pain: The first step in pain neuroscience education. *Physiother Theory Pract.* 2016;32(5):368-384.
- 43. Di Blasi Z, Harkness E, Ernst E, Georgiou A, Kleijnen J. Influence of context effects on health outcomes: a systematic review. *Lancet.* 2001;357(9258):757-762.
- 44. Olsson B, Tibblin G. Effect of patients' expectations on recovery from acute tonsillitis. *Fam Pract.* 1989;6(3):188-192.
- 45. Thomas KB. General practice consultations: is there any point in being positive? *Br Med J* (*Clin Res Ed*). 1987;294(6581):1200-1202.
- 46. Dorflinger L, Kerns RD, Auerbach SM. Providers' roles in enhancing patients' adherence to pain self management. *Transl Behav Med.* 2013;3(1):39-46.
- 47. Robinson V, King R, Ryan CG, Martin DJ. A qualitative exploration of people's experiences of pain neurophysiological education for chronic pain: The importance of relevance for the individual. *Manual Therapy*. 2016;22:56-61.

- 48. Moseley GL. Widespread brain activity during an abdominal task markedly reduced after pain physiology education: fMRI evaluation of a single patient with chronic low back pain. *Aust J Physiother.* 2005;51(1):49-52.
- 49. Shutty MS, Jr., DeGood DE, Tuttle DH. Chronic pain patients' beliefs about their pain and treatment outcomes. *Arch Phys Med Rehabil.* 1990;71(2):128-132.
- 50. Kerns RD, Rosenberg R. Predicting responses to self-management treatments for chronic pain: application of the pain stages of change model. *Pain.* 2000;84(1):49-55.
- 51. Biller N, Arnstein P, Caudill MA, Federman CW, Guberman C. Predicting completion of a cognitive-behavioral pain management program by initial measures of a chronic pain patient' s readiness for change. *Clin J Pain*. 2000;16(4):352-359.
- 52. Creswell JW. *Qualitative Inquiry and Research Design, choosing among five approaches.* Second ed. London: Sage Publications, Inc.; 2007.
- 53. Free MM. Cross-cultural conceptions of pain and pain control. *Proc (Bayl Univ Med Cent).* 2002;15(2):143-145.
- 54. Kawulich B. Participant Observation as a Data Collection Method. Forum Qualitative Social Research, 2005; 6(2), Art. 43. http://nbn-resolving.de/urn:nbn:de:0114-fqs0502430.

Clinical observations of patient and therapist interactions in transdisciplinary Pain Neuroscience Education

Exploring patient-therapist interactions for effective transdisciplinary pain neuroscience education: A qualitative constructivist grounded theory study

Amarins J. Wijma^{1,2}, Tineke Pieterson², Lennard Voogt^{1,3}, Jacqui Clark¹, Doeke Keizer², Jo Nijs¹, C. Paul van Wilgen^{1,2}

Affiliations

¹Pain in Motion research group (www.paininmotion.be), Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium

²*Transcare Transdisciplinary pain management center, Groningen, the Netherlands* ³*Department of Physical Therapy and Research Centre for Healthcare Innovations, Rotterdam University of Applied Sciences, Rotterdam, the Netherlands.*

Status: submitted

Abstract

Purpose

Even though mounting evidence supports the use of Pain Neuroscience Education, the experience of receiving such education has been studied scarcely. As Pain Neuroscience Education is a 'talk-modality' many interpersonal aspects contribute to the outcome. Therefore, the aim of this study was to explore social processes in the practice of providing Pain Neuroscience Education.

Materials and methods

A heterogeneous convenience sample of eight participants with chronic pain were recruited from a transdisciplinary treatment center. Nine sessions of transdisciplinary Pain Neuroscience Education were recorded with audiovisual equipment. Data analysis was performed according to Constructive Grounded Theory.

Findings

Four connected and interactive categories were found in the process of transdisciplinary Pain Neuroscience Education: "generating a safe and comfortable feeling, situation", "the Pain Neuroscience Education process", "process of change of the respondent" and "the outcome of the Pain Neuroscience Education process". The following overarching categories were identified: "communicational skills", "interplay between the healthcare professionals" and "the influence of changeability of the respondents".

Conclusion

A variety of aspects contributing to the experience of transdisciplinary Pain Neuroscience Education were identified. Therapists should be aware to use a personcentered approach, of the importance of the therapeutic alliance, their communication skills, use a step by step introduction of the education and individualized metaphors.

Introduction

Chronic pain is defined as pain that persists beyond normal tissue healing time and/or when pain persists for 3 to 6 months or longer¹. Recent studies show a prevalence rate of moderate to severe chronic non-cancer pain between 19%^{2,3} and 27% in Europe⁴. Chronic musculoskeletal pain is associated with increased medical costs, decreased income and huge economic burdens^{5,6}. Besides that, chronic pain is immensely burdensome to the individual and associated with a negative impact on quality of life^{2,5,7}.

In persons with chronic pain, the persistence of pain, unlike acute pain or a chronic disease, cannot usually be explained by the presence of an obvious anatomical pathology or tissue damage. There is strong evidence that in persons with chronic pain abnormal endogenous pain modulation exists and that the peripheral and central nervous system have become sensitized⁸⁻¹². This complex neurophysiological phenomenon in chronic pain is also known as central sensitization (CS)⁸⁻¹¹. CS entails not only neurophysiological changes, but also differences in autonomic, motor, neuroendocrine and immune systems¹². Recently the clinical phenomenon of CS has been renamed by the International Association for the Study of Pain into "nociplastic pain"¹³. Anywhere in this chapter where is written "CS", it can also be read as "nociplastic pain". In persons with CS the central processing of information is altered and, compared to acute pain, different brain areas become involved in the processing of stimuli¹⁴⁻¹⁸. Together, these pathways and areas are called "the pain matrix"¹⁹ whereby every individual has his/her own personal "dynamic pain connectome"²⁰. Activity in this neuromatrix is influenced by complex interplay between factors such as behavioural, psychological and environmental factors, which can contribute to the continuation and amplification of pain^{12,21,22}.

It is advised to use Pain Neuroscience Education (PNE) as a part of the treatment process for persons with chronic pain. In PNE, the person's understanding of the neurophysiological processes underlying chronic pain and the relevance of such pain in the lives the individual is addressed. Effectiveness of PNE in chronic pain treatment programs holds level A evidence^{23,24} and it is recommended by the International Association for the Study of Pain as an important step in individual treatments for persons with chronic pain. During PNE the neurophysiological-neuroendocrine-neuroimmunology alterations of chronic pain are explained in lay language in

accordance with the patient's own experiences, perspective and contributing factors, whilst integrating the important role of biopsychosocial perpetuating factors^{10,25}. The goal of PNE is to establish a reconceptualization of pain, decrease the perceived threat of pain, reduce pain catastrophizing, and improve functioning and quality of life²⁶⁻²⁸. Several studies have examined the effects of PNE in a variety of chronic pain populations and have shown positive results, especially when added to other treatment modalities^{23,24,26-35}. These positive results include: increased knowledge of pain and pain perceptions, lower pain intensity, normalized endogenous pain inhibition, improved mental health, physical functioning, vitality and self-rated disability, and diminished passive coping, kinesiophobia and pain catastrophizing^{23,24,26-35}. Recent qualitative studies have underlined the importance of the relevance to the individual and individualization of PNE in order to reconceptualize pain³⁶⁻³⁸. Furthermore, an increasing number of healthcare professionals from various health disciplines, working with persons with chronic pain apply PNE as an integrated part of their pain management program^{25,36-40}.

Relevance

As little is known about the clinical application of PNE, it has been recommended to study clinical characteristics around receiving PNE and person-therapist interactions affecting the outcomes of PNE⁴¹. Thus far, most insights into the clinical practice of PNE have been only possible through expert opinion and perspectives by authorities in the application for PNE⁴². However, the "voices" of therapists and persons involved in the clinical practice of PNE have not been extensively heard and exploration of these is warranted. Qualitative research is often proposed for exploring clinical phenomena⁴³. To our knowledge thus far four studies have explored the clinical practice of PNE, through patient interviews^{36-38,40}. As PNE is a "talk-modality" interpersonal aspects^{36-38,40}, such as the therapeutic alliance, contribute to the effectiveness of the treatment⁴⁴. A strong therapeutic alliance is known to increase treatment adherence, treatment satisfaction, as well as to improve treatment outcomes in patients with chronic pain^{45,46}.

The therapeutic alliance consists of verbal, but most importantly, nonverbal communication and often subconscious processes⁴⁴. For instance, the key components of therapeutic alliance, as described by Rogers, are empathy, congruence and unconditional positive regard⁴⁴. These processes can only be captured through observations of clinical practice. To further understand the process of PNE, analysis of clinical practice using video observations was considered in this study to be an

appropriate way to observe social interactions, and with minimal researcher intrusion⁴⁷. Using the video data, analyses of nonspecific factors such as interactions and nonverbal communication can be undertaken⁴⁸.

Research Aim

The primary aim of PNE is to change perceptions of pain, thereby decreasing the threat associated with pain in the individual experiencing chronic pain. Based on previous research three possible outcomes are theorized: 1) the individual agrees with the explanation and reconceptualizes his/her pain, 2) has doubts and partly reconceptualizes or 3) disagrees and does not reconceptualize^{36-38,40}. Important clinical questions are; In what contexts do some individuals agree (outcome 1), while others hesitate or even disagree (outcomes 2 and 3)? We theorized that there are multiple influencing contexts and factors, such as: the patients' biopsychosocial experiences and expectations, and whether or not these internal processes are expressed by the individual. Furthermore, the healthcare professionals' characteristics, such as communication skills, knowledge and expectations may influence the PNE. And lastly, the characteristics of the PNE, the content, process, social context, and delivery may influence the outcomes. These components may all influence each other and consequently also the outcomes of PNE (figure 1). Therefore, the aim of this study is to explore the social process between the individual in chronic pain, their partner and healthcare professionals in the practice of transdisciplinary PNE, to deeply comprehend and theorize what factors lead to a successful PNE.




Figure 1. Primary conceptual framework *HCP's: healthcare professionals

Methods

Design

Based on Constructivist Grounded Theory^{49,50} we conducted an observational qualitative study with video observations to explore the practice of transdisciplinary PNE. In order to develop a theory describing the relationships between what is observed and the outcomes seen in practice. Qualitative research is the most suitable method for inquiry on social phenomena, as it can provide a greater depth of understanding of the phenomenon⁴³. Constructivist Grounded Theory adopts the inductive, comparative, emergent, and open-ended approach of the original Grounded Theory⁵¹, while being aware of the influence of the observer/analysist. Constructivist Grounded Theory disputes the idea of a neutral observer and acknowledges that researchers have preconceptions⁴⁹. These preconceptions influence the construction of the theories induced through the research process and they should therefore be examined, identified and described⁴⁹.

The outcome of this study was a theoretical conceptualization of PNE practice, based on the categories induced from the observations grounded in the video data. Our study was conducted and reported where possible according to the Consolidated criteria for Reporting Qualitative research (COREQ)⁵².

Study location

The current study took place in a transdisciplinary pain management center (Transcare) in the Netherlands. In this transdisciplinary center, the care for individuals with chronic pain is provided by teams of doctors, psychologists and physical therapists. Working transdisciplinary implies that the teams collaborate intensively, have flexible boundaries and roles, learn simultaneously and have a shared biopsychosocial view on chronic pain^{53,54}. When needed collaboration with other medical specialists from nearby hospitals can be initiated. A complete description of the goals, focus of the transdisciplinary treatment center, treatment and transdisciplinary PNE patients received can be found in a previous article by Wijma et al.⁴⁰.

Participants

A heterogenous convenience sample of 8 respondents with various chronic musculoskeletal pain disorders were recruited from a transdisciplinary pain management center between March 2015 and December 2016. The inclusion criteria

were: 1) between 18-85 years of age, 2) experiencing nonspecific chronic pain as defined by the International Association for the Study of Pain¹, and 3) sufficiently able to read, speak and understand Dutch. The exclusion criteria were: 1) identified as experiencing specific nociceptive pain, 2) mentally limited, 3) having dementia, 4) or a serious psychiatric condition according to the Symptom Checklist 90 (SCL-90)⁵⁵.

Twenty-three persons fulfilling the inclusion criteria were asked to participate in this study. For those who agreed to participate, AW contacted them by phone to provide further verbal and written information and an informed consent form. In the moments prior to the PNE session AW unobtrusively installed the camera and received the informed consent forms from the respondents. Confidentiality was guaranteed and all material was handled anonymously. The study received ethical approval from the Maasstad Hospital in Rotterdam, the Netherlands.

Healthcare professionals

The demographic information of the healthcare professionals, in every session a psychologist and physiotherapist, is described in table 1. All healthcare professionals working at the transdisciplinary treatment center were invited to participate, and all provided consent to do so.

Healthcare professional	M/F	Age	Education	Occupation	Years of experience	Years of experience in working with patients with chronic pain	Years working at the trans-disciplinary team	Relevant PNE n courses
	Σ	49	DhD	Physiotherapist/ Psychologist	25	25	∞	PNE course teacher
2	ш	26	University	Psychologist	2	1	٢	Courses Pain in Motion
m	ш	26	Higher professional education	Physiotherapist	4	2,5	2,5	Courses Pain in Motion
4	ш	43	University	Psychologist	20	10	ĿЛ	Courses Pain in Motion, several
								psychological courses
L.	ш	49	University	Psychologist	21	15	D	Courses Pain in Motion, self-study
9	ш	57	Higher professional education	Physiotherapist	34	15	Ŀſ	Courses Pain in Motion
7	Σ	62	Higher professional education	Physiotherapist	37	Ŀ	Ū	Courses Pain in Motion

Clinical observations of patient and therapist interactions in transdisciplinary PNE

Procedure

The practice of transdisciplinary PNE was examined using data from video observations of the interactions between individuals with chronic pain (respondents), their partners and the healthcare professionals (physiotherapist and psychologist) during the second PNE session, whereby the respondents were the main focus. In the preceding session with general practitioner the respondents were asked to bring a relative, friend or partner to this session.

Both the observations and analysis were in Dutch. Observations continued until the researchers AW and TP felt confident that theoretical data saturation was achieved. At this point, there was conceptual depth of each category and it was believed that no new information emerged from the data in the analysis that added to dimensions of the categories and understanding of the phenomenon under study⁴⁹. The theory was then translated from Dutch into English. Throughout the translation efforts were made to convey the sense of the Dutch-language by allowing enough time to carefully and critically reflect on the meaning and wording of the prime language. Furthermore, the translation, performed by AW, was reviewed by two team members (PvW, TP) of which one (TP) had been involved in the complete analysis. Prior and all the way through the analysis all researchers created memos which were later on used in the analysis⁴⁹.

Points of departure – Theoretical framework

Prior to the observations primary conceptual theoretical frameworks (figure 1) were constructed based on our previous knowledge. Furthermore sensitizing concepts (table 2) were developed based on:

- a previous study of the experiences of patients with chronic pain with transdisciplinary PNE⁴⁰:
- previous literature about the expectations of patients with chronic pain influencing treatment outcomes⁵⁶, about the importance of PNE feeling relevant to the individual³⁸ and stages of change⁵⁷;
- and professional and academic experiences of AW, CPvW, LV and TP.

The process of articulating this theoretical framework (figure 1) helped to shape the study theories stated in the research aims section above.

Both the theoretical frameworks and sensitizing concepts were constructed to help structure the analysis of the data according to the Constructivist Grounded Theory by Charmaz⁴⁹ and evolved during the analysis (table 2.).

Table 2. Sensitizing concepts	
Primary sensitizing concepts	sensitizing concepts after the first member check
 How do respondents understand CS? Did they get the message, can they explain this model, how do they explain the message? What are the practical implications for the respondents? And how do the respondents transfer it to daily life? What were their expectations and perceptions? How did the healthcare professionals deal with this? How did the healthcare professionals take care of externalizing internal processes of the respondents to reveal the perceptions of the respondents based on acceptation of the model? (agreement, hesitation, disagreement) 	I. Overall: Does the partner play a role? If so, what kind of role? How do they accept their pain(complaints) and the consequences of those? Do they get advice etc. for the future/is the following treatment discussed? How do the healthcare professionals bond with the respondents? Which conversation techniques were used by the healthcare professionals? Are there clumsy actions/faux pas by the healthcare professionals? How is the PNE structured? What is being said/used?
 2. What are the reelings of the respondents with the message? Do they get emotional? Does it reassure them? Or do they get anxious in any way? Or relieved? Or angry? Is there resistance against the message/CS, or not? If so: which forms of resistance do they show? And how do the healthcare professionals cope? In which manner does the resistance they have, have an effect on the outcome of the PNE session? (agreement, hesitation, disagreement) If a respondent keeps being resistant, how do the healthcare professionals cope with that? 	C. How do respondents understand Cs? Did they get the message? What is their knowledge? How is this expressed? What were their expectations and perceptions? How did the Healthcare Orofessionals deal with this? (is both 2 and 3) How did the Healthcare professionals take care of externalizing internal processes of the respondents to reveal the perceptions of the respondents based on acceptation of the model? (agreement, hesitation, disagreement) (is both 2 and 3)

Primary sensitizing concepts	Sensitizing concepts after the first member check
3. Does the message have any consequences for their behaviour?	3. What are the feelings of the respondents with the message?
-Did the respondent already adapt?	 Do they get emotional? What kind of emotions are shown?
-Does the respondent have any idea how to adjust his/her behaviour?	- Is there resistance against the message/CS, or not? If so: which forms of resistance
(This might be complicated to get from the PNE session)	do they show?
- Is there any sign of changing into another stage of change?	 How do the healthcare professionals cope with resistance?
	-In which manner does the resistance they have, have an effect on the outcome of
	the PNE session?
	- If a respondent keeps being resistant, how do the healthcare professionals cope
	with that?
	-What kind of processes can you see between the respondents and healthcare
	professionals when there is concern/doubt/anxiety/resistance?
4. General:	4 Does the message have any consequences for their behaviour?
- Does the partner play a role? If so, what kind of role?	 Does the respondent have any idea how to adjust his/her behaviour?
	(This might be complicated to get from the PNE session)
	 - Is there any sign of changing into another stage of change?

Prior to the analysis all researchers (AW, TP and CPvW) described their own theoretical background in a reflective paper⁴⁹, including descriptions of their knowledge, thoughts, perceptions and emotions about: the setting (Transcare), CS and PNE, the other professionals involved in the treatment and transdisciplinary collaboration. This description increases the trustworthiness of the study⁵⁸.

Analysis

The analysis was performed in an iterative process, meaning that there was constant comparison between data gathering and analysis, by AW and TP. Both are physiotherapists, respectively with a MSc., PhD-researcher with training in qualitative research and MSc.-student with training in qualitative research. The analysis started with a general impression of the recording. Then the recordings were observed in a focused manner based on the sensitizing concepts. During the third step in a selective observation the differences, contrasts, extremes and paradoxes between the observations of the set were sought. This phase was independently done by AW and TP whereby the actors, activities, jottings and broad lines of the setting where described⁴⁹.

The next step was to transcribe all observations verbatim with transcription software F4 (dr. Dresing & Pehl GmbH, Marburg, Germany)⁵⁹ by AW. Non-verbal actions that were noticed were described by AW and inserted based on consensus by both AW and TP. For example: *"5.41 min: the respondent frowned, interpretation: hesitation"*.

After the transcriptions the data were analyzed according to Constructivist Grounded Theory⁴⁹. First, AW and TP used an interpretive coding process to perform initial coding of the data by applying conceptual labels. A codebook was made in QSR International's NVivo 12 software (QRS International Pty Ltd., Doncaster, Australia). After the initial coding of three observations focused coding was performed. This iterative process was repeated after the initial coding of the subsequent observations. Finally, theoretical coding was categorized to inform construction of a theory and theoretical framework by AW and TP. This was peer reviewed and debriefed by CPvW and LV, CPvW being a psychologist and physiotherapist and LV a physiotherapist and lecturer, both academics with a background in qualitative research: this resulted in the final theoretical findings and framework. 5

After the focused coding of the first four observations a member check of synthesized data⁶⁰ was held with one psychologist, a participating healthcare provider whom had been observed. Another member check was conducted after the focused coding of all observations with another observed psychologist. These psychologists were chosen to reflect the transdisciplinary perspective of the phenomenon under study, as both researchers performing the initial analysis (AW and TP) are physiotherapists. During these member checks the preliminary focused coding was discussed. The member checks resulted in adaptation of the focused codings. For instance, by renaming wording in the codes, whereby 'perceptions' were renamed into 'cognitions', as it was construed as more accurate.

The sensitizing concepts (table 2) and conceptual theoretical frameworks (figure 1) were modified throughout the analysis of the observations, particularly after the first member check.

Trustworthiness

The trustworthiness criteria of Lincoln and Guba⁵⁸ were used to endorse the quality of the study.

As well as using member checks, credibility was facilitated by reaching data saturation and achieving in-depth observations. Negative case analysis was done by falsifications and a heterogeneous sample. Furthermore, anecdotalism was avoided and interpretations were agreed on by all the researchers, who were aware of their own preconceived theoretical perspectives. All data was systematically handled in NVivo. Researcher triangulation was done by two researchers throughout. The codebook and codes were discussed and peer debriefing was done by two individuals.

Transferability was pursued by writing a thick description of the population, field and data. Furthermore, the theoretical implications (patterns) were fully described in this study.

Dependability was sought by a continued monitoring of the process and product of the research by the authors. Furthermore, transparent description of the research steps and memos were made.

Confirmability is known among researchers as being potentially problematic in qualitative observational studies, however, to improve confirmability the following steps were taken: 1) NVivo was used, with complete and meticulous documentation, thereby creating a full audit trail⁵⁸; 2) researcher triangulation was performed; 3) the researchers (AW and TP) constantly applied self-reflection by reflective field notes during the analysis. They were aware of the researchers-effect, the effect of preconceptions, faulty interpretations and familiarity with the setting on the data, recording these in a reflective paper.

Findings

Twenty-three individuals were asked to participate in the study. Twelve were not willing to participate because they felt uncomfortable with the camera or felt they would focus too much on the camera instead of the conversation. In three individuals, the data collection was not completed due to the following reasons: one individual stepped out of the program prior to the PNE, one video recording failed and one person did not arrive at the PNE appointment. Eight individuals remained who were recorded, one individual was recorded twice. While only one psychologist and one physiotherapist attended each session, this represented seven different health professionals across the nine recordings. Further demographics are described in table 3. The length of the observations varied from 30 to 83 minutes per session.

Respondent	M/F	Age	Relational status	Occupation	Education	Location of pain/ complaints	Duration of complaints (months)	Other relevant health problems	Significant other at the PNE
-	Σ	64	Married	Disenabled to work	Junior secondary	Headache, dizziness	36	Eye problems	3
					technical education				
2	ш	65	Divorced	Retired	Junior secondary	Back, hips and knee	60	,	Daughter
					home economics				
					education				
e	ш	33	Living together	Disenabled to work	Intermediate	Stomach, widespread	96		I
					vocational education	pain			
4	ш	35	Married	Housewife	Intermediate	Fatigue, widespread pain,	240	,	I
					vocational education	headache,			
5	Σ	39	Married	Disenabled to work	Intermediate	Backpain towards legs	11	1	Wife
					vocational education	and shoulders			
6	ш	38	Divorced	Disenabled to work	Intermediate	Backpain	38		I
					vocational education				
7	ш	45	Married	Nurse	Higher professional	Widespread pain, legs,	24	1	I
					education	shoulders, fatigue			
8	ш	49	Divorced	Disenabled to work	Intermediate	Widespread pain,	312		I

only the summarized results from the theoretical coding are described, the appendix provides the complete results Due to the thick description of findings,

legs

restless

ocational education

The findings are presented according to the categories that emerged during the data analysis below. When the data represents aspects from the respondent, their partner as well as the healthcare professionals, the aspects from the respondent are described first, in line with patient-centeredness. Whenever is written 'respondent', it can be both respondent and partner, if it was just the partner this is described.

Seven connected, interactive categories emerged from the data, as can be seen in the conceptual theoretical framework (figure 2).

Four of these categories were related to the phases of the PNE session:

- 1) Generating a safe and comfortable feeling and situation
- 2) The PNE proces
- 3) Process of change of the respondent
- 4) The outcome of the PNE process

A further three categories emerged as important factors in all phases of the PNE

- 5) Communication skills of the healthcare professionals
- 6) Interplay between the healthcare professionals
- 7) Influence of changeability of the respondents on the process.

The categories *interplay between the healthcare professionals* (described in the appendix) and communication skills played a role in both generating a safe and comfortable feeling (1) and the PNE process (2). Both the communication skills of, and the interplay between the healthcare professionals made the respondents feel at ease and helped guide them through comprehending the information as presented during the PNE session.

The influence of changeability of the respondents on the process was more related to the PNE process (category 2), the process of change (category 3) and the outcome of the PNE process (category 4). Here, the changeability guided the content of PNE and influenced the process and the outcome.

Every videotaped PNE session was individualized, resulting in a broad variation of data during the session. A general overview of the content addressed in the observed sessions is summarized in Table 4, while change processes of the respondents are detailed in Table 5. Table 6 describes the individual differences in the PNE. An effort is made to convey these individual differences, as well as reflect the similarities in the following report.

Table 3. Demographic data of the study participants (n=8)



Figure 2. Definitive conceptual theoretical framework Yellow stands for themes related to the respondent, blue for themes related to the healthcare professionals and green related to the PNE

Category 1: Generating a safe and comfortable feeling and situation

Everyone in the process invested in creating a warm and friendly environment. Efforts to create and keep an alliance with the respondents was seen from the healthcare professionals. These efforts resulted in the respondents showing both verbal and nonverbal signs of feeling at ease, acknowledged and understood.

Respondent 3: "I do feel as if you understand me" Respondent 1, Psychologist: "Because we don't feel what you feel, but we believe every word you say" Respondent 1: "Yes (nods twice)"

Generating a safe and comfortable feeling was started during the prior intake and continued to be an active process throughout the session with all respondents. Individual differences, however, were too small to describe.

Three sub-categories were identified linked to 'generating a safe and comfortable feeling'. From both the respondents and healthcare professionals: Making jokes to keep the conversation light (described in the appendix). From the healthcare professionals: Building an alliance with the respondents and Faux pas from the healthcare professionals.

Sub-category: Building an alliance with the respondent

Building an alliance with the respondent consisted of multiple efforts: *Creating a safe* environment, Recognizing and understanding personal suffering and pain and Using personal information from the intake in the PNE (the latter overlapping with the next category: The PNE Process).

Creating a safe environment

Creating a safe environment ensured the respondents dared to express themselves. For instance, by telling them they could always ask any question, no matter what, and responding to those questions with ease and understanding. Furthermore, this save environment was created by letting them tell their story and giving the respondents time to express themselves. Expressed or non-verbal shown discomfort or awkwardness as felt by the respondents were handled with care. Discomfort was also handled by making polite jokes by the healthcare professionals, often followed by a gesture that they took them seriously.

Clinical observations of patient and therapist interactions in transdisciplinary PNE

Creating a comfortable space also meant placing the respondent in the center, asking for their preferences.

Physiotherapist respondent 7: "Regarding the future treatment trajectory, it is important that you say what you want .. of course, it's nice that you take our feelings into account, be aware that this process is all about you."

Recognizing and understanding personal suffering and pain

Recognizing and understanding the respondents' pain by the healthcare professionals meant showing an overall empathy for the respondent as a person and their suffering. The healthcare professionals often recognized, emphasized and expressed the frustrations, feelings of heaviness and the hardships of dealing with chronic pain and fatigue the respondents felt.

Some of the most used sentences were: "yes, that is hard", "I get it" and "we understand that it is difficult".

The impact of the pain on the respondent's life and future was often mentioned.

Respondent 8, physiotherapist:" Well, it is complicated to deal, to, deal with this"

The healthcare professionals underlined their empathy for the unfamiliarity of the respondents with CS and contributing factors, acknowledging the doubt some respondents experienced.

Physiotherapist: "That (hay fever) is in essence an overreaction of the nervous system. That is actually what it is about, and that signals pain, right? The pain is real; however, you can't see anything different in those people. That is often what makes you insecure, that is what often makes it incomprehensible: why am I in pain now?"

The healthcare professionals also corroborated on the realness of the complaints. Often by saying things like: "CS pain is a real pain and not "in your head", it is not psychological." "Whatever you feel is real." Furthermore, they explained to the respondents that even though there is no damage, their pain is real. For those respondents who had a biomedical question or quest the healthcare professionals understood and acknowledged this quest. If necessary, and not already performed, biomedical examination, such as bloodwork, was done. In an interactive process with the respondent outcomes of these or previous examinations were explained in the light of PNE.

Among this recognition was also the acknowledgement that the biopsychosocial contributing factors are often factors one does/feels unconsciously and are not easy to change for the respondents. However, it also implied acknowledging that when these biopsychosocial contributing factors changed, life would probably be more enjoyable.

Using personal information from the intake in the PNE

The healthcare professionals used personal, sometimes specific, information which they heard or had seen during the intake. Using this personal story was important to enhance its relevance to the individual, reinforce the explanation and made respondents feel heard. For instance, what they had seen in the physical examination, information about biopsychosocial contributing factors and emotions that where present. Or they just referred to what the respondents said by:

"I've heard you say this before".

Respondent 2, psychologist: "You told me in the first conversation: I am not that much of a worrier, and I don't worry that much, but I do feel that the pain now dominates my life. (R: Yes, yes, yes, yes.)'

Sub-category: Faux pas from the healthcare professionals

Faux pas from the healthcare professionals were unhelpful actions that caused small ruptures in the alliance. For instance, asking many questions in one sentence, using difficult words, introducing the contributing biopsychosocial factors too fast, giving an unclear answer to a question, etc. These were often addressed and the ruptures repaired.

Respondent 5, physiotherapist: "No that is true.. can you grasp what I am saying? How is that for you? .. Because it means a difference in dealing with the pain, right? And how you look at pain. Is that, is that a reassurance or not? How does it feel for you both?" Chapter 5

Respondent 5: "... I don't know how that feels" Respondent 5, partner: "I am so glad that it isn't a herniated disc anymore, yet on the other hand he has something that isn't visible. How do we explain that to others?"

Category 2: The PNE process

The category the PNE process consists of the characteristics of the PNE, in which the content of the PNE (*The steps of the PNE process*), *Individualization of the PNE* and *Examples and metaphors* are described. Among with the PNE there are different processes that are in action, related to the next categories *Process of change of the* respondent and *Outcome of the PNE process*.

Subcategory: The steps of the PNE process

Most often the PNE conversation would involve similar 'steps' in the process, as described in table 4. By applying this stepwise structure respondents were engaged in the explanation of their pain. However, these steps would change according to the individual respondent or preferences of the healthcare professionals (table 6).

Table 4. General lay out of the content of the observed PNE sessions

General content of the PNE session	Healthcare professional explaining (most often)	Particulars
1 a) Welcome back, give a general lay out of this session.	Psychologist	1a and 1b could be exchanged
1 b) Ask how the first session with the GP is experienced by the respondent and what the respondent remembers of the GP's explanation.	Psychologist	
1 c) Explain goals for this session; more clarification, explain their pain in more depth, some guidance in how to deal with his/her pain, and further treatment(s).	Psychologist	
2) Explain difference between acute and chronic pain.	Physiotherapist	Based on the predominant paint mechanism ⁶¹ . Respondents who first experienced acute/nociceptive pain were explained that it started with acute pain, however there is no "damage left" and the metaphors were their own (for instance falling of stairs). Respondents who had a combination (nociceptive pain and/or neuropathic and CS) were explained the combination

Table 4. (Continued)

General content of the PNE session	Healthcare professional explaining (most often)	Particulars
3) From "damage" to CS: explaining the difference between acute and chronic pain for the individual. Based on respondents own description of pain, disabilities, prior treatments and tests.	Physiotherapist	Normalizing the pain. Important in this phase is that the healthcare professionals underlined the credibility of the pain, taking the respondent and his/her pain really serious.
4 a) Explanation that it is an overactive processing/sensitization. Drawing a schematic picture of the nervous system. Explaining neurophysiology of acute pain by an example.	Physiotherapist	
4 b) Explain neurophysiology of chronic pain/ central sensitization, while drawing the changes in the central nervous system. With examples from the individual.	Physiotherapist	
5) Bridging the PNE towards the biopsychosocial factors.	Psychologist	The variations in this "bridge" depended on the healthcare's" preferences and their estimation of the patient's preferences
6) Explain and discuss the biopsychosocial factors that contribute in the respondent's pain experience, while drawing the links in the brain. Individual contributing factors were discussed in a for the respondent non-threatening manner.	Both, however most often the psychologist	
7a) Based on the contributing biopsychosocial factors provide some advice and guidance in dealing with his/her pain.	Both	7a/b/c could be exchanged
7 b) Discuss further treatment options and goals.	Both	
7 c) Give a realistic picture and hope for the future. Give back purpose.	Both	
8) Closure of the PNE session.	Both	

A more detailed, yet general, description of the PNE session is included in the appendix

Subcategory: Individualizing the PNE

Individualizing the PNE was done in a subconscious manner and multiple ways. First, in all sessions the healthcare professionals recalled information and questions from the respondents that were mentioned in the intake. These were answered during the session. Additionally, they copied the words and language of the respondents, with language based on the individual. They explained in a pace adapted to the respondent, and used recognizable metaphors for the individual or their own metaphors. The treatment options were discussed in a shared decision manner, with an individualized level of involvement.

Individualizing the PNE also mentioned explaining all complaints of the individual respondent, not just the pain.

Respondent 1, physiotherapist: "So that is it, this is it. It's not pretention ...It's just, that system is upset, just as it is mixed up with you, and you can notice that in different aspects, it can just be pain, but with you we see a combination of pain, sound, light.."

Further individual differences during the PNE are described in table 5 and 6.

Respondent	Start PNE	During/end PNE	Resistance*	Stage of change prior PNE	Stage of change after PNE
-	Feels urge to explain complaints, didn't	Unknown whether he truly	No obvious resistance against PNE,	Precontemplation	Precontemplation
	quite remember the message of the first PNE.	understands PNE and contributing factors	unknown comprehension, finds it hard to accept his complaints		
2	Understood the prior PNE, fire	Understands PNE, agrees with it.	No resistance against PNE,	Contemplation	Contemplation
	alarm, happy with the diagnosis and	Did have questions on whether	reasonable comprehension and		
	explanation. Wanted to know the	her pain was an inflammation,	acceptation		
	cause of the fire alarm going off.	reassured when explained why not			
		and why CS. Has already changes			
		in mind (yoga, relaxation)			
0	Doubt and hesitation: found the first	Understands PNE, agrees with	No resistance against PNE,	Contemplation	Contemplation
	explanation woolly. Understood it,	it, understands the contributing	accepted it, however, finds it		
	however, can't cope with her pain.	factors.	hard to accept her complaints,		
			resistance towards changing,		
			lesser psychological flexibility		
5	Resistance against prior PNE, felt	Quickly shifted from resistance	First resistance against PNE,	Contemplation	Contemplation
	there weren't enough biomedical	to agreement when "no results	quickly changed, acceptance and		
	examinations done, wanted an easy	from bloodwork" in relation	good comprehension at the end.		
	solution.	to complaints and CS as			
		underpinning chronic pain was			
		explained. Thinking along with			
		contributing factors and how to			
		deal with those.			
5	Unknown whether he really	Both go along with the PNE. Ps and	Large role partner, he has little	Precontemplation	Precontemplation
	comprehended the previous PNE,	Pht (#) frequently try to get him	expressions and involvement. No		
	partner did. Acceptance idem.	involved. When asked they agree	obvious resistance against PNE,		
		with the PNE, contributing factors	probably acceptance, unknown		
		000 + 2000 + 2000	compression of the sector		

\sim
ດນ
Ť
2
-
5
\cup
\sim
പ്
5.
le 5.
ole 5.
ıble 5.
able 5.
Table 5.

162

Respondent	Start PNE	During/end PNE	Resistance*	Stage of change prior PNE	Stage of change after PNE
ба	No resistance against CS, first part previous PNE, however resistance	Reasonable understanding of PNE, acceptation PNE, only fear of	Started with resistance against "being afraid to move" as factor	Contemplation	Contemplation
	against "being afraid to move" as told	movement as contributing factor	in her problem, changed into		
	by doctor being a part of her problem	explained due to lack of time	agreement with it being a factor.		
			No resistance against PNE,		
			reasonable comprehension		
6b	Grasped the essence of the previous	Agrees with the message of the	Reasonable to well	Contemplation	Contemplation
	PNE, accepted it	PNE. Questions whether her	comprehension. No resistance		
		crooked pelvis is of influence,	against PNE and contributing		
		understands explanation of Pht	factors, however finds it hard to		
		that it is not crooked however she	accept her complaints, deal with		
		has a tensed and stiff movement	them and change behaviour		
		pattern.			
		Finds it hard to change her			
		behavioural patterns			
7	Found the first PNE pleasant and	Understands the PNE, agrees	No resistance against PNE, good	Action	Action
	enlightening, understands quite a bit	with it, comes with contributing	comprehension and acceptation		
	already. Already changed behaviour	factors, comes with ideas on how			
	after the PNE. Came with some	to proceed			
	questions written down.				
0	Slight hesitation, wants to agree with	During the complete PNE: keeps	Hesitation and agreement	Contemplation/	Contemplation/
	the previous PNNE/CS, however thinks	: on changing directions from:	with PNE alternate, same with	Preparation/ Action	Preparation/ Action
	"what if there is something else?"	agreement, acceptance knowing	comprehension vs. biomedical		
	Looked up information on Moseley,	and open towards change, to: "but	thinking. However, is accepting		
	already insight in	what if it is", "yes, but", and back	the PNE and open towards		
	receptors and role stress	to agreement	contributing factors and		
			treatment.		

*Resistance against the complete PNE, or a part of PNE, # Ps = psychologist, Pht = physiotherapist

ng the PNE	
differences duri	
Individual o	
Table 6.	

Respondent	Manners in which the PNE was individualized
,	Simple vocabulary. Start with his accident as nociception and explain how it got to CS. Let him speak about his complaints and based on that explain PNE, explain faulty cognitions and why not moving his head increases the problem, as well as taking him seriously and showing understanding; following his rhythm. Only discuss other contributing factors (such as acceptance stress) in limited detail. Urge him and explain why not to pursue biomedical solutions. Further treatment: physiotherapy, graded activity and exposure.
7	Daughter is present, yet not really involved. Respondent mentions burglar alarm in the beginning of the PNE: at her job (now retired) the burglary alarm would go off unexplained. Use example walking from the intake. Use different metaphors: burglar, hot pain, hay fever, load/capacity, diminished physical condition, whiplash as example CS, asthma, headaches, stomachache, phantom pain. Explain limited role arthritis. Explain physical and emotional tensions, and their relationship, and with CS. Talk about role social surroundings; feels taken seriously, feels supported, they have read about CS. Explain her questions on: whether CS is hereditary, why it is just one part of her body, whether she should fitness or do yoga. Only physiotherapy based on relaxation and increasing physical fitness, door is open to psychologist if necessary.
m	Ask if bloodwork was done; yes, no specialties. Explain CS based on hypersensitiveness of sitting on a chair, twice, then based on hay fever. Followed by explanation pain control system and how this system can get repressed by several factors. (total about 20 minutes) Then go into detail about negative emotions, fatigue, pushing boundaries, early life events, caring for oneself, respondent find this part difficult: agrees with it, brings it up herself, then avoids it, this is a dance between the respondent and the healthcare professionals. Advice to start psychotherapy, will think about it, second PNE planned to review and discuss next therapy.
4	Start with discussion bloodwork and X-rays, respondent mentions not being content about results; no abnormalities. Healthcare professionals take her seriously, name her frustration, ask what she would have wanted: an explanation for her complaints, they call it fibromyalgia and then start explaining CS. Respondent has had previous therapy on pain, this PNE focus on both pain and fatigue. Take the burden of pain and fatigue on her very seriously. During the PNE deviate from the neurophysiology when she mentions examples of her disabilities and then go back. First part PNE about 18 minutes, then no to contributing factors. Struggling and difficulties in dealing with different aspects of life, diminished feeling of fulfillment of life, life not being easy and setting the bar high. A lot of reflective questions. Physiotherapy and psychology: graded activity, relaxation overall and in anger, dealing with life.

ndent Manners in which the PNE was individualized	More plain vocabulary, only mention of sensitive ne
Respond	ы

•	
	More plain vocabulary, only mention of sensitive nervous system and not CS. Start with hernia as acute onset, then towards hernia is gone, still in pain: CS.
	Explanation why biomedical therapies such as bloodsuckers won't work in this (CS) system. Explain previous well-functioning pain control system (container
	on hand) and his current pain. Trying to involve him in the PNE with questions, sits huddled, little expressions (difference from at home). Partner is really
	involved and understands explanation. How to explain this to the social surroundings. Example bronchitis, going on and on with an ankle sprain; body also
	needs time to heal. Lesser mentioning of emotions (stress boss, finances, how he used to be a different man), more on normalizing, relaxation in movement,
	trust in the body and steadily going back to work. Goals for further treatment: physiotherapy.
	Had two sessions. In the first session first focus on removing resistance surrounding "afraid to move" via understanding, moving along with her, explaining

9	Had two sessions. In the first session first focus on removing resistance surrounding "afraid to move" via understanding, moving along with her, explaining
	neurophysiology, explanation moving tensed and influence on pain and then towards naming it "being afraid to move". Second session more focus on stress
	factors, such as: start pain and trauma surrounding fall of stairs, divorce, setting the bar high, behavioural influences such as: always being busy, not taking
	rest and sawtooth pattern of daily activities. Goals further treatment: both physiotherapy and psychology, one of the goals: normalizing work.
7	Already understood a great deal of PNE based on the prior session and information leaflet. Brings her own questions to the PNE session. Shorter
	neurophysiology during the PNE, more focus on contributing factors. A great deal of influence and input from the respondent. Together towards shared
	derision making for further treatment. has ed on the question: "what do vou peed?" During the last phase the resonndent indicates that she prefers another

pelvic physiotherapy; she explains her arguments pro/con, based on questions and advice she decides to stop this therapy. Throughout the PNE session the respondent asks questions, comes up with her own examples (walking, singing, burn out, adrenaline) and the PNE is partly based on these questions and examples. The respondent "flutters" between understanding and acceptance, coming up with examples to support this, then hesitation and contradicts herself. The healthcare professionals try to balance the PNE conversation by letting her talk, then questioning, reflective questioning and examples. Further Respondent is offered a stool, because she can't sit in a chair. One mayor question at the beginning of the respondent is whether she should continue other older psychologist. ∞

psychologist and physiotherapist

creatment:

Subcategory: Metaphors and examples

Metaphors and examples were based on the individual person, in plain language, based on the expected health literacy of the respondent. Often the respondents would react in agreement, this could be seen by the non-verbal communication (listening seriously and nodding), or in a verbal manner by "yes", "hmhm" (often in combination with nonverbal communication), or by responding to a metaphor with their own metaphor/ example.

Respondent 7, physiotherapist: 'It is, no, these are none painful signals that go up (points from spine upwards to the brain), they, they come into a network which immediately says: POING, pain! Because you sense pain in your brain, not in your foot'

A complete list of the metaphors used is presented in the appendix.

Category 3: Process of change of the respondent

Understanding, agreeing with and implicating the PNE on his/herself and daily life was a process for the respondent. This change process is a personal iterative process with multiple steps, as displayed in figure 3. Not all steps could be seen in all videos, these individual differences can be seen in table 5, describing the individual differences of the PNE session and table 6 describing the change process of the respondent.

1.	2.	3.	4.	5.
Gain knowledge and understanding	Applying this knowledge to oneself	Acceptance of the situation and dealing with loss	Being able to recognize changes that need to be made	Applying these changes and sticking to it
(changing cognitions)	(individualizing)	(grieve and accepting)	(insight)	(changeability)

Figure 3. Change process surrounding PNE

Within this change process respondents had several struggles which emerged as the sub-categories: Concerns about the complaints they experienced, Reaction of the healthcare professionals towards concerns, Having trouble accepting their complaints and Doubts and resistance towards the PNE and change and Reaction of the healthcare professionals towards doubts and hesitation. Throughout these struggles, a Process of insight in CS and the factors surrounding the experienced complaints also started and in some even changes for daily life were made or planned.

Sub-category: Concerns about the complaints they experienced

All respondents expressed concerns surrounding their complaints. These concerns were often expressed at the start of the PNE. Followed by a cyclical process in the PNE between the explanation for these complaints (CS) and other concerns the respondents perceived. Most often throughout the PNE these concerns decreased or diminished completely, apart from respondent 1 and 8 (table 5,6). These concerns often surrounded the etiology of their complaints: how it could be that they were now experienced chronic pain but not prior, why their complaints were increasing, and how to deal with their pain.

Respondent 8: "and when I get up I get this shooting pain in my foot...then I'm like, well!, just by lying! I'm thinking I'm giving my body a bit of rest and then when I get up the pain is even worse. That's when I'm concerned, I'm like: What is it? What is it?"

Furthermore, some respondents expressed concerns on how people in their social network reacted or would react to their complaints. Concerns surrounding how their social environment reacts to their complaints where often either about their pain not being understood or taken seriously in the past ("chronic pain is not a visible problem like breaking a leg"), or about novelty of CS ("central sensitization is not known in our social environment", and "how do we explain this to our environment?").

Sub-category: Reaction of the healthcare professionals towards concerns

The healthcare professionals dealt with the concerns first by asking the respondents about their concerns, in order to allow them to express their concerns. Subsequently, the healthcare professionals took great care in taking the respondents' concerns seriously. Often, by these actions the concerns would then be gone, or less important.

Respondent 1, physiotherapist: "You say your symptoms have increased, is that something you worry about? Say, what if it continues like this, what then? Or isn't that on your mind?"

Respondent 1: "Well that is something I have been thinking about. What if I have to live like this? It continues and gets worse? (looks sad). That's why I'm here"

If these concerns were driven by inadequate cognitions, they questioned these misconceptions and then proceeded by (re)explaining the PNE in further detail. This cycle was repeated if necessary.

By letting the respondents express their concerns and addressing these, both the therapeutic alliance was reinforced as well as a deepened understanding of the respondents about their pain was achieved.

Sub-category: Having trouble accepting the complaints

Dealing with the complaints and the consequences of them in daily life was hard for the respondents. All respondents expressed that they often felt frustrated, sad, angry, stressed, restricted in life and activities, and often mentioned having trouble accepting their pain and other complaints (fatigue, dizziness, etc.).

Respondent 4: "Eh well, I would like to walk more, I would pretty much like to walk more, however my feet hurt so much when I do so"

There were, however, also some signs from respondents that they did accept the pain and had found ways to deal with the complaints. For instance, by mentioning that they tried to live mindful and expressing that accepting the complaints is a stepwise process they had to go through.

Sub-category: Doubt and hesitation towards the PNE and changing

Doubts and hesitation could be seen in multiple manners, most often by small concealed resistance, such as: *Yes, but.....", "I get it, however"* and non-verbal manners. As can be seen in table 5 and 6, some of the respondents started with some hesitation towards message of PNE from the previous session. Doubts and hesitation could also be related to the contributing factors. As can be seen in table 5 and 6, some respondents found it difficult to face the biopsychosocial factors that appeared to contribute to the persistence of their pain.

There could also be hesitation or resistance about how the healthcare professionals would frame or use certain words and explanations.

Respondent 5, physiotherapist: "It is also about how you deal with your own body, ehm, stress, tension, right? Partner respondent 5: "He is not a stressed person at all. He is, he can be super relaxed about almost everything. He is not somebody who worries easily"

During the start of the PNE session, all respondents had some doubts, hesitation or resistance about the explanation. These diminished as the healthcare professionals

further explained what they were doubtful about, for instance specific somatic perceptions. Most often during the PNE session respondents shifted their aim away from searching for a simple biomedical solution towards a biopsychosocial understanding of their complaints. Apart from respondent one, who found it difficult to comprehend the message that there was no specific biomedical cause, and respondent eight, who alternated between hesitation and agreeing with the PNE throughout the session. Furthermore, there was also resistance associated with behavioural change necessary to deal with pain in the long term in some respondents (especially in respondents three and eight, in smaller manner in respondent six). For instance, respondent three did agree with CS, the PNE and contributing factors, however found it hard to change her behaviour in the future (see table 5 and 6). Doubt and resistance towards change was harder to address and often persisted in those who experienced it. This form of hesitation linked to the category of Changeability of the respondent. There was no obvious resistance to be seen against the healthcare professionals and no major ruptures in the alliance were observed. Based on these video observations it was not possible to tell which actions from the healthcare professionals might have led to a decrease in doubt and hesitations of the respondents, or if other actions would have led to other outcomes. Furthermore, based on these video observations it is only partly possible to tell whether the amount of doubt and resistance had influence on the final outcomes of the PNE session, as we could only observe the doubt and resistance during this session and not what happened during the subsequent treatment.

Sub-category: Reaction of the healthcare professionals towards doubts and hesitation

The healthcare professionals would react in different manners towards doubt and hesitation. First, by asking them how the previous session of PNE with the GP went, they addressed doubts and hesitation surrounding this session in the beginning of the current PNE.

Most and foremost they would listen, acknowledge the respondent, take them seriously, acknowledge that living with these complaints is hard, recognizing what is important for the respondents and that behaviour change is difficult. Sometimes by listening and giving the respondents time to express themselves the resistance diminished. If the respondents didn't express these doubts and hesitations but they were noticeably there (because the nonverbal communication would show a closed posture, frowning, retreating in the chair), the healthcare professionals would ask them about the

resistance they saw or mention them themselves. When they were out in the open, they would then discuss them. Explaining the goals of the session also diminished some doubts and hesitations. Furthermore, a part of the PNE was to explain origin of the complaints the respondents experienced, which also dealt with doubt and resistance. When a word or sentence would cause resistance, they would reframe it and use other words. The same accounted for the examples they used.

Respondent 1, physiotherapist: ".... There are things that can enhance the signal, for instance stress. Such a conversation is a form of stress" Respondent 1: "Well not for me, I am not easily stressed"..

Psychologist: "Effort, maybe we should call it effort" Respondent 1: "Yeah that I realize"

Every now and then they cleared the air by using (appropriate) jokes. To further deal with resistance they also used positive reinforcement of the changes the respondents already made. Furthermore, they would try to bring insight to the respondent by asking in depth questions, having silences, summaries, reflective questions and sometimes more confronting questions, and giving the respondents time to express themselves. Or they would try to let the respondent look at his/her life from a distance. Giving hope for the future was also a way to diminish resistance of the respondents:

Respondent 8: ".. I used to walk so much, where, where is that all that investment? .. I'm so, I find that frustrating." Physiotherapist: "Well maybe that it is still somewhere?" Respondent 8: "[frustration laugh] Yes I hope so, because if it all just gets worse, well then I feel like there's nothing left. [shakes head] That's when I get those thoughts: this is never going to be over and just let it go.." Physiotherapist: "So yes, we should break that cycle"

Category 4: Outcome of the PNE process

Two sub-categories emerged in relation to the outcome of the PNE process: Process of insight into CS and factors surrounding CS and Stages of change of the respondent (table 5).

The process of insight into CS involves two components: the first is gaining knowledge, the second is applying this knowledge to yourself and possibly others. This process is interactive with the previous categories (e.g. *Process of change of the respondent*) and the category *Influence of changeability of the respondents*.

Respondent 1, psychologist: "And how can that (conversation) increase your complaints?" Respondent 1: "Yeah, well, it's probably because I turn my head a lot, I think"

Psychologist: "That is the explanation?"

- Respondent 1: "Yes, I think so" Physiotherapist: "But in a such a conversation you're facing the other right? So, then it isn't the head (turning)"
- Respondent 1: "Yes, that's true"

All respondents showed signs of trying to convey the message and understand what is being said. However, in some respondents it was more difficult to align the individuals' beliefs with the healthcare professionals views regarding the underlying pain mechanisms and contributing factors. As mentioned in table 5 and 6 all respondents gained different levels of knowledge about CS, the contributing factors and applied these differently to themselves. Signs of increased awareness, knowledge and understanding could be seen in the videos (a larger description is in the appendix) and would for instance be:

Respondent 6, session 2: "So you have to work here (points to head, smiles) as it were, to decrease the pain there (points to hip)"

Category 5: Communication skills

Several different communication skills were used and every healthcare professional had his/her own preferences. For example, some of the psychologists" used more "reflective summarizing", where the physiotherapist videotaped in the PNE session of respondent 7 used a lot of silences. Furthermore, the communication style was modified in a patient centered manner to the respondent in front of them. For instance, in persons with less formal education less difficult language was used.

Respondent 2, physiotherapist: "It is actually too strong of a response from the nervous system. What do you think?"

Communication skills and strategies included: encouraging, verbalizing non-verbal communication, parroting, positive reinforcements, silences, asking interactive questions, (reflective) summarizing, reflective and provocative questions and comments. A detailed description with examples can be found in the appendix.

Category 6: Interplay between the healthcare professionals

Throughout the PNE session there was an interplay between the healthcare professionals. For instance, one of the two would confirm the others explanation, deepen the explanation or check the cognitions and emotions of the respondent about that explanation. This interplay was also characterized by the synchronicity and fluidity between the healthcare professionals in the conversations.

Category 7: The influence of changeability of the respondents

As can be seen in figure 2 and 3, throughout the PNE a change process is initiated. The changeability, or psychological/cognitive flexibility, of the respondent, as seen in these videos during this process, encompasses a few steps:

- 1) the ability to exert reciprocity in interpersonal contact and openness;
- 2) the comprehension of the neurophysiology of the PNE and change in cognitions;
- 3) the self-reflectivity about the contributing factors;
- the ability to stop fighting against their complaints and in some way accept these complaints;
- 5) the ability to modify behaviour or accept help regarding this.

Whilst not all respondents were able to progress through all steps, all of the videos mapped onto the steps. For instance, most understood the neurophysiology, however some found it hard to comprehend (the depth of this comprehension remains unknown from these videos). Some respondents found it difficult to be reflective about their contributing factors, while others found it hard to accept their complaints and stop fighting. Last, some found it hard to modify or alter their behaviour in the future (described in table 5 and 6), especially respondent 3, 6 and 8:

Respondent 3: "yes, yeah.. it is about going on and on.. I'm a hard worker so that is what I do. And I know that I have more physical complaints afterwards. It is just... How can I change that? I'm like: yes, on the one hand I want to change because I want to get rid of the pain, however, on the other hand I want to live my life like I do now"

From these videos the changeability of a respondent was seen as something that corresponds with expected health literacy rather than intelligence, as both respondents

with a respectively expected higher or lower intelligence (based on education) could be proactive/cooperative or be resistant towards change.

Discussion

In this study we studied the second PNE session for patients with chronic pain in a transdisciplinary setting. The aim of this study was to explore the social process between the patient, partner and healthcare professionals in the practice of transdisciplinary PNE, in order to try to comprehend and theorize what contexts and influences may lead to effective PNE outcomes.

Several key findings in the process of PNE emerged: Underlying the PNE was the therapeutic alliance, as found in "Generating a safe and comfortable feeling, situation". The video observations showed that effort was put in to achieve and maintain a positive relationship. Important in this category, and overlapping with the "PNE process", was incorporating personal information from the intake in the PNE. These results are in concordance with a previous study highlighting the importance of a good patienttherapist interaction in pain rehabilitation⁶². The patient-therapist interaction is also known as the therapeutic alliance ⁴⁴, with a strong therapeutic alliance able to improve treatment outcome in patients having chronic pain^{45,46}. In correspondence with existing literature regarding the therapeutic "ruptures and repairs" were seen throughout the video observations^{63,64}. Ruptures are part of the therapeutic process and when healthcare professionals are attentive to ruptures, explore the negative feelings of the patient, and respond to those feelings in an open and non-defensive way, ruptures can strengthen the therapeutic relationship^{63,65,66}. Attending to ruptures, as seen during the observations, by the healthcare professionals through "communicational skills", as well as taking the patient seriously, is not only important, it can also contribute to positive patient outcomes⁶³. However, the therapeutic alliance grasps more than the healthcare professionals being caring, sensitive, a personal bond and positive feelings, it also comprehends agreement on tasks and goals and the perception of the patient that the offered interventions have potential⁶⁷⁻⁶⁹. In alignment with the therapeutic alliance, based on the video observations, PNE was used as a steppingstone to reach agreement for the following intervention. When respondents were not yet ready for these future interventions, a next session was scheduled to give the respondent time to think about it and further explore how to proceed.

Secondly, the step by step explanation of the PNE, the metaphors that were used and the individualization of PNE to the respondent, also known as "The PNE process", helped the respondents change their way of thinking ("The process of change of the respondent") from searching for a biomedical, somatic, cause into a more biopsychosocial understanding in which pain does not represent physical harm. The current is in concordance with a previous study, in which the respondents mentioned the education as being comprehensible, especially by the repetition, examples and individualized metaphors⁴⁰. The PNE process sparked a "Process of change of the respondents". This process involved the interactions between the respondents and healthcare professionals on topics such as: concerns about the complaints, trouble with accepting and dealing with the complaints, doubt and resistance towards the new information (most often at the beginning of the session). In most respondents the interaction and message of PNE positively influenced the topics mentioned above. Even though some nuances must be introduced: several respondents did still have some biomedical beliefs and questions at the end of the PNE, for instance: "Are you sure it is not an inflammation?". These guestions can be seen as doubts towards the provided explanation. Previous studies have shown that it is "normal for patients to feel this way": most often the patients have heard numerous theories about pain and reconceptualizing and adopting a new one takes time^{36-38,40}. From these studies, as well as other studies, it is known that patients often had various tests, treatments and diagnosis' and in order to re-engage trust, as well as following a biopsychosocial route requires time and an optimal therapeutic alliance⁴². Furthermore, it is important to keep in mind that relevance to the individual is important and PNE will not suit all patients³⁸, some patients will not agree with the neurophysiological explanation of pain^{40,62}. The PNE process activated the "Outcome of the change process" of the respondents, in which most respondents gained improved understanding of their pain and factors contributing to their pain experience. This is consistent with both previous qualitative research³⁸, an implementation survey⁴¹ and reviews on quantitative studies of PNE, which have shown increased pain knowledge, changed attitudes towards pain, improved coping, decreased kinesiophobia and pain catastrophizing in patients with chronic pain^{24,34}. However, in line with the current observations previous studies also mentioned that follow up during further treatment is needed to further deepen the reconceptualization and enhance these aspects^{33,34,37}.

Some of the respondents did understand the PNE, however found it hard to envision changes in their pain related behaviour, whilst others had already changed their daily practice based on the first PNE session. The "*Influence of changeability of the respondents*"

on the PNE and outcomes, even though present, was a difficult concept to grasp. There was no distinct profile noticed from the video observations, nor characteristics of a "responder" or "non-responder" to the PNE. This is in line with previous studies on the implementation of PNE, which also found that even though there are patients who respond better to PNE defining characteristics of these patients and patients who do not respond so positively continues to be difficult^{37,38,41}.

Limitations

There are several factors to mention with regards to limitations in this study. First, as there were only observations there was no data triangulation. This may have resulted in limited richness of the data as the experiences of the respondents surrounding their PNE have not been investigated. Furthermore, the observations only included the second PNE session. As the intake, first session with the GP and further treatments were not taken into account, the trajectory of the respondents surrounding the understanding and reconceptualization of their pain was not visible. It can be debated to what extent true changes and reconceptualization can be explored in the video recording of one session. The generalizability of the current study is limited to the transdisciplinary context of the observed PNE in which a psychologist and physiotherapist collaborate. Most PNE however, is delivered in practice in a monodisciplinary (physiotherapy) setting. It is known that physiotherapists often struggle with the psychosocial part of the treatment⁷⁰⁻⁷³. However, when they are trained in combining physiotherapy with psychology treatment it improves outcomes^{74,75}. In the current study the physiotherapists were trained in these psychosocial parts and had extensive clinical expertise in chronic pain and (bio)psychosocial treatments. This training might have influenced the categories that were found in this study. Another limitation concerns the translation of the theory. Even though efforts were made to convey the essence of the Dutch-language and the translation was reviewed by two team members, there are some considerations to mention. First, procedures to assure the trustworthiness of the translation, such as for instance forward and backward translation, were not performed. Second, it is previously proposed that any translation of language is problematic, as: 'no one can be sure of which concepts or words differ in meaning across languages and which do not, or if it matters in the context of the translation⁷⁷⁶. However, whilst critically challenging the translation, as mentioned in the method-section, some actions were made to enhance the validity of the translation. And last, the reflexivity of the researchers analyzing the data is a source of limitation⁷⁷. Both researchers performing the initial analysis have experience with providing PNE (one in a private clinical practice, the other both in clinical

practice, education and science, as well as in a transdisciplinary team) and believe PNE is an important part of the intervention of patients with (long term) pain. Furthermore, all other authors are in line with this view and in some form contribute to the field of PNE either in clinical practice, education or science. To ensure that the researchers were not overly involved a reflective paper was written and a neutral stance was sought throughout the analysis.

Implications for science

This study contributes to the research topic of PNE for patients with chronic pain. Future research should indicate, for instance by treatment trajectory studies⁷⁸, how the acceptance of the PNE could influence reconceptualization of cognitions, and how changes in outcomes following PNE unfold in a transdisciplinary setting. Furthermore, it would be interesting to investigate whether there is a difference in outcomes in comparison to similar PNE within monodisciplinary settings.

Implications for practice

It has been suggested previously to study clinical characteristics that may increase the effectiveness of PNE⁴¹. Even though qualitative research has a limited transferability⁴³, with some caution based on the current study several implications for practice can be mentioned that may help clinicians. First, of large importance: clinicians should be aware and invest in the therapeutic alliance between the patient and healthcare professional(s) prior, during and after the PNE. For instance, by taking the patient and his/ her problems seriously, using the aforementioned communicational skills and by using personal information from the intake in the PNE session. Second, individualizing PNE to the patient and the variety of the metaphors by which the healthcare professional can choose individual examples is of importance. And last, introducing the biopsychosocial perspective early on in treatment, whenever possible in an inter- or transdisciplinary setting, might enhance the patients' comprehension of the biopsychosocial aspects of their complaints. This forms a foundation and rationale for the patient to address these in a biopsychosocial-viewed pain management treatment.

Conclusion

In conclusion a variety of factors related to and influencing the practice of PNE emerged in relation to effective PNE outcomes. These included the underlying therapeutic alliance in "Generating a safe and comfortable feeling", situation, the steps, individualization and metaphors used during the PNE session in *The PNE process*, the "Process of change of the respondent" and the interactions of the healthcare professionals influencing this process. This leads to an increased knowledge about CS and biopsychosocial factors contributing to the pain experience in "*The outcome of the PNE process*". Further inter-linking findings were "Communication skills, *The interplay between health care professionals*" and "*The influence of changeability of the respondents*". Due to limitations in the design the transferability of the study findings is limited to the second session of PNE in a transdisciplinary setting. These findings provide the groundwork to inform further research into what is required for effective delivery of PNE services.

Acknowledgments

The authors would like to thank the patients, the persons they brought to the PNE and the healthcare professionals willing to participate in this study. Last, the authors would like to thank the administration of the transdisciplinary outpatient center with their help during the recruitment stage.

References

- 1. Merskey H, Bogduk N., International Association for the Study of Pain. Task Force on Taxonomy. *Classification of chronic pain: descriptions of chronic pain syndromes and definitions of pain terms.* 2 ed. the University of Michigan: IASP Press; 1994.
- 2. Breivik H, Collett B, Ventafridda V, Cohen R, Gallacher D. Survey of chronic pain in Europe: prevalence, impact on daily life, and treatment. *Eur J Pain*. 2006;10(4):287-333.
- Reid KJ, Harker J, Bala MM, et al. Epidemiology of chronic non-cancer pain in Europe: narrative review of prevalence, pain treatments and pain impact. *Curr Med Res Opin*. 2011;27(2):449-462.
- 4. Leadley RM, Armstrong N, Lee YC, Allen A, Kleijnen J. Chronic diseases in the European Union: the prevalence and health cost implications of chronic pain. *J Pain Palliat Care Pharmacother*. 2012;26(4):310-325.
- 5. Bekkering GE, Bala MM, Reid K, et al. Epidemiology of chronic pain and its treatment in The Netherlands. *Neth J Med.* 2011;69(3):141-153.
- van Tulder MW, Koes BW, Bouter LM. A cost-of-illness study of back pain in The Netherlands. Pain. 1995;62(2):233-240.
- GBD 2015 Disease and Injury Incidence and Prevalence Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990-2015: a systematic analysis for the Global Burden of Disease Study 2015. *Lancet.* 2016;388(10053):1545-1602.
- 8. Latremoliere A, Woolf CJ. Central sensitization: a generator of pain hypersensitivity by central neural plasticity. *J Pain*. 2009;10(9):895-926.
- 9. Woolf CJ, Salter MW. Neuronal plasticity: increasing the gain in pain. *Science*. 2000;288(5472):1765-1769.
- 10. van Wilgen CP, Keizer D. The sensitization model to explain how chronic pain exists without tissue damage. *Pain Manag Nurs.* 2012;13(1):60-65.
- 11. Torebjork HE, Lundberg LE, LaMotte RH. Central changes in processing of mechanoreceptive input in capsaicin-induced secondary hyperalgesia in humans. *J Physiol.* 1992;448:765-780.
- Gifford L. Pain, the Tissues and the Nervous System: A conceptual model. *Physiotherapy*. 1998;84(1):27-36.
- 13. Kosek E, Cohen M, Baron R, et al. Do we need a third mechanistic descriptor for chronic pain states? *Pain.* 2016;157(7):1382-1386.
- 14. Puentedura EJ, Louw A. A neuroscience approach to managing athletes with low back pain. *Phys Ther Sport.* 2012;13(3):123-133.

- 15. Moseley GL. A pain neuromatrix approach to patients with chronic pain. *Man Ther.* 2003;8(3):130-140.
- 16. Moseley GL. Widespread brain activity during an abdominal task markedly reduced after pain physiology education: fMRI evaluation of a single patient with chronic low back pain. *Aust J Physiother.* 2005;51(1):49-52.
- 17. Peyron R, Laurent B, Garcia-Larrea L. Functional imaging of brain responses to pain. A review and meta-analysis (2000). *Neurophysiol Clin.* 2000;30(5):263-288.
- Flor H. The functional organization of the brain in chronic pain. *Prog Brain Res.* 2000;129:313-322.
- Nijs J, Clark J, Malfliet A, et al. In the spine or in the brain? Recent advances in pain neuroscience applied in the intervention for low back pain. *Clin Exp Rheumatol.* 2017;35 Suppl 107(5):108-115.
- 20. Kucyi A, Davis KD. The dynamic pain connectome. *Trends in neurosciences*. 2015;38(2):86-95.
- 21. Gracely RH, Geisser ME, Giesecke T, et al. Pain catastrophizing and neural responses to pain among persons with fibromyalgia. *Brain.* 2004;127(Pt 4):835-843.
- 22. Turk DC, Okifuji A. Psychological factors in chronic pain: evolution and revolution. *J Consult Clin Psychol.* 2002;70(3):678-690.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil*. 2011;92(12):2041-2056.
- Louw A, Zimney K, Puentedura EJ, Diener I. The efficacy of pain neuroscience education on musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract.*(2016). 2016;32(5):332-55.
- 25. Nijs J, Paul van Wilgen CP, Van Oosterwijck J, van Ittersum M, Meeus M. How to explain central sensitization to patients with 'unexplained' chronic musculoskeletal pain: practice guidelines. *Man Ther.* 2011;16(5):413-418.
- Gallagher L, McAuley J, Moseley GL. A randomized-controlled trial of using a book of metaphors to reconceptualize pain and decrease catastrophizing in people with chronic pain. *Clin J Pain.* 2013;29(1):20-25.
- 27. Moseley GL. Evidence for a direct relationship between cognitive and physical change during an education intervention in people with chronic low back pain. *Eur J Pain.* 2004;8(1):39-45.
- 28. Moseley GL, Nicholas MK, Hodges PW. A randomized controlled trial of intensive neurophysiology education in chronic low back pain. *Clin J Pain*. 2004;20(5):324-330.
- 29. Meeus M, Nijs J, Van Oosterwijck J, Van Alsenoy V, Truijen S. Pain physiology education improves pain beliefs in patients with chronic fatigue syndrome compared with pacing and self-management education: a double-blind randomized controlled trial. *Arch Phys Med Rehabil.* 2010;91(8):1153-1159.

- 30. Van Oosterwijck J, Meeus M, Paul L, et al. Pain Physiology Education Improves Health Status and Endogenous Pain Inhibition in Fibromyalgia: A Double-Blind Randomized Controlled Trial. *Clin J Pain*. 2013.
- 31. Van Oosterwijck J, Nijs J, Meeus M, et al. Pain neurophysiology education improves cognitions, pain thresholds, and movement performance in people with chronic whiplash: a pilot study. *J Rehabil Res Dev.* 2011;48(1):43-58.
- 32. Moseley L. Combined physiotherapy and education is efficacious for chronic low back pain. *Aust J Physiother.* 2002;48(4):297-302.
- Wood L, Hendrick PA. A systematic review and meta-analysis of pain neuroscience education for chronic low back pain: Short-and long-term outcomes of pain and disability. *Eur J Pain*. 2019;23(2):234-249.
- Watson JA, Ryan CG, Cooper L, et al. Pain neuroscience education for adults with chronic musculoskeletal pain: a mixed-methods systematic review and meta-analysis. *J Pain*. 2019;20(10)1140.e1-1140.
- 35. Geneen LJ, Martin DJ, Adams N, et al. Effects of education to facilitate knowledge about chronic pain for adults: a systematic review with meta-analysis. *Syst Rev.* 2015;4:132.
- 36. King R, Robinson V, Elliott-Button HL, Watson JA, Ryan CG, Martin DJ. Pain Reconceptualisation after Pain Neurophysiology Education in Adults with Chronic Low Back Pain: A Qualitative Study. *Pain Res Manag.* 2018;2018:3745651.
- King R, Robinson V, Ryan CG, Martin DJ. An exploration of the extent and nature of reconceptualisation of pain following pain neurophysiology education: A qualitative study of experiences of people with chronic musculoskeletal pain. *Patient Education and Counseling.* 2016;99:1389-1393.
- 38. Robinson V, King R, Ryan CG, Martin DJ. A qualitative exploration of people's experiences of pain neurophysiological education for chronic pain: The importance of relevance for the individual. *Manual Therapy.* 2016;22:56-61.
- Nijs J, Van Houdenhove B. From acute musculoskeletal pain to chronic widespread pain and fibromyalgia: application of pain neurophysiology in manual therapy practice. *Man Ther.* 2009;14(1):3-12.
- 40. Wijma AJ, Speksnijder CM, Crom-Ottens AF, et al. What is important in transdisciplinary pain neuroscience education? A qualitative study. *Disabil Rehabil*. 2018;40(18):2181-2191.
- 41. Louw A, Puentedura EJ, Zimney K, Cox T, Rico D. The clinical implementation of pain neuroscience education: A survey study. *Physiother Theory Pract.* 2017;33(11):869-879.
- 42. Louw A, Zimney K, O'Hotto C, Hilton S. The clinical application of teaching people about pain. *Physiother Theory Pract.* 2016;32(5):385-395.
- 43. Creswell JW. *Qualitative Inquiry and Research Design, choosing among five approaches.* Second ed. London: Sage Publications, Inc.; 2007.

- 44. Rogers CR. *Client-centered therapy.* American Psychological Association Washington, DC; 1966.
- 45. Kinney M, Seider J, Beaty AF, Coughlin K, Dyal M, Clewley D. The impact of therapeutic alliance in physical therapy for chronic musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract.* 2018:1-13.
- 46. Hall AM, Ferreira PH, Maher CG, Latimer J, Ferreira ML. The influence of the therapist-patient relationship on treatment outcome in physical rehabilitation: a systematic review. *Phys Ther.* 2010;90(8):1099-1110.
- 47. Lofland J, Snow, D., Anderson, L., Lofland, L.H. *Analyzing social settings, A guide to qualitative observation and analysis.* Fourth ed. Wadsworth: Cengage Learning; 2006.
- 48. Kawulich B. Participant Observation as a Data Collection Method. Forum Qualitative Social Research, 6(2), Art. 43. http://nbn-resolving.de/urn:nbn:de:0114-fqs0502430. Published 2005.
- 49. Charmaz K. Constructing Grounded Theory. 2nd ed: Sage Publications Ltd.; 2014.
- 50. Charmaz K. 'Discovering' chronic illness: using grounded theory. *Soc Sci Med.* 1990;30(11):1161-1172.
- 51. Corbin J SA, eds. *Basics of qualitative research: techniques and procedures for developing grounded theory.* London: Sage; 2008.
- 52. Tong A, Sainsbury P, Craig J. *Consolidated criteria for reporting qualitative research (COREQ): a* 32-item checklist for interviews and focus groups. Vol 192007.
- 53. Shaw L, Walker R, Hogue A. The art and science of teamwork: enacting a transdisciplinary approach in work rehabilitation. *Work.* 2008;30(3):297-306.
- 54. Reilly C. Transdisciplinary approach: an atypical strategy for improving outcomes in rehabilitative and long-term acute care settings. *Rehabil Nurs.* 2001;26(6):216-220, 244.
- 55. Ettema JHM, Arrindell, W.A. *SCL-90. Symptom Checklist. Handleiding bij een multidimensionele psychopathologie-indicator.* Lisse: Swets Test Publishers; 2003.
- 56. Cormier S, Lavigne GL, Choiniere M, Rainville P. Expectations Predict Chronic Pain Treatment Outcomes. *Pain*. 2016 Feb;157(2):329-38.
- 57. Prochaska JO, Norcross JC. Stages of change. *Psychotherapy: Theory, Research, Practice, Training.* 2001;38(4):443-448.
- 58. Lincoln Y, Guba, EG. *Naturalistic Inquiry*. Newbury Park, CA1985.
- 59. Transcriptionsoftware f4, Deutschhausstrasse 22a, 35037 Marburg, Germany.
- 60. Birt L, Scott S, Cavers D, Campbell C, Walter F. Member Checking: A Tool to Enhance Trustworthiness or Merely a Nod to Validation? *Qualitative Health Research*. 2016;26:1802-1811.
- Nijs J, Torres-Cueco R, van Wilgen CP, Girbés EL, et al. Applying Modern Pain Neuroscience in Clinical Practice: Criteria for the Classification of Central Sensitization Pain. *Pain Physician*. 2014;17:447-457.

- 62. Oosterhof B, Dekker JHM, Sloots M, Bartels EAC, Dekker J. Success or failure of chronic pain rehabilitation: the importance of good interaction a qualitative study under patients and professionals. *Disability and Rehabilitation*. 2014;36:1903-1910.
- 63. Eubanks CF, Muran JC, Safran JD. Alliance rupture repair: A meta-analysis. *Psychotherapy* (*Chic*). 2018;55(4):508-519.
- 64. McLaughlin AA, Keller SM, Feeny NC, Youngstrom EA, Zoellner LA. Patterns of therapeutic alliance: rupture-repair episodes in prolonged exposure for posttraumatic stress disorder. *J Consult Clin Psychol.* 2014;82(1):112-121.
- 65. Safran JD, Muran JC, Eubanks-Carter C. Repairing alliance ruptures. *Psychotherapy (Chic)*. 2011;48(1):80-87.
- 66. Larsson MH, Falkenstrom F, Andersson G, Holmqvist R. Alliance ruptures and repairs in psychotherapy in primary care. *Psychother Res.* 2018;28(1):123-136.
- 67. Bordin ES. The generalizability of the psychoanalytic concept of the working alliance. *Psychotherapy: Theory, research & practice.* 1979;16(3):252.
- 68. Roth A, Fonagy P. *What Works for Whom?: A Critical Review of Psychotherapy Research.* 1 edition (12 Jan. 2006) ed. London: Guilford Press; 1996.
- 69. Squier RW. A model of empathic understanding and adherence to treatment regimens in practitioner-patient relationships. *Soc Sci Med.* 1990;30(3):325-339.
- 70. Daykin AR, Richardson B. Physiotherapists' pain beliefs and their influence on the management of patients with chronic low back pain. *Spine (Phila Pa 1976).* 2004;29(7):783-795.
- 71. van Wilgen CP, Beetsma A, Neels H, Roussel N, Nijs J. Physical therapists should integrate illness perceptions in their assessment in patients with chronic musculoskeletal pain; a qualitative analysis. *Man Ther.* 2014;19(3):229-234.
- 72. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. *J Physiother*. 2015;61(2):68-76.
- Singla M, Jones M, Edwards I, Kumar S. Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study. *Man Ther.* 2015;20(2):328-334.
- 74. Wilson S, Cramp F. Combining a psychological intervention with physiotherapy: A systematic review to determine the effect on physical function and quality of life for adults with chronic pain. *Physical Therapy Reviews*. 2018;23(3):214-226.
- 75. Silva Guerrero AV, Maujean A, Campbell L, Sterling M. A Systematic Review and Meta-Analysis of the Effectiveness of Psychological Interventions Delivered by Physiotherapists on Pain, Disability and Psychological Outcomes in Musculoskeletal Pain Conditions. *The Clinical Journal of Pain*. 2018;34(9):838-857.

- Temple B, Young A. Qualitative Research and Translation Dilemmas. *Qualitative Research*. 2004;4(2):161-178.
- 77. Jootun D, McGhee G, Marland GR. Reflexivity: promoting rigour in qualitative research. *Nurs Stand.* 2009;23(23):42-46.
- 78. Caneiro JP, Smith A, Linton SJ, Moseley GL, O'Sullivan P. 'How does change unfold?' an evaluation of the process of change in four people with chronic low back pain and high pain-related fear managed with Cognitive Functional Therapy: A replicated single-case experimental design study. *Behav Res Ther.* 2019;17:28-39.

Appendix 1: Comprehensive results

The findings that are presented here are the comprehensive results from the theoretical coding of the research subject: *"Exploring patient-therapist interactions for effective Transdisciplinary Pain Neuroscience Education: A qualitative Constructivist Grounded Theory study"*. The tables and figures mentioned in the appendix correspond with the tables and figures mentioned in the research subject.

Category 1: Generating a safe and comfortable feeling and situation

Everyone in the process invested in creating a warm and friendly environment. Efforts to create and keep an alliance with the respondents was often seen from the healthcare professionals. These efforts resulted in the respondents showing both verbal and non-verbal signs of feeling at ease, acknowledged and understood. Signs in non-verbal communication were such as 'sighs', 'relaxing the shoulders', often 'nodding', 'gentle smiles' and getting emotional. It was also seen in verbal communications such as '*yes that is it!*, '*exactly*' or for instance finishing the sentence of the healthcare professional.

Respondent 3: "I do feel as if you understand me" Respondent 1, Psychologist: "Because we don't feel what you feel, but we believe every word you say" Respondent 1: "Yes (nods twice)"

Generating a safe and comfortable feeling was started during the prior intake and continued to be an active process throughout the session with all respondents. Even though there were small individual differences, they were too small to describe as differences between respondents.

Three subcategories were identified. From both the respondents and healthcare professionals: *Making jokes to keep the conversation light*. From the healthcare professionals: *Building an alliance with the respondents* and *Faux pas from the healthcare professionals*.

Making jokes to keep the conversation light

Making jokes was used by everyone to keep a light atmosphere. Whenever respondents felt a bit tensed or a bit awkward, any one of them would 'laugh it off' or joke. This could be when the respondent didn't really understand something or when a difficult topic was addressed by anyone of them, when they felt less comfortable talking about themselves.

Respondent 6: 'yeah, that sawthooth pattern (in daily activities) (laughing)'

Building an alliance with the respondent

Building an alliance with the respondent consists of multiple efforts: *Creating a safe environment*, in which the healthcare professionals *Showed understanding, recognition and empathy,* and *acknowledged the authenticity of pain*.

Creating a safe environment

Creating a safe environment ensured the respondents dared to express themselves. This environment was created non-verbal by for instance offering a comfortable chair and creating a comfortable space. Verbally it was done telling them they could always ask any question, no matter what for question or how critical, during the session or in later treatment and responding to those questions with ease and understanding. Furthermore, this save environment was created by letting them tell their story and giving the respondents time to express themselves. Expressed or non-verbal shown discomfort or awkwardness as felt by the respondents were handled with care. For instance, respondents would be joking, or telling something that was hard for them:

Respondent 6: 'I have just browsed through it (booklet), because I have to concentrate very hard when I am reading. I am a bit dyslexic and I often have something like: 'what is written there?' And I read everything wrong again' Psychologist: 'You have to take the time for it.'

Discomfort was also handled by making polite jokes by the healthcare professionals, often followed by a sentence or gesture to reassure the respondent they took them seriously.

Respondent 1: (mentions that he had a lot of pain after the movement assessment of the intake) Psychologist to physiotherapist: (jokes) 'What did you do? Did you pull?' Respondent 1: ' No, I had to do this and this (turns head slowly, moves arms slowly), and well' Physiotherapist: 'That does sound like a funny movement' Psychologist: 'I think I'll quit our collaboration (all laugh) ... But (serious) the physio made a movement with your neck you don't usually make'

Creating a comfortable space meant placing the respondent in the center. For instance by first asking them if it was ok to contact other healthcare professionals related to the

treatment and what kind of information the respondent wanted to be shared. It was also done by asking the respondent for his/her preferences:

Physiotherapist respondent 7: ' Regarding, the future treatment trajectory, it is important that you say what you want .. of course, it's nice that you take our feelings into account, be aware that the process is all about you.'

Recognizing and understanding personal suffering and pain

Recognizing and understanding the respondents' pain by the healthcare professionals meant showing an overall empathy for the respondent as a person and their suffering. The healthcare professionals often recognized, emphasized and expressed the frustrations, feelings of heaviness and the hardships of dealing with chronic pain and fatigue the respondents felt.

For instance, by being caring, wanting to gain an in-depth knowledge of the complaints and emotions, acknowledging mixed emotions, underlining emotions, renaming emotions and giving the respondents time to deal with their emotions.

Respondent 3: 'Now I'm getting emotional again ..' Psychologist: 'That really touches you, right?'

As well as expressing what these emotions did to them as healthcare professionals.

Respondent 4, physiotherapist: 'Could it also be eehm .. your story is so moving, .. euhm sorry. A very moving story, yes, it keeps you busy, kids, they keep you busy'

Some of the most used sentences were: 'yes, that is hard', 'I get it' and 'we understand that it is difficult'.

The impact of the pain on the respondent's life and future was often mentioned.

Besides an overall empathy, the healthcare underlined their empathy for the unfamiliarity of the respondents with CS and contributing factors. They recognized that is not easy to understand and that it is normal to not know all that information. Thereby they acknowledged the doubt some respondents had towards the PNE.

Respondent 2, physiotherapist: 'Well, the hard thing is, you can't see anything right? Someone with hay fever has tears in her eyes and know it's hay fever' Respondent 2: 'Yes, tears and snotty, she gets a cold, she has red eyes and that (hay fever) is why, right?' Physiotherapist: 'That is in essence an overreaction of the nervous system. That is actually what it is about and now about pain signals pain. The pain is real; however, you can't see anything different in those people. That is often what makes you insecure, that is what often makes it incomprehensible: why am I in pain now?' Respondent 2: 'Yes, yes' (nods twice)

Furthermore, the healthcare professionals showed an understanding, recognition and empathy for the suffering of the respondents.

Respondent 5, physiotherapist towards partner: 'So he's a completely different man you say'

The healthcare professionals often expressed the frustrations, feelings of heaviness and the hardships of dealing with chronic pain and fatigue the respondents felt. They recognized that pain costs a lot of energy, is frustrating, feels uncontrollable, disappointing, often endures, demands attention, and is difficult to deal with.

Respondent 8, physiotherapist:' Well, it is complicated to deal, to, deal with this'

The impact of the pain on the respondent's life and future was often mentioned, for instance how activities were limited, normal activities where now full of effort and pain and weighed upon them.

Besides acknowledging the emotional impact of pain on a person, the healthcare professionals also corroborated on the realness of the complaints. By explaining to the respondents that: *'CS isn't something you do to yourself', 'it is a process that happens within your body and outside someone's will'. 'Just because you can't see it on the outside, doesn't mean it is not real'. 'Pain is not something someone imagines', 'nor is someone a wimp for having CS pain'. 'CS pain is a real pain and not 'in your head'', 'it is not psychological', 'whatever you feel is real'.*

For those respondents (respondents 1, 2, 4 and 8) who had a biomedical question or quest the healthcare professionals understood and acknowledged this quest and that

for them biomedical explanations needed to be excluded. If necessary, and not already performed, biomedical examination, such as bloodwork, was done. In an interactive process with the patient outcomes of these or previous examinations were explained in the light of PNE.

They also understood that respondents find it hard to differentiate between 'nociceptive' and 'CS' pain and it is hard to deal with chronic pain:

Respondent 3, physiotherapist: 'If only I would give my arm some rest and it would become less painful, however, that is not the case (in your situation)'

One of the ways in which they acknowledged the realness of the complaints is by explaining to the respondents that around 20% of the population experiences chronic pain. Furthermore, they explained to the respondents that even though there is no damage, their pain is real.

Among this category was also the acknowledgement that the contributing factors, such as frustration, negative feelings towards pain, tension, avoidance, are often factors one changes/does/feels unconsciously and not are easy to change. However, also acknowledging that when these contributing factors were changed, life would be more enjoyable.

Respondent 1, physiotherapist: 'Exactly, however, hurt, or screeching sounds scream for attention, because it is just very annoying and often your thoughts go straight to those feelings.. however, that is also a factor which maintains the problem. So try distraction, that's easy to say, huh, from the distraction, but (...) it's a given. If you manage to find distractions, for example with music, which makes you distract from the sound, it eventually gives you peace.'

Using personal information from the intake in the PNE

The healthcare professionals used personal sometimes specific information which they heard or had seen during the intake. Using this personal story was important to enhance its relevance to the individual, reinforce the explanation and made respondents feel heard. For instance, the healthcare professionals would explain what they had seen in the physical examination (moving tensed, being tensed), repeat information from the intake (trauma/memories surrounding trauma, always pushing yourself, struggling,

trying everything to control the pain, working nightshifts, going beyond ones limits, finding it hard to be relaxed) and/or emotions that where present during the intake. Or they just referred to what the respondents said by: '*I've heard you say that before*'.

Respondent 2, psychologist: 'You told me in the first conversation: I am not that much of a worrier, and I don't worry that much, but I do feel that the pain now dominates my life. (R: Yes, yes, yes, yes.) And that is a very frightening thought. Respondent 2: Yes. Well that is the case at this moment'

Faux pas from the healthcare professionals

Faux pas from the healthcare professionals were unhelpful actions that caused small ruptures in the therapeutic alliance. On a communicational level these were for instance asking many questions in one sentence and using difficult words.

Respondent 5, physiotherapist: 'No that is true.. can you grasp what I am saying? How is that for you? .. Because it means a difference in dealing with the pain, right? And how you look at pain. Is that, is that a reassurance or not? How does it feel for you both?'

Respondent 5: '... I don't know how that feels'

Respondent 5, partner: 'I am so glad that it isn't a herniated disc anymore, yet on the other hand he has something that isn't visible. How do we explain that to others?'

Furthermore, faux pas would be when the healthcare professional quickly went through the contributing biopsychosocial factors:

Respondent 7, physiotherapist: 'Again, raising the bar high, right? (Yes) and constantly over asking your body (Yes) that is not something you can always control (No) not within your family, within your work (No, no) the nightshifts you work (Yes) Yes, the irregular work shifts (business) Yes' Respondent: 'Yes, well it is..' Physiotherapist: 'It is quite a lot' Respondent: 'Yes, well in terms of work, it is kind of a lot'

Physiotherapist:' Well, (pointing to the board) and thoughts also go with the anger it evokes'

Sometimes there would be an unclear answer to a question or topics and questions mentioned by the respondent or partner were not readdressed.

Furthermore, faux pas also happened when the healthcare professional was not at his/ her best game explaining the neurophysiology of PNE, had a misapprehension on a source of stress of the respondent, or mentioned arthritis all of the sudden somewhere in between the PNE.

Some other faux pas were: bridging the contributing factors in the PNE was explained in a inartful manner, a metaphor was used that the respondent didn't understand, and in one of the respondents two appointments were scheduled wrong.

These faux pas were often addressed and the ruptures repaired.

Category 2: The PNE process

The category the PNE process consists of the sub-category 'characteristics of the PNE, in which the content of the PNE, individualization of the PNE and examples and metaphors are described. Among with the PNE there are different processes that are in action, related to the next categories 'Process of change of the respondent' and 'Outcome of the PNE process'.

The steps of the PNE process

Most often the PNE conversation would involve similar steps in the process, as described in table 4 and in the table below (supplementary table 1). By this stepwise structure respondents were engaged in the explanation of their pain/complaints.

Supplementary table 1. Detailed description of the general lay out of the PNE session

General content of the PNE session	Healthcare professional explaining (most often)	Particulars
1 a) Welcome back, general lay out of this session	Psychologist	1a and 1b could be exchanged
1 b) Ask how the first session with the GP is experienced by the respondent and what the respondent remembers of the GP's explanation.	Psychologist	
1 c) Explain goals for this session; more clarification, explain their pain in more depth, some guidance in how to deal with his/her pain, and further treatment(s).	Psychologist	Reaction of the respondents on the goals were expectations and questions: which treatment trajectory, why am I in pain/what is the cause of CS, the desire to have more energy, more tools to deal with pain, desire to get rid of the pain
2) Explain difference between acute and chronic pain; 20% of the population experiences chronic pain, examples. There is a difference in cause, duration, progression, efficacy of NSAIDs, often increased pain. Both are alarming, however in acute pain there is harm, in chronic not. There is an overactive processing of (danger) signals, fire alarm metaphor.	Physiotherapist	Respondents who first experienced acute pain were explained that it started with acute pain, however there is no 'damage left' and the metaphors were their own (for instance falling of stairs) Based on the classification of pain ^{61.}
3) From 'damage' to CS: explaining the difference between acute and chronic pain for the individual. No damage, however: the pain signal is real, credible. Biomedical cause are ruled out, or not enough to explain the pain (arthrosis), and/or: it started with acute pain or neuropathic pain, however there is now an overactive processing system. Based on patients' own description of pain, disabilities, prior treatments and tests.	Physiotherapist	Normalizing the pain. Important in this phase was that the healthcare professionals underlined the credibility of the pain, taking the respondent and his/her pain really serious.
4 a) Explanation that it is an overactive processing/ sensitization. By drawing a schematic picture of the nervous system: brain, spinal cord, and peripheral nerves. Explaining neurophysiology of acute pain by an example (for instance a cut, individual example), while drawing ascending and descending pathways, in order to normalize acute pain (helpful).	Physiotherapist	
4 b) Explain neurophysiology of chronic pain/central sensitization, while drawing the changes in the central nervous system: ascending and descending overactive interpretations of sensory input, decreased filter for sensory input in the spinal cord and brain, whilst using individual examples (for instance hyperalgesia, pain when sitting).	Physiotherapist	
5) Bridging the PNE towards the biopsychosocial factors in various ways, for instance: now we are going to see what might induce this system to get overactive, you have previously said Do you think this could influence this system? Do you have an idea what could influence this? Or via predisposition, or naming the first contributing factor.	Psychologist	The variations in this 'bridge' depended on the healthcare's' preferences and their estimation of the patients' preferences

Supplementary table 1. (Continued)

General content of the PNE session	Healthcare professional explaining (most often)	Particulars
6) Explain and discussing the biopsychosocial factors that contribute in the respondent's pain experience, while drawing the links in the brain. Individual contributing factors were discussed in a for the respondent non-threatening manner, such as: previous operations, continuing searching for the cause/solution, questions about their pain, maladaptive cognitions, not accepting pain/fighting, emotions surrounding pain (frustration, anger, sadness, compulsiveness, fear, fear of movement), emotions surrounding other aspects of life (traumas, family, work, etc), not ventilating emotions, stress, physical tensions, changed behavioral patterns (small: not lifting, larger: doing a lot one day and nothing the other), avoidance, sleep, irregular working schedules, not taking care of oneself, putting the bar high, trouble dealing with life, life's purpose/fulfillment, financial problems, pressure to go back to work.	Both, however most often the psychologist	
7a) Based on the contributing biopsychosocial factors some individual advice and guidance in dealing with his/ her pain was provided. Such as: this needs a different kind of handling than acute pain, stop looking for the fire, stop looking for a magic pill, take the pain into account but not too much, don't do all or nothing/ try to minimize sawtooth patterns in activities, regain structure, seek distraction, you can influence your pain; it's not easy but you can, simple therapies won't work in this system and will disappoint you, allow yourself time to grow, it's a learning process, taking care of yourself is important, dealing with stress and tension, relaxation (overall, and when in pain) is important, it's important to break this spiral, it is important to enhance movement/ activities, smooth reintegration to work is important, accepting the pain vs. fighting it, create your own vision on how to deal with chronic pain.	Both	7a/b/c could be exchanged

Supplementary table 1. (Continued)

Supplementary table 1. (Continued)

General content of the PNE session	Healthcare professional explaining (most often)	Particulars	
7 c) Give a realistic picture and hope for the future. Thereby giving back purpose: These complaints are not something that will disappear in an instant. This situation (CS) is reversible or partly reversible, but we don't know where we'll end up or how long it will take. We are aiming to decrease pain in the long term. It's not about being completely pain free. The pain itself is hard to treat, however by treating the factors surrounding it, the pain-system might get calmer. This all depends on the person, complaint, history, factors surrounding it. We are aiming to give back control. The better one deals with pain/life, the better it is to bare, the more energized we feel, the more satisfied we are with life, the more the pain is manageable and sometimes fades away. It's not easy, but possible. Individual responses: 'You have already changed some factors (work, psychological treatment), those surrounding things are what makes it better.' 'Your body is strong, that is important to remember, the pain signal might stay but that is a writing error, maybe it is about giving life meaning even though you are in pain.'	Both		
8) Closure of the PNE session: There is a lot we have talked about today. These factors, treating those is what we aim to focus on. Are there any questions? Is everything clear? Sometimes: shortly explain what is going to happen in the next sessions. Advice to write down any thoughts and questions for the next sessions.	Both		

In most occasions these steps would change according to the individual respondent or preferences of the healthcare professionals. As described in the next category.

Individualizing the PNE

As can be seen in table 5 and 6 every PNE session was individualized according to the respondent. Individualizing the PNE was done in a subconscious manner and multiple ways. First, in all sessions the healthcare professionals recalled information and questions from the respondents that were mentioned in the intake. Furthermore, if the respondent had mentioned questions or expectations during the intake or PNE session, these were answered during the session. The individual contributing biopsychosocial factors were explained during the second part of the PNE. Additionally, the healthcare professionals copied the words and language of the respondents, with language based on the individual. They explained in a pace adapted to the respondent

and used recognizable metaphors for the respondents. The treatment options were discussed in a shared decision manner, with an individualized level of involvement.

Individualizing the PNE also meant keeping it interactive by asking questions to the respondent, checking whether the respondent understood the previous information.

Respondent 8, psychologist: 'Could you think of something? What could also be a factor is the thoughts you have about the pain ... could you give an example of that?'

Furthermore, in the beginning of the PNE the healthcare professionals would often ask them what their knowledge about (chronic) pain was or mentioned that the respondent might already know some things about (chronic) pain from previous treatments/sessions. Based on their answers they would then proceed. Sometimes reducing the amount of neurophysiology because the respondent already knew a fair deal about it. In some PNE sessions the neurophysiology part was decreased and less complicated vocabulary, in respondent five without even calling it 'sensitization', to keep it understandable. If a respondent mentioned their own metaphors for the explanation the healthcare professionals would incorporate these in the further explanation.

Individualizing the PNE also mentioned explaining all complaints of the individual respondent, the pain, but also fatigue, dizziness, (physical and mental) tension, seeing flashes, hearing sounds and trouble sleeping.

Respondent 1, physiotherapist: 'So that is it, this is it. It's not affectation.. It's just, that system is upset, just as it is mixed up, with you, and you can notice that in different aspects, it can just be pain, but with you we see a combination of pain, sound, light..'

Metaphors and examples

Metaphors and examples were based on the individual person, in plain language, based on the expected health literacy of the respondent. Often the respondents would react in agreement, this could be seen by the non-verbal communication (listening seriously and nodding), or in a verbal manner by 'yes', 'hmhm' (often in combination with nonverbal communication) or by responding to a metaphor with their own metaphor/ example. Respondent 7, physiotherapist: 'It is, no, these are none painful signals that go up (points from spine upwards to the brain), they, they come into a network which immediately says: POING, pain! Because you sense pain in your brain, not in your foot'

Metaphors and examples related to the neurophysiology were: examples of acute pain (falling on your knee, spraining an ankle); our body has no difference between acute pain and chronic pain; alarm bell going off; a burglary/fire alarm that goes off for nothing/ wind/a fly, which is not necessary; fire alarm is too sensitive; hay fever as an example of amplified reactions; there is pain however no damage; when in pain one searches for the fire but it is not there; you are put on the wrong foot (by own cognitions of the pain: bone fragment, something being loose in the skull); the problem is real, however there is no physical cause, the problem is the alarm system; the problem is not the fire; the system is the problem; the system is over excited; the system is under pressure; the filter doesn't filter signals enough; a volume button that is set open; amplified signals; sensitizing of the body and processing system; everything that enters that system is interpreted as pain, while there is none there is no damage; brain is on the wrong track; misinterpretation; example of how he sees light flashes in the dark as an example of how the nerves are overstimulated, same as for pain and sounds, it's something he sees/hears yet is not there, overactive nervous system that makes own sounds and light, started with blow to the head, now overactive; the stimuli from her legs are interpreted as pain, there is nothing wrong with the legs it is the system behind it that interprets the stimuli as painful, that is where you feel pain; it's the signals that are coming up, enter the network, which 'poing' marks pain, you feel pain in your brain not in your foot; examples on how normal daily activities (sitting, using the stairs, etc.) are painful due to amplified signals; example of how cutting yourself doesn't hurt when your children are crying, because then the filter does work and the brain prioritizes; example of browsing through vacation photo's vs not seeing them for a while; the pain system is wired, however not a solid thing. Metaphors and examples related to the contributing factors were: pain as an unwelcome guest; how stress negatively affects to your cognitive functioning; examples how their pain control system used to be effective; how persistence in survival mode negatively affects pain; examples on how continuing to persevere with bronchitis or a sprained ankle is not useful; it is a bucket with water that is overflowing (always trying to please others); how 'normal fatigue' is concurred in a few weeks of holiday, however hers' continues and is therefore not just a wrought-up phase; a lid that opens up every time (traumas); operations are often part

of the problem, because they are such heavy input, that's why they often stop after 2-3 surgeries; operations can be useful however, are also detrimental for this system; your brain keeps on going (thinking, worrying), and is linked to all sorts of connections in your body, so 'pfieuw' there goes the signal again, it is all connected; asthma as a physical condition which is affected by stress; how not taking care of yourself increases the system; physical and psychological taking care of oneself is the same thing.

Metaphors linking the neurophysiology and contributing factors and further treatment were: about this alarm system, if it's a little clearer, more understandable, you can take advantage of it, it's not something that doesn't change, you are not crazy, this is the explanation for your complaints; the cause is in the system, not in that something is damaged, this other interpretation of your complaints could be the first step; it's about quieting down this system; the more relaxed it is up there, the better the filter works, that's how the system works; lyrica (medication) tries to ignite the filter, which helps in some people; when your emotions calm down a bit you might notice that you're functioning improves.

Category 3: Process of change of the respondent

Understanding, agreeing with and implicating the PNE on his/herself and daily life was a process for the respondent. This change process is a personal iterative process with multiple steps, as displayed in figure 3.

Not all steps could be seen in all videos: some respondents found it more difficult to comprehend the information, others found it hard to accept their situation, or found it hard to change, whereas others already understood the knowledge, had no trouble accepting, or already made some differences in their lives. These individual differences can be seen in table 6, describing the individual differences of the PNE session and table 5, describing the outcome of the PNE process.

Within this change process respondents had several struggles, which emerged as the sub-categories, as can be seen in the theoretical framework. These struggles surrounded 'concerns about the complaints they experienced', 'having trouble accepting the complaints' and 'doubts and resistance towards the PNE and change'. Throughout these struggles, a 'process of insight in CS and the factors surrounding the experienced complaints' also started and in some even changes for daily life were made or planned for in the future.

Concerns about the complaints they experienced and how the healthcare professionals react to this

There was a cyclical process between the concerns about the complaints the respondents perceived and the explanation for this (CS) during the PNE. All respondents expressed concerns surrounding their complaints, however respondents 3,4,6 and 8 more often. Furthermore, some respondents expressed concerns surrounding how their environment reacts to their complaints, either by themselves or because the healthcare professionals asked them. The healthcare professionals would often react to these concerns with conversational skills.

Concerns about the complaints

Respondents expressed their concerns about their complaints often in the start of the PNE. Most often throughout the PNE this decreased or diminished completely, apart from respondent one and eight (table 5 and 6). These concerns surrounded the etiology of their complaints: how it could be that they were now experienced chronic pain but not prior, why their complaints were increasing, and how to deal with their pain.

Respondent 8: 'and when I get up I get this shooting pain in my foot...then I'm like, well!, just by lying! I'm thinking I'm giving my body a bit of rest and then when I get up the pain is even worse. That's when I'm concerned, I'm like: What is it? What is it?'

Furthermore, they expressed concerns surrounding how their social environment reacts to their complaints. Either about their complaints in the past: problems with employers that didn't show understanding, some experienced incomprehension and not being taken seriously by their social environment/family/friends (others did experience being understood by their surroundings), some said they felt their social environment thought 'it's in your head' and 'chronic pain is not a visible problem like breaking a leg'. And in the present: 'central sensitization is not known by our social environment', and 'how do we explain this to our environment?'

Respondent 2, psychologist: 'Miss told me that she practiced the word 'sensitization' quite a lot at home (all laugh) and she noticed that in her environment it is a new thing'

Respondent 2: Yes, I hadn't heard of it before, nobody, nobody I talked to had heard of it before'

Respondent 4, partner: 'I'm really glad that it is not a hernia.. however, now he has something that is not visible .. how do you explain that? .. towards another?'

Reaction of the healthcare professionals towards concerns

The healthcare professionals dealt with the concerns first by asking the respondents about their concerns, so they would out them. Subsequently, healthcare professionals took great care in taking the respondents and their concerns seriously. Often, by these actions the concerns would then be gone, or less important.

Respondent 1, physiotherapist: 'You say your symptoms have increased, is that something you worry about? Say, what if it continues like this what then? Or isn't that on your mind?'

Respondent 1: 'Well that is something I have been thinking about. What if I have to live like this? It continues and gets worse? (looks sad). That's why I'm here'

If these concerns were driven by inadequate cognitions, they questioned these misconceptions and proceeded by (re)explaining the PNE in further detail. This cycle was repeated if necessary. By letting the respondents express their concerns and addressing these both the therapeutic alliance was reinforced as well as a deepened understanding of the respondents about their pain was achieved.

Having trouble accepting the complaints

Dealing with the complaints and the consequences of these complaints in daily life was hard for the respondents. All respondents expressed that they often felt frustrated, sad, angry, stressed, restricted in life and activities, and often mentioned having trouble accepting their pain and other complaints (fatigue, dizziness, etc.).

Respondent 4: 'Eh well, I would like to walk more, I would pretty much like to walk more, however my feet hurt so much when I do so'

The complaints also had consequences which they found hard to deal with, such as a loss of income, financial problems, struggles with bosses and being less involved in life. Furthermore, respondents often mentioned having trouble accepting the complaints they perceived. In this they expressed a desire to experience less complaints, be able to do more, hope for an easy solution and hope for less pain.

Respondent 6: '.. Yeah, that's what I said earlier, even if I have to be crippled for the rest of my life, I don't care, as long as I get rid of the pain'

Having trouble accepting the complaints often let to internal struggles. For instance, between wanting to do more and then perceiving pain, or having the household done in a fast pace and feeling less energetic, having to accept help from others, knowing that in order to be able to deal better with the pain certain internal changes had to be made.

There were, however, also some sings from respondents that they did accept the pain and had found ways to deal with the complaints. For instance, by mentioning that they tried to live mindful, expressed that accepting the complaints is a stepwise process they had to go through, had ideas on what could help them deal with their complaints, or because they already made some changes and had a positive reaction on those changes.

Doubt and resistance towards the PNE and changing

There was no obvious resistance to be seen against the healthcare professionals. Based on these video observations no major ruptures in the alliance was observed, nor increase in doubt and hesitation towards the healthcare professionals. If respondents had doubt and resistance towards the message of the PNE it often diminished throughout the PNE session. Meaning that respondents hadn't already changed their mind, then most often during the PNE session they went from searching for a simple biomedical solution towards a biopsychosocial understanding of their complaints. Apart from respondent one, who found it difficult to comprehend the message that there was nothing biomedically wrong. Doubt and resistance towards change was harder to address and as can be read in table 5, often persisted in those who experienced it. This form of hesitation linked to the category of *'changeability of the respondent'*.

Based on these video observations it was not possible to tell which actions from the healthcare professionals might have led to a decrease in doubt and hesitations of the respondents. Or if other actions would have led to other outcomes. Based on

these video observations it is only partly possible to tell whether the amount of doubt and resistance had influence on the final outcomes of the PNE session, as we could only observe the doubt and resistance towards the during this session and not what happened during the subsequent treatment. Therefore, it was not possible to tell which doubts and hesitations might have risen after this session and brought back to the healthcare professionals in the next treatment session. Furthermore, as PNE is at the beginning of the treatment and not a stand-alone treatment modality it is unsure which other treatments influence these doubts and hesitations later on.

Doubts and hesitation from the respondents

Doubts and hesitation could be seen in multiple manners. Most often by small concealed resistance, for instance in the form of: 'Yes, but....', 'I get it, however'. By increasing the pitch and loudness of their voice, or by non-verbal manners: swallowing words, getting a bit restless, sitting with a closed attitude and by sitting back and creating more space between them and the healthcare professionals.

As can be seen in table 5 and 6 some of the respondents started with some hesitation towards message of PNE from the previous session. Respondent two, for instance, did agree with her pain being an alarm signal going off without a burglar present, however had some resistance as to whether she caused the alarm going off. Once it was neurophysiological explained that she did not the hesitation was gone. Respondent three had some resistance because there was no biomedical cause, which, once taken seriously and explained was gone. Respondent four kind of understood the previous session, however found it 'a bit wooly', didn't really grasp it and once it was explained that there is something neurophysiological changed had less resistance towards the message. However, her resistance against changing behaviours in order to deal with pain stayed. Respondent six had some resistance disappeared. Respondent eight had some hesitation, 'what if there is something else going on?' and switched in this hesitation throughout the PNE.

During the PNE session, all respondents who first had some doubt, hesitation or resistance about the PNE diminished in this as the healthcare professionals explained the part they were doubtful of. Besides respondent eight, who alternated between hesitation and agreeing with the PNE throughout the session. Doubts could come from: finding the first explanation a bit wooly; disappointment that there was no biomedical

cause; doubts about the diagnosis: isn't there something damaged/a biomedical explanation; hope for an easy solution; I'm not the cause of this (the alarm going off) right?; previous operations being iatrogenic; why do I feel this now?.

Doubts and hesitation could also be related to the contributing factors. As can be seen in table 5 and 6, some found facing their biopsychosocial contributing factors difficult. There could also be hesitation or resistance about how the healthcare professionals would frame or use certain words and explanations: stress, life being heavy: 'yes, but I don't experience life as being heavy, to me it (feeling distant to herself in conversations) is strange', your back is strong and not crooked, there is nothing wrong/damaged.

Respondent 5, physiotherapist: 'It is also about how you deal with your own body, ehm, stress, tension, right?

Partner respondent 5: 'He is not a stressed person at all. He is, he can be super relaxed about almost everything. He is not somebody who worries easily'

Or hesitation surrounding the cause of the complaints being traumatic (fall from the stairs), being worried about their complaints (dizziness, fatigue, pain, etc) and/or how they dealt with their complaints (avoidance, finding it hard to accept their pain, keeping on searching for a cause of the pain, the importance of regulating emotions). Last, there was also resistance associated with behavioural change necessary to deal with pain in the long term in some respondents (especially in respondents 3 and 8, in smaller manner in respondent 6). Either in behavioural change that already happened, however did not diminish the complaints, or even though they understood that behavioural change was important yet finding it hard to change. For instance, respondent three did agree with CS, the PNE and contributing factors, however found it hard to change her behaviour in the future.

Doubt and hesitations were linked to the 'influence of changeability of the respondents'.

Reaction of the healthcare professionals towards doubts and hesitation

The healthcare professionals would react in different manners towards doubt and hesitation. First, by asking them how the previous session of PNE with the GP went, they addressed doubts and hesitation surrounding this session in the beginning of the current PNE.

Respondent 6, physiotherapist:' Could she (doctor) explain what she was thinking about as cause for your complaints, did she tell you that?'

Respondent 6: 'Euh, well, no. I told her in my previous rehabilitation treatment they also said that euh, I had to learn to deal with the pain. They already told me that I am scared to move. Well, I said: that's not completely correct, because if that would be true, then I would have used the crutches all the time and wouldn't have used my right foot.' Physiotherapist: 'Yes, so you don't recognize yourself in that description?' Respondent: 'No, because I walk about 20 minutes every day, preferably without crutches.. because I feel like, you need to get some movement otherwise you get stiff' Physiotherapist: 'Yes, that is really good of you to do that'.. continues about not being afraid to move ..

Physiotherapist: 'You feel like doesn't cover it (psychologist: No) (Respondent: No, no [shakes head]). I get that.'

Psychologist: 'It's really just a part of the story actually, a part of the bigger picture, so it is important that we explain the bigger picture to you, so this part can fall into place, but we have to talk about it together, right?' Respondent: 'Yes' [nods, non-verbal no resistance]

Most and foremost they would acknowledge the respondent, take them seriously, acknowledge that living with these complaints is hard, it is not something someone chooses, that these contributing factors are not thoughts/emotions/behaviours/ experiences someone has or does on purpose, acknowledging what is important for the respondents and that it is not easy to change. Sometimes by listening and giving the respondents time to express themselves the resistance diminished in itself. If the respondents didn't express these doubts and hesitations but they were noticeably there (because the nonverbal communication would show a closed posture, frowning, retreating in the chair), the healthcare professionals would ask or mention about the resistance they saw themselves. When they were out in the open, they would then talk these over with the respondents. Explaining the goals of the session also diminished some doubts and hesitations. A part of the PNE was to explain origin of the complaints the respondents experienced, which also dealt with doubt and resistance. For instance, why it was not an inflammation, why it was only one spot, why they believed it was CS pain, why it got worse, why sitting was painful, why physical tension and moving stiff increases pain, and why they believed the respondent was feeling tired. Sometimes when a word or sentence, such a 'stress', would cause resistance they would reframe it and use other words. The same accounted for the examples they used, they would

use examples and check by verbal and non-verbal signs that the respondents could understand them, if not, they would use different examples.

Respondent 1, physiotherapist: '.... There are things that can enhance the signal, for instance stress. Such a conversation is some sort of stress' Respondent 1: 'Well not for me, I am not easily stressed'.. Psychologist: 'Effort, maybe we should call it effort' Respondent 1: 'Yeah that I realize'

Every now and then they cleared the air by using (appropriate) jokes. They also used positive reinforcement of the changes the respondents already made to deal with resistance. Furthermore, they would try to bring insight to the respondent by asking in depth questions, having silences, summarizations, reflective questions and sometimes more confronting questions, and giving the respondents then time to express themselves. Or they would try to let the respondent look at her life from a distance, as if it was someone else's' life, to recognize that the way the respondent was dealing with things only made it harder. Giving hope for the future was also a way to help respondents deal with the hardships surrounding the complaints, contributing factors and having to change.

Respondent 8: '.. I used to walk so much, where, where is that all that investment? .. I'm so, I find that frustrating.' Physiotherapist: 'Well maybe that's still somewhere?' Respondent 8: '[frustration laugh] Yes I hope so, because if it all just gets worse, well then I feel like there's nothing left. [shakes head] That's when I get those thoughts: this is never going to be over and just let it go..' Physiotherapist: 'So yes, we should break that cycle'

Giving them hope that by changing their behaviour, often not extremely but in some amount, they could regain some perspective for the future.

Category 4: Outcome of the PNE process

Two sub-categories emerged in relation to the outcome of the PNE process: "Process of insight into CS and factors surrounding CS" and "stages of change of the respondent" (table 5).

Process of insight into CS and factors surrounding CS

The process of insight into CS involves two components: the first is gaining knowledge, the second is applying this knowledge to oneself and possible others. This process is

interactive with the previous sub-categories 'concerns about the complaints', 'doubt and hesitation', and the reaction of the healthcare professionals towards this, 'having trouble accepting the complaints' and the category 'influence of changeability of the respondents'.

Respondent 1, psychologist: 'And how can that (conversation) increase your complaints?' Respondent 1: 'Yeah, well, it's probably because I turn my head a lot, I think' Psychologist: 'That is the explanation?' Respondent 1: 'Yes, I think so' Physiotherapist: 'But in a such a conversation your facing the other right? So, then it isn't the head (turning)' Respondent 1: 'Yes, that's true'

In some respondents it was more difficult to align the respondents' beliefs with the healthcare professionals views regarding the underlying pain mechanisms and contributing factors. In respondent five it was harder to notice as he had minimal nonverbal expressions, however, he and all other respondents showed signs of trying to convey the message and understand what is being said. These expressions could be noticed by showing signs of actively and seriously listening and thinking.

As is mentioned in table 5 and 6, all respondents gained different levels of knowledge about CS, the surrounding factors and applied these differently to themselves. Most of them would have some doubts, biomedical ideas or uncertainties at the beginning of the session and steadily progressed towards an increased knowledge, understanding of CS. As well as in applying it to themselves. Concerning the factors surrounding CS, by the explanation most understood why and how these factors influenced their pain experience. However, as previously explained some found it hard to deal with changing their behaviour/themselves in order to cope with these factors.

Signs of increased awareness, knowledge and understanding would for instance be: explaining acute and CS pain in his/her own terms, asking questions about the explanations/metaphors, acknowledging his/her own contributing factors, agreeing with/acknowledging that the explanation they are receiving is 'their pain', coming up with treatment options or agreeing with the proposed treatment options, seeing changes for the future, coming up with own examples:

Respondent 5, partner: 'So then, everything you feel is in stereo?' Respondent 6, session 2: 'So you have to work here (points to head, smiles) as it were, to decrease the pain there (points to hip)' Respondent 7: 'Yes, absolutely, that's what I mean by focus. That is, if I'm, if it (the pain) is there, then, then, yes, I increase it by frustrating myself over it'

Category 5: Communication skills

Several different communication skills were used. Whereby every healthcare professional had his/her own preferences. As an example, some of the psychologists' used more 'reflective summarizing', whereas the physiotherapist videotaped in the PNE session of respondent 7 used a lot of silences. Furthermore, the communication style was modified in a patient centered manner to the respondent in front of them. For instance, in persons with less formal education less difficult language was used.

The healthcare professionals used the following communication techniques:

Encouraging

Using words and expressions such as: 'hmhm', 'yes', 'exactly', 'no', 'that's correct', 'precisely', 'yes I agree with that', 'ha', 'good', as well as nodding and putting question marks at the end of these short expressions showed understanding and encouraged respondents to talk.

Verbalizing non-verbal communication

Non-verbal communication, often sighs, were checked or interpreted.

Respondent 3: 'And also over time, that it just takes so long, (PS: Yes) that's why I got so emotionally flattened. (PS: Yes, yes.) I'm just .. [looks down to the right, sighs, mouth corners down, eyes hanging lightly]' Psychologist: 'Sigh, downhearted'.

Parroting the respondent

Parroting what the respondent just said in the wording of the respondent was used to encourage the respondent to keep on talking/explaining.

Positive reinforcements

Positive reinforcements of gained insight, changed knowledge and behavioural change that respondents have in mind or already happened was used to increase the alliance.

Respondent 7, physiotherapist: 'And that is important, that you (see something has to change). That's why we try to map it all out, even though you already know a lot ... and you already know a big deal (about her triggers)'

Silences

Silence was used to give the respondents time to think, reflect, and express themselves.

Asking (interactive) questions about cognitions, knowledge, understanding, emotions and behaviour

One of the reasons these questions were asked was to verify and increase the understanding and knowledge of the respondents with the previous and current PNE. They often checked whether the given information changed the knowledge and understanding of the respondents about their pain.

Respondent 2, physiotherapist: 'It is actually a too strong of a response from the nervous system. What do you think? Do you think, well I understand that? Or not at all?'

By asking these questions, they became more aware of the knowledge the respondents gained, but also how they incorporated this knowledge and whether there was resistance or doubt and hesitation about the explanation. They also often used this technique to question faulty cognitions such as 'my vertebrae is out of line', 'my SI joint is crooked' or the cognition that 'there is a bone fragment in my brain'. Furthermore, they asked questions about the emotions and linked behaviours of the respondents. Asking them and their partners about their lived experience with pain, so they felt understood as well as letting them gain insight in how these emotions and behaviours can increase pain. For instance, about how the pain affected them, irritations, fear, the ability to feel relaxed, emotions surrounding reintegration of work and how they handled their pain. At the end of the PNE the questions were more directed to how they could help the respondents deal with their pain. One important aspect they questioned was the emotions surrounding the PNE explanation and further treatment. Besides getting to know hesitation by asking for the knowledge of the respondents they often asked questions surrounding doubt and hesitation:

Respondent 6, psychologist: 'How did the explanation of the doctor seem to you?'

One of the most used communication strategies was asking clarifying questions. Either open- or closed ended, with a non-judgmental character. For instance:

Psychologist: 'So it is not as if you do a little less today then you would feel better tomorrow?' Respondent 6: 'Well really, yes [laughs]. It makes me feel like: hé, I feel a little better-' Psychologist: 'And then, what are you going to do?' Respondent 6: 'Well then, I will do some "heavier" work. ' Psychologist: 'Exactly' Respondent 6: 'Well, and then I have to pay for it the next day.'

There was a slight difference in asking questions and interactive questions and comments, because interactive questions and comments requires more of an action in the form of thinking and response of the respondent. Furthermore, interactive questions were often asked to involve the respondent into the neurophysiology of the PNE. For instance:

Respondent 7, physiotherapist: 'Are there other topics (contributing factors) we didn't mention, but you think hé that is important?'

Summarizing

Summarizing was used by the healthcare professionals to build rapport, check whether they understood the respondent correctly and take the next step in the PNE.

Reflective summarizations

Reflective summarizations are more influential than normal summarizations. Besides giving the respondent a feeling of being understood, they aim to give insight to the respondent, are more metaphorically and often contain emotions. For instance:

Respondent 7, physiotherapist: 'Again, it is about putting the bar high, right? (R: yes). And constantly over asking yourself .. family, work, nightshifts, irregularity. it's quite a lot'

Respondent 4, psychologist: 'For you beauty is very important, right?, insight and out. And you find it difficult to accept, let's say, to accede that not all is beauty, that its' not all perfect and beautiful. How beautiful you show it to be on the outside, yet on the inside you are in pain. That doesn't match at all, they have no idea how you feel in the inside, how much effort it costs you.'

Reflective and provocative questions and comments

The healthcare professionals often used reflective and provocative questions and comments. These are shorter than the summaries and invite the respondent to reflection and can be somewhat provocative. They were often used to let respondents gain insight in their behaviour, cognitions, emotions and how this negatively impacts their pain. These questions and comments were often accompanied by 'soothing words', such as: *'it is not criticism', 'we understand why you would think/do/feel that way'*, in order to not disrupt the rapport.

Some examples:

Respondent 6, physiotherapist: 'You recognize the struggle?'

- Respondent 6, psychologist: 'So you are often under time pressure?'
- *Respondent 3, psychologist: 'It quite heavy, your load ... so you are continuously balancing'*

Respondent 5, psychologist: 'Well, it's not going that well because the gas bill is not being paid!'

Respondent 8, psychologist: 'Yes, so you start to question yourself a little more do you mean?'

Category 6: Interplay between the healthcare professionals

Throughout the PNE session there was an interplay between the healthcare professionals and sometimes afterwards they reflected on the PNE together. For instance, the physiotherapist would ask the psychologist if he/she missed anything. Or one of the two would confirm the others explanation, deepen the explanation or check the cognitions and emotions of the respondent about that explanation. This interplay was also characterized by the synchronicity and fluidity between the healthcare professionals in the conversations. Often the physiotherapist would explain the neurophysiology and the psychologist would have an observing role and check how the PNE landed.

Category 7: Influence of the changeability of the respondents

As can be seen in figure 2 and 3, throughout the PNE a change process is started. The changeability, or psychological/cognitive flexibility, of the respondent as seen in these videos during this process encompasses a few steps:

- the ability to have interpersonal contact with the healthcare professionals and be open;
- the manner in which the respondent was able to understand and embody the knowledge of the PNE and change his/her cognitions about the complaints they experienced;
- the self-reflectivity about his/her contributing factors, such as emotions, cognitions, stress and behaviour;
- the ability to stop fighting against his/her complaints and their influence on his/ her daily life, and in some way accept these complaints;
- 5) the ability to modify or alter his/her behaviour towards behaviour that would be more helpful or accept help regarding this

Whilst not all respondents were able to progress through all steps, all of the videos mapped onto the steps. For instance: respondent five had little interpersonal contact with the healthcare professionals (however did partake in step 4 and 5). Or that most understood the neurophysiology, however some found it hard to embody the knowledge. Furthermore, the depth of this comprehension remains unknown as that can only be seen in future sessions. Some respondents found it difficult to be reflective about their contributing factors, while others found it hard to accept their complaints and stop fighting. Last, some found it hard to envision how to, and/or to modify their behaviour in the future.

Most often, those respondents who found it hard to modify their behaviour, respectively respondent 3, 6 and 8, somewhere already knew that in order to deal with their complaints they had to change, however found it really difficult.

Respondent 3: 'yes, yeah.. it is about going on and on.. I'm a hard worker so that is what I do. And I know that I have more complaints afterwards. It is just... How can I change that? I'm like: yes, on one hand I want to change because I want to get rid of the pain, however, on the other hand I want to live my life like I do now'

From these video's the changeability of a respondent was seen as something that corresponds with health knowledge rather than intelligence, as both respondents with a respectively expected higher or lower intelligence (based on education) could be proactive/cooperative or be resistant towards change.

Moderators of change in self-reported signs of central sensitization following an individualized transdisciplinary pain management program

6

The moderating effects of pain catastrophizing on the relationship between changes in illness perceptions and self-reported sings of central sensitization in patients with chronic pain

Amarins J. Wijma^{1,2}, Rinske Bults^{1,2,3}, Jo Nijs¹, C.Paul van Wilgen^{1,2}

Affiliations

¹Pain in Motion research group (www.paininmotion.be), Department of Physiotherapy, Human Physiology and Anatomy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium

²*Transcare Transdisciplinary pain management center, Groningen, the Netherlands* ³*Department of Rehabilitation Sciences and Physiotherapy, Faculty of Physical Education and Physiotherapy, Vrije Universiteit Brussel, Brussels, Belgium*

Status: Modified from the following publication: International Journal of Rehabilitation Research, 43(4):347-354, Dec 2020

Abstract

Purpose

Two factors related to the continuation of chronic pain are pain catastrophizing and illness perceptions. Pain neuroscience education is known to positively influence both in patients with chronic pain. As the integration of pain neuroscience education monodisciplinary physiotherapy treatments is effective, integrating in transdisciplinary cognitive behavioural treatments seems recommendable. When doing so, the moderating effect of pain catastrophizing and perceptions on treatment results has to be examined, as these provide valuable information under what conditions treatment works.

Methods

A bottom up retrospective observational study evaluated the changes in clinical outcomes, and relationships between clinical outcomes and cognitive and emotional factors in patients with chronic pain. Multiple regression analysis, PROCESS macro, explored the moderating effects of pain catastrophizing on the relationship between illness perceptions and self-reported symptoms of central sensitization.

Results

In total 78 patients were included in the study. A correlation between pretreatment scores and change-scores in illness perceptions and self-reported symptoms of central sensitization following treatment were found (resp. *R-sq* .407, *F*(10,99) = .638, *p*=.000, *R-sq* .361, *F*(5, 54) = .609, *p*=.000 and *R-sq* .314, *F*(4,55) = .560, *p*=.00). However, moderation of pain catastrophizing scores on these correlations were not found.

Conclusions

Even though a correlation between changes in pain catastrophizing and illness perceptions in patients with chronic pain was found, the direction or strength between the changes in illness perceptions and changes self-reported symptoms of central sensitization was not influenced by pretreatment scores of pain catastrophizing.

Introduction

'Pain is a mutually recognizable somatic experience that reflects a person's apprehension of threat to their bodily or existential integrity'¹. This new definition of the International Association for the study of Pain (IASP) emphasizes the complex nature of pain: pain and the intensity of pain is a subjective experience of threat, rather than an objective measure of tissue damage. Pain that perseveres beyond the normal time for tissue healing and/or longer than 3-6 months is defined as chronic pain².-

Two factors that negatively influence chronic pain are pain catastrophizing and maladaptive illness perceptions³⁻⁷. Pain catastrophizing is characterized by cognitive and emotional processes surrounding irrational negative thoughts, magnified threat value of pain and experienced helplessness in the context of pain⁸. These three factors have been operationalized as magnification (*'I wonder whether something serious might happen'*), rumination (*'I keep thinking about how much it hurts'*) and helplessness (*'I feel I can't stand it anymore'*)^{8,9}. Pain catastrophizing is associated with, or predicts the degree of pain and disability¹⁰⁻¹³ and delayed recovery¹² in various musculoskeletal pain problems. In addition, pain catastrophizing has been linked to activation of brain areas involved in pain processing, pain perception, modulation, attention to pain, and affective components of pain, leading to reduced top-down inhibition^{8,13,14}.

Important in any health behaviour are the beliefs a person holds on their 'illness or condition'¹⁵. These illness perceptions will influence the perceived threat, used coping strategies and outcomes of a patients' illness¹⁵. In patients with chronic pain, illness perceptions account for a large amount of depressive and anxiety symptoms¹⁶⁻¹⁸. As a consequence of maladaptive illness perceptions, the cognitive and emotional representation of these illness perceptions represent threat and fear. Which then amplify the neural signaling, thereby intensifying the experienced pain¹⁹.

One way to deal with these maladaptive illness perceptions and pain catastrophizing is by educating patients. Also known as pain neuroscience education (PNE), the neurophysiological explanation of pain is well studied and has a high level of evidence when combined with other treatment modalities in the treatment of patients with chronic pain²⁰⁻²³. A recent review has shown that PNE produces clinically significant reductions in pain catastrophizing in patients with chronic pain.²². Furthermore, PNE improves most patients' knowledge on pain and reconceptualizes illness perceptions
about their chronic pain^{24,25}. However, PNE should be considered as a steppingstone and should be integrated in further treatment²².

The effect of pain catastrophizing on treatment outcomes is increasingly studied. This effect can be studied from multiple perspectives: In a predictive manner: high pain catastrophizing scores are associated with more pain and disability at follow up in patients with low back pain¹⁰. From a mediator perspective: a decrease in pain catastrophizing is associated with better outcomes in daily activity, pain control, disability and pain in patients with low back pain¹⁰. Furthermore, evidence from cognitive behavioural therapy (CBT) for patients with chronic pain, including addressing maladaptive cognitions, shows that decreased pain catastrophizing is, amongst other things, linked to decreased activity in the dynamic pain connectome²⁶⁻²⁸. These findings suggest that CBT-induced brain changes relate to a reduction of CS in patients with chronic pain.

From a moderator perspective: higher pain catastrophizing scores significantly moderate the association between pain and secondary hyperalgesia in healthy indiviuals²⁹. Furthermore, PNE combined with cognition-targeted exercise therapy is more effective in decreasing self-reported symptoms of CS in patients with high levels of pain catastrophizing³⁰. Both show the connection between these cognitive-emotional factors and symptoms of CS. However, a review on treatment of low back pain concluded that the assessment of moderating effects was inconclusive¹⁰.

Moderator analyses provide valuable information on for whom and under what conditions treatment works³¹. Therefore, it is suggested to further investigate the role of pain catastrophizing as a moderator for the outcomes of treatments of patients with chronic pain¹⁰. Understanding for whom and how changes in outcomes following transdisciplinary CBT treatment, including PNE, might achieve its effects is of importance for both current practice and future science.

Aim

The aim of this observational study is to explore whether the relationship between changes in illness perceptions and changes in self-reported symptoms of CS following transdisciplinary CBT treatment, among which PNE, is moderated by pain catastrophizing in patients with chronic pain.

The following hypotheses were stated prior to the analysis:

- Pretreatment levels of pain catastrophizing moderate the relationship between pretreatment levels of illness perceptions and pretreatment levels of self-reported signs of central sensitization in patients with chronic pain;
- Pretreatment levels of pain catastrophizing moderate the relationship between changes in illness perceptions and changes in self-reported signs of central sensitization in patients with chronic pain undergoing transdisciplinary CBT treatment;
- Changed of levels of pain catastrophizing moderate the relationship between changes in illness perceptions and changes in self-reported signs of central sensitization in patients with chronic pain undergoing transdisciplinary CBT treatment.

Methods

A bottom up retrospective observational study evaluated the changes in clinical outcomes, relationships between clinical outcomes and illness perceptions and pain catastrophizing, and moderating effect of pain catastrophizing scores following treatment.

Patients

Patient data were gathered and retrospectively analyzed from an individual outpatient CBT treatment program at Transcare pain management center. Prior to the intake and post treatment, patients filled out a battery of questionnaires which is part of standard care. Data were gathered from 112 patients pre-treatment and 78 post-treatment. 34 patients dropped out either because they did not continue treatment, did not want to complete the post treatment questionnaires or because questionnaires were not submitted.

Patients were eligible to participate if they: 1) were between 18 and 85 years of age, 2) were suffering from nonspecific chronic pain as defined by the International Association for the Study of Pain (IASP)², and 3) were sufficiently able to read, speak and understand Dutch. Patients were excluded from this study when they: 1) were diagnosed with a specific medical condition (e.g., Parkinson's disease, rheumatoid arthritis, stroke), 2) had cognitive impairments, 3) had dementia, 4) or had a serious psychiatric condition as

identified through the Symptom Checklist 90 (SCL-90)³². All patients signed an informed consent form on which they agreed to the collection of their data. Confidentiality was guaranteed and all materials were handled anonymously. This study was conducted in compliance with the principles of the Declaration of Helsinki. Based on Dutch laws, ethics committee approval was considered unnecessary due to the nature of this study.

Intervention

All patients followed a transdisciplinary CBT treatment, as described in a previous study²⁵. This treatment starts with a three-hour assessment – one hour with a medical doctor, a psychologist and physiotherapist. After this extensive intake, the healthcare professionals discuss the patient analysis containing the pre-dominant pain mechanism and biopsychosocial contributing factors. About a week later the medical doctor explains the pre-dominant pain mechanism and contributing factors to the patient, the first PNE session, in about 20-30 minutes. Often the patient is accompanied by their partner/spouse. After this session all patients receive written PNE information. In a second session, about a week later, the psychologist and physiotherapist elaborate in a second PNE session. During this session treatment goals are discussed based on shared decision making. The intervention consists of a patient centered CBT. Meaning that components of CBT, the content and duration of the therapy is adjusted to the patient and his/her needs. The CBT, without assessment, consisted of on average four psychology sessions and five to six physiotherapy sessions. The CBT, without assessment, consisted of on average four psychology sessions and five to six physiotherapy sessions.

Measures

Sociodemographic data Sociodemographic data were collected with a standardized questionnaire on age, gender, educational level, relationship status, profession and employment status.

Pain characteristics and intensity

The following characteristics were gathered: duration, intensity and location. Patients were asked to rate their average pain on a Numeric Rating Scale (NRS), ranging from 1 (no pain) to 10 (severe pain). The psychometric properties of the NRS are good^{33,34}.

The number of painful body regions was measured with the Widespread Pain Index (WPI). Which is a well-known measure for the distribution of pain. The score from none (0) to the complete body, including the head (20). In this study patients described where the pain is located on an illustration of the human body³⁵. The WPI was then summed up and analyzed as a categorical variable. The WPI has good psychometric properties in juveniles with pain³⁶.

Central sensitization

Symptoms of central sensitization were measured using the Dutch version of the Central Sensitization Inventory (CSI). The CSI appears to be a valid, reliable, usable, and diagnostically relevant questionnaire assessing common symptoms and facilitating factors to CS³⁷⁻⁴⁰. A cutoff score of 40 points indicates the symptoms are due to predominant CS pain^{41,42}. Only total CS scores should be used⁴³. Furthermore, scores on the CSI merely reflect general distress rather than a direct measure of CS⁴⁴.

Pain catastrophizing

Pain catastrophizing was measured using the Dutch version of the Pain Catastrophizing Scale (PCS). The PCS is a self-reported questionnaire with 13 items on a 0 (not at all) to 4 (all the time) scale, n which patients are asked to report what they feel and think when they are in pain. The three subscales consist of rumination, magnification and helplessness. Higher scores relate to more pain catastrophizing, with a cut off score of 24⁴⁵ and a score of 21.6 serves as the reference value from the Dutch Dataset Pain Rehabilitation⁴⁶. The PCS has adequate to excellent internal consistency, good concurrent validity and good responsiveness^{47,48}.

Illness perceptions

Illness perceptions were measured with the Dutch version of the Brief Illness Perception Questionnaire (IPQ-B). The IPQ-B is developed to assess the cognitive representations of illnesses⁴⁹ based on the Common Sense Model of Self-regulation by Leventhal. The first eight items, as used in this study, are scored on a 0 to 10 scale and represent five scales that consist of identity, cause, timeline, consequences and cure-control. Retest reliability and internal consistency is good, except for the causal item. It has a good construct validity, sound discriminant and predictive validity, but the factor structure is unstable in patients with musculoskeletal pain⁵⁰. In the current study a total score of the IPQ-B was used. An overall score can be computed by reversing items 3, 4, and

7 and then summing all items. A higher score reflects a more threatening view of the illness.

Disability

Disability was measured using the Dutch version of the Disability Rating Index (DRI). The DRI is 12-item questionnaire measuring limitations in daily physical functioning in patients with pain, scored on a visual analogue scale from 0 to 10 (total score 0-120). It's internal consistency is good, with good reliability, good content validity, reasonable construct validity and good responsivity^{34,51,52}.

Health related quality of life

The Dutch version of the RAND-36 was used to measure health related quality of life (HrQoL). A higher score indicates a perceived better HrQoL. The generic questionnaire RAND-36 is almost identical to the Short-Form-36 (SF-36). Because calculating the total score of the RAND-36 for HrQoL is under discussion only the subscales were used⁵³. These consist of scales for physical functioning, social functioning, and limitations due to physical and emotional problems, mental health, energy, pain and overall health perception . The RAND-36 has a high convergent validity, however low discriminant validity in an overall Dutch population⁵⁴.In three chronic diseases reliability showed strong unidimensional hierarchical scales⁵⁵.

Data analysis

For all statistical analyses SPSS version 24 statistical software was used (IBM SPSS Statistics for Mac, version 24, IBM Corp., Armonk, N.Y., USA).

Demographic variables and outcome measures were described using descriptive statistics.

Educational levels were classified according to the International Classification Education (ISCED) system, 2011. Due to the small amount (N=3) of post-treatment WPI results, the WPI was only used for baseline.

Normality was checked with Kolmogorov-Smirnov tests. Because the data were not normally distributed non-parametric tests were used. To assess pre- and post-treatment changes Wilcoxon's Signed Rank Tests were performed. For all analyses *p*< 0.05 was considered significant. Complete case analysis was performed.

Based on an a-priori sample size calculation, calculated using G^*Power^{56} for multiple linear regression analysis, random model, based on a 0.05 alpha error probability, and a proposed tentative effect size for moderation analysis set on f2=0,15, with a (1-ß) power of 0,80 and number of tested moderators 3, a sample size of 69 deemed sufficient.

Prior to the moderator analysis data were checked for, and the assumption of homoscedasticy was met. Residuals were normally distributed. Associations were tested using Spearman Rank Correlations. Variance inflation factor was used to check for multicollinearity, which was not found.

The PROCESS-macro v. 3.4 was used for moderation analysis (model 1)⁵⁷. All variables were centered to the mean and the Johnson-Neyman-technique was used. Only complete data were taken into account.

Key co-variates were pretreatment and changes in pain intensity, age, sex, education, employment status, widespread pain index and duration of the complaints. As only the widespread pain index and changes in pain intensity were significant contributors in the primary model they were kept.

Results

In total 78 patients with chronic pain were included in the study. Sociodemographic characteristics are described in table 1. The majority of the sample was female (51.3%) with a mean age of 48.75 (SD=15.61). The duration of complaints was on average 91.91 months (SD=113.18), with an average pain intensity of 6.61 (SD=1.85) and 5.66 (SD=4.27) pain reported sites. The average score of self-reported symptoms of central sensitization was 33.93 (SD=15.34), which is below the cut off score of 40; 30.1% of the patients scored above the cut off score (\geq 40) on the CSI⁴¹. The average score on the pain catastrophizing (PCS) was 19.31 (SD=10.47), 44.53% scored above the Dutch reference value of 21.6⁴⁵.

Table 1. Sociodemographic and pain characteristics of the studied patients with chronic pain (n=78)

Characteristic		
Age (mean ± SD)	48.75 ± 15.61	
Gender (<i>m/f, %</i>)	38/40 (48.7/51.3)	
Relationship status (N,%)		
Single	16 (20.5)	
Married	40 (51.3)	
Cohabiting	13 (16.7)	
Divorced	6 (7.7)	
Widowed	3 (3.8)	
Education (N,%)		
Primary school or less	3 (3.8)	
Lower secondary education	26 (33.3)	
Higher secondary education	28 (35.9)	
Post-secondary non tertiary education	8 (10.3)	
Tertiary education bachelor	10 (12.8)	
Tertiary education master	2 (2.6)	
Missing	1 (1.3)	
Employment status (N,%)		
Employed	40 (51.3)	
Unemployed	8 (10.3)	
Studying	3 (3.8)	
Welfare or social assistance	21 (26.9)	
Retired	6 (7.7)	
Duration of complaints in months (mean ± SD)	91.91±113.18	
Pain intensity last week (mean ± SD)	6.61±1.85	
WPI (mean ± SD)	5.66 ± 4.27	

Educational levels were classified according to the International Classification Education (ISCED) system, 2011. Due to the small amount (N=3) of post-treatment WPI results, the WPI was only used for baseline. WPI: Widespread pain index

Pre-post treatment changes

On average pre- and posttreatment measures were taken 193 days (SD 110) apart from each other.

Pre- and posttreatment differences as analyzed with Wilxocon's Signed Rank Tests are depicted in table 2.

Results suggest significant changed levels of pain, NRS scores and in the RAND subscale pain, (both moderate effect size), in health (RAND subscale health change and vitality, both moderate effect size) and a significant decrease in CSI (small effect size). Besides, reported disabilities improved significantly (DRI), RAND subscales physical functioning, Role limitations physical functioning, social functioning (all small effect sizes) after treatment.

PCS decreased significantly after treatment by 28.18%, and after treatment 19.23% of the patients scored above the Dutch reference value⁴⁵.

The total score in Illness perceptions (IPQ-B) improved significantly. Illness perceptions about consequences, personal control, treatment control, identity, concerns, coherence and emotional representation significantly changed and became less negative.

There were no significant differences after treatment on the RAND subscales role limitations emotional problems, mental health, general health perception and item 2 of the IPQ-B on duration of pain.

Table 2. Outcomes of pre-post measures

Characteristic	N⁵	Pre (M±SD)*	Post (M±SD)*	Τ°	Z¥	р	r #
NRS	77	6.61±1.85	4.86±2.37	225.50	-5.50	0.000	-0.44
CSI	71	33.93±15.34	28.99±13.85	554.50	-3.91	0.000	-0.33
DRI	73	52.53±25.37	45.40±28.29	735	-2.83	0.005	-0.23
PCS							
Total	77	19.32±10.47	14.04±12.04	691.50	-3.63	0.000	-0.29
Rumination	77	7.51±4.18	5.59±4.65	642	-3.39	0.001	-0.27
Magnification	77	2.22±1.98	1.71±2.10	443	-2.08	0.037	-0.17
Helplessness	77	9.58±5.61	6.74±6.47	596	-3.66	0.000	-0.29
IPQ-B							
Total	73	47.97±9.85	40.12±15.51	430	-4.76	0.000	-0.39
Consequences	78	6.64±1.99	5.15±2.46	270	-4.99	0.000	-0.40
Timeline	77	7.24±2.07	6.68±2.94	584	-1.76	0.078	
Personal control	77	4.24±2.36	5.09±2.54	1412	-2.506	0.012	-0.20
Treatment control	75	3.99±1.89	5.19±2.85	1706	-2.46	0.014	-0.20
Identity	78	6.68±1,72	5.15±2.36	184	-5.32	0.000	-0.43
Illness concern	78	6.00±2,15	4.26±2.93	265	-5.22	0.000	-0.42
Coherence	77	5.31±2,38	6.04±2.71	1426	-2.06	0.040	-0.17
Emotional	77	6.06±2,83	4.82±2.76	383.50	-3.68	0.000	-0.30
representation							
RAND							
Physical functioning	78	57.37±23.25	64.23±25.09	1579.50	-3.33	0.001	-0.27
Social functioning	75	60.23±25.30	71.38±28.05	1414	-4.05	0.000	-0.33
Role limitations	77	26.92±35.98	50.65±42.53	1023	-4.13	0.000	-0.33
physical functioning							
Role limitations	77	70.13±41.03	76.07±38.33	379	-1.07	0.284	
emotional functioning							

Table 2. (Continued)

Characteristic	N§	Pre (M±SD)*	Post (M±SD)*	Τ°	Z¥	р	r#
Mental health	77	71.85±15.53	73.71±17.09	1495	-1.02	0.308	
Vitality	76	46.22±18.90	52.63±18.47	1282.50	-2.71	0.007	-0.22
Pain	75	39.36±18.84	58.51±20.46	1751.50	-6.18	0.000	-0.50
General health	76	58.27±20.38	61.12±21.61	1407	-1.69	0.092	-0.14
perception							
Health change	78	34.92±24.47	60.58±25,33	185.50	-5.31	0.000	-0.43

*All values are presented as mean \pm standard deviation two sided, as continuous non-normal distributed data. SN = number of patients, m T = sum of positive ranks, 4Z = Z-value, # r = effect size, \pm HQoL = Health related Quality of Life, CSI = Dutch version of the Central Sensitization Inventory, DRI= Dutch version of the Disability Rating Index, PCS = Dutch version of the Pain Catastrophizing Scale, RAND = Dutch version of the RAND36, a higher score reflects a better outcome, IPQ-B = Dutch version of the Illness Perception Questionnaire Brief

Correlations

As can be seen in Table 3, there were significant associations between the three baseline scores of pain catastrophizing, illness perceptions and symptoms of central sensitization (PCS and IPQ-B (r=.582, p=.000), PCS and CSI (r=.261, p=0.027), IPQ-B and CSI (r=.266, p=.0.023)).

Pretreatment scores and change-scores always strongly correlate and were therefore excluded from the analyses. There were no associations between pretreatment measures and the change-scores. Within change-scores there were significant associations between all measures (changes in PCS and changes in IPQ-B (r=.453, p=.000), changes PCS score and CSI score (r=.337, p=.005), changes IPQ-B and changes CSI score (r=.387, p=0.02)).

Table 3. Correlations at baseline and changes scores

		IPQ-B	CSI	CSI	PCS	IPQ-B
		baseline	baseline	changes	changes	changes
PCS	Correlation Coefficient	,582**	,261*	-,184		,050
baseline	Sig. (2-tailed)	,000	,027	,133		,676
	Ν	77	72	68		71
IPQ-B	Correlation Coefficient		,266*	-,078	-,019	
baseline	Sig. (2-tailed)		,023	,525	,872	
	Ν		73	69	76	
CSI	Correlation Coefficient				-,022	,022
baseline	Sig. (2-tailed)				,853	,856
	Ν				72	68
CSI	Correlation Coefficient				,337**	,387**
changes	Sig. (2-tailed)				,005	,002
	Ν				68	64
PCS	Correlation Coefficient					,453**
changes	Sig. (2-tailed)					,000
	Ν					70

As pretreatment and change scores of the same measures are always correlated these were not taken into account. * Correlation is significant at the 0.05 level (2-tailed), ** Correlation is significant at the 0.01 level (2-tailed), CSI = Dutch version of the Central Sensitization Inventory, PCS = Dutch version of the Pain Catastrophizing Scale, IPQ-B = Dutch version of the Illness Perception Questionnaire Brief

Moderation analysis

The first hypothesis, *i.e.* moderation of pretreatment levels of PCS on the relationship between pretreatment IPQ and pretreatment CSI scores (table 4) was not supported. There was a significant association between change in CSI and change in IPQ-B. The variance was significantly explained in the model, *R-sq* .407, *F*(10,99) = .638, p=.000. However, both pretreatment scores of PCS and IPQ-B were no predictors for pretreatment scores on CSI. The association between pretreatment IPQ-B and pretreatment PCS scores was not significant and did not add to the model, *R-sq* 0.009, *F*(1,64)=.962, *p*=0.33.

The second hypothesis, *i.e.* moderation of pretreatment levels of PCS on the relationship between changes in IPQ-B and changes in CSI (table 5) was not supported. There was a significant association between change in CSI and change in IPQ-B. The variance was significantly explained in the model, *R-sq* .361, *F*(5, 54) = .609, *p*=.000. However, pretreatment PCS and the change in IPQ-B were no predictors for change in CSI. Furthermore, the association between changes in IPQ-B score and pretreatment

PCS scores was not significant and did not add to the model, *R-sq* 0.002, F(1, 54)=.187, p=0.70.

The third hypothesis, *i.e.* moderation of changes in PCS on the relationship between changes in IPQ and changes in CSI (table 6) was not supported. There was a significant association between change in CSI and change in IPQ-B. The variance was significantly explained in the model, *R-sq* .314, F(4,55) = .560, p=.00. However, only change in IPQ-B was a predictor for change in CSI. Furthermore, the association between changes in IPQ-B score and changes in PCS scores was not significant and did not add to the model, *R-sq* 0.00, F(1, 55)=.0.07, p=0.79.

Table 4. Hypothesis 1: Linear model of predictors of pretreatment CSI, N=60

	b	SE B	Т	p
Constant	23.941 [18.088, 29.073]	2.569	9.318	.000
Baseline score IPQ-B (centered)	.055 [315, .425]	.185	.297	.766
PCS baseline score (centered)	.294 [048, .636]	.171	1.718	.091
Baseline score IPQ-B x PCS baseline score	015 [045, .015]	.015	981	.331
WPI	2.02 [1.328, 2.719]	.348	5.812	.000*

*significant at the level of p < 0.05 $\,$

 R^2 =.36, CSI = Dutch version of the Central Sensitization Inventory, PCS = Dutch version of the Pain Catastrophizing Scale, IPQ-B = Dutch version of the Illness Perception Questionnaire Brief, WPI= Dutch version of the Widespread Pain Index

Table 5. Hypothesis 2: Linear model of predictors of changes in CSI, N=60

	b	SE B	т	р
Constant	299 [-3.87, 3.268]	1.779	168	.867
Changes in IPQ-B (centered)	.133 [057, 0.323]	.095	1.398	.168
PCS baseline score (centered)	085 [268, .098]	.091	929	.357
Changes in IPQ-B x PCS baseline score	.003 [268, .098]	.008	.433	.667
WPI	636 [-1.104,171]	.232	-2,742	.008*
Changes in NRS score	-1.185 [-2.261,108]	.537	-2.206	0.03*

*significant at the level of p <0.05

 R^2 =.36, CSI = Dutch version of the Central Sensitization Inventory, PCS = Dutch version of the Pain Catastrophizing Scale, IPQ-B = Dutch version of the Illness Perception Questionnaire Brief, WPI= Dutch version of the Widespread Pain Index, NRS = Dutch version of the Numeric Rating Scale

Table 6. Hypothesis 3: Linear model of predictors of changes in CSI, N=60

	b	SE B	т	р
Constant	-2.076 [-5.344, 1.193]	11.631	1273	.208
Changes in IPQ-B (centered)	.199 [.006, 0.393]	.096	2.063	.044
Changes in PCS (centered)	152 [059, .343]	.101	1.411	.164
Changes in IPQ-B x Changes in PCS	001 [012, .09]	.005	-2.684	.791
WPI	634 [-1.107,161]	.236	0.009	.009*

*significant at the level of p <0.05 R^2 =.36, CSI = Dutch version of the Central Sensitization Inventory, PCS = Dutch version of the Pain Catastrophizing Scale, IPQ-B = Dutch version of the Illness Perception Questionnaire Brief, NRS = Dutch version of the Numeric Rating Scale

Discussion

This study retrospectively examined a sample of patients with chronic pain who received patient centered transdisciplinary CBT, containing PNE. The results show improvements of patients with chronic pain in self-reported pain severity, self-reported symptoms of central sensitization, disability and pain catastrophizing, as well as improved illness perceptions after receiving transdisciplinary CBT including PNE. There was an association between changes in pain catastrophizing and illness perceptions, however, moderation of baseline pain catastrophizing on the relationship between changes in illness perceptions and changes in self-reported symptoms of central sensitization was not found.

We found moderate associations between pretreatment scores of pain catastrophizing and illness perceptions, suggesting a bilateral relationship between catastrophic thinking and negative illness perceptions in patients with persistent pain. Additionally, within those measures reflecting change-scores there were moderate associations between all measures (pain catastrophizing, illness perceptions and self-reported symptoms of sensitization). These associations are in agreement with previous reviews, which have shown that PNE for patients with chronic pain improve illness perceptions, decrease pain catastrophizing, pain and disability in patients with chronic pain^{20,23,58}. As well as with results from CBT treatments, which show improved pain, disability and health-related quality of life and decreased depression in patients with chronic pain⁵⁹.

Furthermore, decrease in the self-reported symptoms of central sensitization following CBT treatment is in line with current research on cognitive-emotional sensitization¹⁹. It is

known that CBT decreases pain catastrophizing, which is linked to positive neuroplastic alterations in the dynamic pain connectome²⁶⁻²⁸. This effect is most present in the connectivity between the anterior default mode network and the amygdala, the 'threat-center' of the brain²⁸.

The results indicate that even though there is an association between changes in illness perceptions and self-reported symptoms of central sensitization in patients with chronic pain, the direction nor strength between of the relationship between these two was influenced by pain catastrophizing. Low pain catastrophizing scores at baseline (19.55, 44.54% above the Dutch reference value) and small changes (14.04, 19.23% above the Dutch reference value) might explain non-moderating findings¹⁰. Reason for these low initial PCS scores, and subsequently lesser changes, could be the communal aspects of pain catastrophizing⁶⁰. Higher pain catastrophizing scores are associated with more submissive interpersonal interactions and high levels of dependency⁶¹. Geographical interpersonal culture has an influence on how we experience and deal with pain. The Netherlands, where the current study took place (e.g. the province of Friesland), is known for its interpersonal culture surrounding being independent and 'acting strong and sturdy'. This could have played a role in the low scores on pain catastrophizing.

Several limitations of this study can be distinguished. First, as the design was only a pre-post treatment design instead of a randomized clinical trial design, no assumptions can be made regarding the effectiveness of the treatment, nor that the content of the intervention influenced illness perceptions and pain catastrophizing. Besides, as this intervention was a patient-tailored approach some patients only received PNE while others also received CBT containing both physiotherapy and psychology. Therefore, not only the intervention differed, but the measurement points (T0 and T1) also differed between patients. As the current design of this study introduces the possibility of selection bias and limits the generalizability of the data we suggest viewing the current as a scoping study. Additionally, there was a limited amount of posttreatment data as 34 were lost to follow up. From the 78 patients with post treatment data, some questionnaires had missing values, affecting the number of patients in the moderator analysis (n=60). It is previously suggested that moderator analysis should include enough patients, as any moderating effect could be missed with too few patients to detect a moderating effect¹⁰. The current amount of data in the moderator analysis was below the a priori estimated power calculation (n=69), which could have contributed to not finding a difference. Currently, the handling and imputation of missing data

in moderation analysis is in its first steps and, with caution, Bayesian Estimates are recommended for imputation of missing data⁶². However, when distributions are non-parametric, such as in our study, Bayesian Estimates methods have a big change to misspecify. Therefore, before we can use it to impute missing data, more robust Bayesian Estimates methods are needed^{62,63}.

Scientific implications

Due to the current limitations of this study no conclusions can be drawn on the effectiveness of the treatment. However, the results do indicate changes within these patients and we suggest to further evaluate the effectiveness of transdisciplinary pain management treatment for patients having chronic pain. As suggested previously, the role of pain catastrophizing as a moderator for the outcomes of treatments¹⁰ should be further investigated. For instance, by studying the influence of pain catastrophizing as a moderator in disability and other treatment outcomes¹⁰. When doing so, one has to keep in mind that in patients with chronic pain there is an interplay of many factors, both positive and negative, that influencing the patients' pain experience. Other, more evolved approaches, for instance network analysis, might be more suitable for to further investigate these complex interplaying factors⁶⁴.

Clinical implications

As the design of the current study limits generalizations, clinical implications are scarce. However, some can be mentioned. First, based on the results from this study changes were observed in pain, disability, self-reported symptoms of central sensitization, pain catastrophizing and illness perceptions for patients with chronic pain. In accordance with previous literature, when providing a patient centered transdisciplinary CBT treatment program with PNE, these patients can indeed improve on these outcome measures. And second, in accordance with previous research, this study shows that changes in illness perceptions are associated with changes in pain catastrophizing in patients having chronic pain. As PNE addresses both illness perceptions and pain catastrophizing, physiotherapists should be aware of the importance of providing PNE in the treatment of patients with chronic pain.^{20-22,65}

Conclusion

This is, to our knowledge, the first study investigating the within treatment changes of patients with chronic pain who received a patient centered transdisciplinary CBT, among which PNE. There was an association between changes in pain catastrophizing and illness perceptions, suggesting that both were influenced by the treatment. However, the results indicate that even though there is an association between pretreatment and change scores in illness perceptions and self-reported symptoms of central sensitization in patients with chronic pain, neither the direction nor strength of the relationship between these two was influenced by pain catastrophizing. In future research the analysis of the data should incorporate the multi-faceted and complex aspect of chronic pain, for instance by complex network analysis.

References:

- Treede RD, Rief W, Barke A, et al. Chronic pain as a symptom or a disease: the IASP Classification of Chronic Pain for the International Classification of Diseases (ICD-11). *Pain.* 2019;160(1):19-27.
- 2. Merskey H, Bogduk N., International Association for the Study of Pain. Task Force on Taxonomy. *Classification of chronic pain: descriptions of chronic pain syndromes and definitions of pain terms.* 2 ed. the University of Michigan: IASP Press; 1994.
- 3. Zusman M. Forebrain-mediated sensitization of central pain pathways: 'non-specific' pain and a new image for MT. *Man Ther.* 2002;7(2):80-88.
- Trost Z, Strachan E, Sullivan M, Vervoort T, Avery AR, Afari N. Heritability of pain catastrophizing and associations with experimental pain outcomes: a twin study. *Pain*. 2015;156(3):514-520.
- Edwards RR, Dworkin RH, Sullivan MD, Turk DC, Wasan AD. The Role of Psychosocial Processes in the Development and Maintenance of Chronic Pain. J Pain. 2016;17(9 Suppl):T70-92.
- 6. Turner JA, Jensen MP, Romano JM. Do beliefs, coping, and catastrophizing independently predict functioning in patients with chronic pain? *Pain*. 2000;85(1-2):115-125.
- Lee MC, Zambreanu L, Menon DK, Tracey I. Identifying brain activity specifically related to the maintenance and perceptual consequence of central sensitization in humans. *J Neurosci.* 2008;28(45):11642-11649.
- 8. Quartana PJ, Campbell CM, Edwards RR. Pain catastrophizing: a critical review. *Expert Rev Neurother*. 2009;9(5):745-758.
- 9. Sullivan MJL, Bishop SR, Pivik J. The Pain Catastrophizing Scale: Development and validation. *Psychological Assessment*. 1995;7(4):524-532.
- 10. Wertli MM, Burgstaller JM, Weiser S, Steurer J, Kofmehl R, Held U. Influence of catastrophizing on treatment outcome in patients with nonspecific low back pain: a systematic review. *Spine (Phila Pa 1976).* 2014;39(3):263-273.
- Martinez-Calderon J, Struyf F, Meeus M, Luque-Suarez A. The association between pain beliefs and pain intensity and/or disability in people with shoulder pain: A systematic review. *Musculoskelet Sci Pract.* 2018;37:29-57.
- 12. Martinez-Calderon J, Jensen MP, Morales-Asencio JM, Luque-Suarez A. Pain Catastrophizing and Function In Individuals With Chronic Musculoskeletal Pain: A Systematic Review and Meta-Analysis. *Clin J Pain*. 2019;35(3):279-293.
- 13. Galambos A, Szabo E, Nagy Z, et al. A systematic review of structural and functional MRI studies on pain catastrophizing. *J Pain Res.* 2019;12:1155-1178.

- 14. Malfliet A, Coppieters I, Van Wilgen P, et al. Brain changes associated with cognitive and emotional factors in chronic pain: A systematic review. *Eur J Pain*. 2017;21(5):769-786.
- 15. Leventhal H, Brissette I, Leventhal EA. The common-sense model of self-regulation of health and illness. . In: Cameron LD L, ed. *The self-regulation of health and illness behaviour*. New York: Routledge; 2003:42–65.
- Costa ECV, Vale S, Sobral M, Graca Pereira M. Illness perceptions are the main predictors of depression and anxiety symptoms in patients with chronic pain. *Psychol Health Med*. 2016;21(4):483-495.
- de Heer EW, Vriezekolk JE, van der Feltz-Cornelis CM. Poor Illness Perceptions Are a Risk Factor for Depressive and Anxious Symptomatology in Fibromyalgia Syndrome: A Longitudinal Cohort Study. *Front Psychiatry*. 2017;8:217.
- 18. Jaremo P, Arman M, Gerdle B, Larsson B, Gottberg K. Illness beliefs among patients with chronic widespread pain associations with self-reported health status, anxiety and depressive symptoms and impact of pain. *BMC Psychol.* 2017;5(1):24.
- 19. Brosschot JF. Cognitive-emotional sensitization and somatic health complaints. *Scand J Psychol.* 2002;43(2):113-121.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil.* 2011;92(12):2041-2056.
- Wood L, Hendrick PA. A systematic review and meta-analysis of pain neuroscience education for chronic low back pain: Short-and long-term outcomes of pain and disability. *Eur J Pain*. 2019;23(2):234-249.
- Watson JA, Ryan CG, Cooper L, et al. Pain neuroscience education for adults with chronic musculoskeletal pain: a mixed-methods systematic review and meta-analysis. *J Pain.* 2019; 2019;20(10):1140.e1-1140.e22.
- Louw A, Zimney K, Puentedura EJ, Diener I. The efficacy of pain neuroscience education on musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract.* 2016;32(5):332-355.
- King R, Robinson V, Ryan CG, Martin DJ. An exploration of the extent and nature of reconceptualisation of pain following pain neurophysiology education: A qualitative study of experiences of people with chronic musculoskeletal pain. *Patient Education and Counseling*. 2016;99:1389-1393.
- 25. Wijma AJ, Speksnijder CM, Crom-Ottens AF, et al. What is important in transdisciplinary pain neuroscience education? A qualitative study. *Disabil Rehabil*. 2018;40(18):2181-2191.
- 26. Seminowicz DA, Shpaner M, Keaser ML, et al. Cognitive-behavioral therapy increases prefrontal cortex gray matter in patients with chronic pain. *J Pain*. 2013;14(12):1573-1584.

- 27. Cunningham NR, Kashikar-Zuck S, Coghill RC. Brain mechanisms impacted by psychological therapies for pain: identifying targets for optimization of treatment effects. *Pain Rep.* 2019;4(4):e767.
- 28. Shpaner M, Kelly C, Lieberman G, et al. Unlearning chronic pain: A randomized controlled trial to investigate changes in intrinsic brain connectivity following Cognitive Behavioral Therapy. *Neuroimage Clin.* 2014;5:365-376.
- 29. Pressman AJ, Peterlin BL, Tompkins DA, et al. Pain catastrophizing may moderate the association between pain and secondary hyperalgesia. *J Appl Biobehav Res.* 2017;22(1).
- Malfliet A, Kregel J, Meeus M, et al. Patients With Chronic Spinal Pain Benefit From Pain Neuroscience Education Regardless the Self-Reported Signs of Central Sensitization: Secondary Analysis of a Randomized Controlled Multicenter Trial. *PM R.* 2018; 10(12):1330-1343.e1.
- 31. Kraemer HC, Wilson GT, Fairburn CG, Agras WS. Mediators and moderators of treatment effects in randomized clinical trials. *Arch Gen Psychiatry*. 2002;59(10):877-883.
- 32. Ettema JHM, Arrindell, W.A. *SCL-90. Symptom Checklist. Handleiding bij een multidimensionele psychopathologie-indicator.* Lisse: Swets Test Publishers; 2003.
- 33. Ferraz MB, Quaresma MR, Aquino LR, Atra E, Tugwell P, Goldsmith CH. Reliability of pain scales in the assessment of literate and illiterate patients with rheumatoid arthritis. *J Rheumatol.* 1990;17(8):1022-1024.
- Grotle M, Brox JI, Vollestad NK. Concurrent comparison of responsiveness in pain and functional status measurements used for patients with low back pain. *Spine (Phila Pa 1976)*. 2004;29(21):E492-501.
- 35. Wolfe F, Clauw DJ, Fitzcharles MA, et al. The American College of Rheumatology Preliminary Diagnostic Criteria for Fibromyalgia and Measurement of Symptom Severity. *Arthritis Care* & *Research*. 2010;62(5):600-610.
- 36. Dudeney J, Law EF, Meyyappan A, Palermo TM, Rabbitts JA. Evaluating the psychometric properties of the Widespread Pain Index and the Symptom Severity Scale in youth with painful conditions. *Canadian Journal of Pain*. 2019;3(1):137-147.
- Kregel J, Vuijk PJ, Descheemaeker F, et al. The Dutch Central Sensitization Inventory (CSI): Factor Analysis, Discriminative Power and Test-Retest Reliability. *Clin J Pain*. 2015.
- 38. Mayer TG, Neblett R, Cohen H, et al. The development and psychometric validation of the central sensitization inventory. *Pain Pract.* 2012;12(4):276-285.
- 39. van Wilgen CP, Vuijk PJ, Kregel J, et al. Psychological Distress and Widespread Pain Contribute to the Variance of the Central Sensitization Inventory: A Cross-Sectional Study in Patients with Chronic Pain. *Pain Pract.* 2018;18(2):239-246.

- 40. Kregel J, Vuijk PJ, Descheemaeker F, et al. The Dutch Central Sensitization Inventory (CSI): Factor Analysis, Discriminative Power, and Test-Retest Reliability. *Clin J Pain*. 2016;32(7):624-630.
- 41. Neblett R, Cohen H, Choi Y, et al. The Central Sensitization Inventory (CSI): establishing clinically significant values for identifying central sensitivity syndromes in an outpatient chronic pain sample. *J Pain.* 2013;14(5):438-445.
- 42. Neblett R, Hartzell MM, Cohen H, et al. Ability of the Central Sensitization Inventory to Identify Central Sensitivity Syndromes in an Outpatient Chronic Pain Sample. *Clin J Pain*. 2013;14(5):438-445.
- 43. Cuesta-Vargas AI, Neblett R, Chiarotto A, et al. Dimensionality and Reliability of the Central Sensitization Inventory in a Pooled Multicountry Sample. *J Pain.* 2018;19(3):317-329.
- Kregel J, Schumacher C, Dolphens M, et al. Convergent Validity of the Dutch Central Sensitization Inventory: Associations with Psychophysical Pain Measures, Quality of Life, Disability, and Pain Cognitions in Patients with Chronic Spinal Pain. *Pain Pract.* 2018;18(6):777-787.
- 45. Scott W, Wideman TH, Sullivan MJ. Clinically meaningful scores on pain catastrophizing before and after multidisciplinary rehabilitation: a prospective study of individuals with subacute pain after whiplash injury. *Clin J Pain*. 2014;30(3):183-190.
- 46. Köke AJ, Smeets RJ, Schreurs KM, et al. Dutch Dataset Pain Rehabilitation in daily practice: Content, patient characteristics and reference data. *Eur J Pain*. 2017;21(3):434-444.
- 47. Osman A, Barrios FX, Gutierrez PM, Kopper BA, Merrifield T, Grittmann L. The Pain Catastrophizing Scale: further psychometric evaluation with adult samples. *J Behav Med.* 2000;23(4):351-365.
- 48. Osman A, Barrios FX, Kopper BA, Hauptmann W, Jones J, O'Neill E. Factor structure, reliability, and validity of the Pain Catastrophizing Scale. *J Behav Med.* 1997;20(6):589-605.
- 49. Broadbent E, Petrie KJ, Main J, Weinman J. The brief illness perception questionnaire. *J Psychosom Res.* 2006;60(6):631-637.
- 50. Leysen M, Nijs J, Meeus M, et al. Clinimetric properties of illness perception questionnaire revised (IPQ-R) and brief illness perception questionnaire (Brief IPQ) in patients with musculoskeletal disorders: A systematic review. *Man Ther.* 2015;20(1):10-17.
- 51. Grotle M, Garratt AM, Krogstad Jenssen H, Stuge B. Reliability and construct validity of selfreport questionnaires for patients with pelvic girdle pain. *Phys Ther.* 2012;92(1):111-123.
- 52. Salen BA, Spangfort EV, Nygren AL, Nordemar R. The Disability Rating Index: an instrument for the assessment of disability in clinical settings. *J Clin Epidemiol*. 1994;47(12):1423-1435.
- 53. Lins L, Carvalho FM. SF-36 total score as a single measure of health-related quality of life: Scoping review. *SAGE Open Med.* 2016;4:2050312116671725.

- VanderZee KI, Sanderman R, Heyink JW, de Haes H. Psychometric qualities of the RAND 36-Item Health Survey 1.0: a multidimensional measure of general health status. *Int J Behav Med.* 1996;3(2):104-122.
- 55. Moorer P, Suurmeije Th P, Foets M, Molenaar IW. Psychometric properties of the RAND-36 among three chronic diseases (multiple sclerosis, rheumatic diseases and COPD) in The Netherlands. *Qual Life Res.* 2001;10(7):637-645.
- 56. Faul F, Erdfelder E, Buchner A, Lang AG. Statistical power analyses using G*Power 3.1: tests for correlation and regression analyses. *Behav Res Methods*. 2009;41(4):1149-1160.
- 57. Hayes AF. PROCESS: A versatile computational tool for observed variable mediation, moderation, and conditional process modeling [White paper]. Retrieved from http://www. afhayes.com/ public/process2012.pdf. 2012.
- King R, Robinson V, Ryan CG, Martin DJ. An exploration of the extent and nature of reconceptualisation of pain following pain neurophysiology education: A qualitative study of experiences of people with chronic musculoskeletal pain. *Patient education and counseling.* 2016;99(8):1389-1393.
- 59. Pincus T, McCracken LM. Psychological factors and treatment opportunities in low back pain. *Best Practice and Research: Clinical Rheumatology.* 2013;27:625-635.
- 60. Sullivan MJ, Thorn B, Haythornthwaite JA, et al. Theoretical perspectives on the relation between catastrophizing and pain. *Clin J Pain.* 2001;17(1):52-64.
- 61. Lackner JM, Gurtman MB. Pain catastrophizing and interpersonal problems: a circumplex analysis of the communal coping model. *Pain*. 2004;110(3):597-604.
- 62. Zhang Q, Wang L. Moderation analysis with missing data in the predictors. *Psychol Methods*. 2017;22(4):649-666.
- 63. Zhang Q, Yuan K-H, Wang L. Asymptotic bias of normal-distribution-based maximum likelihood estimates of moderation effects with data missing at random. *British Journal of Mathematical and Statistical Psychology.* 2019;72(2):334-354.
- 64. Borsboom D, Cramer AO. Network analysis: an integrative approach to the structure of psychopathology. *Annu Rev Clin Psychol.* 2013;9:91-121.
- 65. Louw A, Zimney K, Puentedura EJ, Diener I. The efficacy of pain neuroscience education on musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract.* 2016;32(5):332-355.



General discussion

In the last decades there has been an improved understanding of the biopsychosocial mechanisms underlying chronic pain. Based on this increased knowledge we can now say that in patients with chronic pain centrally mediated mechanisms, also known as central sensitization (CS)¹⁻⁵, play an important role. Furthermore, as we are humans: body and mind cannot be separated. Therefore, bio, psycho as well as social factors influence the experience and continuation of pain⁶⁻¹⁰. Consequently, the treatment of patients with chronic pain has evolved. In this treatment, the explanation of CS and contributing biopsychosocial factors to patients, also known as PNE, has a level A evidence^{11,12} and has become a key element in the treatment.

In the current dissertation transdisciplinary PNE was studied to further improve our understanding of the processes surrounding patient-centeredness and PNE. The first part of the dissertation focused on patient centered care and assessments performed by physiotherapists in patients with chronic pain. The second part of the dissertation consisted of qualitative studies focusing on understanding the processes involved in PNE. The last part investigated the when and how of changes in outcomes following a transdisciplinary cognitive behavioural pain management program.

The following research questions were addressed in the thesis:

- To what extent is patient-centeredness examined in physiotherapy in qualitative research and can a theoretical framework be constructed for patient-centeredness in physiotherapy?
- What could be an outline of the clinical biopsychosocial physiotherapy assessment of patients with chronic pain?
- What are the experiences of patients with chronic pain who recently received PNE in a transdisciplinary setting?
- What are the clinical observations of transdisciplinary PNE? With the purpose to comprehend and theorize the social process of transdisciplinary PNE.
- What are the changes after patient centered transdisciplinary CBT, among which PNE, and is the relationship between illness perceptions and self-reported signs of CS moderated by pain catastrophizing?

In this chapter we will discuss the outcomes of these questions by deliberating on the outcomes of the studies included in this dissertation, followed by reflecting on these results in a broader context of healthcare practice, teaching and future research.

Main findings

A theoretical framework of patient-centeredness in physiotherapy

Besides being a moral philosophy of healthcare professionals¹³, the utilization of patient centered care is promising as it has shown to improve the therapeutic alliance, patient satisfaction and patient outcomes¹⁴⁻¹⁷. We conducted a qualitative literature review¹⁸, in which the phenomenon of patient-centeredness in physiotherapy was studied from the perspective of both patients and physiotherapists. The results from these studies indicate a synthesis of interrelated themes with subthemes and a proposed conceptual framework. The themes we found were: either related to the patient-therapist interaction in forms of *"individuality"*, *"communication"*, *"education"*, *"goal setting"*, *"support"* or related to the physiotherapists' ability to practice patient-centeredness in forms of the *"social characteristics of a patient-centered physiotherapist"*, *"a confident physiotherapist"*, and *"knowledge and skills of a physiotherapist in patient-centeredness"*.

The concepts of this review are quite similar to previous frameworks, with small variations between the current model and patient-centeredness in overall care (The Picker Institute's principles¹⁹), medicine ^{13,20} and nursing²¹. For instance, in contrast to overall care, medicine and nursing in the articles in the review environmental aspects were not an important part of patient-centeredness in physiotherapy^{19,21}. One reason for this could be that in contrast to our review aspects like transitions, policy, coordination of care, social involvement, etc. are more important in hospital settings^{19,21}. However, in a recent study on the patients' view on patient-centeredness in physiotherapy some environmental aspects, such as having privacy, were important²². Additionally, contextual factors, in forms of the healthcare setting and rituals of the therapists, are known to influence clinical outcomes²³. Therefore, it can be assumed that environmental aspects are most certainly of importance in patient-centeredness in physiotherapy. Our review sought to describe the phenomenon of patient-centeredness in physiotherapy, however, as a limitation it did only describe facilitators, not the barriers. Yet, knowing what is perceived as not being patient centered by patients is also important. Barriers such as having a negative attitude, lack of self-confidence, inappropriate (non-verbal) communication, lack of biopsychosocial sensitivity,

environmental factors (over crowdedness, lack of privacy) have a negative effect on patient centered physiotherapy²². Furthermore, recent research shows that in clinical practice the use of patient centeredness is to say the least lacking^{22,24-26}. Therefore, to improve patient-centeredness in physiotherapy, critical reflections on the professional practice and communication skills of physiotherapists should be increasingly taught. As well as teaching physiotherapists in these skills, barriers of patient-centeredness should also be addressed, thereby improving their patient centered skills. The current can be taught, for instance, during undergraduate education and, for those already working as physiotherapists, by coaching of professional organizational societies. By doing so the knowledge and skills necessary for patient-centeredness in physiotherapy can be improved.

When teaching physiotherapists about patient-centeredness one should keep in mind that the proposed synthesis of patient centeredness in physiotherapy in this dissertation has not been researched on its efficacy and use in clinical practice. Therefore, future studies should examine 1) to what extent teaching physiotherapists about patient-centeredness improves their knowledge and skills; 2) whether this improved knowledge and skills increases the use of patient-centeredness in practice; 3) improves clinical outcomes, 4) increases patient satisfaction and 5) decreases health care costs in patients received patient centered care.

A practical tool for the clinical biopsychosocial physiotherapy assessment in patients with chronic pain

In patient centered physiotherapy it is important to gain a complete picture of the patients' problem in a biopsychosocial way. However, physiotherapists struggle with identifying and treating psychosocial factors^{27:35}, and for instance rarely question the patients' beliefs about their health care problem³³. Biopsychosocial factors contribute to an amplified pain experience by increasing activity in the 'dynamic pain connectome'^{6-10,36}. Therefore, a thorough assessment of these factors is required to understand the individuals' pain, provoking and perpetuating factors. Furthermore, this assessment allows for an individualized patient centered PNE. To aid physiotherapists in this biopsychosocial physiotherapy assessment we conducted a narrative review in which we proposed a model to investigate these biopsychosocial factors. Based on this review the PSCEBSM-model emerged, which comprises the following factors: predominant Pain mechanism, Somatic, Cognitive, Emotional, Behavioural, Social and Motivational factors. First, the predominant type of pain is established, followed by

identifying the different factors associated with chronic pain and ending with the stage of motivation of the patient. Together with this model a pain analysis sheet is provided to help the clinician gain a clear overview of the PSCEBSM factors of the individual with chronic pain and guide the further treatment. As described in our 'Response Letter to the Editor'³⁷ (chapter 3), based on the correspondence with Oostendorp et al.³⁸, the layout of our current PSCEBSM-model somewhat originated from the SCEBS method as described by Spaendonck and Bleijenberg^{39,40}. However, there are two main differences. First, the PSCEBSM-model contains two extra factors to asses, the Pain- and Motivation-factors. Identifying the primary pain mechanism allows the physiotherapist to tailor the PNE to the patients' problem. Furthermore, Motivation is added to examine the patients' treatment expectations, psychological flexibility to change and stage of behavioural change. Second, the SCEBS method focused on providing healthcare professionals with questions for biopsychosocial history taking (for instance: "can you move your neck/back?" or "what do you think when you experience pain?"), whereas the current PSCEBS-model also includes guestionnaires. The guestionnaires were added to aid physiotherapists in recognizing psychosocial factors.

There are some limitations to mention. Chronic pain is a complex phenomenon in which research studying factors influencing pain is continuously evolving. Therefore, the practical guides for healthcare professionals in the assessment of patients should evolve. In our PSCEBSM-model we did not assess all factors, such as sleep, diet or coping. In recent years the importance and reciprocal relationship of sleep on pain has become increasingly acknowledged.⁴¹ The contribution of sleep in patients with chronic pain has been described in the key-set as developed by the American Pain Society (APS) Pain Taxonomy (AAPT) by Turk et al.⁴². When combining the key-set by Turk et al⁴² and the PSCEBSM-model, keeping in mind practicality, a more overarching tool for healthcare professionals, among which physiotherapists, could be made. Furthermore, there are no studies performed into practical application and the effectiveness of this model, the SCEBS-model⁴³, and the overviews provided by Turk et al.^{42,44}. Future research needs to investigate the appropriateness and possible effects on patient outcomes of these assessments.

Perspectives of patients with chronic pain who recently received Pain Neuroscience Education in a transdisciplinary setting

PNE has been recognized as an important element in the treatment of patients with chronic pain^{11,12}. However, as chronic pain is complex, its treatment is preferred to be

multimodal. In this treatment of patients with chronic pain effective team approaches can exceed the expertise of monodisciplinary care⁴⁵. In recent years transdisciplinary teamwork has emerged. Transdisciplinary teamwork meaning, among other aspects, that there is an intensive collaboration, with shared responsibilities^{46,47} and flexible boundaries⁴⁶⁻⁴⁹.

It is important to recognize patients' experiences of the transdisciplinary Pain Neuroscience Education process. Therefore, based on the methods of Grounded Theory⁵⁰, we conducted a qualitative study to comprehend and theorize how patients with chronic pain experienced transdisciplinary PNE.

The first topic that emerged was "the pre-PNE phase", the phase prior to the PNE in which the respondents met the healthcare professionals during a broad biopsychosocial intake session. The second found topic "a comprehensible PNE", which contained an "understandable explanation" and the "interaction between the physiotherapist and psychologist". The third topic involved the "outcomes of PNE", including the subthemes "awareness", "finding peace of mind", and "fewer complaints". The fourth topic, "scepticism", contained "doubt towards the diagnosis and PNE", "disagreement with diagnosis and PNE", and "PNE can be confronting".

Even though qualitative research is limited in generalization, the results reveal the importance of the therapeutic alliance between the patient with chronic pain and healthcare professionals. A broad biopsychosocial intake, with empathic healthcare professionals, who take time to listen, is known to enhance the therapeutic alliance^{51,52} and of importance in the rehabilitation of patients with chronic pain⁵³. The biopsychosocial view on pain by the transdisciplinary team, their active involvement, and interpersonal skills are crucial to the outcomes of the PNE. Furthermore, the patient centered aspect of the treatment (*i.e.*, providing patient-centred care, taking time, building rapport and building a therapeutic alliance)^{51,54,55}, helps to tailor the comprehensible PNE to the individual, thus enhancing the perceived relevance. However, even with this perceived relevance, some patients with chronic pain did not or partly reconceptualized their pain. As King et al. mentioned before: pain-related beliefs of respondents prior to the PNE can be barriers for the PNE⁵⁶. Furthermore, patients' readiness to change, an increased commitment to self-management, and initial hesitation about the treatment predicts the satisfaction and likelihood of completion of a pain management program⁵⁷⁻⁵⁹.

However, most respondents with chronic pain in our study did mention perceptual changes about pain in a cognitive, emotional and behavioural manner. From previous research it is known that those patients with chronic pain who reconceptualized their pain mention an increased coping with their condition and improved physical activity levels⁶⁰. The latter is similar to our study, in which most respondents mentioned increased coping, as well as mentioning to have an elicited biopsychosocial awareness. For instance, respondents mentioned to take better care of themselves.

Generalizing study results is always a delicate issue in qualitative research⁶¹. Generalizability in this study is an concern as there was an individualized transdisciplinary intake, the repetition of PNE by the general practitioner, booklet, and combined physiotherapist/ psychologist session which are not common for clinical practice, nor has been studied. Therefore, even though the current design is promising, one should be careful in generalizing the study findings to other settings. Furthermore, in the current study a classical Grounded Theory approach was used⁵⁰. Grounded Theory is a qualitative research methodology used to deeply analyse and develop theoretical explanations emerging from the participants' unique perspectives⁶². Within Grounded Theory there is a distinctive difference between classical Grounded Theory and constructive Grounded Theory, especially concerning the role of the researcher. Whereby in classical Grounded Theory the researcher tries to take a neutral, objective, more distance stance towards the data and their preconceptions^{50,62}. Starting in 2013, this study was, to our knowledge, the first study exploring the experiences of patients with chronic pain with PNE. Therefore, the choice for classical Grounded Theory was based on the following arguments: 1) no previous closely related literature was available on the subject, 2) an open stance in which any experience of the patients with chronic pain was analysed and theorized was preferred, and 3) the patients with chronic pain voices needed to be adequately reflected in the theory. However, as all researchers were working academically and/or clinically with PNE the neutral, an objective stance can be doubted and in retrospect a Constructive Grounded Theory approach might have been more appropriate.

Clinical observations of patient and therapist interactions of Pain Neuroscience Education in a transdisciplinary setting

Because PNE is a "talk-modality" many interpersonal aspects, such as the therapeutic alliance, contribute to the treatment effect. These social processes can best be captured through qualitative observational studies. Therefore, based on Constructive

Grounded Theory^{63,64} we conducted an observational gualitative study to comprehend and theorize the practice of transdisciplinary PNE. The outcome was a theoretical construct and framework of transdisciplinary PNE sessions, grounded in the data of the observations. Four connected and interactive themes where found to be of importance for transdisciplinary PNE, with three overarching themes. The first theme we found was "generating a safe and comfortable feeling, situation" as a prerequisite. Important in this theme and overlapping with the second theme the "PNE process", was incorporating personal information from the intake in the PNE. Individualizations, as well as the step by step explanation and the used metaphors, sparked "the process of change of the respondent" (third theme) thereby leading to the fourth theme: "outcome of the change process", meaning that most respondents gained improved biopsychosocial understanding of their pain, whereby some had a partial reconceptualization. One of the overarching themes, the "influence of changeability of the respondents" on the PNE and outcomes, even though obviously present, was a difficult concept to grasp. There was no distinct profile detected from the video observations of those who would "respond" fully, partially or were "non-responders". Two other overarching themes were: "interplay between the healthcare professionals" and "communication skills" of the healthcare professionals.

The results of this study are comparable to studies highlighting the importance of a good patient-therapist interaction in pain rehabilitation^{53,65}, in the form of a good therapeutic alliance (TA)⁶⁶. Attending to, and healing ruptures in the TA^{67,68}, as seen during the observations, can strengthen the therapeutic relationship^{67,69,70} and contribute to positive patient outcomes^{67.}

Whereas the TA is a basis for the PNE, reconceptualizing their pain based on PNE is sometimes difficult for patients with chronic pain: adopting a new concept on pain takes time^{56,71,72}. In line with the current study, reconceptualization is often partial and follow up during further treatment is needed to further deepen the reconceptualization^{56.} Therefore, it is necessary to keep in mind the patient centered nature of PNE⁷³: one size will not fit all. Furthermore, even though reconceptualization is the first step, nonetheless dealing with the biopsychosocial factors that influence the pain experience is the next⁷⁴. In the current study some of the patients with chronic pain did understand the PNE, but found it hard to envision changes in their pain related behaviour. Whilst others had already changed their daily practice based on the first PNE session. It is

important to keep in mind that PNE is an important part of the treatment of patients with chronic pain, it is not a stand-alone intervention^{12,60,75}.

A few limitations can be mentioned from this study: First, the observations only included the second PNE session of the respondents. Hence, the trajectory of the respondents' reconceptualization and further implications in an emotional, cognitive and behavioural manner could not be investigated. However, as the reconceptualization was only a part of the research question and we wanted to deeply understand the social aspects surrounding those who had doubt/resistance, the current format was more suitable.

Second, as previously mentioned generalizability in qualitative research is always a delicate issue⁶¹. The generalizability of this study is limited to the transdisciplinary context of the observed PNE in which a psychologist and physiotherapist collaborate. Whereas the evidence of the effectiveness of PNE however, comes from studies based on a monodisciplinary (physiotherapy) setting. And third, the reflexivity of the researchers analyzing the data is a source of limitation⁷⁶. All authors in some form contribute to the field of PNE either in clinical practice, education or science. Based on the epistemology of constructive Grounded Theory the idea of being a completely neutral observer without preconceptions was chucked. Instead, these conceptions construct the research and they were therefore examined, identified and described⁶³.

Moderators of change of self-reported signs of central sensitization following an individualized transdisciplinary pain management program

As the integration of PNE in monodisciplinary physiotherapy is effective^{11,12}, integrating it and studying this integration in multi- or transdisciplinary care for patients with chronic pain seems recommendable. Hence, a scoping evaluation by means of a retrospective observational study of current practice of a patient centered transdisciplinary pain management program for patients with chronic pain, including PNE, was performed. Subsequently, it is important to comprehend which factors moderate the changes in outcomes following this program. The relationships between these clinical outcomes and cognitive and emotional factors and changes in these outcomes following treatment were analyzed. Multiple regression analysis, by the PROCESS macro⁷⁷ explored the moderating effects of pain catastrophizing on the relationship between illness perceptions and self-reported signs of CS in patients with chronic pain. We found a moderate association between pretreatment scores of pain catastrophizing and illness perceptions in patients with chronic pain, suggesting a bilateral relationship between catastrophic thinking and negative illness perceptions. Additionally, there were significant, however moderate, correlations between pain catastrophizing, illness perceptions and self-reported symptoms of sensitization in the change-scores of patients with chronic pain following a patient centered CBT.

However, contrary to our hypothesizes, the results from our moderation analyses indicate that the direction or strength between the baseline scores and change scores of illness perceptions and self-reported symptoms of CS in patients with chronic pain was not influenced by baseline, nor change scores in pain catastrophizing. One reason could be that patients with chronic pain had low pain catastrophizing at baseline and only small changes⁷⁸.

Important to mention as a limitation is that the current design was a pre-post treatment design instead of an RCT design. Therefore, no hard assumptions can be made regarding the effectiveness of the treatment, nor a part of the treatment (PNE) can be made. Besides, as this intervention was a patient-tailored approach not all patients with chronic pain received the same treatment and measurement points (T0 and T1) differed between patients. In an ideal world a double blind RCT design, with prior intentions for effect moderation analysis, would have been performed. Furthermore, there was a significant amount of data lost to follow-up: 34 patients dropped out and a further 18 patients did not complete all questionnaire items. The data collection was allocated to a secretary, without knowledge of data collection nor was it his/her core business, and post-intervention data collection was performed by mailing these questionnaires. Even though patients were called and asked to fill in these questionnaires, some patients found it too burdensome to complete them, whilst others did not return them. In future this task should be structurally embedded within routine clinical practice, for instance by using online questionnaires in a system that allows state-of-the-art data protection and data management. Due to these dropouts and missing data the current number of data in the moderator analysis was below the number obtained through the a priori conducted power calculation. As suggested previously by Wertli et al.⁷⁸ moderator analyses should include enough patients with chronic pain, as any moderating effects could be missed with too few patients to detect a difference. The actual number of participants one should include under debate. For instance, a review on moderators in response to treatment in low back pain by Gurung et al. included,

based on a priori sample size calculation, only trials with a minimum dataset of 503 participants⁷⁹. However, they did use an effect size of 0.5 in the sample size calculation, which is large. Furthermore, they included studies whom performed multiple moderator analyses with multiple moderators, thus requiring more participants. And last, within the statistical community a sample size calculation, as we performed, indicating the number of participants needed is deemed sufficient and post-hoc power analysis not recommended⁸⁰.

Finally, as the data were not normally distributed, non-parametric testing was used for the analysis of pre- versus post-treatment differences and correlations. For the moderation analysis, however, it is advised to use the parametric regression analysis PROCESS-macro, by Hayes⁷⁷, which we have used. It is uncommon to use this combination of statistic testing. This choice was made because of the following arguments: 1) non-parametric moderation analysis is fairly new⁸¹ and was not available at the time of the analysis, 2) transformation of data is currently under debate, and 3) there was no skewness of the data. The PROCESS macro bootstrapping does not require normal distribution if the sample size is adequate and skewness is small⁸².

And now what? Philosophizing the PNE/physiotherapy future

Overthinking

Based on the current dissertation some implications are formulated. However, prior to reading these implications one should realize that these studies are mostly qualitative in nature. Therefore, the reader is encouraged to be cautious with generalizing this information towards clinical practice. One should rather consider the information and then reflect on his/her own practice and what it might mean to them.

Furthermore, the findings from the qualitative studies should be interpret with caution and reflexitivity is warranted. First, even though all studies were teamwork with different team-members in each study, I was the main researcher. The chronological sequence of the studies (systematic literature review on patient centeredness in physiotherapy, GT study on the perspectives of patients on transdisciplinary PNE, and then the CGT with clinical observations of transdisciplinary PNE), could have influenced the successive studies. Nevertheless, we applied the trustworthiness criteria of Lincoln and Guba⁸³ in these studies to reduce such influences. Furthermore, according to the used methodologies we tried to bracket out our perspectives as much as possible in the (first) GT-study. In the (second) CGT-study the perspectives of the authors influenced the theory, as is usual in CGT. Therefore, these perspectives were written down in a reflective paper, member checking was performed and a neutral stance was sought and debated within the research-team. Second, throughout the years of my early research career, my professional career also changed: from working in private practice and hardly using PNE, to working in both private practice, using PNE, working at the transdisciplinary pain management center and teaching other healthcare professionals in PNE/working with patients with chronic pain. Following the chronological sequence of the above-mentioned studies, I started working at the transdisciplinary pain management center of this study. To enhance open-mindedness we performed researcher-triangulation, with the second researcher not being involved in the treatment, as well as write a reflective paper, perform peer debriefing and other aspects of the trustworthiness criteria.

Considering this reflexivity, to broaden it: we now have all these findings, compared to known research, yet what does it mean in the grand scheme of our profession(s)? Of PNE? So, to take it one step further: can we helicopter these findings?

As we have seen throughout this dissertation, communication skills are vital in the delivery of PNE. As PNE is currently under debate, especially in social media, I find it important to reflect on this part of PNE. First, PNE is under debate as the latest reviews have only shown small improvements in outcomes when providing PNE. One could ask: why? The first RCTs showed promising results. If I remember correctly, my first encounter with PNE was by a video of Lorimer Moseley, one of the authors of the first RCTs on PNE. Besides his fascination for neurophysiology of pain he encompasses great skills in communicating. His enthusiasm and 'plain language', as well as engagement with persons, might be one of the factors of the success reported in these first RCTs. As especially in a talk-modality, the person delivering the PNE is of major importance. Second, reviews on the effect of providing only PNE show limited effects. Pain, particularly chronic pain, is often a complex phenomenon. Therefore, the treatment of patients with chronic pain should consist of multiple treatment modalities and not just PNE. There are few patients with chronic pain who are helped with only PNE, however, most will need more than that. In my opinion PNE can be used as psychologists use psychoeducation: an evidence based neurophysiological explanation of their condition and contributing factors, which serves as a 'coat rack' (figure 1) to

'hang' the patient centered evidence-based treatment modalities on. For instance, to use as a steppingstone for Graded Exposure, in which the patient with chronic pain is encouraged to reengage in fearful activities which they have previously believed to cause damage to them and/or were fearful of.



Figure 1. PNE as a 'coat-rack' to explain and use other treatment modalities (not all modalities are mentioned)

Furthermore, it provides patients with chronic pain with an explanation for why they are in pain. Which brings us to the third concern: some (clinicians, researchers) question whether we can accurately enough explain the neurophysiological processes concerning pain to patients with chronic pain without oversimplifying it, whether PNE is not too positivistic. On which I humbly disagree. Of course, we should always be reflective on how we verbalize psychoeducation. The term Pain <u>Neuroscience</u> Education might imply to much of a focus on the neurophysiology-part. So clinicians might only educate patients on neurophysiological changes related to CS. Whereas in a truly patient centered psychoeducation the neurophysiology is adapted on the person, and the most important part of the education is the explanation of the (changeable) biopsychosocial factors involved in the continuation of CS and (chronic) pain. Therefore, Pain Education

might reflect a more inclusive terminology of this psychoeducation. Additionally, I agree with the critics that any metaphor used will never truly reflect the complex neuroendocrine-immunological and perceptual processes involved when one is in (chronic) pain. As well as that the therapist should be careful not to 'medicalize' these processes. However, as we are human beings, our brains desire for 'logical' explanations of what is happening and what we experience. De-educating patients' maladaptive cognitions of these painful experiences ('my pain means my back muscles tear when I bend') is an important first step. Yet, we still have to explain to them why they are still in pain. Otherwise their brain will want to provide them another 'logical explanation'. By PNE we can help patients with chronic pain reconceptualize these explanations towards less threatening ones ('my back is sensitive, but not damaged, bending is a healthy step towards recovery'). Furthermore, as pain is a complex and individual experience, the narrative of the patient with chronic pain preferably constructs the PNE. The wording and experiences of the patient with chronic pain should be used during the PNE, allowing the metaphors to be patient centered and dynamical. Furthermore, the metaphors used should be thoughtfully and carefully selected. One might even discuss that therefore PNE or PE is constructivist in manner...

The latter brings us to the last reflective point of the current dissertation: chronic pain is often a complex phenomenon. Therefore, what are our professional boundaries as physiotherapists when seeing a patient with chronic pain? Is monodisciplinary physiotherapy care enough? Or should every patient with chronic pain be seen by an inter-/transdisciplinary team? These are difficult questions to answer. On a practical level one could say: no, our healthcare system is not equipped enough, nor does every patient with chronic pain need to be seen in an inter-/transdisciplinary setting. However, it is difficult to answer what the grades of complexity are and at which grade the patient with chronic pain needs to be referred from mono- to inter-/transdisciplinary. Even though a stepped care principle is recommended in the Netherlands, the levels of complexity for mono/inter/transdisciplinary have not been described. It is a dilemma, we, as scientists, professionals and patients with chronic pain, need to discuss and philosophize about to further improve the care for patients with chronic pain.

Implications for practice – based on the overthinking

Let us bring it back to the practical implications of this dissertation. First, an increasing bulk of evidence keeps on showing what philosophers have told us all along: the relationship between healthcare professional (physiotherapist) and patient is the most important pre-requisite for the treatment of any patient. Especially in patients with chronic pain who have often had other (negative) experiences. Or as mentioned by the founding father of the TA, Carl Rogers: 'Before every session I take some time to remember my humanity'. To further improve this relationship, the themes mentioned in our qualitative study defining patient-centeredness in physiotherapy can be used. However, this does require a self-reflective communicationally skilled physiotherapist. Second, this relationship is nurtured during the initial biopsychosocial assessment of the patient with chronic pain. During this assessment the PSCEBSM-model can be used to assess the individuals' predominant-pain mechanism, somatic, cognitive, emotional, behavioural, social and motivational factors. Furthermore, sleep and diet should be taken into account. The individuals' contributing factors can then be filled in in the provided pain-analysis form, to have a clear overview of the individuals' problems. Third, PNE should preferably be the next step in the treatment. Explaining the predominant pain-mechanism, pain neurophysiology, and contributing/provoking biopsychosocial factors in a patient centered, step-by-step manner, with lay terms adjusted to the individual. Throughout the (if possible transdisciplinary) PNE the therapist is advised to be aware of his/her communicational skills in order to provide a safe and comfortable environment and engage the patient with chronic pain to the PNE. When doing so patients with chronic pain mention to experience being seen, taken seriously and cognitions on the biopsychosocial nature of pain can improve. Most patients with chronic pain, however, not all, reconceptualize their pain from 'damage' into 'an overprotective state'. Reconceptualization and changes in outcomes are enhanced not only during the PNE session, but also in the period and treatment after the PNE. Therefore, the repetition of the PNE during the further treatment is important, for instance, as an explanation of why relaxation exercises or CBT are of importance.

And last, throughout the PNE and further treatment some doubt and scepticism about whether 'there isn't something damaged' are normal. When this occurs therapists should be aware of these so called 'ruptures' and carefully re-explain PNE whilst restoring the TA.

Implications for future research - based on the overthinking

As known, when performing research, the answers we get always lead to more questions. Therefore, some recommendations for further research are suggested. By now there are quite some 'communication-aspects' known that surround patient

care, such as patient centeredness and the therapeutic alliance as outlined in this dissertation, but also Shared Decision Making, the working alliance, narrative medicine and Socratic dialogue. These constructs show considerable overlap and can be quite confusing for healthcare professionals. Especially when these healthcare professionals, such as physiotherapists, find these communication-aspects difficult. As research should be practical, it is important to help healthcare professionals see the maze of these constructs. Therefore, one could suggest studying the overarching construct of these communication aspects and provide a clinical framework, which then should be investigated for effectiveness, to help healthcare professionals further improve the most important aspect of the treatment. Furthermore, the biopsychosocial physiotherapy assessment for patients with chronic pain as suggested in this dissertation has not been investigated on its efficacy. However, prior to investigating its efficacy it is important to further supplement the assessment with factors known from more recent studies to influence the pain experience. Such as, but not exclusively, sleep and diet. Investigating the efficacy of the biopsychosocial assessment could be done in multiple manners. First, the psychometric properties of the biopsychosocial assessment could be studied by test-retest observational studies. Second, as most physiotherapists find it hard to implement psychosocial factors in the biopsychosocial assessment, behavioural change of physiotherapists after training should be assessed. This could be done by Quality Indicators (QI's), as previously used by Oostendorp et al.⁸⁴ QI's evaluate the implementation and change in quality of provided care and measure the behavioural change of physiotherapists. And third, the added value of the biopsychosocial assessment on healthcare outcomes could be studied in clinical trials. Preferably in mixed methods designs whereby the evaluation of patients with chronic pain on the assessment is taken into account.

The current dissertation outlined patient centered transdisciplinary PNE and treatment, by experiences, observations and moderators. However, to date, most healthcare professionals treating patients with chronic pain are working in a monodisciplinary setting. In the Netherlands, where the current dissertation took place, patients with chronic pain who do not respond to monodisciplinary treatment are often referred to multi- or interdisciplinary treatment centers. As outlined in the General Introduction, there are distinct differences between multi-/interdisciplinary treatment programs and transdisciplinary treatment programs. Thus, generalizing the outcomes of this dissertation to other (mono-, multi- or interdisciplinary) settings is difficult. Therefore, two implications for future research can be listed: It would be interesting to investigate

the effectiveness of working transdisciplinary, for instance by comparison to multior interdisciplinary settings. Throughout these studies a mixed methods design can be implemented. For instance, by including gualitative observations of not only the second session of PNE (as with ours) but also the intake, first session, and further treatment. To further theorize how reconceptualization of pain takes place, how healthcare professionals help reconceptualize pain by PNE whilst also maintaining the TA, and what the implications of these reconceptualizations are in terms of changes in cognitions, emotions, behaviour, coping, acceptance, etc. Furthermore, as patients with chronic pain are, and will be continued to, be treated in a monodisciplinary setting: further studying the implementation and effect of PNE in clinical practice is important. By means of gualitative and guantitative research methods several topics could be investigated: 1) the current implementation of PNE in monodisciplinary practice (qualitatively, quantitatively), 2) the experiences of patients with chronic pain with PNE in a monodisciplinary setting, and 3) how and when changes occur in patients with chronic pain receiving PNE in practice (for instance by investigating latent changes trajectories by latent grow curves and growth mixture modeling analysis), 4) as well as studying how we can help guide physiotherapists with this population.

To conclude

Treating patients with chronic pain with patient centered PNE entails much more than 'the talk'. It is about patient centered-working, the therapeutic alliance, a broad intake based on the PSCEBSM-model and an interactive education within a (transdisciplinary) treatment setting. As academics we have multiple obligations, and not only to our patients with chronic pain. It is not solely to study interventions, but also to support our fellow healthcare professionals. To help them implement these treatments. As they are the ones delivering the care for the patients with chronic pain. With this dissertation I hope to have unraveled some of the aspects surrounding PNE and further spark the treatment of patients with chronic pain: for them and for our fellow healthcare professionals.

References

- 1. Woolf CJ, Salter MW. Neuronal plasticity: increasing the gain in pain. *Science*. 2000;288(5472):1765-1769.
- 2. Latremoliere A, Woolf CJ. Central sensitization: a generator of pain hypersensitivity by central neural plasticity. *J Pain*. 2009;10(9):895-926.
- 3. LaMotte RH, Shain CN, Simone DA, Tsai EF. Neurogenic hyperalgesia: psychophysical studies of underlying mechanisms. *J Neurophysiol*. 1991;66(1):190-211.
- Koltzenburg M, Torebjork HE, Wahren LK. Nociceptor modulated central sensitization causes mechanical hyperalgesia in acute chemogenic and chronic neuropathic pain. *Brain.* 1994;117 (Pt 3):579-591.
- 5. Torebjork HE, Lundberg LE, LaMotte RH. Central changes in processing of mechanoreceptive input in capsaicin-induced secondary hyperalgesia in humans. *J Physiol.* 1992;448:765-780.
- 6. Zusman M. Forebrain-mediated sensitization of central pain pathways: 'non-specific' pain and a new image for MT. *Man Ther.* 2002;7(2):80-88.
- Lee MC, Zambreanu L, Menon DK, Tracey I. Identifying brain activity specifically related to the maintenance and perceptual consequence of central sensitization in humans. *J Neurosci.* 2008;28(45):11642-11649.
- 8. Brosschot JF. Cognitive-emotional sensitization and somatic health complaints. *Scand J Psychol.* 2002;43(2):113-121.
- 9. Turk DC, Okifuji A. Psychological factors in chronic pain: evolution and revolution. *J Consult Clin Psychol*. 2002;70(3):678-690.
- 10. Moseley GL. A pain neuromatrix approach to patients with chronic pain. *Man Ther.* 2003;8(3):130-140.
- Louw A, Diener I, Butler DS, Puentedura EJ. The effect of neuroscience education on pain, disability, anxiety, and stress in chronic musculoskeletal pain. *Arch Phys Med Rehabil*. 2011;92(12):2041-2056.
- Louw A, Zimney K, Puentedura EJ, Diener I. The efficacy of pain neuroscience education on musculoskeletal pain: A systematic review of the literature. *Physiother Theory Pract.* 2016;32(5):332-355.
- 13. Epstein RM, Franks P, Fiscella K, et al. Measuring patient-centered communication in patient-physician consultations: theoretical and practical issues. *Soc Sci Med.* 2005;61(7):1516-1528.
- 14. World Health Organization. *WHO global strategy on people-centred and integrated health services: interim report. World Health Organization.* 2015.

- 15. Dwamena F, Holmes-Rovner M, Gaulden CM, et al. Interventions for providers to promote a patient-centred approach in clinical consultations. *Cochrane Database Syst Rev.* 2012;12:CD003267.
- 16. Rees S, Williams A. Promoting and supporting self-management for adults living in the community with physical chronic illness: A systematic review of the effectiveness and meaningfulness of the patient-practitioner encounter. *JBI Libr Syst Rev.* 2009;7(13):492-582.
- 17. Plewnia A, Bengel J, Korner M. Patient-centeredness and its impact on patient satisfaction and treatment outcomes in medical rehabilitation. *Patient education and counseling*. 2016;99(12):2063-2070.
- 18. Wijma AJ, Bletterman AN, Clark JR, et al. Patient-centeredness in physiotherapy: What does it entail? A systematic review of qualitative studies. *Physiother Theory Pract.* 2017;33(11):825-840.
- 19. Gerteis M, Edgman-Levitan, S., Daley, J., Delbanco, T.L. *Through the Patient's Eyes: Understanding and Promoting Patient-Centered Care.* 2 ed: Wiley; 2002.
- 20. Mead N, Bower P. Patient-centredness: a conceptual framework and review of the empirical literature. *Soc Sci Med.* 2000;51(7):1087-1110.
- Kitson A, Marshall A, Bassett K, Zeitz K. What are the core elements of patient-centred care? A narrative review and synthesis of the literature from health policy, medicine and nursing. J Adv Nurs. 2013;69(1):4-15.
- 22. Morera-Balaguer J, Botella-Rico JM, Catalan-Matamoros D, Martinez-Segura OR, Leal-Clavel M, Rodriguez-Nogueira O. Patients' experience regarding therapeutic person-centered relationships in physiotherapy services: A qualitative study. *Physiother Theory Pract.* 2019:1-11.
- 23. Testa M, Rossettini G. Enhance placebo, avoid nocebo: How contextual factors affect physiotherapy outcomes. *Man Ther.* 2016;24:65-74.
- 24. Hall AJ, Burrows L, Lang IA, Endacott R, Goodwin VA. Are physiotherapists employing person-centred care for people with dementia? An exploratory qualitative study examining the experiences of people with dementia and their carers. *BMC Geriatrics*. 2018;18(1):63.
- 25. Cruz EB, Moore A, Cross V. Clinical reasoning and patient-centred care in musculoskeletal physiotherapy in Portugal--a qualitative study. *Man Ther.* 2012;17(3):246-250.
- 26. Hiller A, Guillemin M, Delany C. Exploring healthcare communication models in private physiotherapy practice. *Patient Educ Couns.* 2015;98(10):1222-1228.
- 27. Daykin AR, Richardson B. Physiotherapists' pain beliefs and their influence on the management of patients with chronic low back pain. *Spine (Phila Pa 1976)*. 2004;29(7):783-795.
- 28. Haggman S, Maher CG, Refshauge KM. Screening for symptoms of depression by physical therapists managing low back pain. *Phys Ther.* 2004;84(12):1157-1166.
- 29. Valjakka AL, Salanterä S, Laitila A, Julkunen J, Hagelberg NM. The association between physicians' attitudes to psychosocial aspects of low back pain and reported clinical behaviour: A complex issue. *Scandinavian Journal of Pain*.4(1):25-30.

- 30. Synnott A, O'Keeffe M, Bunzli S, Dankaerts W, O'Sullivan P, O'Sullivan K. Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a systematic review. *J Physiother*. 2015;61(2):68-76.
- 31. Singla M, Jones M, Edwards I, Kumar S. Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study. *Man Ther.* 2015;20(2):328-334.
- Overmeer T, Linton SJ, Boersma K. Do physical therapists recognise established risk factors? Swedish physical therapists' evaluation in comparison to guidelines. *Physiotherapy*. 2004;90(1):35-41.
- 33. van Wilgen CP, Beetsma A, Neels H, Roussel N, Nijs J. Physical therapists should integrate illness perceptions in their assessment in patients with chronic musculoskeletal pain; a qualitative analysis. *Man Ther.* 2014;19(3):229-234.
- 34. Franca AA, Santos VD, Filho RL, Pires KF, Lagoa KF, Martins WR. 'It's very complicated': Perspectives and beliefs of newly graduated physiotherapists about the biopsychosocial model for treating people experiencing non-specific low back pain in Brazil. *Musculoskelet Sci Pract.* 2019;42:84-89.
- 35. Driver C, Lovell GP, Oprescu F. Physiotherapists' views, perceived knowledge, and reported use of psychosocial strategies in practice. *Physiother Theory Pract.* 2019:1-14.
- 36. Kucyi A, Davis KD. The dynamic pain connectome. Trends in neurosciences. 2015;38(2):86-95.
- 37. Wijma AJ, Paul van Wilgen CP, Meeus M, Nijs J. Response to Letter. *Physiotherapy Theory and Practice*. 2017;33(3):263-265.
- 38. Oostendorp RAB, Elvers H, Mikolajewska E, Laekeman M, van Trijffel E, Samwel H. "Clinical biopsychosocial physiotherapy assessment of patients with chronic pain: The first step in pain neuroscience education" by Amarins J Wijma et al., 2016. *Physiotherapy Theory and Practice*. 2017;33(3):260-262.
- van Spaendonck KPM, Bleijenberg G. Biopsychosociale klachten. Het biopsychosociale model (SCEGS) http://www.wikifysio.nl/index.php/Biopsychosociale_klachten,_SCEGS. Published 2010.
- 40. van Spaendonck KPM, van de Lisdonk EH. Biopsychosociale klachten, SCEGS. In: *Het biopsychosociale ziektemodel: een proeve van onderwijskundige operationalisatie. Gezond Onderwijs-4*. Houten/Zaventem: Bohn Stafleu Van Loghum; 1995:272-276.
- 41. Nijs J, Mairesse O, Neu D, et al. Sleep Disturbances in Chronic Pain: Neurobiology, Assessment, and Treatment in Physical Therapist Practice. *Phys Ther.* 2018;98(5):325-335.
- 42. Turk DC, Fillingim RB, Ohrbach R, Patel KV. Assessment of Psychosocial and Functional Impact of Chronic Pain. *J Pain.* 2016;17(9 Suppl):T21-49.

- 43. van Spaendonck K, van de Lisdonk, EH. Biopsychosociale klachten, SCEGS. In: *Het biopsychosociale ziektemodel: een proeve van onderwijskundige operationalisatie. Gezond Onderwijs-4.* Houten/Zaventem: Bohn Stafleu Van Loghum; 1995:272-276.
- 44. Dansie EJ, Turk DC. Assessment of patients with chronic pain. Br J Anaesth. 2013;111(1):19-25.
- 45. Gordon DB, Watt-Watson J, Hogans BB. Interprofessional pain education-with, from, and about competent, collaborative practice teams to transform pain care. *Pain Rep.* 2018;3(3):e663.
- 46. Shaw L, Walker R, Hogue A. The art and science of teamwork: enacting a transdisciplinary approach in work rehabilitation. *Work*. 2008;30(3):297-306.
- 47. Reilly C. Transdisciplinary approach: an atypical strategy for improving outcomes in rehabilitative and long-term acute care settings. *Rehabil Nurs.* 2001;26(6):216-220, 244.
- 48. Vyt A. Interprofessional and transdisciplinary teamwork in health care. *Diabetes Metab Res Rev.* 2008;24 Suppl 1:S106-109.
- 49. Cartmill C, Soklaridis S, David Cassidy J. Transdisciplinary teamwork: the experience of clinicians at a functional restoration program. *J Occup Rehabil.* 2011;21(1):1-8.
- 50. Bryant A CK, eds. *The sage handbook of grounded theory*. London: Sage; 2007.
- 51. Di Blasi Z, Harkness E, Ernst E, Georgiou A, Kleijnen J. Influence of context effects on health outcomes: a systematic review. *Lancet.* 2001;357(9258):757-762.
- 52. Dorflinger L, Kerns RD, Auerbach SM. Providers' roles in enhancing patients' adherence to pain self management. *Transl Behav Med.* 2013;3(1):39-46.
- 53. Oosterhof B, Dekker JH, Sloots M, Bartels EA, Dekker J. Success or failure of chronic pain rehabilitation: the importance of good interaction a qualitative study under patients and professionals. *Disabil Rehabil*. 2014;36(22):1903-1910.
- 54. Olsson B, Tibblin G. Effect of patients' expectations on recovery from acute tonsillitis. *Fam Pract.* 1989;6(3):188-192.
- 55. Thomas KB. General practice consultations: is there any point in being positive? *Br Med J* (*Clin Res Ed).* 1987;294(6581):1200-1202.
- King R, Robinson V, Ryan CG, Martin DJ. An exploration of the extent and nature of reconceptualisation of pain following pain neurophysiology education: A qualitative study of experiences of people with chronic musculoskeletal pain. *Patient Education and Counseling.* 2016;99:1389-1393.
- 57. Shutty MS, Jr., DeGood DE, Tuttle DH. Chronic pain patients' beliefs about their pain and treatment outcomes. *Arch Phys Med Rehabil.* 1990;71(2):128-132.
- 58. Kerns RD, Rosenberg R. Predicting responses to self-management treatments for chronic pain: application of the pain stages of change model. *Pain.* 2000;84(1):49-55.

- 59. Biller N, Arnstein P, Caudill MA, Federman CW, Guberman C. Predicting completion of a cognitive-behavioral pain management program by initial measures of a chronic pain patient' s readiness for change. *Clin J Pain.* 2000;16(4):352-359.
- Watson JA, Ryan CG, Cooper L, et al. Pain neuroscience education for adults with chronic musculoskeletal pain: a mixed-methods systematic review and meta-analysis. *J Pain*. 2019;20(10)1140.e1-1140.
- 61. Creswell JW. *Qualitative Inquiry and Research Design, choosing among five approaches.* Second ed. London: Sage Publications, Inc.; 2007.
- 62. Corbin J SA, eds. *Basics of qualitative research: techniques and procedures for developing grounded theory.* London: Sage; 2008.
- 63. Charmaz K. *Constructing Grounded Theory.* 2nd ed: Sage Publications Ltd.; 2014.
- 64. Charmaz K. 'Discovering' chronic illness: using grounded theory. *Soc Sci Med.* 1990;30(11):1161-1172.
- 65. Oosterhof B, Dekker JHM, Sloots M, Bartels EAC, Dekker J. Success or failure of chronic pain rehabilitation: the importance of good interaction – a qualitative study under patients and professionals. *Disability and Rehabilitation*. 2014;36:1903-1910.
- 66. Rogers CR. Client-centered therapy. American Psychological Association Washington, DC; 1966.
- 67. Eubanks CF, Muran JC, Safran JD. Alliance rupture repair: A meta-analysis. *Psychotherapy* (*Chic*). 2018;55(4):508-519.
- McLaughlin AA, Keller SM, Feeny NC, Youngstrom EA, Zoellner LA. Patterns of therapeutic alliance: rupture-repair episodes in prolonged exposure for posttraumatic stress disorder. *J Consult Clin Psychol.* 2014;82(1):112-121.
- Safran JD, Muran JC, Eubanks-Carter C. Repairing alliance ruptures. *Psychotherapy (Chic)*. 2011;48(1):80-87.
- 70. Larsson MH, Falkenstrom F, Andersson G, Holmqvist R. Alliance ruptures and repairs in psychotherapy in primary care. *Psychother Res.* 2018;28(1):123-136.
- 71. Wijma AJ, Speksnijder CM, Crom-Ottens AF, et al. What is important in transdisciplinary pain neuroscience education? A qualitative study. *Disabil Rehabil*. 2018;40(18):2181-2191.
- 72. King R, Robinson V, Elliott-Button HL, Watson JA, Ryan CG, Martin DJ. Pain Reconceptualisation after Pain Neurophysiology Education in Adults with Chronic Low Back Pain: A Qualitative Study. *Pain Res Manag.* 2018;2018:3745651.
- 73. Robinson V, King R, Ryan CG, Martin DJ. A qualitative exploration of people's experiences of pain neurophysiological education for chronic pain: The importance of relevance for the individual. *Manual Therapy*. 2016;22:56-61.
- 74. Nijs J, van Wilgen CP, Van Oosterwijck J, van Ittersum M, Meeus M. How to explain central sensitization to patients with 'unexplained' chronic musculoskeletal pain: practice guidelines. *Man Ther.* 2011;16(5):413-418.

- Wood L, Hendrick PA. A systematic review and meta-analysis of pain neuroscience education for chronic low back pain: Short-and long-term outcomes of pain and disability. *Eur J Pain*. 2019;23(2):234-249.
- 76. Jootun D, McGhee G, Marland GR. Reflexivity: promoting rigour in qualitative research. *Nurs Stand.* 2009;23(23):42-46.
- 77. Hayes AF. PROCESS: A versatile computational tool for observed variable mediation, moderation, and conditional process modeling. [White paper]. Retrieved from http://www. afhayes.com/ public/process2012.pdf. 2012.
- 78. Wertli MM, Burgstaller JM, Weiser S, Steurer J, Kofmehl R, Held U. Influence of catastrophizing on treatment outcome in patients with nonspecific low back pain: a systematic review. *Spine (Phila Pa 1976).* 2014;39(3):263-273.
- Gurung T, Ellard DR, Mistry D, Patel S, Underwood M. Identifying potential moderators for response to treatment in low back pain: A systematic review. *Physiotherapy*. 2015;101(3):243-251.
- 80. Hoenig JM, Heisey DM. The Abuse of Power. The American Statistician. 2001;55(1):19-24.
- 81. George Assaf A, Tsionas MG, Andrikopoulos A. Testing moderation effects using nonparametric regressions. *International Journal of Hospitality Management*. 2020;86:102441.
- Preacher KJ, Rucker DD, Hayes AF. Addressing Moderated Mediation Hypotheses: Theory, Methods, and Prescriptions. *Multivariate Behav Res.* 2007;42(1):185-227.
- 83. Lincoln Y, Guba, EG. Naturalistic Inquiry. Newbury Park, CA. 1985.
- 84. Oostendorp RA, Elvers H, Mikolajewska E, et al. Manual physical therapists' use of biopsychosocial history taking in the management of patients with back or neck pain in clinical practice. *ScientificWorldJournal*. 2015;2015:170463.



Summary

Whether we want to or not, pain is an inevitable part of life. As on average twenty percent of the population suffers from chronic pain, chances are we might even experience chronic pain in our lifetime ourselves. When one is in chronic pain centrally mediated mechanisms, also known as central sensitization, play an important role in the pain experience. Central sensitization related pain, or recently called nociplastic pain by the International Association for the study of Pain, is characterized by hyperexcitability of the central nervous system. Leading to increased responsiveness and eventually causing pain without noxious peripheral stimuli. Especially in chronic pain it is known that biopsychosocial factors provoke and contribute to the pain one experiences. Whilst we are humans, body and mind cannot be separated. As a result, the treatment of patients with chronic pain has evolved and multifactorial cognitive behavioural treatments are advised. In the current dissertation an outline of such a treatment for patients with chronic pain, by means of patient centered transdisciplinary Pain Neuroscience Education as integration in the cognitive behavioural treatment for patients with chronic pain is explored.

A theoretical framework of patient centeredness in physiotherapy

Research has shown that improvements in patient centered healthcare improves the therapeutic alliance, satisfaction of patients, patient outcomes and decreases healthcare costs. Patient-centeredness has possibilities to further enhance the treatment of patients with chronic pain. Previous gualitative studies have shown that for instance 'relevance for the individual', which is a central part of patient-centeredness, is of importance during Pain Neuroscience Education. Pain Neuroscience Education is an important aspect in the treatment of patients with chronic pain. Within the healthcare system, Pain Neuroscience Education is most often delivered by physiotherapists. However, to our knowledge, theoretical models proposing patient centeredness have thus far been studied in overall care (The Picker Institute's principles), medicine and nursing. Without a commonly accepted definition of patient-centeredness in physiotherapy, research into its implementation and outcomes is stranded. Therefore, we conducted a systematic qualitative literature review in which the phenomenon of patient-centeredness in physiotherapy was studied from both the perspective of patients and physiotherapists. From these studies a synthesis of interrelated themes ("individuality", "communication", "education", "goal setting", "support", "social characteristics of a patient-centered physiotherapist", "a confident physiotherapist", "knowledge and skills of *a physiotherapist in patient-centeredness"*) with subthemes and a proposed conceptual framework of patient-centeredness in physiotherapy was constructed. Patient-centeredness in physiotherapy entails the characteristics of offering an individualized treatment, continuous communication (verbal and non-verbal), education during all aspects of treatment, working with patient defined goals, a treatment in which the patient is supported and empowered and a physiotherapist with patient centered social skills, confidence and knowledge.

A practical tool for the clinical biopsychosocial physiotherapy assessment of patients with chronic pain

To individualize the Pain Neuroscience Education prior to the education the physiotherapist should perform a thorough patient centered assessment of the patient with chronic pain and his/her complaints. Besides the 'usual physiotherapy intake' psychosocial factors ought to be examined during this intake, for instance factors such as cognitions, pain catastrophizing and emotional problems. However, research shows that physiotherapists find it difficult to investigate and deal with psychosocial problems, especially when they did not receive additional post graduate training in it. Therefore, to aid physiotherapists we conducted a narrative review in which an outline of the clinical biopsychosocial assessment to be used by physiotherapists in the encounter of patients with chronic pain was delineated. During this assessment the PSCEBSM-model can be used by physiotherapists, containing: Pain mechanism, Somatic, Cognitive, Emotional, Behavioural, Social and Motivational factors. First the dominant pain mechanism (predominant nociceptive, neuropathic or non-neuropathic central sensitization pain) is established according to the classification algorithm. Followed by an evaluation of the perpetuating biopsychosocial factors. The patient' specific pain mechanism and perpetuating factors can be filled pain analysis sheet, to be used in the following individualized Pain Neuroscience Education session.

Perspectives of patients with chronic pain who recently received Pain Neuroscience Education in a transdisciplinary setting

As chronic pain is often a complex problem effective team approaches frequently exceed the expertise of monodisciplinary care. More team collaboration, for instance by working transdisciplinary, has the potential for better outcomes of the treatment of patients with chronic pain. Transdisciplinary teamwork outperforms multi- and interdisciplinary in that professionals collaborate intensively, with shared responsibilities. Even though every team member has his/her expertise, there are flexible boundaries and roles, and team members learn simultaneously. It is important to comprehend the experiences of patients with these new approaches and interventions, such as transdisciplinary Pain Neuroscience Education. Therefore, to study the clinical application of this transdisciplinary teamwork and patient centered Pain Neuroscience Education, the experiences of patients with chronic pain were examined by a qualitative classical Grounded Theory study. Following Grounded Theory, 15 patients with chronic pain were interviewed and the data was analysed, of which four interacting topics emerged: The first topic was "the pre-Pain Neuroscience Education phase", the phase prior to the Pain Neuroscience Education, which comprised "a broad intake" and "the healthcare professionals". The second found topic "a comprehensible Pain Neuroscience Education" contained an "understandable explanation" and the "interaction between the physiotherapist and psychologist". The third topic involved the "outcomes of Pain Neuroscience Education", including the subthemes "awareness", "finding peace of mind", and "fewer complaints". The fourth topic, "scepticism", contained "doubt towards the diagnosis and Pain Neuroscience Education", "disagreement with diagnosis and Pain Neuroscience Education", and "Pain Neuroscience Education can be confronting".

The results reveal the importance of the therapeutic alliance between the patient and healthcare professional, taking time, listening, providing a clear explanation and valuable team interaction. When doing so, most of the respondents with chronic pain reported improved cognitions about their pain and pain related behaviour.

Clinical observations of patient and therapist interactions in transdisciplinary Pain Neuroscience Education

It has been suggested to study clinical characteristics that may increase the effectiveness of PNE. As little is known about the clinical application of Pain Neuroscience Education. Because Pain Neuroscience Education is a "talk-modality" interpersonal aspects, such as the therapeutic alliance contribute to the treatment. The important aspects of social processes, such as the therapeutic alliance in clinical practice, can best be captured through qualitative studies in the form of observations. Therefore, based on Constructive Grounded Theory we conducted an observational qualitative study to try to grasp the practice of transdisciplinary Pain Neuroscience Education. With the aim of the study to explore the social processes between the patient with chronic pain, partner and healthcare professionals in the practice of transdisciplinary Pain Neuroscience Education. The outcome was a theoretic construct, grounded in the data of the observations. Nine Pain Neuroscience Education sessions from eight participants with

chronic pain who were treated in a transdisciplinary treatment center were recorded with audiovisual equipment. These sessions were provided by one psychologist and physiotherapist per session. Four connected and interactive themes were found to be of importance for transdisciplinary PNE, with three overarching themes. The first theme we found was "generating a safe and comfortable feeling, situation" as a prerequisite. The second theme was the "PNE process". The third theme emerged as "the process of change of the respondent", leading to the fourth theme: "outcome of the change process". With overarching themes: the "influence of changeability of the respondents", "interplay between the healthcare professionals" and "communication skills" of the healthcare professionals.

Therapists should be aware of the importance of using a patient centered approach, in which they continuously monitor the therapeutic alliance and of their communicational skills. Furthermore, a step by step introduction of the Pain Neuroscience Education is suggested. And last, individualized metaphors are of importance to provide a patient centered Pain Neuroscience Education.

Moderators of change of self-reported signs of central sensitization following an individualized transdisciplinary pain management program

Two factors that greatly influence chronic pain and the activity in the dynamic pain connectome are pain catastrophizing and illness perceptions. Both influence each other and contribute to what is known as 'cognitive emotional sensitization'. Pain Neuroscience Education is known to improve the knowledge on pain of patients with chronic pain, as well as improve their illness perceptions. Thereby decreasing the threat of pain and pain catastrophizing. As the integration of Pain Neuroscience Education in monodisciplinary physiotherapy is effective, integrating it in multi- or transdisciplinary care seems recommendable, yet has not been studied before. Subsequently, it is important to comprehend which factors moderate these changes following a transdisciplinary cognitive behavioural therapy treatment for patients with chronic pain. A bottom up retrospective quantitative observational study evaluated the differences in clinical outcomes, among which pain catastrophizing, illness perceptions and self-reported signs of central sensitization. Furthermore, the relationship between these variables and the moderating effects of pain catastrophizing on the relationship between illness perceptions and self-reported signs of central sensitization in patients with chronic pain following a transdisciplinary cognitive behavioural therapy treatment were investigated. We found a moderate association between pretreatment scores of

pain catastrophizing and illness perceptions, suggesting a bilateral relationship between catastrophic thinking and illness perceptions. Furthermore, there were significant, yet moderate, associations between change scores of pain catastrophizing, illness perceptions and self-reported symptoms of central sensitization. However, the results indicate that even though there is an association between pretreatment and change scores in illness perceptions and self-reported symptoms of central sensitization in patients with chronic pain, neither the direction nor strength of the relationship between these two was influenced by both pretreatment or change scores of pain catastrophizing. Due to the design of the study no hard conclusions can be drawn. As pain, and patients with chronic pain are multi-faceted, multifaceted everchanging factors influence their pain experience. Therefore, in future research the interplay of

these factors should be studied by evolved approaches such as network analysis.



Samenvatting

Of we het willen of niet, pijn is een onvermijdelijk onderdeel van het leven. Gemiddeld lijdt twintig procent van de bevolking aan chronische pijn, dus is er zelfs een kans dat wij in ons leven chronische pijn ervaren. Wanneer iemand chronische pijn ervaart spelen centraal mediërende mechanismen, ook wel bekend als centrale sensitisatie, een belangrijke rol in de pijnervaring. Centrale sensitisatie gerelateerde pijn, of recent door de International Association for the Study of Pain nociplastische pijn genoemd, wordt gekenmerkt door hyperexcitabiliteit van het centrale zenuwstelsel. Dit zorgt voor een verhoogde responsiviteit, leidend tot pijn zonder dat er sprake is van schadelijke perifere stimuli. Omdat bij de mens lichaam en geest niet aparte entiteiten zijn is het bekend dat biopsychosociale factoren de ervaring en voortzetting van pijn beïnvloeden. Als gevolg hiervan is de behandeling van patiënten met chronische pijn geëvolueerd en een multifactoriële cognitief gedragsmatige behandeling is geadviseerd. In deze dissertatie is een synopsis van de behandeling voor patiënten met chronische pijn, in de vorm van patiënt gecentreerde intake, transdisciplinaire pijneducatie en interventie van patiënten met chronische pijn, onderzocht.

Een theoretisch kader van patiënt gecentreerde fysiotherapie

Onderzoek heeft aangetoond dat positieve veranderingen in patiënt gecentreerde zorg de therapeutische alliantie, de patiënt tevredenheid en de patiënt gerelateerde uitkomsten verbetert en daarnaast de zorgkosten verlaagt. Tevens hebben eerdere kwalitatieve onderzoeken aangetoond dat bijvoorbeeld 'relevantie voor het individu', een centraal onderdeel van patiënt gecentreerde zorg, belangrijk is in de uitleg over pijn aan patiënten met chronische pijn (Pijneducatie). Pijneducatie is een belangrijk onderdeel van de behandeling van patiënten met chronische pijn. In de huidige gezondheidszorg wordt Pijneducatie meestal uitgevoerd door fysiotherapeuten. Echter, het onderzoek naar de theoretische modellen van patiënt gecentreerde zorg is tot dus ver, voor zover onze kennis strekt, alleen bestudeerd in de algehele zorg (de principes van het Picker Institute), de geneeskunde en de verpleging. Zonder een algemeen aanvaarde definitie van patiënt gecentreerde fysiotherapie kan het onderzoek naar de implementatie en effecten van patiënt gecentreerde fysiotherapie zich niet verder ontwikkelen. Daarom hebben wij een systematisch kwalitatief literatuuronderzoek uitgevoerd waarin het fenomeen van patiënt gecentreerde fysiotherapie is bestudeerd. In het onderzoek is zowel het perspectief van patiënten, als het perspectief van fysiotherapeuten meegenomen. Uit deze studie kwamen verschillende onderling samenhangende thema's naar voren, namelijk: "individualiteit", "communicatie", "educatie", "doelen stellen", "ondersteuning", "sociale kenmerken van een patiënt gecentreerde fysiotherapeut", "een zelfverzekerde fysiotherapeut", "kennis en vaardigheden van een patiënt gecentreerde fysiotherapeut". Daarnaast zijn de verbanden in de vorm van een conceptueel framework van patiënt gecentreerde fysiotherapie weergegeven. Patiënt gecentreerde fysiotherapie omvat een geïndividualiseerde behandeling, continue communicatie (verbaal en non-verbaal), educatie tijdens alle aspecten van de behandeling, werken met door de patiënt gedefinieerde doelen, een behandeling waarbij de patiënt wordt ondersteund en bekrachtigd door een fysiotherapeut met patiënt gecentreerde sociale vaardigheden, zelfvertrouwen en kennis.

Een raamwerk voor de biopsychosociale intake van patiënten met chronische pijn binnen de fysiotherapie

Om pijneducatie geïndividualiseerd te kunnen geven is het van belang dat de fysiotherapeut voorafgaand aan de educatie een grondige patiënt gecentreerde intake van de patiënt met chronische pijn uitvoert. Naast de 'gebruikelijke fysiotherapeutische intake' is het belangrijk dat psychosociale factoren worden onderzocht, zoals cognities, pijn catastroferen en emotionele problemen. Uit onderzoek blijkt echter dat fysiotherapeuten het moeilijk vinden om met psychosociale problemen om te gaan, vooral als ze hier geen aanvullende opleiding in hebben gevolgd. Om fysiotherapeuten te helpen is er een narratief review uitgevoerd, waarin een opzet van de klinische biopsychosociale intake van patiënten met chronische pijn is beschreven voor de fysiotherapeutische praktijk.

Tijdens deze intake kan het PSCEGSM-model worden gebruikt door fysiotherapeuten, bestaande uit: predominant Pijn mechanisme, Somatische, Cognitieve, Emotionele, Gedragsmatige, Sociale en Motivationele factoren. Als eerste wordt hierin het predominante pijnmechanisme (overheersend nociceptieve, neuropatische of nietneuropatische centrale sensitisatie pijn) vastgesteld volgens het classificatie-algoritme. Gevolgd door een evaluatie van de onderhoudende en beïnvloedende biopsychosociale factoren. Het specifieke pijnmechanisme van de patiënt met chronische pijn en de beïnvloedende factoren kunnen vervolgens worden ingevuld in een pijnanalyse formulier. Deze kan gebruikt worden in de daarop volgende geïndividualiseerde patiënt gecentreerde Pijneducatie-sessie.

De ervaring van patiënten met chronische pijn die recent transdisciplinaire Pijneducatie hebben gehad

Omdat chronische pijn een complex probleem is, overtreffen teambenaderingen vaak de expertise van monodisciplinaire zorg. Hoe intensiever de samenwerking, bijvoorbeeld door transdisciplinair werken, des te groter zijn de mogelijke effecten van de behandeling van patiënten met chronische pijn. Transdisciplinair teamwerk overtreft hierin multi- en interdisciplinair teamwerk, doordat professionals intensief samenwerken met gedeelde verantwoordelijkheden en hoewel elk teamlid zijn/haar expertise heeft zijn er flexibele grenzen en rollen en leren teamleden tegelijkertijd. Het is echter van belang om de ervaringen van patiënten over deze intensievere samenwerkingen te onderzoeken.

Om de klinische toepassing van dit transdisciplinaire teamwerk en patiënt gecentreerde transdisciplinaire Pijneducatie te bestuderen, werden de ervaringen van patiënten met chronische pijn over transdisciplinaire pijneducatie onderzocht middels een kwalitatieve klassieke Grounded Theory studie.

15 patiënten werden geïnterviewd en deze data zijn geanalyseerd, hieruit kwamen vier interacterende topics naar voren: Het eerste topic was "de fase voor de pijneducatie", waarin de basis gelegd werd voor de PE, met sub thema's "een brede intake" en "de zorgprofessionals". Het tweede topic was "een begrijpelijke pijneducatie" met "begrijpelijke verklaring" en "interactie tussen de fysiotherapeut en psycholoog". Het derde topic bevatte de "uitkomsten van de pijneducatie" inclusief "bewustwording", "gemoedsrust vinden" en "minder klachten". En het laatste topic bevatte "scepsis" met "twijfel over de diagnose en pijneducatie", "niet eens zijn met de diagnose en pijneducatie" en "pijneducatie kan confronterend zijn".

Uit de resultaten blijkt het belang van de therapeutische alliantie tussen de patiënt en de zorgprofessionals, zoals de tijd nemen voor de patiënt, luisteren naar de patiënt, een duidelijke uitleg geven aan de patiënt en goede teaminteractie. Indien dit gebeurt, noemden de meeste respondenten met chronische pijn positief veranderde cognities over hun pijn en verbeterd pijn gerelateerd gedrag.

Klinische observaties van de patiënt en therapeut interacties van transdisciplinaire Pijneducatie

Er is tot heden weinig bekend over de klinische toepassing van Pijnedcuatie. Daarom is voorgesteld om de klinische karakteristieken die de toepassing van Pijneducatie mogelijk verbeteren, te onderzoeken. Omdat Pijneducatie een 'praatmodaliteit' is, dragen verschillende interpersoonlijke aspecten, zoals de therapeutische alliantie bij aan de behandeling. De belangrijke aspecten van sociale processen, zoals de therapeutische alliantie, kunnen het best geanalyseerd worden in kwalitatief onderzoek door middel van observaties. Daarom is er een kwalitatief observatief onderzoek uitgevoerd, op basis van Constructive Grounded Theory, waarin de getracht werd de praktische toepassing van transdisciplinaire Pijneducatie te doorgronden. Met als doel om de sociale processen tussen patiënt, partner en zorgprofessionals te onderzoeken en theoretiseren. De uitkomst was een theoretisch construct, gegrond in de data van de observaties.

Negen Pijneducatie-sessies van acht respondenten met chronische pijn die werden behandeld in een transdisciplinair behandelcentrum werden opgenomen middels audiovisuele apparatuur. Deze sessies werden gegeven door één een psycholoog en een fysiotherapeut per sessie.

Vier samenhangende en interactieve thema's kwamen naar voren. Het eerste thema was "het genereren van een veilig en comfortabel gevoel en situatie". Het tweede thema dat naar voren kwam was "het Pijneducatie-proces". Het derde thema was het "veranderingsproces van de respondent", waaruit voort vloeide het vierde thema: "de uitkomst van het Pijneducatie-proces". De volgende overkoepelende thema's werden geïdentificeerd: "de invloed van de veranderbaarheid van de respondenten", het "samenspel tussen de zorgprofessionals" en "communicatieve vaardigheden".

Het is van belang dat therapeuten een patiënt gecentreerde benadering gebruiken, waarbij ze de therapeutische alliantie bewaken en hun communicatieve vaardigheden onderkennen. Daarnaast is een stapsgewijze opbouw van de Pijneducatie aan te bevelen. Ten slotte zijn geïndividualiseerde metaforen van belang om een patiënt gecentreerde Pijneducatie te geven.

Moderatoren van verandering van zelf-gerapporteerde symptomen van centrale sensitisatie na een geïndividualiseerd transdisciplinair pijn management programma

Twee factoren die chronische pijn sterk beïnvloeden door een verhoogde activiteit in het pijn-connectoom zijn pijncatastrofering en ziektepercepties. Beide factoren hebben invloed op elkaar en dragen bij aan wat bekend staat als 'cognitieve emotionele sensitisatie'. Het is bekend dat Pijneducatie de kennis van patiënten met chronische pijn over pijn verbetert, ziektepercepties verandert, waardoor de dreiging van pijn en pijn catastroferen als gevolg van de Pijneducatie mogelijk afneemt. Omdat de integratie van Pijneducatie in monodisciplinaire fysiotherapie effectief is lijkt integratie in multiof transdisciplinaire zorg aan te bevelen. Dit is echter nog niet eerder onderzocht. Daarbij is het belangrijk om het modererende effect van pijn catastroferen op behandelingsresultaten te onderzoeken.

Een bottom-up retrospectief kwantitatief observationeel onderzoek is uitgevoerd om de klinische resultaten te evalueren van onder andere pijn catastroferen, ziektepercepties en zelf-gerapporteerde symptomen van centrale sensitisatie. Tevens is onderzocht wat relaties tussen en de modererende effecten van pijn catastroferen op de relatie tussen ziektepercepties en zelf gerapporteerde symptomen van centrale sensitisatie zijn.

We vonden een gematigde associatie tussen voormetingen van pijn catastroferen en ziektepercepties. Dit wijst op een bilaterale relatie tussen catastrofaal denken en ziektepercepties. Verder waren er significante, maar gematigde, associaties tussen de verander-scores van pijn catastroferen, ziektepercepties en zelf gerapporteerde symptomen van centrale sensitisatie.

Als laatste, hoewel er een associatie was tussen baseline scores en scores in veranderingen in ziektepercepties en zelf gerapporteerde symptomen van centrale sensitisatie, werd deze niet gemodereerd door zowel baseline scores als scores in verandering van pijn catastroferen. Dit kan mogelijk verklaard worden door het feit dat de beginscores van de patiënten met chronische pijn op pijn catastroferen lager waren dan de cut off score voor pijn catastroferen en er slechts kleine veranderingen aanwezig waren.

Vanwege het design van de studie kunnen geen harde conclusies getrokken worden. Omdat pijn en patiënten met chronische pijn multidimensionaal zijn, beïnvloeden interactieve complexe factoren de pijnervaring. In toekomstig onderzoek is het daarom van belang om het samenspel van meerdere factoren, in plaats van alleen pijn catastroferen, te bestuderen door bijvoorbeeld netwerkanalyses.



Acknowledgements

This chapter was one of the most challenging to write. The pas seven years were incredible. Many people helped shape this dissertation, either by contributing in the research, mentoring me, discussing, providing feedback, visiting me in the practice with a complaint, and/or just by being there for me whenever I needed support. The notion that I would miss mentioning someone, not thanking you for the influence you had, directly or indirectly on this dissertation, is getting to me.

So, first off: Thank you ALL! Your name may not be listed, know how much you mean to me. With that out of the way comes the second point: Who comes first? Again, an unbearable task. So please, don't be offended if your name is not at the top of the list, a bit lower on the page, it does not mean I value you any less.

I'll start with my colleagues at Fysio de Tille and Transcare. Your knowledge, kindheartedness, A shoulder when needed, fun and laughter has helped shape me as a human being, clinician and researcher.

The patients (with long-term pain) which I have had the chance to meet throughout my career and those patients who were willing to cooperate in our studies. Thank you for being honest, FOR trusting me, and showing me what you value, and what we can accomplish together.

Jo Nijs and Paul van Wilgen. I wondered whether I should mention you separately. But, you two complement each other, therefore this felt more natural. Thank you for letting me swim on my own, you were my buoys. Even though every now and then I swam my own (different) direction. Thank you for allowing me to follow my own route. Thank you for teaching me how (letting me) to become a critical thinker. Thank you for all the opportunities in learning new skills (teaching, clinical, research). And formost, for being friendly, approachable, down to earth, wise (but not old ;)) men.

Doeke Keizer: Al is Fryslân plat, it hat syn hichtepunten. You're one of them. Thanks for being the 'native' in my PhD-team. Thank you for your interest, ever kind words and support whenever I needed it. You made my seven-year journey feel like walk in the park. Of course I won't forget our Rotterdammer, Lennard Voogt (ps: I kind of think he has a bit of Fryslân in him). Thank you for your knowledge on qualitative research. Putting time and effort in our collaboration when I knew your time was precious, yet never letting on it was that way. And for reading these never-ending long documents of mine.

Of course all PiM-ers, but especially: Rinske Bults, Sanneke Don, Hester den Bandt and Margot de Koning. A favorite quote of mine: Strong women lift each other up. Let's keep lifting each other up and keep having each other's back. I've enjoyed the time we spend together: travelling, working, the discussions, philosophies, good food and wine! Here's to more of all that.

My friends, outer- and inner circle, I can't name all of you, just know I love you. A few names I do have to mention. Christina and Tineke, Marloes, my dear, dear neighbors and godchildren. All of you, thank you for supporting me, listening to me, the cups of tea/coffee, comfort food, but also showing me there is a world outside of work filled with kind, loving people.

My family, heit and mem, brothers and sister, their partners and my nephews: thank you. I know I have a lot to be grateful for, and one of the things I am most grateful of is to have such a loving, warm and supportive family.

It is probably not customary to thank an animal in your PhD-dissertation, but I would like to thank Bella Donna, my horse. Beside teaching me so much about myself, there is nothing better than to snuggle up, cuddle and smell her 'horse-perfume'.

Last, but most certainly not least! Eelko, my hubby. Thank you for being the solid rock, whenever I 'butterfly' into all the different things I like to do. We complement each other. Without the numerous ways in which you supported me the last seven years, things would've been very different. I am looking forward towards whatever the future holds for us and our little family.



About the author

Amarins Jelly Wijma was born on February 17th 1987 in Bakkeveen, the Netherlands. She joined the bachelor program Physiotherapy at the Hanze University of Applied Sciences, Groningen, from 2004-2008. During her final years of the bachelor program and first year of working in clinical private practice her interest in the 'when' and 'how' of physiotherapy was sparked. Therefore, she decided to further enhance her scientific knowledge by enrolling the pre-master and master program Clinical Health Sciences, Physiotherapy Science, at the Utrecht University Medical Center (2010-2013). During the Master program she met Paul van Wilgen and 'thus the science journey began'. This journey started with a systematic qualitative literature review on patient centeredness in physiotherapy, the foundation for her PhD dissertation. During the last phase of her master's degree a qualitative Grounded Theory study was conducted, with the subject-matter 'The experiences of patients with transdisciplinary PNE', the next chapter of her PhD dissertation. In March 2014 she started her PhD at The Vrije Universiteit Brussel, Brussels, Belgium. Throughout her PhD trajectory, with the mentorship of Paul van Wilgen, Jo Nijs, Doeke Keizer and Lennard Voogt she further developed her qualitative research skills, writing (inter)national scientific publications, book chapters, blogs, teaching - and project-management skills. As can be read throughout the thesis, she finds it important to perform 'bottom up research', in which patients' voices are expressed to translate and improve clinical care for patients with chronic pain.

She is a member of the Pain in Motion research group. She has co-authored several chapters in (inter) national books. Furthermore, she has authored 8 vSCI-indexed full-text papers of which she published 5 as first author.

Besides her PhD she has continued to work in clinical practice, both in private practice and in a secondary outpatient treatment center for patients with chronic pain. Furthermore, she has taught, and continues to teach clinicians working with patients with long-term pain about PNE and other aspects involving the management of longterm pain throughout Europe. In the future she hopes to be able to continue her clinical work in combination with research and teaching.



List of publications

List of publications in journals with peer review (chronological)

Pain following cancer treatment: Guidelines for the clinical classification of predominant neuropathic, nociceptive and central sensitization pain.

Nijs J, Leysen L, Adriaenssens N, Aguilar Ferrándiz ME, Devoogdt N, Tassenoy A, Ickmans K, Goubert D, van Wilgen CP, Wijma AJ, Kuppens K, Hoelen W, Hoelen A, Moloney N, Meeus M. *Acta Oncol.* 2016 Jun;55(6):659-63.

Clinical biopsychosocial physiotherapy assessment of patients with chronic pain: The first step in pain neuroscience education. Wijma AJ, van Wilgen CP, Meeus M, Nijs J. Physiother Theory Pract. 2016 Jul;32(5):368-84.

What is important in transdisciplinary pain neuroscience education? A qualitative study.

Wijma AJ, Speksnijder CM, Crom-Ottens AF, Knulst-Verlaan JMC, Keizer D, Nijs J, van Wilgen CP.

Disabil Rehabil. 2018 Sep;40(18):2181-2191.

Patient-centeredness in physiotherapy: What does it entail? A systematic review of qualitative studies.

Wijma AJ, Bletterman AN, Clark JR, Vervoort SCJM, Beetsma A, Keizer D, Nijs J, van Wilgen CP.

Physiother Theory Pract. 2017 Nov;33(11):825-840

Explaining pain following cancer: a practical guide for clinicians. Nijs J, Wijma AJ, Leysen L, Pas R, Willaert W, Hoelen W, Ickmans K, van Wilgen CP. *Braz J Phys Ther.* 2019 Sep - Oct;23(5):367-377.

Between the lines: a qualitative phenomenological analysis of the therapeutic alliance in pediatric physical therapy. Crom A, Paap D, Wijma AJ, Dijkstra PU, Pool G.

Phys Occup Ther Pediatr. 2020;40(1):1-14

Integrating motivational interviewing in pain neuroscience education for people with chronic pain: a practical guide for clinicians.

Nijs J, Wijma AJ, Willaert W, Huysmans E, Mintken P, Smeets R, Goossens M, van Wilgen CP, Bogaert W, Louw A, Cleland J, Donaldson M. *Phys Ther.* 2020 May 18;100(5):846-859.

The moderating effects of pain catastrophizing on the relationship between illness perceptions and self-reported signs of central sensitization in patients with persistent pain.

Wijma AJ, Bults RM, Nijs J, van Wilgen CP Int J Rehabil Res. 2020 Dec;43(4):347-354.

Published letters, short communications and national publications

"Pijneducatie als basis voor de behandeling van subacute lage rugklachten" Wijma A, van Wilgen CP, Nijs J. *Fysiopraxis* 2014;23(8):24-28.

"Hoe is pijneducatie te gebruiken bij patiënten met chronische pijn?" Wijma A. *Physios* 2016; 1 (8): 52-53.

"Response to Letter" Wijma AJ, van Wilgen CP, Meeus M, Nijs J. Physiother Theory Pract. 2017; March;33(3):263-265.

"Pijn na kanker: behandeling in de fysiotherapeutische praktijk" Nijs J, Leysen L, Pas R, Wijma A, van Wilgen CP, Hoelen W. *Fysiopraxis* 2017; July-August: 28-29.

"Whiplash Associated Disorder: 'the bigger picture'" Don S, Wijma A, Voogt L, Nijs J. *Fysiopraxis* 2017; November: 21-23

"Centrale sensitisatie: de diagnostiek en pijneducatie als eerste stap in de behandeling" van Wilgen CP, Bults RM, Wijma A. *Tijdschrift voor Bedrijfs- en Verzekerzekeringsgeneeskunde* 2018; November; 26(9):458-461
"Addertjes onder het gras: stappen in de behandeling van chronische pijn in het biopsychosociale model"

Wijma AJ and van Wilgen CP. Fysiopraxis 2018; December-January:24-27

Contribution to books

"Van acute naar chronische pijn, van biomedisch naar biopsychosociaal" van Wilgen CP, Wijma A, Keizer D, Nijs J. Jaarboek Ergotherapie 2013 Uitgeverij ACCO Leuven/Den Haag; Hoofdstuk 8: p139-146.

"Pijneducatie voor patiënten met pijn na kanker"

Nijs J, Leysen L, Pas R, Wijma A, Ickmans K, Hoelen W, van Wilgen CP.

Chapter 7 in: "Pijneducatie: een praktische handleiding voor (para)medici" Second edition. van Wilgen CP, Nijs J (eds.). Houten, the Netherlands: Bohn Stafleu van Loghum, 2018. pp. 97-110.

"Pijneducatie in een transdisciplinaire setting: een nieuwe, innovatie aanpak voor de toekomst?"

van Wilgen CP, Wijma A, Bults RM.

Chapter 9 in: "Pijneducatie: een praktische handleiding voor (para)medici" Second edition. van Wilgen CP, Nijs J (eds.). Houten, the Netherlands: Bohn Stafleu van Loghum, 2018. pp. 131-137.